Foreign-Trade Zones Manual





U.S. Customs and Border Protection Office of Field Operations Publication#: 0000-0559A 2011

FOREWORD

The purpose of the Foreign Trade Zone Manual is to place in one document, the various laws, regulations, policies and procedures that Customs and Border Protection personnel, grantees, operators and users need to know in the daily operation of Foreign Trade Zones. It may also be useful to brokers, freight forwarders, bonded and international carriers and other members of the import-export community.

Many laws and regulations in this manual are paraphrased for the sake of simplicity and easier reading. Grantees, operators, users and other readers are urged to study the actual text of the cited laws and/or regulations before making any costly investment decisions initially based on information in this manual. This manual is not itself law, it is intended merely as a guide. Readers may seek rulings from the Office of International Trade concerning matters covered by this manual pursuant to the procedures of 19 CFR 177.

This manual supersedes FTZ Manual, Publication #: 0000-0559, issued in March, 2003. CBP internal instructions, not intended for public disclosure are not included in this manual, but are still in effect. Any new measures taken by Customs and Border Protection affecting the Foreign Trade Zone program, after issuance of this manual will be announced through various methods, e.g., electronic bulletin board, e-mail and facsimile transmission, directives and rulings. These issuances will be provided as supplements for inclusion in this manual upon release. This manual is intended to be maintained with current information to better serve all parties.

This manual has been compiled by Customs and Border Protection as a joint effort with the Foreign-Trade Zones Board, the U.S. Census Bureau and the National Association of Foreign Trade Zones. Their contributions to this document are appreciated.

Thomas S. Winkowski Assistant Commissioner Office of Field Operations

TABLE OF CONTENTS

Chapter 1

INTRODUCTION, PURPOSE, AND INTERPRETATIONS

1.1	Introduction	16
1.2	Purpose	16
1.3	Referenced material	16
1.4	Instructions vs. Guidelines	17
1.5	Port Director Authority	17
1.6	CBP Issuances	18
1.7	Rulings and Disagreements	18
1.8	Disclosure of Information	18
1.9	Local CBP Procedures	18
1.10	O Automated Procedures	18
	Chapter 2	
I.	AWS, REGULATIONS, ORGANIZATION, DUTIES, AND O	ZENERAI.
	INFORMATION CONCERNING FOREIGN TRADE ZO	
2.1	Laws and Regulations	19
	(a) Foreign-Trade Zones Act	
	(b) Foreign-Trade Zone Regulations	
	(c) CBP Laws and Regulations	19
	(d) Other Federal Laws	20
	(e) State and Local Laws and Regulations	20
	(f) Grantee Schedules	21
2.2	The FTZ Board	21
	(a) Makeup and Authority of The FTZ Board	21
	(b) Presidential Direction to Furnish Records	22
	(c) Committee of Alternates	22
	(d) Executive Secretary	22
	(e) CBP Role	22
	(f) Other Agencies	23
2.3	Grantee, Operator, and User Duties and Responsibilities	23
	(a) Grantees	
	(b) Operators	24
	(c) Users	25
2.4	Subzones	25
2.5	Vessels and other Carriers	25
	(a) Movement of Zone Merchandise	25
2.6		
	(a) Receipt of Supplies	

(b) Use	26
2.7 Production Equipment	
2.8 CBP Forms	
2.9 Place of Filing and Merchandise Examination	
(a) Exceptions	
(b) Delivery Procedures	
(b) Delivery Frocedures	20
Chapter 3	
CUDEDVICION OF FODEICN TO A DE ZONEC CD ANTEE AND	ODEDATOD
SUPERVISION OF FOREIGN-TRADE ZONES, GRANTEE AND RESPONSIBILITIES AND CBP FEES	OPERATOR
RESPONSIBILITIES AND CDF FEES	
3.1 General	29
3.2 Audit-Inspection Supervision	
3.3 CBP Supervision	
(a) Physical Supervision	
(b) Security Service	
(c) Access to Zones	
3.4 Audits	
(a) Notification and Arrival	
(b) Closeout	
3.5 Compliance Reviews	
(a) Notification and Arrival	
(b) Duration, Frequency, and Conduct	
(c) Completion and Closeout	
(d) Best Practices: C-TPAT	
(e) Best Practices: IMAGE	
· /	
3.6 Coordination of Audits and Compliance Reviews3.7 Monitoring Foreign Trade Zone Regulations	
3.8 Supervision by Operator	
3.9 Operator Responsibilities	
(a) Enforcement of Operator's Requirements	
3.10 Grantee Liability	
(a) Activation and Alteration Fees	
(b) Merchandise Processing Fee (MPF)	
(c) Harbor Maintenance Fee (HMF)	
(c) Harbor Maintenance Fee (HMF)	30
Chapter 4	
ESTABLISHMENT, ACTIVATION, ALTERATION, AND OTHE	R CHANGES
4.1 General	
4.2 Application and Approval of Foreign-Trade Zone	
(a) Location Requirements	39

	(b) Grantee Sponsorship	40
	(c) Qualification of Applicant	
	(d) Subzones	40
	(e) Application Fees	
	(f) Criteria for Grants	40
	(g) Initiation of Procedure	
	(h) Application Review	42
4.3	Grant Limits Lapse and Termination	45
4.4	Expansion or Boundary Modification	46
	(a) Minor Boundary Modification	
	(b) Description of Approved Zone Area	46
4.5	Other Changes to the Zone Project	46
	(a) Change to Grantee Organizations	
4.6	Commencement of Operations	
4.7	Application for Activation	
	(a) Initiation of Procedure	
	(b) Format of Application	
	(c) Treatment of Operator's Procedures Manual	47
	(d) Review of Application	
	(e) Background Investigations	
	(f) Cargo Security Survey	
	(g) Other Considerations	
	(h) Decision on Application	
	(i) Admission of Merchandise Already in Zone	
	(j) Foreign-Trade Zone Operator's Bond	
4.8	11	
	(a) Application Procedure	
	(b) Bonded Warehouse Within Approved Zone	
	Deactivation of Zone Site	
	Reactivation	
	New Bond	
	New Zone Operator	
	(a) Interim Responsibility of Existing Operator	
4.13	Other Changes	52
	(a) Change in Ownership of Operator	
	(b) Name, Address, or Organizational Status Change	
4.14	Permanent File Requirements	53
	Chapter 5	
	•	
5.1	ZONE STATUS General	54
	Merchandise Not in Zone Status	
	Application for Zone Status or Status Change	
	Foreign Status Merchandise	

 (a) Application Requirements (b) Determination of Duties and Taxes (c) Privileged Foreign Status Binding 5.6 Non-Privileged Foreign Status 5.7 Domestic Status (a) Merchandise Qualifying for Domestic Status (b) Admission Requirements (c) Recordkeeping Requirements (d) Loss of Identity as Domestic Status Merchandise (e) Commissioner's Order to Required Permits (f) Return of Domestic Status Merchandise to Customs Territory 5.8 Zone-Restricted Status 5.9 (a) Application for Zone-Restricted Status (b) Merchandise Considered Exported 	565657575758585858
(c) Privileged Foreign Status Binding 5.6 Non-Privileged Foreign Status 5.7 Domestic Status (a) Merchandise Qualifying for Domestic Status (b) Admission Requirements (c) Recordkeeping Requirements (d) Loss of Identity as Domestic Status Merchandise (e) Commissioner's Order to Required Permits (f) Return of Domestic Status Merchandise to Customs Territory 5.8 Zone-Restricted Status 5.9 (a) Application for Zone-Restricted Status (b) Merchandise Considered Exported	5656575758585858
5.6 Non-Privileged Foreign Status 5.7 Domestic Status (a) Merchandise Qualifying for Domestic Status (b) Admission Requirements (c) Recordkeeping Requirements (d) Loss of Identity as Domestic Status Merchandise (e) Commissioner's Order to Required Permits (f) Return of Domestic Status Merchandise to Customs Territory 5.8 Zone-Restricted Status 5.9 (a) Application for Zone-Restricted Status (b) Merchandise Considered Exported	56575758585858
5.7 Domestic Status (a) Merchandise Qualifying for Domestic Status (b) Admission Requirements (c) Recordkeeping Requirements (d) Loss of Identity as Domestic Status Merchandise (e) Commissioner's Order to Required Permits (f) Return of Domestic Status Merchandise to Customs Territory 5.8 Zone-Restricted Status 5.9 (a) Application for Zone-Restricted Status (b) Merchandise Considered Exported	575758585858
 (a) Merchandise Qualifying for Domestic Status (b) Admission Requirements (c) Recordkeeping Requirements (d) Loss of Identity as Domestic Status Merchandise (e) Commissioner's Order to Required Permits (f) Return of Domestic Status Merchandise to Customs Territory 5.8 Zone-Restricted Status 5.9 (a) Application for Zone-Restricted Status (b) Merchandise Considered Exported 	57 58 58 58 58
(b) Admission Requirements (c) Recordkeeping Requirements (d) Loss of Identity as Domestic Status Merchandise (e) Commissioner's Order to Required Permits (f) Return of Domestic Status Merchandise to Customs Territory 5.8 Zone-Restricted Status 5.9 (a) Application for Zone-Restricted Status (b) Merchandise Considered Exported	5758585858
(c) Recordkeeping Requirements	58 58 58 58
(d) Loss of Identity as Domestic Status Merchandise	58 58 58
 (e) Commissioner's Order to Required Permits (f) Return of Domestic Status Merchandise to Customs Territory 5.8 Zone-Restricted Status 5.9 (a) Application for Zone-Restricted Status (b) Merchandise Considered Exported 	58 58
 (f) Return of Domestic Status Merchandise to Customs Territory 5.8 Zone-Restricted Status 5.9 (a) Application for Zone-Restricted Status (b) Merchandise Considered Exported 	58 58
 (f) Return of Domestic Status Merchandise to Customs Territory 5.8 Zone-Restricted Status 5.9 (a) Application for Zone-Restricted Status (b) Merchandise Considered Exported 	58 58
5.9 (a) Application for Zone-Restricted Status	
(b) Merchandise Considered Exported	59
	59
(c) Merchandise Entered Into a Bonded Warehouse	60
(d) Destruction	60
(e) Prohibition Against Manufacture, Processing or Manipulation	60
(f) Transfer to Customs Territory	
ADMISSION OF MERCHANDISE	
	62
6.1 General	
6.1 General	62
6.1 General6.2 Prohibited Merchandise6.3 Conditionally Admissible Merchandise	62 62
 6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 	62 62
 6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 6.4 Time of Admission 	62 62 62
 6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 	62 62 62 63
6.1 General	62 62 62 63
6.1 General	62 62 62 63 63
 6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 6.4 Time of Admission (a) Domestic Status Merchandise 6.5 Non-Zone Status Transactions 	62 62 63 63 63
6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 6.4 Time of Admission (a) Domestic Status Merchandise 6.5 Non-Zone Status Transactions (a) Temporary Deposit for Manipulation (b) Merchandise Transiting Zone	62 62 63 63 63 64
6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 6.4 Time of Admission (a) Domestic Status Merchandise 6.5 Non-Zone Status Transactions (a) Temporary Deposit for Manipulation (b) Merchandise Transiting Zone (c) Temporary Deposit to Complete Documentation 6.6 Arrival and Retention of Merchandise	62 62 63 63 63 64 64
6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 6.4 Time of Admission (a) Domestic Status Merchandise 6.5 Non-Zone Status Transactions (a) Temporary Deposit for Manipulation (b) Merchandise Transiting Zone (c) Temporary Deposit to Complete Documentation 6.6 Arrival and Retention of Merchandise (a) Movements within a CBP Port of Entry	
6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 6.4 Time of Admission (a) Domestic Status Merchandise 6.5 Non-Zone Status Transactions (a) Temporary Deposit for Manipulation (b) Merchandise Transiting Zone (c) Temporary Deposit to Complete Documentation 6.6 Arrival and Retention of Merchandise (a) Movements within a CBP Port of Entry	62 62 63 63 63 64 65 65
6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 6.4 Time of Admission (a) Domestic Status Merchandise 6.5 Non-Zone Status Transactions (a) Temporary Deposit for Manipulation (b) Merchandise Transiting Zone (c) Temporary Deposit to Complete Documentation 6.6 Arrival and Retention of Merchandise (a) Movements within a CBP Port of Entry 6.7 Application for Admission and Permit	
6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 6.4 Time of Admission (a) Domestic Status Merchandise 6.5 Non-Zone Status Transactions (a) Temporary Deposit for Manipulation (b) Merchandise Transiting Zone (c) Temporary Deposit to Complete Documentation 6.6 Arrival and Retention of Merchandise (a) Movements within a CBP Port of Entry 6.7 Application for Admission and Permit (a) Application	
6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 6.4 Time of Admission (a) Domestic Status Merchandise 6.5 Non-Zone Status Transactions (a) Temporary Deposit for Manipulation (b) Merchandise Transiting Zone (c) Temporary Deposit to Complete Documentation 6.6 Arrival and Retention of Merchandise (a) Movements within a CBP Port of Entry 6.7 Application for Admission and Permit (a) Application (b) Statistical Reporting	
6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 6.4 Time of Admission (a) Domestic Status Merchandise 6.5 Non-Zone Status Transactions (a) Temporary Deposit for Manipulation (b) Merchandise Transiting Zone (c) Temporary Deposit to Complete Documentation 6.6 Arrival and Retention of Merchandise (a) Movements within a CBP Port of Entry 6.7 Application for Admission and Permit (a) Application (b) Statistical Reporting (c) Supporting Documents (d) Selectivity Processing, Document Review and Examination (e) Issuance of Permit for Admission	
6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 6.4 Time of Admission (a) Domestic Status Merchandise 6.5 Non-Zone Status Transactions (a) Temporary Deposit for Manipulation (b) Merchandise Transiting Zone (c) Temporary Deposit to Complete Documentation 6.6 Arrival and Retention of Merchandise (a) Movements within a CBP Port of Entry 6.7 Application for Admission and Permit (a) Application (b) Statistical Reporting (c) Supporting Documents (d) Selectivity Processing, Document Review and Examination (e) Issuance of Permit for Admission	
6.1 General 6.2 Prohibited Merchandise 6.3 Conditionally Admissible Merchandise (a) Conditionally Admissible to Zone 6.4 Time of Admission (a) Domestic Status Merchandise 6.5 Non-Zone Status Transactions (a) Temporary Deposit for Manipulation (b) Merchandise Transiting Zone (c) Temporary Deposit to Complete Documentation 6.6 Arrival and Retention of Merchandise (a) Movements within a CBP Port of Entry 6.7 Application for Admission and Permit (a) Application (b) Statistical Reporting (c) Supporting Documents (d) Selectivity Processing, Document Review and Examination	

5.8	Inbond transfer to Zone	73
	(a) Intra-Port Transfer Procedure	73
	(b) Inter-Port Transfer Procedure	73
5.9	Receipt into Zone	74
	(a) Supervision and Receipt Procedures	74
	(b) Liability for Merchandise	
	(c) Recordkeeping upon Admission	75
	(d) Receipt Admission of Domestic Status Merchandise	
5.10	Direct Delivery of Merchandise	77
	(a) Application for Direct Delivery Procedures	
	(b) Direct Delivery Procedure	
5.11	Certificate of Arrival	
	TTB Requirements	
	Drawback upon Admission	
	(a) Manufacturing Drawback	
	(b) Drawback from Continuous CBP Custody	
	(c) Unused or Rejected Merchandise	
	(d) Person Entitled to Claim Drawback	
j. 14	Harbor Maintenance Fee	
ΡI	Chapter 7 ERATOR RECORDKEEPING AND REPORTING RESPONSIBI	LITIES
	ERATOR RECORDKEEPING AND REPORTING RESPONSIBI	
7 .1	ERATOR RECORDKEEPING AND REPORTING RESPONSIBIL	86
'.1 '.2	General	86 86
7.1	General Authority for Operator Recordkeeping Authority for Importer Recordkeeping Authority for Importer Recordkeeping	86 86
7.1 7.2 7.3 7.4	General Authority for Operator Recordkeeping Authority for Importer Recordkeeping CBP Authority to Examine Zone Records	86 86 86
7.1 7.2 7.3 7.4 7.5	General	86 86 86
7.1 7.2 7.3 7.4 7.5	General Authority for Operator Recordkeeping Authority for Importer Recordkeeping CBP Authority to Examine Zone Records Foreign-Trade Zone Board Authority to Examine Zone Records Accessibility and Retention of Records	86 86 86 86
7.1 7.2 7.3 7.4 7.5 7.6	General Authority for Operator Recordkeeping Authority for Importer Recordkeeping CBP Authority to Examine Zone Records Foreign-Trade Zone Board Authority to Examine Zone Records Accessibility and Retention of Records (a) Records Retention Period	86 86 86 86 86
.1 .2 .3 .4 .5 .6	General Authority for Operator Recordkeeping Authority for Importer Recordkeeping CBP Authority to Examine Zone Records Foreign-Trade Zone Board Authority to Examine Zone Records Accessibility and Retention of Records (a) Records Retention Period Safekeeping of Records and Proprietary Information	86 86 86 86 86 87
.1 .2 .3 .4 .5 .6	General Authority for Operator Recordkeeping Authority for Importer Recordkeeping CBP Authority to Examine Zone Records Foreign-Trade Zone Board Authority to Examine Zone Records Accessibility and Retention of Records (a) Records Retention Period Safekeeping of Records and Proprietary Information Inventory Control and Recordkeeping System	86 86 86 86 86 87 87
7.1 7.2 7.3 7.4 7.5 7.6	General	86 86 86 86 87 87 87
7.1 7.2 7.3 7.4 7.5 7.6	General	86 86 86 86 87 87 87
.1 .2 .3 .4 .5 .6	General Authority for Operator Recordkeeping Authority for Importer Recordkeeping CBP Authority to Examine Zone Records Foreign-Trade Zone Board Authority to Examine Zone Records Accessibility and Retention of Records (a) Records Retention Period Safekeeping of Records and Proprietary Information Inventory Control and Recordkeeping System (a) Operator's Procedures Manual (b) Receipt and Admission to Zone (c) Accountability for Zone Merchandise	86 86 86 86 87 87 87 89
7.1 7.2 7.3 7.4 7.5 7.6	General Authority for Operator Recordkeeping Authority for Importer Recordkeeping CBP Authority to Examine Zone Records Foreign-Trade Zone Board Authority to Examine Zone Records Accessibility and Retention of Records (a) Records Retention Period Safekeeping of Records and Proprietary Information Inventory Control and Recordkeeping System (a) Operator's Procedures Manual (b) Receipt and Admission to Zone (c) Accountability for Zone Merchandise (d) Transfer of Merchandise from Zone	86 86 86 86 87 87 87 88 89
.1 .2 .3 .4 .5 .6	General Authority for Operator Recordkeeping Authority for Importer Recordkeeping CBP Authority to Examine Zone Records Foreign-Trade Zone Board Authority to Examine Zone Records Accessibility and Retention of Records (a) Records Retention Period Safekeeping of Records and Proprietary Information Inventory Control and Recordkeeping System (a) Operator's Procedures Manual (b) Receipt and Admission to Zone (c) Accountability for Zone Merchandise (d) Transfer of Merchandise from Zone (e) Physical Inventory	86 86 86 86 87 87 87 89 90
7.1 7.2 7.3 7.4 7.5 7.6	General Authority for Operator Recordkeeping Authority for Importer Recordkeeping CBP Authority to Examine Zone Records Foreign-Trade Zone Board Authority to Examine Zone Records Accessibility and Retention of Records (a) Records Retention Period Safekeeping of Records and Proprietary Information Inventory Control and Recordkeeping System (a) Operator's Procedures Manual (b) Receipt and Admission to Zone (c) Accountability for Zone Merchandise (d) Transfer of Merchandise from Zone (e) Physical Inventory (f) Annual Reconciliation	86 86 86 86 87 87 87 89 90 94
7.1 7.2 7.3 7.4 7.5 7.6	General Authority for Operator Recordkeeping Authority for Importer Recordkeeping CBP Authority to Examine Zone Records Foreign-Trade Zone Board Authority to Examine Zone Records Accessibility and Retention of Records (a) Records Retention Period Safekeeping of Records and Proprietary Information Inventory Control and Recordkeeping System (a) Operator's Procedures Manual (b) Receipt and Admission to Zone (c) Accountability for Zone Merchandise (d) Transfer of Merchandise from Zone (e) Physical Inventory (f) Annual Reconciliation (g) Annual System Review	86 86 86 86 87 87 87 89 90 94 94
7.1 7.2 7.3 7.4 7.5 7.6	General Authority for Operator Recordkeeping Authority for Importer Recordkeeping CBP Authority to Examine Zone Records Foreign-Trade Zone Board Authority to Examine Zone Records Accessibility and Retention of Records (a) Records Retention Period Safekeeping of Records and Proprietary Information Inventory Control and Recordkeeping System (a) Operator's Procedures Manual (b) Receipt and Admission to Zone (c) Accountability for Zone Merchandise (d) Transfer of Merchandise from Zone (e) Physical Inventory (f) Annual Reconciliation (g) Annual System Review (h) Sanctions for Recordkeeping Violations	86 86 86 86 87 87 87 89 90 94 94 94
7.1 7.2 7.3 7.4 7.5 7.6	General	86 86 86 86 87 87 87 89 90 94 94 94 95
7.1 7.2 7.3 7.4 7.5 7.6	General Authority for Operator Recordkeeping Authority for Importer Recordkeeping CBP Authority to Examine Zone Records Foreign-Trade Zone Board Authority to Examine Zone Records Accessibility and Retention of Records (a) Records Retention Period Safekeeping of Records and Proprietary Information Inventory Control and Recordkeeping System (a) Operator's Procedures Manual (b) Receipt and Admission to Zone (c) Accountability for Zone Merchandise (d) Transfer of Merchandise from Zone (e) Physical Inventory (f) Annual Reconciliation (g) Annual System Review (h) Sanctions for Recordkeeping Violations	86 86 86 86 87 87 87 89 90 94 94 95 95

7.11 Record of Foreign-Trade Zone Activity under CBP Permit	98
(a) Blanket Application	
7.12 Reporting of Merchandise Discrepancies	98
7.13 Grantee Responsibilities	99
(a) Accounts and Records	99
(b) Annual Reports	99
(c) Operator Responsibilities to Grantee	99
Chapter 8	
OPERATIONS IN ZONES	
8.1 General	
(a) Operations Involving Nonexempt Merchandise	
(b) Supervision by Operators	
(c) Safekeeping of Merchandise and Records	
(d) Transfer of Title	
(e) Third Party/Toll Processing	
(f) Storage	
8.2 Storage Conditions (a) Sofaty and Sonitory Conditions	
(a) Safety and Sanitary Conditions	
(b) Separate Storage	
8.3 Permit for Manipulation, Processing, or Manufacture	
(a) Application	
(b) Decision on Application	
(c) Revocation of Approval	
(d) Recordkeeping	
8.4 Destruction Under Permit	
(a) Alcoholic Beverages	
(b) Application for Destruction	
(c) Decision on Application	
(d) Supervision	106
(e) Removal from Zone for Destruction	106
(f) Recordkeeping of Destruction	106
8.5 Merchandise Marking Requirements	106
(a) U.S. Origin Marking for Goods Manufactured in a Zone	107
(b) False Country of Origin Marking	107
(c) Copyrighted Articles	107
8.6 Abandonment, Bankruptcy, and Absconding	107
(a) Abandonment	
(b) Insolvency, Absconding, or Arrearage	
(c) Bankruptcy	
8.7 Security of Zone	
(a) CBP Security Requirements	108

(b) Guard Service	108
(c) Cargo Security Standards and Specifications	108
(d) High-Security Storage for Title II Firearms	
(e) Compliance Reviews of Cargo Security	
(f) Full Security Surveys	
(g) Action on Cargo Security Deficiencies	110
(h) Theft Reporting	
(i) List of Persons	110
Chapter 9	
TRANSFER OF MERCHANDISE FROM ZONE	
9.1 General	112
9.2 Temporary Removal	112
(a) Application	
(b) Conditions	
(c) Return to Zone	113
(d) Violation of Conditions	113
9.3 Domestic Status Merchandise	
9.4 Articles Exempt from Entry Requirements	113
9.5 Constructive Transfer	114
(a) Procedure	114
9.6 Entry Procedures	
(a) Place of Filing	114
(b) Administrative Requirements	
(c) Penalties for False or Inaccurate Information	
9.7 Entry for Consumption	
(a) Entry Filing Procedure	
(b) Entry Bond	
(c) Deposit of Duties	
(d) Entry Processing and Release	
(e) Admissibility and Restricted Merchandise	
(f) Merchandise Examination	
(g) Tariff Classification	
(h) Valuation	
(i) Liquidation	
(j) Antidumping and Countervailing Duties	
9.8 Weekly Entries for other than Quota Class Merchandise	
(a) Purpose of Weekly Entries	
(b) Specific Estimate of Quantities and Value	
(c) Filing of Entry Summary	
(e) Exceptions to Other Provisions	
(f) Inventory Relief Point	
(1) Inventory Rener 1 out	120

9.9 Procedures for FDA Weekly Entry Filing	126
9.10 Entries for Bonded Warehouse, TIB, Trade Fair, or U.S. Government	
Importations	127
(a) Bonded Warehouse Entries	127
(b) Entry Under Temporary Importation Bond	128
(c) Trade Fair Entries	128
(d) Entries of U.S. Government Importations	129
9.11 Entries for Diplomatic and Foreign Military Use	129
(a) Regulatory Provisions for Exemption from Duties and Taxes	129
(b) Zone Admission File Records	
(c) Transfer to Other Ports after Entry	130
(d) Refused Merchandise	
9.12 Transfer to another Zone or Port	130
(a) Procedures for QP/WP and QX/WX (CBP In-Bond System)	131
(b) Transfer to Another Zone in Same Port	131
(c) Transfer to Zone Site with Same Operator	132
(d) Transfer to Zone at Another Port	
(e) Transfer to Another Port for Entry for Consumption or	
Warehouse	134
(f) Liability and Payment of Internal Revenue Tax	135
9.13 Transfer from Zone for Exportation	135
(a) Export Statistical Reporting	136
(b) EEI/AES Filing Requirements	
(c) Exportation for Purposes of Other Agency Laws	137
(d) Shipments to Insular Possessions	137
(e) Direct Exportation	
(f) Immediate Exportation	
(g) Indirect Exportation	
(h) Filing of Entry	
(i) Exportation by Mail	139
(j) Supervision and Control of Exports	140
(k) Merchandise Not Laden	140
(l) Relanding of Goods	
(m)Reimportation of Textile Articles Changed in Zone	140
(n) Reimportation of Merchandise Manufactured in Zone	
9.14 Weekly Entry for Transportation or Exportation	
(a) Purpose of Weekly Permit	
(b) Preparation of Application	
(c) Individual Entries on CBP Form 7512	
(d) Weekly Statement of Merchandise Entered	
9.15 Transfers for Vessel or Aircraft Supply	
(a) Entry Procedures	
(b) Eligibility for Duty-Free Entry	
(c) Supply Use vs. Exportation Requirements	
(d) Filing of Withdrawal	145

(e) Bond Requirements	145
(f) Delivery to and Retention on Dock	
(g) Special Permit and Procedure for Fishing Vessels	146
(h) Special Procedures for Cigars and Cigarettes	
(i) Supervision and Cancellation of Bonds	
9.16 Transfer of Zone-Restricted Status Merchandise	147
(a) Entry for Consumption	147
(b) Entry for Warehouse Pending Exportation	
(c) Transfer to Customs Territory	147
(d) Transfer for Other Purposes	148
9.17 Release and Transfer from Zone	147
(a) Liability for Discrepancies	148
(b) Sealing of Conveyances	148
(c) Recordkeeping and Accounting for Transfers	148
(d) Time Limit for Removal from Zone	148
(e) Time of Transfer from Zone	149
9.18 Retention or Return of Zone Merchandise	149
(a) Purposes	150
(b) Exceptions	
(c) Entry into Commerce	150
(d) Actions in Lieu of Entry Cancellations	151
(e) Erroneous Entry	
(f) Readmission in Domestic Status	
9.19 Drawback on Zone Merchandise	151
9.20 NAFTA Duty Deferral Requirements	
(a) Manufacturing Operations	152
Chanton 10	
Chapter 10	
DISCREPANCY REPORTING AND LIABILITY	
10.1 General	155
10.2 Foreign-Trade Zone Operator Responsibility	
10.3 Sealing of Containers and Conveyances	
10.4 Discrepancies Found upon Receipt	
(a) When Operator Becomes Responsible	
(b) Manifest Discrepancy Reporting	
(c) Breakbulk, Conveyance Not Sealed	
(d) Concealed Within-Case Shortage or Discrepancy in Sealed	
Conveyance	157
(e) Conveyance with Seals Not Intact	
(f) Bulk Merchandise	
(g) Overages upon Admission	158
(h) Damage upon Admission	
(i) Temporary Deposit and Direct Delivery Procedures	158

10.5	Discrepancies Found After Admission	158
10.5	(a) Operator Responsibility after Admission	
	(b) Discrepancy Reporting and Recording Requirements	
	(c) Identification of Discrepancy	
	(d) Netting Policy	
	(e) Offsetting Error	
	(f) Shortages Over 1 percent and \$100 Duty	
	(g) Shortages Not Over 1 percent and \$100 Duty	
	(h) Overages Found in Zone	
	(i) Casualty and Similar Losses	
	(j) Damage to Merchandise in Zone	
10.6	Discrepancies Found Upon Transfer to Customs Territory	
10.0	(a) Seal Concurrence	
	(a) Scar Concurrence (b) CBP Physical Supervision	
	(c) When Operator Is Relieved of Responsibility	
10.7	Discrepancy Recordkeeping	
10.7	Discrepancy Recordsceping	103
	Chapter 11	
	PROHIBITIONS, RESTRICTIONS, AND	
	SPECIAL MERCHANDISE PROVISIONS	
		1 - 1
	General	
	Prohibited Merchandise	
	Public Interest Exclusion	
	Limitations to Scope of Authority	
11.5	Prohibitions and Restrictions on Admission	
	(a) Bonded Warehouse Shipments.	
	(b) Grant Restricted Steel Products	
	(c) Motor Vehicles for EPA/DOT Conversion in Zone	
	(d) Non Tax-paid Alcohol and Tobacco Products	
	(e) Firearms and Other Weapons	166
	(f) Merchandise Subject to Antidumping or Countervailing Duties	
	(g) Other Foreign-Trade Zone Grant Restrictions	
11.6	Prohibitions and Restrictions on Zone Activities	
	(a) Residence in Zone	
	(b) Retail Trade	
	(c) Motor Vehicles Owned by Individuals	
	(d) Articles Subject to Internal Revenue Code	
	(e) Steel Products	
	(f) Textiles and Textile products	
	(g) Sugar and Sugar Products	
	(h) Petroleum Products	
	(i) Alcohol/Gasohol	171
	(j) Explosives	171

	(k)	Firearms	171
	(1)	Meat Processing	172
	(m)	Pigments and Printers Ink	172
	(n)	Tires	172
	(o)	Chain Saws	172
	(p)	Golf carts	172
	(q)	Television Tubes	172
	(r)	Pistachios/Nuts	172
	(s)	False Marking of Merchandise	172
	(t)	Zone-Restricted Status Merchandise	173
11.7	Res	trictions on Transfer to Customs Territory	173
	(a)	Entry for Warehouse	
	(b)	Entry under Temporary Importation Bond	173
	(c)	Zone-Restricted Status Merchandise	173
	(d)	Textiles and Textile Products	173
	(e)	Export-Only Restrictions	174
	(f)	Meat Subject to Quota	174
	(g)	Toxic Substances	
	(h)	Motor Vehicles Pending EPA/NHTSA Approval	
	(i)	Sugar and Sugar-Containing Products	
	(j)	Petroleum Products	
	(k)	Alcohol/Gasohol	
		e Limits on Grant	
	-	ce Limits on Grants	
11.10)Spe	cial Merchandise Provisions	
	(a)		
	(b)	Bulk Concentrated Orange Juice	176
		Chapter 12	
		BONDS AND LIQUIDATED DAMAGES	
		BONDS AND ENGODATED DIAMINGES	
12.1	Gen	eral	177
		d Administration	
12.3	Agr	eements of Zone User in Importation and Entry Bond	178
	(a)	Agreements to Pay Duties, Taxes, and Charges	
	` /	Reimbursement and Exoneration of the United States	
	, ,	Agreement on Duty-Free Entries or Withdrawals	
	, ,	Consequences of Default	
12.4		eement of Operator in Operators Bond	179
		Agreement Concerning Receipt Handling and Disposition of	
		chandise	
		Agreement to Pay Duties, Taxes, and Charges	
		Reimbursement and Exoneration of the United States	
	(d)	Consequences of Default	179

12.5	Consequences of Default	179
	(a) Consequences of Merchandise Default	
	(b) Consequences of Non-Merchandise Default	
	(c) CBP Determination of Nature of Default	
12.6	What Constitutes Merchandise Default	180
12.7	What Constitutes Non-Merchandise Default	180
	(a) Application and Examples of Non-merchandise Defaults	180
12.8	What Constitutes Restricted Merchandise	181
12.9	Value of Merchandise	181
12.10	Payment of Duties by Operator on Missing Merchandise	182
12.11	Issuance of Claims of Liquidated Damages	182
	(a) Liquidated Damage Claims	182
	(b) Petitions for Relief	182
	(c) Where Filed	183
	(d) Supplemental Petitions for Relief	183
12.12	Waiver or Cancellation of Claim without Payment	183
	(a) Mitigation or Cancellation	183
	(b) Mandatory Cancellation	183
	(c) Discretionary Cancellation for Merchandise Default	
	(d) Discretionary Waiver for Minor Losses	183
	(e) Discretionary Cancellation for Non-merchandise Default	184
	(f) Discretionary Cancellation (Entry Bond)	184
	(g) Effect of Discretionary Waiver or Cancellation	185
12.13	Prior Disclosure/Voluntary Notification	185
12.14	Cancellation upon Payment of Smaller Amount Operators Bond)	
	(a) Defaults Involving Merchandise	185
	(b) Defaults Not Involving Merchandise	
12.15	Cancellation upon Payment of Smaller Amount (Entry Bond)	
	(a) Defaults Involving Merchandise	
	(b) Defaults Not Involving Merchandise	
	Offers in Compromise	
12.17	Suspension of Activation for Nonpayment of Claim	189
	Chapter 13	
	FOREIGN-TRADE ZONE ENFORCEMENT ACTI	ONC
	FUREIGN-TRADE ZUNE ENFURCEMENT ACTI	ONS
13.1	General	
	Warning Notices	
13.3	Assessment of Liquidated Damages	
13.4	Fines	
	(a)Notice of Fine	
	(b)When Fine Assessed	
	Revocation of Blanket Permit	
13.6	Physical Supervision	191

13.7 New Bond Requirement	191
13.8 Suspension Procedure	192
(a) Grounds for Suspension	192
(b)Show Cause Notice to Operator	193
(c)Notice to Grantee	193
(d)Hearing	193
(e)Decision After Hearing	
(f) Placing Suspension in Effect	194
(g)Extension of Suspension by Board	193
13.9 Suspension of Privilege	194
(a) Notice to Show Cause	194
(b)Effect of Suspension of Privilege	194
13.10 Suspension of Activation	
(a) Effect of Suspension of Activated Status	194
13.11 Revocation of Activated Status or Privilege	
(a)Restoration of Approval	
13.12 Revocation of Grant	
(a) Recommendation of Port Director for Revocation	
(b)Procedure and Decision	
13.13 Removal of Merchandise	196
(a)Procedure	
13.14 Responsibility of Operator after Removal	
13.15 Penalties for Fraud, Gross Negligence, or Negligence	197
(a) Application to Zones	197
(b)Procedural Protections	
(c)Payment of Duties	198
13.16Criminal Violation	
13.17Search, Arrest, and Seizure	201
(a)Search Authority	201
(b)Arrest Authority	201
(c)Seizure Authority.	201
APPENDICES	203
GLOSSARY OF FOREIGN-TRADE ZONE TERMINOLOGY	239
KEV WORD INDEX	244

Chapter 1

INTRODUCTION, PURPOSE, AND INTERPRETATIONS

1.1 Introduction – A Foreign Trade Zone (FTZ) is a secure area under the supervision of the Bureau of Customs and Border Protection (CBP). FTZs are considered to be outside of the Customs territory of the United States for the purposes of payment of duty. The authority for establishing zones is granted by the Foreign Trade Zone Board (FTZ Board), under the Foreign-Trade Zones Act of 1934, (FTZ Act) as amended (19 U.S.C. 81a-81u), and the FTZ Board's regulations (15 C.F.R. 400).

As stated in the FTZ Act, the purpose of establishing FTZs is to "expedite and encourage foreign commerce and other purposes." The FTZ Act is administered in the context of evolving U.S. economic and trade policy and economic factors relating to international competition. The Secretary of Homeland Security assigns CBP personnel to FTZs to protect the revenue and provide for entry of foreign merchandise into the Customs territory from FTZs. FTZS are operated pursuant to public utility principles under the sponsorship of a corporation granted authority by the FTZ Board. Under FTZ procedures, foreign and domestic merchandise may be admitted into FTZs for operations such as storage, exhibition, manipulation, destruction, assembly, manufacture and processing, without being subject to formal CBP entry procedures and payment of duties, unless and until the foreign merchandise enters the Customs territory for domestic consumption. The importer ordinarily has a choice of paying duties either at the rate applicable to the foreign material in its admitted condition, or, if used in manufacturing or processing, at the rate applicable to the emerging product. The FTZ Board can deny or limit the use of zone procedures in specific cases on public interest grounds. Merchandise moved into zones for export (zone-restricted status) may be considered exported for purposes such as federal excise tax rebates and drawback. Foreign merchandise (tangible personal property) admitted to a zone and domestic merchandise held in a zone for exportation are exempt from certain state and local ad valorem taxes (19 U.S.C. 81o(e)). Articles admitted into zones for purposes not specified in the FTZ Act shall be subject to the tariff laws and regular entry procedures, including the payment of applicable duties, taxes, and fees (15 CFR 400.1(c)).

- 1.2 Purpose The purpose of this manual is to set forth in one document all the instructions, guidelines, and other information that CBP personnel and existing and prospective Grantees, Operators and Users need to know for effective operation, supervision, and use of FTZs and FTZ procedures. This manual itself does not have the force of law, and is intended only as a convenient compilation of the referenced material cited in Section 1.3 of this manual, and of CBP policies and procedures included, or previously included, in the CBP Issuance System.
- 1.3 Referenced Material Portions of this manual are taken directly from the FTZ Act, FTZ Board regulations, CBP regulations, and other sources. In other cases, this manual paraphrases the appropriate regulation for clearer understanding. However, in all cases, the applicable reference will be cited accordingly. The reader may wish to review the referenced citation

before taking any definitive action on a particular matter. The references contain abbreviations as follows:

Abbreviation	Reference
FTZ Board	Foreign Trade Zone Board
CBP	U.S. Customs and Border Protection
CD	CBP Directive
CFR	Code of Federal Regulations
C.S.D.	CBP Legal Decision published in CBP Bulletin
FR	Federal Register
FTZ Act	Foreign-Trade Zones Act of 1934, as amended (19 §81a-81u)
FTZM	Foreign-Trade Zone Manual
FTZO	Foreign-Trade Zone Board Order
HQ. Ltr.	Unpublished CBP Headquarters Letter
HTS	Harmonized Tariff Schedule of the U.S.
HQR	Published Headquarters Rulings
MS	CBP Manual Supplement
OCOD	Office of Commercial Operations Decision published in CBP Bulletin
PL	Public Law
Slip Op	Slip Opinion of the Court
TA	Tariff Act of 1930, as amended (Title 19 U.S.C. and 18 U.S.C. §541-552)
TD	Treasury Decision published in CBP Bulletin
U.S.C.	United States Code
UST	United States Treaties in Force

- 1.4 Instructions vs. Guidelines As previously stated in Section 1.2 herein, this manual does not have the force of law and is intended only as a convenient compilation of cited reference materials. Generally, mandatory instructions for CBP personnel, Grantees or Operators are conveyed through use of the word "shall" and are based on requirements of law, regulation, or in the case of CBP personnel, direction from CBP Headquarters. Guidelines are also provided in this manual. Guidelines do not have the force of law; however, they do express CBP preference or an area in which CBP management may exercise some discretion or judgment. These guidelines are conveyed through use of the word "may" or "should," the wording of section headings, or the context of the section or paragraph. While every effort was made to carefully choose the correct words for this manual, it must be understood that distinctions between instructions and guidelines may not be easy to determine. The word "CBP" may still read as "Customs" in other correspondences printed before this manual. All questions concerning what is or is not legally required should be referred to the Office of Field Operations, CBP Headquarters at 202-344-1620.
- 1.5 Port Director Authority Whenever the term "Port Director" appears in this manual, it refers not only to Port Directors themselves, but also to any CBP personnel to whom a Port Director delegates authority to carry out a task. Thus, the term may refer to Port Directors, CBP

Officers (CBPOs), Import Specialists, Entry Specialists, Agricultural Specialists or any other titles.

- 1.6 Customs and Border Protection Issuances The Records and Policy Management Program (RPM) system contains four principal types of documents; Directives, Delegation Orders, Handbooks, and Information Notices. Delegation Orders, Handbooks, and Directives are issued primarily in electronic format on the CBPNET. Information Notices are issued in paper form and are in effect for a limited period of time. The CBP RPM Officer should be contacted for individualized guidance in the preparation of RPM documents before they are submitted for signature. Specific forms used to issue RPM documents are available from the CBP RPM Officer.
- 1.7 Rulings and Disagreements Anyone who desires an interpretation or explanation of any law or regulation administered by CBP may request a ruling on the matter under the procedures in 19 CFR 177. A ruling issued under this procedure represents the official position of CBP in the matter. Any question or disagreement concerning a CBP issuance should be directed to the issuing office. Any disagreement or question concerning a matter involving the judgment or discretion of the Port Director should be directed to the Director, Field Operations, who has oversight responsibility for that port after the matter has been fully discussed with the Port Director.
- 1.8 Disclosure of Information Information in CBP possession affecting Grantees, Operators and Users is subject to the Freedom of Information Act (FOIA) (5 U.S.C. §552). Information exempt from disclosure is listed in 19 CFR 103.12, which includes trade secrets, confidential financial and business information and material involving personal privacy.
- 1.9 Local CBP Procedures This manual contains and describes instructions, guidelines, and procedures authorized by CBP for use throughout the Customs territory. Ports may adopt additional service-wide procedures which are within the confines of the regulations and are not covered by the manual. If local procedures conflict with service-wide procedures and/or this manual, Operators and Users are invited to identify the conflict to CBP Headquarters, Office of Field Operations, for resolution, or request that the port resolve the matter through an Internal Advice procedure under 19 CFR 177.11. Users are also encouraged to confer with their Grantees and/or Operators to ensure that issues of conflict are not simple misunderstandings.
- 1.10 Automated Procedures Certain activity involving zone operations may utilize federal agency automated procedures. CBP entries for merchandise transferred to the Customs territory may utilize the CBP electronic systems. The CBP Form (CBPF) 7501 may be utilized for CBP entry with the entry being treated as a live entry (entry summary filed at the time of entry with estimated duties attached or electronic payment). CBP encourages Operators and Users to use the electronic 214 (Application for Foreign Trade Zone Admission and or Designation) (e214) process which is effective in reducing unnecessary paperwork, simplifying procedures, and reducing the need for direct CBP supervision. However, Operators and Users must understand that any procedures may be superseded by future automation efforts developed by CBP.

Chapter 2

LAWS, REGULATIONS, ORGANIZATION, DUTIES, AND GENERAL INFORMATION CONCERNING FOREIGN-TRADE ZONES

- 2.1 Laws and Regulations Zones operate pursuant to the FTZ Act and two sets of regulations; the FTZ Board regulations (15 CFR 400) and CBP regulations (19 CFR 146). They are also affected by other laws and regulations. This section deals with the interplay among the various laws and regulations and how they are applied in zones. Whenever the term "zone or subzone" is used in this manual, it is understood to include subzones, unless the context indicates otherwise.
 - (a) Foreign-Trade Zones Act of 1934 (FTZ Act) The principal statute governing zones which has been codified in the United States Code (U.S.C.) as 19 U.S.C. §81a-81u. The FTZ Act has been periodically amended and covers the establishment and administration of zones, how they are administered, and what may and may not be done in them. Every Grantee and Operator, and most Users, should have a copy of the FTZ Act, updated with recent amendments.
 - (b) FTZ Board regulations Board regulations provide guidelines and procedures on the establishment and modification of zones; their management and administration by Grantees and Operators; and how the FTZ Board conducts its business.
 - (c) CBP Laws and Regulations CBP laws are codified in 19 U.S.C., including the Harmonized Tariff Schedule of the United States. Merchandise of every description, except that prohibited by law, may be brought into a zone and stored or processed there under certain circumstances without being subject to CBP laws of the United States (19 U.S.C. §81c(a)). The FTZ Act itself is a CBP law, since it is codified in Title 19. CBP laws also include 18 U.S.C. §541-552 since they were included in the original Tariff Act of 1930, most of which was codified in Title 19.
 - (1) Applicability of CBP Laws Except as otherwise provided, CBP laws are not usually applicable in zones. However, they are applicable to merchandise passing through the Customs territory before admission to, or after transfer from, zones. CBP laws are also applicable to prohibited merchandise in a zone (see Section 6.2 FTZM), vessels and aircraft entering or leaving an FTZ (see Section 2.5 FTZM), and merchandise which has been brought into an FTZ for purposes not specified in the FTZ Act (15 CFR 400.1(c)).
 - (2) Significance of CBP Regulations Pursuant to the general authority to establish regulations necessary for the protection of the revenue and to ensure compliance with the FTZ Act, regulations have been established in 19 CFR 146, requiring Operators to maintain certain records and make certain reports (See 19 U.S.C. §66, 81h, 81p, and 1624).

- Other Federal Laws Since the FTZ Act specifically excludes, under certain circumstances, only the application of CBP laws, most other federal laws are applicable in zones, such as those affecting public health, immigration, labor, welfare, and income tax. Many laws and regulations governing exportation which are enforced by CBP are applicable in zones, such as those in 15 CFR 772 and in 22 CFR 121 to 22 CFR 130 (see Section 9.12(2)(b) FTZM). Furthermore, various federal regulations may be applicable to FTZs, many of them dealing with specific kinds of merchandise or activities concerning merchandise in zones. The FTZ Board shall cooperate with CBP and such other federal agencies that have jurisdiction in ports of entry (19 U.S.C. §81i and 15 CFR 400.12(e)). All FTZ participants should carefully consider the impact of federal agency laws. Federal agencies with the most direct involvement with FTZs include Alcohol & Tobacco Tax and Trade Bureau (TTB); Federal Communications Commission (FCC); Environmental Protection Agency (EPA); Food and Drug Administration (FDA); Department of Agriculture (USDA); and Fish and Wildlife Service (USF&W). All such agencies normally have specific regulations, guidelines, etc., that directly relate to FTZ activity.
- (e) State and Local Laws and Regulations Generally, state and local laws are applicable in zones, except to the extent they are preempted by federal laws or the constitution. The FTZ Board is directed to cooperate with the state, subdivision, and municipality in which an FTZ is located in the exercise of their police, sanitary, and other powers in connection with the FTZ (19 U.S.C. §81i and 15 CFR 400.41).
 - (1) Commerce Clause of the Constitution The Constitution reserves to Congress the power to regulate commerce with foreign nations (Article I, Section 8, Clause 3, U.S. Constitution) and prohibits states from levying duties on imports and exports (Article 1, Section 10, Clause 2, U.S. Constitution). State laws and regulations are not applicable in zones to the extent that they would contravene the Constitution and federal laws under the Constitution. Federal law may implicitly preempt state law to the extent that state law conflicts with federal regulations. To the extent that a state law would encumber ease of legal transshipment through a zone by requiring unnecessary regulation of goods in which the state has no interest, the law frustrates the goal of the FTZ Act to facilitate use of United States ports for the legal transshipment of goods in foreign commerce [3M Health Care, Ltd. v. Grant, 908 F.2d 918 (11th Cir. 1990)].
 - (2) State and Local Ad Valorem Taxes The FTZ Act specifically prohibits state and local ad valorem taxation of imported tangible personal property stored or processed in a zone, or of tangible personal property produced in the U.S. and held in a zone for exportation in either its original or its processed form (19 §81o(e); Deer Park Independent School District v. Harris County Appraisal Dist., 132 F. 3d 1095 (5th Cir. 1998) cert denied, 118 S. Ct. 2343, 141 L. Ed 2d 714).
 - (3) State Tax Laws Some states have specific legislation conferring certain benefits on Users.

- (4) Ad Valorem Personal Property Taxes The FTZ Act provides an exemption for certain merchandise held in an FTZ. Some states have other laws that may apply to the merchandise.
- (5) Enabling Legislation FTZ Board regulations require that all states have specific legislation conferring authority to submit applications to the FTZ Board. As part of its review, the FTZ Board will analyze the legislation and specific state-by-state requirements to ensure that the organization has the legal authority from the state to submit an application to the FTZ Board (15 CFR 400.22(b) and (c)).
- (f) Grantee Schedules A zone shall be operated, maintained, and administered by the Grantee in accordance with the schedule fixed by the Grantee and reviewed and accepted by the board prior to the start of operations. The schedule shall be available for review at the offices of the Grantee and Operator. All persons entering the zone, for any reason whatsoever, are bound by the schedule in addition to applicable government regulations. Specifications for Grantee schedules are set forth in 15 CFR 400.42(b).
- 2.2 The FTZ Board The FTZ Board is responsible for the establishment, maintenance, and administration of zones under the FTZ Act. In addition, any questions regarding the specific scope of authority in terms of products, components and activities allowed are determined by the FTZ Board. CBP is responsible for carrying out the Tariff Act and various other laws and regulations in respect to zones. This section identifies the duties and responsibilities of the members of the FTZ Board and the role of CBP in FTZs.
 - (a) Makeup and Authority of the FTZ Board The FTZ Board consists of the Secretary of Commerce, who is chairman and executive officer and the Secretary of the Treasury (19 U.S.C. §81a(b)). As set forth in 15 CFR 400.11 the FTZ Board has authority to:
 - (1) Prescribe rules and regulations concerning zones;
 - (2) Issue grants of authority for FTZs and subzones, and approve modifications to the original zone project;
 - (3) Approve manufacturing and processing activity in FTZs and subzones as described in subpart D of 15 CFR 400;
 - (4) Make determinations on matters requiring Board decisions pursuant to 15 CFR 400;
 - (5) Decide appeals in regard to certain decisions of the Commerce Department's Assistant Secretary for Import Administration or the Executive Secretary;

- (6) Inspect the premises, operations and accounts of Grantees and Operators;
- (7) Require Grantees to report on FTZ operations;
- (8) Report annually to the Congress on zone operations;
- (9) Restrict or prohibit zone operations;
- (10) Impose fines for violations of the FTZ Act and this part;
- (11) Revoke grants of authority for cause; and,
- (12) Determine, as appropriate, whether zone activity is or would be in the public interest or detrimental to the public interest.
- (b) Presidential Direction to Furnish Records The President may direct federal agencies to cooperate with the FTZ Board to facilitate FTZ Board investigations and duties in the establishment, operation, and maintenance of FTZs (19 U.S.C. §81j). Through Executive Order 7104, the President did direct that federal agencies furnish records and detail personnel as may be required by the FTZ Board.
- (c) Committee of Alternates The Secretaries of Commerce and Treasury have rarely, if ever, convened in person to conduct FTZ Board business. Instead, they are authorized to appoint alternates authorized to act in an official capacity for them (15 CFR 400.11(c)). The secretaries or their alternates meet periodically as necessary to consider policy issues involving the FTZ Board. However, routine official FTZ Board business is conducted by memoranda and other correspondence circulated among the alternates.
- (d) Executive Secretary The Executive Secretary of the FTZ Board is appointed by the Secretary of Commerce, as chairman of the FTZ Board, to be the chief operating official of the Board. The principal duties of the Executive Secretary are set forth in 15 CFR 400.12.
- (e) CBP Role CBP carries out most of the duties and responsibilities of the Secretary of the Treasury, under the FTZ Act. This includes the preparation and implementation of the rules and regulations approved by the Secretary of the Treasury under 19 U.S.C. §81c, e, and o (b) of the FTZ Act and the assignment of the necessary CBP personnel to perform zone related work under 19 U.S.C. §81d. However, CBP does not represent the Secretary of the Treasury in any matter pertaining to zones which is (1) of a policy-making nature; or (2) within the authority of another agency of the Department of the Treasury, for which authority has not been delegated to CBP, such as TTB or the Internal Revenue Service (IRS).

- (1) Principal CBP Concern The principal interest and concern of CBP in zones is control of merchandise moving to and from the zone, the protection of the revenue, and to ensure that zone procedures are in compliance with the FTZ Act and all laws and regulations pertaining to zone use.
- (2) Port Director as Representative of the FTZ Board The Port Director is the local representative of the FTZ Board. The Port Director may call upon local representatives of other government agencies for advice in matters pertaining to the operation, maintenance, and administration of zones (15 CFR 400.41 and 19 CFR 146.2).
 - (a) Duties as Representative The Port Director is responsible for conducting general oversight of the zone, its processes and procedures and the activities of the Grantee, Operator and Users, and to report exceptions to the FTZ Board. The Port Director will consult with and make recommendations to the FTZ Board on any zone boundary modification upon request of the Grantee or the FTZ Board staff (15 CFR 400.26(a)(2) and Section 4.4(a) FTZM). CBP's approval is always subject to activation conditions, and shall be so stated in the Port Director's comments. The Port Director has no enforcement authority as local representative of the FTZ Board beyond that conferred by the FTZ Board or FTZ Board Order. However, the FTZ Board usually relies on the Port Director's advice on local zone matters (C.D. 089-3210-025).
- (3) CBP Neutrality Policy Consistent with its role as a law enforcement agency, it is CBP's policy to neither encourage nor discourage the establishment or the use of zones. CBP personnel are not authorized to advise any party whether to use one zone in preference to another, or to use a zone at all instead of another facility or administrative mechanism in the Customs territory.
- (f) Other Agencies Since the FTZ Board and other federal agencies are directed to cooperate with one another (19 U.S.C. §81i and 19 U.S.C. §81j), Operators and Users may expect to deal with a number of agencies, other than CBP, concerning merchandise or activities in the zone falling within their jurisdiction.
- 2.3 Grantee, Operator, and User Duties and Responsibilities The FTZ Board does not own or operate any zones; rather, it provides grants of authority to applicants to establish, operate, and maintain zones. The Grantee may execute a contract with another party, usually in the private sector, for the operational management of the zone. Various tenants may lease space or construct buildings in the zone and physically manage their operations in the buildings, while others may pay an Operator a fee for handling their merchandise and performing related services.
 - (a) Grantees A Grantee is a public or private corporation, as defined in 15 CFR 400.2, to which the privilege of establishing, operating, or maintaining a zone project has been given. The principal responsibilities of a Grantee are to:

- (1) Provide and maintain facilities in connection with a zone according to the provisions of 19 U.S.C. §81(b);
- (2) Operate the zone as a public utility with fair and reasonable rates and charges for all zone services and privileges, and afford to all who apply for the use of the zone and its facilities and provide uniform treatment under like conditions, in accordance with 19 U.S.C. §81n and 15 CFR 400.42(b);
- (3) Make annual reports (and at other such time as it may prescribe) to the FTZ Board containing such information as the FTZ Board may require (19 U.S.C. §81p (b) and 15 CFR 400.46(d));
- (4) Maintain books, records, and accounts in accordance with the provisions of 15 CFR 400.46(a), (b);
- (5) Apply to the FTZ Board for a grant of authority to establish a subzone or to expand or otherwise modify its zone project (15 CFR 400.22(d), 400.26(a)(1));
- (6) Permit the erection of buildings necessary to carry out approved zone projects in accordance with 19 U.S.C. §81m and 15 CFR 400.28(a)(6);
- (7) Operate, maintain, and administer the zone project under the FTZ Act, and laws and regulations administered by CBP for other agencies or administered directly by other agencies, and the schedules of rates and charges made and fixed by the Grantee, in accordance with 15 CFR 400.41 and 15 CFR 400.42(b)(1);
- (8) Make written application to the Port Director for approval of a new Operator, pursuant to 19 CFR 146.7(e), (f);
- (9) If acting as the Operator makes application or must provide concurrence to a request for activation, de-activation or reactivation.
- (b) Operators An Operator is a corporation, partnership, or person that operates a FTZ or FTZ subzone under the terms of an agreement with the Grantee (19 CFR 400.2(q)). For CBP purposes, the term "Operator" includes a Grantee that operates its own zone (19 CFR 146.1(b)). Activation of a zone requires the execution of an Operator's bond on a CBPF 301, which contains the bond conditions required by 19 CFR 113.73 (19 CFR 146.6(d) and (e)). The bond insures the principal's agreement to comply with the pertinent laws and CBP regulations and delineates the particular responsibilities of the Operator.
- (c) Users A User is a corporation, partnership, or person that uses a zone under agreement with the Grantee or Operator for storage, handling, processing, or manufacturing of merchandise in zone status, whether foreign or domestic (15 CFR

- 400.2(t) and 19 CFR 146.1(b)). Usually, the User is the entity which requests a permit to admit, process, or remove zone status merchandise. In subzones, the Operator and User are usually the same entity. They may be permitted by the Grantee to construct their own buildings or structures in which they conduct their own business, according to the provisions of 19 U.S.C. §81m and 15 CFR 400.28(a)(6). They are sometimes referred to as "tenants", but the more inclusive term "Users" is used in this manual.
- 2.4 Subzones An FTZ subzone (subzone) is a special-purpose FTZ established as part of a zone project for a limited purpose that cannot be accommodated within an existing FTZ (19 CFR 146.1(b)(17) and 15 CFR 400.2(n)). A subzone may be authorized if the Board finds that the operation cannot be served in the existing zone project and meets the required public interest test in 15 CFR 400.23(b). Subzones were originally authorized in 1952 under FTZO 29, and the courts have approved subzone designations (See, e.g., <u>Armco Steel v. Stans</u>, 431 F2d 779 (2d Cir. 1970).
- 2.5 Vessels and Other Carriers Vessels and carriers entering and leaving a zone are subject to the laws of the United States, except as otherwise provided in the FTZ Act. Vessels leaving a zone and arriving in the Customs territory are subject to regulation pursuant to 19 CFR 4 to protect the revenue and carry out laws concerning navigation. Foreign-flag vessels are not permitted to carry goods shipped from one zone to another zone or port or points within the zone in the protected U.S. coastwise trade (19 U.S.C. §81e and 19 CFR 146.12(b)). Through the operation of 19 U.S.C. 1644a, the foregoing vessel provisions are applicable also to aircraft entering and leaving a zone. The docking, loading, and unloading facilities of a zone are intended primarily for the use of vessels, aircraft, and vehicles loading or unloading zone merchandise. Their use for other purposes may be terminated if found to endanger the revenue or impede the primary use of the facility (19 CFR 146.12(a)).
 - (a) Movement of Zone Merchandise Imported merchandise, including that imported in mail shipments, being transported to and from zones through the Customs territory must be transported in-bond under the provisions of 19 CFR 18, 19 CFR 112, 19 CFR 125, 19 CFR 146.11 and 19 CFR 144.37(g). Domestic goods transported to a zone or goods removed from a zone comprised wholly of domestic status products need not be transported in-bond.
- 2.6 Supplies Used in Zone Supplies brought into a zone for use in a manner other than those uses enumerated in 19 U.S.C. §81c(a) are not entitled to exemption from applicable duties, taxes and fees under the FTZ ACT, and therefore do not have zone status. Such supplies may include, but are not limited to, office furniture, office machines, office equipment, food to be eaten in the zone, and water and fuel which do not become part of a zone product (except certain fuel noted in Section 2.6(a) FTZM which is exempt under the FTZ Act).
 - (a) Receipt of Supplies If such supplies are imported, they cannot be brought into a zone without first being entered for consumption. Supplies which are produced in a zone from imported material and consumed in the zone are considered by CBP to have lost their exemption from duty under the FTZ Act, and must be entered for consumption,

except for certain petroleum refinery fuels covered by the fact situation cited in <u>Hawaiian Independent Refinery v. United States</u>, 460 F. Supp. 1249 (Cust. Ct. 1978) (See C.S.D. 79-418 and Section 11.6(j) FTZM).

- (b) Use If imported merchandise in stock is withdrawn from stock for a use which causes it to lose its exemption from duty under the FTZ Act, the article must be entered for consumption. An example of this use would be a zone wholesaler dealing in office furniture takes an office desk from stock and uses it in its own office operation in the zone. What uses in a zone cause merchandise to be subject to, or exempt from, duty under the FTZ Act are determined from the particular facts in a situation. Borderline issues that may be raised concerning zone use should be referred to the Office of International Trade, for a ruling under the procedures in 19 CFR 177.
- 2.7 Production Equipment 19 U.S.C. §81c(e) allows Users to defer duty on production equipment that is intended for use in zones. The Board has authorized this activity. The Board allows a User to admit equipment into the zone and assemble the equipment prior to entry for consumption. Payment of duty would be deferred until such equipment goes into use as production equipment as part of zone activity, at which time the equipment is entered for consumption as equipment. Alternatively, production equipment that is not for use within a zone may continue to be treated as normal merchandise. Questions regarding the definition of production equipment shall be directed to the Board staff.
- 2.8 CBP Forms - CBP forms unique to zones are the CBPF 214 and CBPF 216 (Application for Foreign Trade Zone Activity Permit). Their uses are described in Parts 6 and 8 and the Appendix to the FTZM. These forms are authorized to be printed by Grantees and Operators, brokers, importers, and other parties using zones, in a format which complies with the forms as approved by the Office of Management and Budget and on a quality of paper commensurate with that used by CBP. The letterhead of the Grantee or Operator, in its own design, may appear in the space designated in these forms for zone location and address. The forms may be printed in color-coded copies, in multiple carbon copies, or as continuous forms for automated use, so long as they conform to the format of the approved forms. Continuation sheets are also authorized to be printed, so long as the header information and the format of the sheet clearly identify the continued information. Operators and Users are authorized to use computer-generated forms, as long as the forms comply with the provisions listed above. Additional authorization must be obtained from The Bureau of the Census (Census) to ensure that any computer-generated information meets the requirements of that agency. CBP prints and maintains a small stock of zone forms for occasional use by the public at its port offices and as a starter supply for newly-established zones for copying and reproduction purposes. A camera ready copy of CBP forms, e.g., CBPF 214, CBPF 214A, CBPF 216, etc., may be obtained upon request from the following by mail:

U.S. Customs and Border Protection Forms and Records Management Section Informational Services Branch Asset Management Office of Administration 1300 Pennsylvania Avenue, NW Washington, D.C. 20229 http://www.CBP.gov

Note: The camera copy will provide the User with the current version of the requested CBP Form and ensure clarity in reproduction and copying.

2.9 Place of Filing and Merchandise Examination - It is CBP policy that all admission and entry documentation, and all other applications for permits, for merchandise destined to, in, or being transferred from a zone will be presented to CBP at a location within a port of entry, even if the zone is outside the port of entry. Such documentation may be delivered to CBP by brokers, messengers, couriers, electronic transmission, facsimile or any other means that is administratively acceptable and within the law and regulations (See e.g., C.S.D. 87-2). It is also CBP policy that, if merchandise is to be physically examined or inspected before admission to a zone or upon transfer from a zone, the examination or inspection, with the exceptions noted below, will be conducted only within a port of entry. CBP may choose to examine zone merchandise within the zone (See Sections 6.7(d), 9.6(a), and 9.7(f) FTZM). Otherwise, a bonded carrier or cartman will be required to deliver the merchandise to a location designated by CBP for the examination of the merchandise. 19 CFR 112.2 governs procedures by which Operators can move merchandise without utilizing a carrier or cartman.

The port of entry covered by this policy will normally be the port of entry having supervisory responsibility for the zone where the merchandise is destined or located. However, at the discretion of the affected Port Directors, the documentation or merchandise may be delivered to a different port of entry for processing or examination.

- (a) Exceptions Zone merchandise may be examined or inspected outside a port of entry when:
 - (1) it is in the interest of CBP;
 - (2) proper examination or inspection requires the use of special facilities, or equipment which the Port Director is satisfied cannot be made available within a port of entry;
 - (3) the merchandise is incapable of being delivered to the port of entry, e.g., oversized article or component; or,
 - (4) there is a concurrence, under unusual and isolated circumstances, by the Assistant Commissioner, Office of Field Operations, in such an examination or inspection.
- (b) Delivery Procedures Merchandise covered by this policy will be delivered to and from the examination or inspection site by a bonded carrier or licensed cartman under the

procedures specified in 19 CFR 18 and 19 CFR 125. Operators who carry their own merchandise may be designated as private carriers under the procedures of 19 CFR 112, Subpart B. Any qualifying Operator may be licensed as a cartman under the procedures of 19 CFR 112, Subparts C and D. As set forth in 19 CFR 112.2(b), Operators may transport merchandise to their zone from anywhere within the geographic boundaries of the port where the zone is located, under their respective bonds and without the necessity of obtaining a cartman's license.

Chapter 3

SUPERVISION OF FOREIGN-TRADE ZONES, GRANTEE AND OPERATOR RESPONSIBILITIES AND CBP FEES

- 3.1 General The Commissioner, through authority delegated from the Secretary of the Treasury, assigns the necessary CBP personnel to protect the revenue and to provide for the entry of foreign merchandise into the Customs territory (See 19 U.S.C. §81d). The cost of maintaining the additional CBP service required under the FTZ Act will be paid by the Operator of the zone (19 U.S.C. §81n). However, see Section 3.11(a) FTZM concerning the suspension of zone fees and charges. For many years, CBP protected the revenue in zones through physical supervision of all merchandise receipts and deliveries to and from the zone, and of all zone processing operations. Zones and subzones are now managed by CBP under an audit-inspection method which means that CBP does not maintain inventory records and CBP personnel are not routinely physically assigned to the premises (TD 86-16).
- 3.2 Audit-Inspection Supervision The Operator exercises physical supervision of zones, while CBP exercises general supervision through the audit-inspection program. Audit-inspection is a comprehensive program based on the following 6 principles:
 - (a) Determination by CBP of the identity and nature of merchandise before or upon its deposit in the zone so that the initial responsibility of the Operator for the merchandise can be reliably determined;
 - (b) Issuance of a prior permit by CBP to the Operator for receipt, delivery from the zone, and any processing in the zone;
 - (c) Assumption of responsibility by the Operator for the merchandise in the zone, including physical supervision, security, recordkeeping, storage conditions, and processing (quantities of merchandise received or delivered are determined jointly by the Operator and the delivering or receiving carrier);
 - (d) Performance by CBP of audits and compliance reviews to verify whether the Operator is properly supervising the zone and maintaining the records in compliance with the laws and regulations;
 - (e) Adequate bonding of the Operator and assessment of liquidated damages by CBP to assure compliance with the requirements for proper supervision by the Operator; and
 - (f) Authority for CBP to suspend activated status for a period not to exceed 90 days, or by Board order, to continue suspension of activated status of any zone operation that cannot or does not comply with the laws and regulations for zones (19 CFR 146.82). The Port Director may at any time recommend to the Board that the grant of authority be revoked for willful and repeated violations of the FTZ Act (19 CFR 146.83).

- 3.3 CBP Supervision The extent and character of CBP supervision of zones and zone transactions shall be in accordance with 19 U.S.C. §81a-u, 19 CFR 1646a, 19 CFR 161.2, and 19 CFR 146. The Port Director may detail, assign or request CBP personnel to supervise any transaction or procedure at a zone. CBP supervision may be exercised through but not limited to audits of records, quantity counts of merchandise in inventory, compliance reviews of selected transactions or procedures, or reviews of record-keeping, security, or storage conditions (19 CFR 146.3).
 - (a) Physical Supervision The Port Director may order the physical supervision by CBP personnel of any merchandise transaction in a zone regarding zone status merchandise for which a CBP permit is required, (19 CFR 146.3(b)), in the same manner as such supervision was performed prior to the inception of the audit-inspection program under TD 86-16 (see Section 3.1 FTZM). However, supervision is usually conducted through the audit-inspection method (See Sections 10.4(e) and 13.6 FTZM).
 - (b) Security Service Although the Operator may contract for guard service to provide security service at zones (19 CFR 146.4(g)) this does not limit the authority of the Port Director to assign CBP personnel to the zone to protect the revenue.
 - (c) Access to Zones - The Operator must permit CBP personnel access to the zone (19 CFR 146.4(b)). CBP personnel will properly identify themselves to company security personnel at the company's initial security point. To permit proper access, the Operator must furnish CBP personnel with special equipment (and, where applicable, the personnel to properly use the equipment) for conducting compliance reviews or audits, such as weighing, gauging, and measuring equipment; protective clothing (for cold storage, hazardous substances, safety hardhat, etc.); and provide access to available computers and peripheral equipment. Providing proper access also includes the explanation by the Operator of procedures and transactions in the Operators or Users inventory and recordkeeping system. The Operator should cooperate in any compliance review, audit, or other lawful CBP action, including the location of records and merchandise, so as to achieve its prompt completion and assure a rapid return to normal zone operation. CBP should respect Operator and User security health, safety, and union work rule procedures, to the extent that they do not interfere with a proper audit or compliance review of the zone. At the discretion of CBP personnel, zone personnel may accompany CBP personnel during a compliance review.
- 3.4 Audits An audit is a systematic and thorough check of the Operators (and sometimes a Users) inventory and financial records against zone lot numbers (see Section 6.7(a)(4) FTZM) or unique identification number (UINs) and against the actual quantity of merchandise in the zone. Audits are conducted by non-uniformed CBP regulatory auditors and are conducted less frequently than compliance reviews (See Section 3.5 FTZM). Their frequency is closely held information, which is not disclosed by CBP. Audits are much more intense and of longer duration than compliance reviews, and may generally be expected to last from a week to several months, depending on the size of the inventory, initial problems disclosed, kinds of merchandise in the zone, and many other factors. Auditors may be expected to look at all kinds of zone

transactions, conditions, and requirements, for not only the current year but also for past years. For companies using lot systems, auditors may request access to closed as well as open lot files.

- (a) Notification and Arrival Auditors, in contrast to CBPOs conducting compliance reviews, provide advance notification of an audit, except when it is in CBP interest to conduct an unannounced audit. Upon arrival, auditors shall announce themselves to the appointed representative of the Operator or Users. The Operator or User may request appropriate government identification of the auditors.
- (b) Closeout Auditors shall provide a closeout interview with the Operator and User to discuss their findings. Their findings are tentative, since they must be reviewed by Field Regulatory Audit management before presentation to the Port Director. The Port Director, not the auditor, takes management action on the audit findings.
- 3.5 Compliance Reviews Port Directors are required to conduct compliance reviews based on the port's risk assessment of the individual zone, as outlined in the FTZ Compliance Review Handbook, HB 3500-10, dated July 2008 (19 CFR 146.3(b)). A compliance review is a visit by one or more CBP personnel (usually uniformed personnel), from a port office to physically observe or examine transactions, records, procedures, or conditions in the zone. The purposes of compliance reviews are to (1) determine whether the Operator and Users are in compliance with applicable laws, regulations, and CBP procedures; (2) provide CBP auditors with information in planning and conducting audits of the zone records; and (3) to determine the risk associated with each zone within the ports' jurisdiction. CBP personnel shall not conduct audits or compliance reviews under unsafe conditions, e.g., where there are tottering stacks of merchandise or damaged and exposed containers of hazardous substances, until the deficiency has been corrected (19 CFR 146.4 (f)).
 - (a) Notification and Arrival Compliance reviews are made without advance notification to the Operator or User except when prior notification is in the interest of CBP. However, the personnel shall announce their arrival to the manager or person in charge at the zone and at the User's premises before the compliance review is started. Operators and Users may request identification by badge number from uniformed personnel and by U.S. Government identification card from non-uniformed personnel. The personnel may invite the manager or person in charge to have a representative present during the compliance review, but they will not be required to have a representative present, except as needed to locate records or merchandise, move merchandise, operate special equipment, or otherwise provide access to the merchandise and records. CBP personnel should seek to minimize disruption of the day-to-day operations of the zone (FTZ Compliance Review Handbook, HB 3500-10, dated July 2008, page 16).
 - (b) Duration, Frequency, and Conduct A compliance review will generally be limited to a day or two, to allow the widest possible CBP presence at zones. However, a compliance review may be extended in duration, at the discretion of the Port Director, for a further check on the basis of initial findings. The frequency of compliance reviews will

depend on CBP assessment of the risk represented by the zone. Risk is the degree of exposure to the chance of non-compliance that would result in loss or injury to trade, industry or the public (FTZ Compliance Review Handbook, HB 3500-10, dated July 2008, pages 12-14.

The following four steps should be applied to the risk management process:

Step 1: Collect Data and Information

Step 2: Analyze and Assess Risk

Step 3: Prescribe Action

Step 4: Track and Report

Based on the ports risk assessment, the annual minimum frequency is: High - not fewer than three compliance reviews per year; Medium - not fewer than two compliance reviews per year; and Low – for low risk manufacturing subzones, schedule at least one compliance review every two years. CBP personnel will usually conduct a quantity count of selected zone lots or inventory categories of merchandise (UINs), and reconcile the quantity count with the quantity reported in records in the zone admission file and/or the Operators inventory and recordkeeping system. However, they may at times focus compliance review efforts on security, safety, housekeeping, receipt or delivery procedures, deletion of merchandise from zone inventories, or other conditions or transactions considered significant in their assessment of the zone's risk. Compliance reviews in manufacturing subzones generally focus on the Operators records and de-emphasizes quantity counts of physical inventory.

- (c) Completion and Closeout If CBP personnel are unable to reconcile a discrepancy between their quantity count, and the balance in the Operators inventory recordkeeping system, they shall give the Operator (and User, if applicable) an opportunity to resolve the discrepancy before the compliance review is completed. At the end of the compliance review, the personnel shall conduct a closeout interview with the person in charge at the zone facility to relate their findings. In some cases, the personnel may issue an oral warning to correct a minor violation. Their findings are tentative, since they must be reviewed by port management before any specific action is taken. The person in charge may provide comment to port management during the closeout interview or later to explain apparent violations or to seek to avoid a claim for liquidated damages. CBP personnel shall make follow-up visits to the zone to determine whether minor violations have been corrected or to verify assertions made by the Operator (FTZ Compliance Review Handbook, HB 3500-10, dated July 2008, and pages 18-19).
- (d) Best Practices The Customs-Trade Partnership Against Terrorism program is considered for the entire supply chain of all shipments to the final destination in the United States. CBP recognizes the role zones play in enhancing security and consider it to be a best practice when the supply chain ends with the in-bond shipment arriving at a

zone due to the safeguards that are in place for checking seals and other security measures.

- (e) Best Practices To combat unlawful employment, the Bureau of Immigration and Customs Enforcement (ICE) has introduced the ICE Mutual Agreement between Government and Employers). This voluntary partnership initiative applies to companies in FTZs just as it does to the public at large. For details, see www.ice.gov/partners/opaimage."
- 3.6 Coordination of Audits and Compliance Reviews Audits and compliance reviews are separate and independent, but mutually supporting, methods of verifying compliance by the Operator's with the laws, regulations, and procedures.
- 3.7 Monitoring Foreign-Trade Zone Regulations The Port Director, may inspect zone operations for the purpose of informing the Board whether the provisions of the FTZ Act are being met (15 CFR 400.41). The Operator shall comply with the Port Director 's requirements, and make available such records and reports as are necessary for a thorough and proper inspection of zone activities (15 CFR 400.46(a), (b)). The Grantee shall act in compliance with the Board Regulations (15 CFR 400.41).
- 3.8 Supervision by Operator Under audit-inspection, the Operator is responsible for all aspects of physical supervision of the zone, except to the extent that the Port Director chooses to conduct physical supervision. Generally, the Operator is responsible for supervising all receipts, deliveries, sampling, recordkeeping, manipulation, destruction, manufacturing, security, and storage conditions, in compliance with the laws and regulations. The degree of supervision is that expected of a prudent manager of a storage, manipulation, or manufacturing facility, and may take into account the degree of supervision exercised by the User having physical possession of zone merchandise (19 CFR 146.4(a)).
- 3.9 Operator Responsibilities The Operator's responsibilities are set out in a number of provisions of 19 CFR 146 and 19 CFR 113. These responsibilities are significant in that failure to carry them out may result in claims of liquidated damages against the Operator, or fines against the Operator, as an agent of the Grantee (See Section 13.4 FTZM, 19 CFR 146.81).
 - (a) Enforcement of Operator 's Requirements To assist in overall zone supervision, all applications for permits to admit, manipulate, manufacture, exhibit, or destroy merchandise in a zone must include the written concurrence of the Operator (19 CFR 146.9). Provision is made in the CBPF 214 and the CBPF 216 for the Operator's signature for concurrence. However, concurrence may be indicated by a separate specific or blanket concurrence which states clearly what the concurrence covers. No concurrence is required when the Operator is the owner or purchaser of the merchandise for which the permit is requested. No Operator concurrence is required for transfers of merchandise from a zone because, pursuant to 19 CFR 146.71(a), the merchandise is released by CBP to the Operator, not to the person making entry. The Operator may subsequently release the merchandise to the appropriate party (19 CFR 146.9, 146.71; Section 8.6 FTZM). An Operator may refuse to release merchandise to any person

unwilling to be bound by the proper regulations or in violation of those regulations. The Port Director shall deny any application for a permit which does not have the concurrence of the Operator when required.

3.10 Grantee Liability - A Grantee is a corporation to which the privilege of establishing, operating, and maintaining a zone has been granted (19 CFR 146.1(a); 15 CFR 400.2(p)). The grant shall not be sold, conveyed, transferred, set over, or assigned (15 CFR 400.28(a)(8)). CBP holds the Operator and surety responsible for compliance in the conditions of the Operator's bond (19 CFR 113.73). However, the execution of the bond by the Operator does not lessen the liability of the Grantee to comply with the FTZ Act and the implementing regulations, in 15 CFR and 19 CFR 146.6(e)). The Grantee is not automatically liable for violations by Operators, Users, or other parties when it delegates its authority to operate and maintain a zone to another person and does not itself operate the zone (15 CFR 400.28(a)(9); 19 U.S.C. 81g).

3.11 CBP Reimbursement and Fees -

- (a) Activation and Alteration Fees There are no fees associated with an application to activate, alter or relocate the activated area of a zone. (See Omnibus Budget Reconciliation Act).
- (b) Merchandise Processing Fee (MPF) Importers of record are responsible for paying the merchandise processing fee as authorized under 19 U.S.C. §58c(a)(9) and 19 U.S.C. §(b)(8). The amount of the MPF is 0.21 percent of the value of the merchandise for a formal entry or release; however, the amount of the MPF may not exceed \$485 or be less than \$25. A \$3 surcharge is added for any manual entry or release (19 CFR 24.23(b)(B); 19 U.S.C. 58c(b)(8)(A)(ii)). The amount of the MPF for an informal entry is \$2 for an automated entry or release, and \$6 for a manual entry or release not prepared by CBP, or \$9 for any automated or manual release if it is prepared by CBP personnel. Except as noted in Section 3.11(b)(2) FTZM, the value of merchandise subject to the MPF is determined according to the provisions of 19 U.S.C. §1401a (i.e. dutiable value of the foreign non-duty paid merchandise, in the case of zones). The MPF shall be deposited at the same time as the applicable duties as set forth in 19 CFR 141.101 (formal entries) or 19 CFR 143.28 (informal entries), and reported on CBPF 7501 (Entry Summary) under Code 499 (formal entry), 311 (informal entry), or 500 (manual surcharge), as applicable. (See sample in the Appendix to the FTZM)
 - (1) Exemptions: See 19 CFR 24.23(c). The merchandise processing fee may not be charged for the entry or release of any article that is:
 - (i) Set forth in Chapter 98 HTS, except subheading 9802.00.60 or 9802.00.80 (see Section 3.11(b)(2)(iii) and (iv) FTZM);
 - (ii) A product of an insular possession of the United States;

- (iii) Products of least-developed beneficiary developing countries (General Note 4(b)(i), HTS) and of beneficiary countries under the Caribbean Basin Economic Recovery Act (General Note 7, HTS); 19 U.S.C. 58c (b) (8) (B);
- (iv) A product of Israel if an exemption with respect to the fee is implemented under Section 112 of the CBP and Trade Act of 1990 (i.e., through negotiation of U.S.-Israel Free Trade Agreement); 19 U.S.C. §58c(b)(11);
- (v) Covered by an entry for consumption filed to document shortages reported to CBP under the provisions of 19 CFR 146.53, since these are not considered to be true entries or releases as envisioned in 19 U.S.C. §58c; or
- (vi) Merchandise described in General Note 19 HTSUS, merchandise released under 19 U.S.C. §1321, and merchandise imported by mail.
- (2) Partial Exemptions A reduced merchandise processing fee is paid as set forth in this subsection
 - (i) Domestic Agricultural Products in Zone In the case of agricultural products of the U.S., which have been processed and packed in a zone, the fee will be applied only to the value of the foreign non-duty paid material used to make the containers for such merchandise (19 U.S.C. §58c(b)(8)(D)(v)).
 - (ii) North American Free Trade Agreement If PF status was attained prior to June 30, 1999, the Merchandise Processing Fee is owed. If PF status was attained on or after June 30, 1999, no Merchandise Processing Fee is owed. For NPF status merchandise, no Merchandise Processing Fee is owed (19 U.S.C. 58c(b)(8)(D)(vi)).
 - (iii) Metal Articles Exported and Returned In the case of merchandise classifiable under subheading 9802.00.60 (metal articles which were exported for repairs or alterations and have been re-imported for further processing), the fee will be applied only to the value of the foreign repairs or alterations.
 - (iv) Articles Exported for Assembly and Returned In the case of merchandise classifiable under subheading 9802.00.80 (articles which were exported for assembly and have been re-imported), the fee will be applied to the full value of the product less the cost or value of the component U.S. products.

- (3) Weekly Entries In the case of weekly entries under 19 CFR 146.63(c), the \$485 maximum and \$25 minimum fees are applicable to the total amount covered by the entry summary, and not to the individual zone transfers under 19 CFR 146.63(c), since there is only one entry or release. However, a separate fee will be applied to any supplemental entries filed if actual removals from the zone will exceed the amount estimated for the weekly entry (See Section 9.8 FTZM).
- (c) Harbor Maintenance Fee (HMF) When imported cargo is unloaded from a commercial vessel at a qualifying U.S. port and admitted into a zone, the Applicant for admission of that cargo into the zone may be subject to the harbor maintenance fee as set forth in 19 CFR 24.24 (19 CFR 146.22(e)). It is important to note that the zone status of the merchandise is irrelevant to the application of the HMF.
 - (1) Amount of the Fee Commercial cargo loaded or unloaded from a commercial vessel is subject to a port use fee, of .125 percent of its value (26 U.S.C. §4461 and 19 CFR 24.24(a)). The fee applies only if the cargo is loaded or unloaded at certain ports listed in 19 CFR 24.24(b)(1). Fees shall be paid for all shipments unloaded and admitted to the zone during the quarter, or in the case of direct deliveries as noted in 19 CFR 146.39 and 19 CFR 146.40, unloaded and received in the zone, under the bond of the Operator .
 - (2) Responsible Party Users, not Operators (unless, they are also the Applicant for admission, i.e., subzone Operator), are subject to the fee, although Operators are encouraged to see that Users are aware of their responsibility to pay the fee. Users may also be subject to the fee for domestic shipment(s) (19 CFR 24.24(e)(1)).
 - (3) Domestic Vessel Movements If merchandise is transferred from a zone for transport on a commercial vessel to another U.S. port, the fee is payable by the shipper (the person who pays the freight) on a quarterly basis in the manner specified in 19 CFR 24.24(e)(1), and reported on CBPF 349 (see sample in the Appendix to the FTZM), or on CBPF 350 (see sample in the Appendix to the FTZM), if applicable. Special intraport rules apply to domestic movements as specified in 19 CFR 24.24(d)(1), (2).
 - (4) Method of Payment The fee shall be paid on a quarterly basis either electronically on www.pay.gov or by mailing a check or money order payable to CBP. Payment shall be accompanied by a CBPF 349 (Harbor Maintenance Fee Quarterly Summary Report) (see sample in the Appendix to the FTZM) covering the monthly aggregate fee. If the quarterly HMF payment is amended, it must be done on a CBPF 350 (Harbor Maintenance Fee Amended Quarterly Summary Report) (see sample in the Appendix to the FTZM). The forms are available on-line at pay.gov. If sending a check or money order, it should be sent to CBP, 6650 Telecom Drive, Suite 100, Indianapolis, IN 46278.

- (i) www.pay.gov is a secure web-based collection portal that will allow customers to complete HMF forms and make payments without mailing paper checks. Pay.gov requires that customers use ACH debit as their method of payment when making quarterly HMF payments (19 CFR 24.25).
- (ii) Information for setting up electronic payment can be obtained at hmf.dhs.gov.
- (5) Time of Payment Quarterly payments are due no later than 31 days after the close of the quarterly period ending on the last day of March, June, September, and December (19 CFR 24.24(f)).
- (6) Recordkeeping Each Applicant for admission responsible for payment of the fee must maintain records necessary for CBP to verify the accuracy of the fee computations and to otherwise determine compliance with the law. These records must be maintained for five years from the date of the fee calculation, and must be made available for inspection or other official use by CBP if requested (19 CFR 24.24(g)).
- (7) Penalty for Late Payment The penalty for failure to pay the harbor maintenance fee and file the summary sheet at the time specified is an amount equal to the liquidated damages assessable for late filing of an entry summary under 19 CFR 24.24(h)(1).
- (8) Exemptions and Special Rules
 - (i) Cargo first unloaded from aircraft, land vehicles, or otherwise not from a commercial vessel is exempt from the HMF.
 - (ii) Cargo first unloaded from a vessel at a port not listed in 19 CFR 24.24(b)(1) is also exempt from the HMF.
 - (iii) The HMF will not be assessed on any shipment entitled to be entered under the informal entry procedures of 19 CFR 143.21, nor collected in the quarterly submission (19 CFR 24.24(d)(3)(i)).
 - (iv) The HMF is not assessed on any import vessel movement or domestic shipment whose aggregate value does not exceed \$1000, nor collected if the aggregate value of all shipments for which a HMF was assessed for the quarter does not exceed \$10,00 (19 CFR 24.24(d)(3)(i); 19 CFR 24.24(d)(4)).
 - (v) Special rules applicable to intraport activities are set forth in 19 CFR 24.24(d)(1)(2).

(vi) Other exemptions set forth in 19 CFR 24.24(c)

The Applicant for admission to the zone must advise the Director, Revenue Division, the name, address, EIN/IRS number, and telephone number of a responsible officer of the payer who will verify records required to be maintained as well as any changes to this information (19 CFR 24.24(g)). The address is:

U.S. Customs and Border Protection 6650 Telecom Drive Indianapolis, IN 46278

Chapter 4

ESTABLISHMENT, ACTIVATION, ALTERATION AND OTHER CHANGES

- 4.1 General Putting a zone into operation is a two-stage process. The first stage is gaining approval by the Board for a grant to establish, operate, and maintain the zone. The Board issues guidelines for all types of applications. The guidelines can be found at www.ita.doc.gov (Board website). The second stage is gaining approval by CBP for activation to allow merchandise to be admitted to the zone in zone status. Only after the approval of activation will Users gain the benefits conferred under the FTZ Act. The same two-stage process applies to changes in the approved area of the zone.
- 4.2 Application and Approval of Foreign-Trade Zone Zone applications are approved under the authority and procedures of the FTZ Act (19 U.S.C. §81), and the Board (15 CFR 400.21 400.33). Zone economic, financial, and physical requirements are set forth in 15 CFR 400.21 400.33. However, Board policy may call for other requirements, and Applicants are advised to consult with the Board early in the planning stage to determine the specific filing and exhibit requirements for their zone or subzone.
 - (a) Location Requirements Zones are located in or adjacent to a port of entry (19 U.S.C. §81b) and are listed in 19 CFR 101.3. Zones may be located in communities that have CBP stations. The Board considers a general-purpose location "adjacent" to a port of entry if it is within 60 statute miles or 90 minutes driving time from the outer limits of a port of entry as determined by the Port Director. This mileage and driving time limitation is specific to general-purpose zones, and not subzones (15 CFR 400.21(b)(2)(i)). Subzones may be approved at more distant locations. A subzone meets the adjacency requirement if; (a) proper CBP oversight can be accomplished with physical and electronic means; (b) all electronically produced records are maintained in a format compatible with CBP for the duration of the record period; and (c) the Grantee/Operator agrees to present merchandise for examination at a CBP site selected by CBP when requested, and further agrees to present all necessary documents directly to the CBP oversight office (15 CFR 400.21(b)(2)(ii)).

Each port of entry is entitled to at least one zone project. Additional zones may be approved if the port of entry is located within more than one state, if two cities separated by a body of water are embraced in one port of entry, or if the Board finds that existing zones will not adequately serve the convenience of commerce (19 U.S.C. §81b and 15 CFR 400.21(a)).

(1) User Fee Airports - User fee airports established under 19 U.S.C. §58b may be accepted by the Board as the equivalent of ports of entry for the purposes of the location requirements of 19 U.S.C. §81b). Operator s of such airports must reimburse CBP for the cost of CBP services to the airport, including any service to a zone established at or in the vicinity of the airport (19 U.S.C. §81b; 15 CFR

- 400.2(i)). This is an exception to the general prohibition against charging any CBP costs to zones (19 U.S.C. §58c(e)(2) and (6) and see Section 3.11 FTZM).
- (b) Grantee Sponsorship The Board regulations define eligible Grantees when there is more than one zone project for a subzone project. Normally, the closest Grantee in the same state would be the sponsor of a subzone. However, 15 CFR 400.22(d)(ii) specifically provides that the Grantee of another zone in the same state is eligible to apply for a grant of authority if the Board finds that such sponsorship better serves the public interest.
- (c) Qualifications of Applicant Grants to public and private corporations will not be approved unless the Applicant has been authorized by an act of the state legislature (15 CFR 400.22(b)(c)). The Applicant must demonstrate its ability to adequately finance and conduct the undertaking (15 CFR 400.23(a)(2)).
- (d) Subzones Applications for subzones must be submitted by a Grantee of an existing general-purpose zone within the same state, or a state agency specifically authorized to submit such an application by an act of the state legislature (15 CFR 400.22(d)(i)(iii)). A subzone may be authorized by the Board if it determines that the operation cannot be accommodated in the existing general purpose zone facilities within the zone project and meets the required public interest test in 15 CFR 400.31 (c) and 15 CFR 400.23(b). In practice, the Board requires additional information from the Applicant zone to describe the particular zone benefit(s) sought for the subzone and why such benefits are in the public interest.
- (e) Application Fees In certain circumstances, a fee may be required with the submission of the application to the Board. Information regarding the fees may be found in 15 CFR 400.29.
- (f) Criteria for Grants When determining whether to issue a grant, the Board will consider the need for zone services in the port area, adequacy of operational and financial plans, suitability of site and facilities, extent of state and local support, and views of persons or firms likely to be affected by the project when determining whether to issue a grant. Any zone project that involves manufacturing will have to be evaluated against the criteria in 15 CFR 400.31.
- (g) Initiation of Procedure The application procedures and requirements are set out in detail at 15 CFR 400.24 400.26. All applications will be filed with the Secretary of Commerce, Attention: Executive Secretary, Foreign-Trade Zones Board.
 - (1) Zone applications will consist of a letter of transmittal, executive summary, and five exhibits (15 CFR 400.24).

- (2) Subzone applications will be the same as above except Exhibit #4 will show the specific activity proposed for the subzone and its net economic effect (15 CFR 400.25).
- (3) Modifications of existing zone projects fall into two categories, major and minor (15 CFR 400.26). Major modifications follow the procedures outlined in Section 4.2(h). When such an application is filed with the Board, the Port Director will be notified and provided a date by which CBP technical comments on the application need to be submitted. Minor modifications to the zone boundaries (as further described in Section 4.4) are evaluated in a simpler, administrative process by the Board. Because of the shorter processing time, CBP technical comments are submitted to the Board with the Grantee's request. Minor modifications are generally processed by the Board within 30 days of the receipt of a completed request.
- (4) Temporary/Interim Manufacturing Authority (T/IM) To be eligible for T/IM consideration, the activity must occur within an approved zone or subzone site and the products and components must be similar to a recent Board approval. Any controversial trade issue and products are excluded from T/IM. The T/IM database (available on the Board Web site) includes eligible products and components. If the 6-digit HTSUS is not included in the database, but the products and components fall within included 4-digit HTSUS categories, a company may provide evidence that the activity is substantially similar to the already approved activity to be eligible for T/IM.

T/IM requests are handled administratively by the Board, and are generally processed within 75 days. Because of the shorter processing time, CBP technical comments are submitted to the Board with the application requesting T/IM authority. In some cases, a company may be applying simultaneously for T/IM as well as permanent manufacturing authority. In those cases, the Port Director may choose to indicate that the technical comments submitted on the T/IM request, would also apply to the request for permanent manufacturing authority.

(5) Alternative Site Framework (ASF) - The ASF is an optional site management framework that was designed to give equal weight to increasing flexibility for Grantees and businesses and to addressing oversight concerns by focusing zone designation primarily on sites where zone activity actually occurs. The ASF is an alternative practice that is consistent with the FTZ ACT and the Board regulations.

In essence, the ASF simply modifies traditional practice by allowing minor boundary modifications that draw on un-allotted acreage from a standard 2,000-acre activation limit for an individual zone. The ASF proposal makes the MBM practice far more predictable and flexible by building into the "plan" adopted for a participating Grantee the ability to designate one

category of new sites through MBM action (as long as acreage from the 2,000-acre activation limit is available to be allotted). However, the ASF proposal's increased flexibility for Grantees is tied to several specific limitations that will increase zones focus on active sites. The most important of these limitations is consistent application of "sunset" limits, as described in more detail below. As with traditional practice, no proposal to add or modify a zone site will be approved unless the appropriate Port Director has concurred on the proposal.

There are several key terms to understand regarding the ASF:

"Service Area" is the geographic area (such as specific counties) where a Grantee intends to be able to propose future zone sites. If approved under the ASF, the zone could serve sites throughout the service area based on companies' needs for zone designation.

"Magnet" sites are sites intended to attract multiple potential Operators/Users.

"Usage-Driven" sites are designated to meet a specific Operator /User's present need for zone designation (rather than being designated speculatively to try to attract future Operator s/Users). A Usage-Driven site is tied to the specific Operator /User for which it was designated and could not be used by any other entity. A separate minor boundary modification action would be required to redesignate a Usage-Driven site for another Operator /User.

"Activation Limit" is a cap on the amount of space that can be in CBP "activated" status simultaneously. The Board's practice has long involved a 2,000-acre activation limit for any Grantee. The ASF also applies activation limits on a site-specific basis (for example, a 500-acre industrial park site could have a 200-acre activation limit so that no more than 200 acres of the site could be in activated status simultaneously). ACE and the Board's web site will track all sites and activation limits

A "Sunset Limit" for a site means zone designation self-removes if there is no zone activity at the site before the sunset date. For Magnet sites, the default sunset period is 5 years (with variation possible based on circumstances, including possible permanent designation for one Magnet site). Usage-Driven sites have a three-year sunset period. FTZ activity at a site during sunset period simply resets the sunset date for an additional 5 years (Magnet) or additional 3 years (Usage-Driven).

To move their FTZ into the ASF, a Grantee would submit an ASF reorganization application to the Board. The application format is available at www.trade.gov/ftz and would follow the procedures described in Section 4.2. Once such an application has been filed by the Board, notification requesting technical comments would be sent to the Port Director. Once an FTZ has been reorganized under the ASF, requests for Usage-Driven Sites could be filed. Requests for usage-driven sites would be processed under the minor boundary modifications procedures (see Section 4.4) and CBP technical comments would accompany the request.

(h) Application Review – The examiner, appointed by the Executive Secretary, thoroughly reviews the application and the comments presented orally or in writing. If

there public interest issues involved, the Board may solicit the views of Federal or State government agencies or trade associations. The Board will ask that the Port Director to submit their written comments views on the application. The Port Director may be asked to include in their comments any special information they consider relevant to approval of the application. When these letters are received and the application has been fully reviewed, the examiner submits a recommendation in writing to the Board.

- (1) Procedures for Review Board procedures for review of applications set forth standard time frames for each step of procedural review. Applications involving manufacturing or processing may be completed within 1 year. Those applications not involving such activity may be completed within 10 months. The Board will determine if all applications satisfy the requirements of 15 CFR 400.
 - (i) The Board will provide notice to the Applicant within 20 days of receipt of the application should any deficiencies need correcting. The Applicant will make any necessary corrections and return the application within 30 days (15 CFR 400.27(b)(1)).
 - (ii) If the application needs no corrections, the Board will formally file the application, assign a case docket number and notify the Applicant within 45 days (15 CFR 400.27(b)(2)).
- (2) Comment Period The Board will designate an examiner, and publish a notice of the application in the Federal Register for public comments.
 - (i) The comment period will normally close 60 days after the date the notice appears. In the event a hearing is scheduled, the general comment period does not close until 15 days after the hearing (15 CFR 400.27(c)(2)).
 - (ii) An additional 15 days will be allowed for rebuttal comments after the closing date of the public comment period. (15 CFR 400.27(c)(2)).
 - (iii) The Port Director will review the application and submit a technical report to the Board within 45 days of the close of the public comment period (15 CFR 400.27(d))
 - (1) CBP Special Requirement CBP requires for its review, a statement of commitment to electronic interface with CBP in the following form:

"(Name of Applicant) hereby commits itself to the establishment of an electronic interface with CBP through its electronic data interchange system, when the specifications and requirements of that interface have been developed and implemented by CBP. We understand that this commitment is applicable to our agents and any Operator. It is also understood that, if the interface has not been established, CBP may not activate the applicable portion of the approved zone covered by this application."

This statement should be submitted with the application, but will be accepted later during the review. If submitted later, it shall be provided to the examiner for transmittal to the Executive Secretary.

- (2) CBP will conduct a background investigation of the zone applicant, their personnel, and their key employees, as a routine part of the review of the zone application. If the investigation discloses derogatory information, e.g. a criminal record involving theft or smuggling, the Board will be advised of the facts, to the extent permitted under the Freedom of Information and Privacy Acts, as amended (5 U.S.C. §552 and 5 U.S.C. §552a). Derogatory information is an important factor in CBP's decision whether to support the application (CD 3210-07).
- (3) Examiner's Review The examiner will prepare a report with recommendations to the Board and submit it to the Board within 120 to 150 days of the close of the period for public comment.
 - (i) Report unfavorable to Applicant:
 - (1) Applicant notified within 5 days;
 - (2) Applicant has 30 to 45 days from notification date to present new evidence.
 - (ii) Applicant presents new evidence:
 - (1) New evidence may be published in Federal Register for a 30-day comment period followed by a 15-day period for rebuttals.
 - (iii) CBP shall submit any additional comments within 45 days after notification by the Board if there are additional CBP issues (15 CFR 400.27(d)(2)(i)).
 - (iv) The examiners report shall be submitted to the Board within 30 days after receipt of additional evidence.
- (4) Special Conditions Public interest issues are normally resolved before they get to the final approval stage. The Board may require special conditions be placed in the grant for a zone to protect the public interest, health or safety (15 CFR 400.33(a)). Reviews of these special conditions can be instituted by the

Board to ensure that they continue to be in the public interest (15 CFR 400.12u, 400.43)).

- (5) Amendment or Withdrawal The Board may permit the amendment or withdrawal of the application during the review period (15 CFR 400.24).
- (6) Completion of Review A copy of the examiners report is sent for review to CBP. Once CBP concurs with the findings, the report is forwarded to the Deputy Assistant Secretary (Enforcement) of the Treasury. If the findings are accepted, a resolution to approve the application is signed by Treasury and forwarded to the Executive Secretary. The time period for CBP and Treasury review is thirty (30) days (15 CFR 400.27(e)).
- (7) Final Action When the Treasury resolution is received, the Board will prepare final action papers for review by the Secretary of Commerce or its designee. After final action, a copy of the Board Order is sent to the Applicant. A public notice of the decision on the application is published in the Federal Register notifying the public of the decision on the application. The total period from the date of filing until receipt of the grant is normally from 10 to 12 months.
- 4.3 Grant Limits Lapse and Termination Grants are limited by the provisions of the FTZ ACT and the Board, and by any special conditions which were inserted in the grant. Grants may not be sold, conveyed, transferred, set over, or assigned by the Grantee (19 U.S.C. §81q, 15 CFR 400.28(a)(8)). However, the Board may reissue a grant from one Grantee to another upon approval of a zone application by the intended recipient of the grant. Grants are issued for an indefinite period of time, unless a special time limit has been placed in the grant. Grants may lapse, terminate or be limited in any one of the following ways:
 - (a) Revocation by the Board for willful and repeated violations of the FTZ Act by the Grantee or subzone Operator, under 19 U.S.C. §81r and 15 CFR 400.28(c);
 - (b) Grants lapse unless the zone or subzone is activated and in operation not later than 5 years after the Board order establishing a zone. "In operation" has been defined by the Board as open for operations and does not require the admission of foreign status merchandise (61 FR 53305, "Rule Related Notice," dated 10/11/96).
 - (c) Voluntary relinquishment of a grant by the Grantee (there is no mention of this method in the FTZ Act or by the Board, but it has been accepted by the Board on several occasions); or
 - (d) Expiration without renewal of a special time limit placed in the grant (see Section 11.5(g) FTZM).
 - (e) Restrict or prohibit activity for a zone site or subzone for public interest reasons (15 CFR 400.43).

- 4.4 Expansion or Boundary Modification An expansion of the area of an established zone shall be made and approved in the same manner as an original application (19 U.S.C. §81f(b) and procedures are noted in 15 CFR 400.26).
 - (a) Minor Boundary Modification Grantees may submit requests to the Board for minor modifications of zone boundaries. Because this is an expedited procedure, Grantees normally discuss such proposals with the Port Director and seek comments prior to submission to the Board. The Port Director's comments normally accompany the Grantee's request. CBP's approval is always subject to activation conditions, and shall be so stated in the Port Director's comments. The Board is authorized to determine the requirements for exhibits for such applications. These exhibits will include a report from the Port Director. If the Port Director recommends approval, the Board is authorized to approve the application (15 CFR 400.26(c)). If the proposed boundary change is not a minor modification, generally, the procedure in 15 CFR 400.24 will be followed (15 CFR 400.26(b)).
 - (b) Description of Approved Zone Area Copies of maps, charts, layouts, and legal descriptions of metes and bounds of the approved zone area, including any expansions or boundary modifications shall be retained in a permanent file by the Port Director and shall be retained also by the Operator or Grantee, so it is clear at all times what the approved area covers (15 CFR 400.46(c)). Any application for activation should be compared with the approved zone area to assure that the activated area will be within with the approved zone area.
- 4.5 Other Changes to the Zone Project There are other changes which affect the zone project which are not considered boundary modifications or expansions.
 - (a) Change to Grantee Organizations Board regulations allow a Grantee of a subzone project to be replaced with another Grantee subject to Board approval.
 - (i) A request with necessary documents must be submitted in writing to the Board explaining the circumstances, with a copy to the Port Director (15 CFR 400.26.
- 4.6 Commencement of Operations An FTZ may commence operations after approval by the Port Director of an application to activate (19 CFR 146.6(a)). Any proposed manufacturing or processing activity must be approved in advance by the Board. The definition of manufacturing by the Board is any activity involving the substantial transformation of a foreign article resulting in a new and different article having a different name, character, and use (15 CFR 400.2(g)). The definition of processing by the Board is any activity involving a change in condition of merchandise, other than manufacturing, which results in a change in the CBP classification of an article or in its eligibility for entry for consumption (15 CFR 400.2(k)). Prior to commencement of operations in a zone, the Grantee must have available a zone schedule (See Section 2.1 (f) FTZM).

- 4.7 Application for Activation Board regulations require approval from the Grantee and the ort director pursuant to 19 CFR 146, prior to the activation of any portion of a zone project (15 CFR 400.28(a)(1)). CBP regulations define activation as approval by the Grantee and the Port Director for operations and for the admission and handling of merchandise in zone status (19 CFR 146.1(b)(2)). The term activation also includes the resumption of operations under a new Operator of a previously deactivated zone (CFR 146.1(b).
 - (a) Initiation of Procedure (See 19 CFR 146.6) An Operator (or where there is no Operator, a Grantee/Operator) shall make written application to the local Port Director to obtain approval for activation of an FTZ or FTZ site. The area to be activated may be all or any portion of the zone approved by the Board. The application must include a description of all the zone sites covered by the application for activation, any operation to be conducted therein, and a statement of the general character of the merchandise to be admitted. No fee is charged or collected by CBP in connection with an application for activation (Omnibus Reconciliation Act of 1987). The application must be accompanied by the following:
 - (i) A blueprint of the area approved by the Board to be activated showing area measurements, including all openings and buildings; and all outlets, inlets, and pipelines to any tank for the storage of liquid or similar product, that portion of the blueprint certified to be correct by the Operator of the tank;
 - (ii) A gauge table, when appropriate, showing the capacity in metric units of any tank certified to be correct by the Operator of the tank;
 - (iii) A procedures manual in the English language describing the inventory control and recordkeeping system that will be used in the zone, certified by the Operator or Grantee to meet the requirements of 19 CFR 146, Subpart B;
 - (iv) Written concurrence of the Grantee when the Operator applies for activation (19 CFR 146.6(a) and (b)(5)).
 - (b) Format of Application No particular form is required for an application for activation; a written request on the letterhead of the Applicant will suffice. The application procedure will be required of any new zone or subzone, or of any zone site having a different Operator from that of an already-activated zone site.
 - (c) Treatment of Operator's Procedures Manual The Operator's procedures manual is furnished for CBP use in conducting compliance reviews and audits. CBP personnel may initially read the manual for familiarization purposes. Receipt of the manual by CBP does not constitute approval or certification of the procedures manual nor is approval of the manual allowed (19 CFR 146.21(b)(4)). The Port Director may advise the Applicant, as a matter of courtesy, of any apparent failure(s) of the system to meet the criteria. Neither the Port Director's reading, courtesy advice nor approval of the

application for activation constitutes any representation by CBP that the system does or does not meet the criteria of 19 CFR 146, Subpart B. The Applicant's certification that the recordkeeping system meets the requirements of Subpart B will be taken at face value by CBP at the time of application, and will be evaluated only through Compliance Assessments, Compliance Reviews, and Audits conducted after approval of activation. At the option of the Port Director, the procedures manual may be returned to the Operator for retention at the zone as CBP's copy of the manual. The Operator shall keep the CBP copy updated (19 CFR 146.21(b)(2)) and furnish it to the appropriate CBP personnel upon demand. The Port Director will retain the Operator's certification of the procedures manual in a permanent file. The procedures manual is exempt from disclosure under 5 U.S.C. §552(b)(4) and 19 CFR 103.12(d), and CBP will maintain its confidentiality. Sanctions for violation of confidentiality by CBP employees are set forth in 19 CFR 103.34.

- (d) Review of Application As a condition of approval of the application, the Port Director will order an inquiry by CBP personnel into:
 - (i) the qualifications, character, and experience of an Operator and/or Grantee, principle personnel, and/or key employees that are responsible for the zone;
 - (ii) the security, suitability, and fitness of the facility to receive merchandise in zone status (security survey) (19 CFR 146.6(c)).
- Background Investigation The background investigation is made of the Operator (e) firm itself, management involved in the operation of the zone, and key employees that have access to or could cause changes to the inventory and recordkeeping system. If a background investigation was done of the Operator in connection with the original zone application, no additional investigation will be made if the information from the original investigation is still current. Checks are not usually made of User firms. Checks of employees in subzones are limited to key management and employee positions. The background investigation should be limited to those employees that have access to or could cause changes to the inventory and recordkeeping system. investigation may be required, as a condition of approval of the application for activation, to complete CBP fingerprint cards and establish their true identity for investigation purposes (19 CFR 146.6(a), TD 93-18). Any information disclosed in this inquiry is confidential and will not be disclosed to unauthorized persons, except as set forth in the Freedom of Information and Privacy Acts, as amended (5 U.S.C. 552 and 552a). An employer may wish to obtain a Privacy Act waiver, which would allow disclosure of information from CBP to the employer, from employees at the time of employment. If the background investigation discloses derogatory information, the Port Director may deny the application to activate. "Derogatory information" includes such information as:
 - i. fraud or misstatement of a material fact in the application;

- ii. failure to furnish, upon request, a complete and accurate list of persons as specified in 19 CFR 146.7(g);
- iii. presence in the management of the Operator of a person who has been convicted of a felony or misdemeanor involving theft, smuggling, fraud or similar property crime; or
- iv. employment of persons convicted of such crimes that would jeopardize the security of merchandise in the zone.
- (f) Cargo Security Survey The surveying personnel may prepare a survey report addressing CBP physical and procedural standards. The surveying personnel shall present his or her findings to the Applicant in a conference with the understanding that the survey findings do not constitute a decision on the activation application. The Applicant shall be afforded an opportunity to correct any deficiencies noted in the conference before the Port Director makes a decision on approval of the application.

After presentation of the survey report to the Operator, the facility shall be re-inspected at an appropriate date to determine whether recommendations for improvements have been carried out. If the Operator declines or is unable to institute improvements to correct significant deficiencies, the Port Director shall deny the application.

- (g) Other Considerations Other factors may be considered by the Port Director in making a decision whether to approve the application (Examples: Before the application is approved, the Operator firm is sold to another party; or the Operator manifests or demonstrates an inability or unwillingness to comply with the law, regulations, or grant conditions). There is no specific limit on the grounds for denial of an application to activate, except that the Port Director may not deny it in an arbitrary or capricious manner.
- (h) Decision on Application The Port Director shall promptly notify the Applicant in writing of the decision to approve or deny activation of the zone. If the application is denied, the notification will state the grounds for denial, which need not be limited to those listed in 19 CFR 146.82(a). The decision of the Port Director will be the final CBP administrative determination in the matter. On approval of the application, an Operator's bond shall be executed on CBPF 301 (Customs Bond), containing the bond conditions of 19 CFR 113.73. Upon the Port Director's approval of the application and acceptance of the executed bond, the zone or zone site will be considered activated and merchandise may be admitted to the zone in zone status. Execution of the bond by an Operator does not lessen the liability of the Grantee to comply with the FTZ Act and implementing regulations (19 CFR 146.6(d) and (e)).

Because of the requirements for the background inquiry, applicants may normally expect the Port Director's decision on the application any time from 1 to 4 months after the application was submitted to CBP. The application will not be provisionally approved pending the completion of a regulatory or administrative requirement, e.g. the completion of a background inquiry. Thus, if an application is approved, it will be suspended or revoked only as set forth in 19 CFR 146.82 or 19 CFR 146.83, (see Sections 13.8 through 13.11 FTZM).

- (i) Admission of Merchandise Already in Zone Domestic and duty-paid merchandise which is already in the activated area at the time of activation will be treated as having been admitted to the zone in domestic status as authorized in 19 CFR 146.43, and the provisions of 19 CFR 146, Subpart B shall become applicable to the merchandise. Merchandise not qualifying for domestic status which is in the area at the time of activation shall be admitted to the activated area under the procedures in 19 CFR 146.32 (and see Section 6.7 FTZM), or removed therefrom for other disposition according to law.
- Foreign-Trade Zone Operator's bond The Operator's bond shall be in the (i) amount set by the Port Director, but will not be less than the amount set by the bonding guidelines. The bond may cover all zones of the same Operator in the United States. The amount of the bond shall be determined by the Port Director in accordance with the guidelines set forth in 19 CFR 113.13(b) and CD 3510-004. The initial standard should be based upon CBP duties and fees owed on the average value of foreign status non-duty paid merchandise held in the zone. An Operator may engage in cartage or lighterage under his bond only for merchandise destined for his zone and may also transport merchandise to his zone from anywhere within the district boundaries where the zone is located (19 CFR 112.2(b)). Any other bonded operations by the Operator, such as of a private carrier, container freight station, or bonded warehouse may not be incorporated in the Operator 's bond, but rather must be covered under a separate custodial bond, containing the conditions set forth in 19 CFR 113.63. No foreign or zone-restricted status merchandise will be admitted to a zone until the Operator's bond has been properly executed and is on file with CBP. An entry of merchandise for consumption into the U.S. must be covered by a separate entry bond (See Section 9.7(b) FTZM).
- 4.8 Application for Alteration "Alteration" means a change in the boundaries of an activated zone or subzone; activation of a separate site of an already activated zone or subzone with the same Operator at the same port; or the relocation of an already activated site with the same Operator (19 CFR 146.1(b)(4)). The term also includes discontinuance of activated status of part of a zone site (19 CFR 146.1(b)(8)) and resumption of activated status of a previously deactivated area, but with different boundaries (19 CFR 146.1(b)(17)). Because the boundary of the activated area is new or changed, the Port Director may review the security, suitability and fitness of the area and shall reply to the applicant (19 CFR 146.7(a)).

A zone may be altered so as to have multiple Operators, so long as responsibility and liability for the goods under each Operator's bond is clearly identified.

(a) Application Procedure - An Operator shall make written application to the Port Director for approval of an alteration of an activated area, including an alteration

resulting from a zone boundary modification (19 CFR 146.7(a) and Section 4.4(a) FTZM).

No CBP fee is charged for an application for alteration of an activated area while Merchandise Processing Fees are active (The Omnibus Reconciliation Act of 1987). The application must be accompanied by a blueprint or layout of the area to be activated, showing the changes from the presently activated area, and showing all openings and buildings. If the newly activated area contains any tanks for the storage of liquids, all inlets, outlets, and pipelines to the tank must be shown, and a gauge table showing the capacity of the tank in metric units, certified by the Applicant to be correct, must be attached. If the inventory control and recordkeeping system is different from that of the already-activated area, a certified procedures manual for the newly activated area must accompany the application. Because no background investigation is required, and a security survey is at the discretion of the Port Director, approval of an application for alteration can be expected to be approved or denied in about 2 or 3 weeks after application.

- (b) Bonded Warehouse Within Approved Zone Whenever any contiguous non-activated area is a portion of a building which has been approved as a bonded warehouse, the Port Director shall designate the means for effective separation of the bonded warehouse from the rest of the facility (19 CFR 19.4(b)(6)). Any portion of an approved general-purpose zone or subzone that is not activated may be utilized as a bonded warehouse, container freight station, or centralized examination station upon approval of the application by CBP and filing of the appropriate CBPF 301.
- 4.9 Deactivation of Zone Site "Deactivation" means voluntary discontinuation of the activation of an entire zone or subzone site by the Grantee or Operator (19 CFR 146.1(b)(8)). A Grantee or an Operator with the concurrence of a Grantee shall make written application to the Port Director for deactivation of a zone site, indicating by layout or blueprint the exact site to be deactivated. The Port Director shall not approve the application unless all merchandise in the site in zone status (other than domestic status) has been removed at the risk and expense of the Operator, or a consumption entry has been filed for the merchandise. The Port Director may require an accounting of all merchandise in a zone as a condition of approving the deactivation (19 CFR 146.7(b)). No fee is charged for deactivation.
- 4.10 Reactivation "Reactivation" means a resumption of the activated status of an entire area that was previously deactivated without any change in the Operator or the area boundaries. If the boundaries are different, the action is an alteration, not a reactivation. If the Operator is different, it is an activation, not a reactivation (19 CFR 146.1(b)(17)). A zone or zone site may be reactivated through the same application procedure that was used for deactivation, if a sufficient Operator's bond is on file (19 CFR 146.7(b)). No CBP fee is charged for reactivation.
- 4.11 New Bond The Port Director may require an Operator to furnish, on 10 days notice, a new Operator 's bond on a CBPF 301 set forth in 19 CFR 146.7(d). Alternatively, the Port Director may require, on 30 days notice, a new Operator 's bond set forth in 19 CFR 113.13(c).

If the Operator fails to furnish the new bond within the allotted time, no more merchandise shall be received in the zone in zone status. Merchandise in zone status (other than domestic status) shall be removed at the risk and expense of the Operator, or the merchandise shall be entered for consumption. A new bond may be required if (1) the activated zone area is substantially altered; (2) the character of merchandise admitted to the zone or operations performed in the zone are substantially changed; (3) the existing bond lacks good and sufficient surety; or (4) for any other reasons substantially affecting liability of the Operator under the bond (19 CFR 146.7(d), see Section 13.7 FTZM).

- New Zone Operator It is permissible to change Operators. In the existing zone operation, the Grantee sponsor should be careful not to terminate contractual relationships until the Port Director has approved a new Operator, background investigations have been completed, and an Operator's bond has been accepted and is in force for an agreed amount. A contract between the Grantee and Operator should govern the relationship between the parties. A Grantee of an activated zone site shall make written application to the Port Director for approval of a new Operator, submitting with the application a certification by the new Operator that the inventory control and recordkeeping system meets the requirements of 19 CFR 146 Subpart B and a copy of the procedures manual if different from the previous Operator's manual. The Port Director may order an inquiry into the qualifications, character, and experience of the Operator and its principal personnel or key employees responsible for the zone. The bond specified in 19 CFR 146.6(d) shall be submitted by the Operator before the operating agreement may become effective in respect to merchandise in zone status. The Port Director shall promptly notify the Grantee, in writing, of the approval or disapproval of the application (19 CFR 146.7(e), (f)). The same background inquiry considerations apply to background investigations of the new Operator as noted in Section 4.7(b)(1) FTZM.
 - (a) Interim Responsibility of Existing Operator The existing Operator remains responsible for merchandise in zone status and for compliance with the laws and regulations, under its Operator's bond until the new Operator is approved and a new bond is executed. The existing Operator is relieved of responsibility in the interim only if the zone is deactivated or activated status is suspended, and all merchandise in zone status (except domestic status merchandise for which no permit is required) has been removed from the zone or entered for consumption.
- 4.13 Other Changes There are other changes which affect the activated status of a zone which do not necessarily constitute alteration, relocation, deactivation, or reactivation:
 - (a) Change in Ownership of Operator If ownership of the zone changes hands through sale or other transfer, the procedure to be followed depends on whether the zone is individually-owned, a partnership, or a corporation.

If the zone is individually owned or a partnership, a new application for activation shall be made under the procedures in 19 CFR 146.6 and Section 4.12 FTZM. If the zone is a corporation and the change in ownership does not result in a new corporate entity with a different corporate charter, the change will be treated as a name change as described in

Section 4.13(b), below, FTZM. If the firm is a corporation and the change results in a new corporate entity, a new application for activation shall be made under the procedures in 19 CFR 146.6 and Section 4.12 FTZM.

- (b) Name, Address, or Organizational Status Change If the name of the firm is changed, the Operator should see that CBP is notified through a rider to the Operator's bond under the procedure set forth in 19 CFR 113.24. The same procedure should be used if there is a change in the address of the principal or a change in the trade names and unincorporated divisions of an Operator, which is a corporate principal on the Operator's bond. If an address change is due to relocation of the zone, the procedures for alteration of the activated zone shall be followed in 19 CFR 146.7(a) and Section 4.8(a) FTZM shall also be followed.
- 4.14 Permanent File Requirements Port Directors shall retain copies of applications for activation, alteration, deactivation, and reactivation; applications for new Operators; and the current Operator's bond; in an updated permanent file for the use and reference of appropriate CBP personnel pursuant to 19 CFR 146, Subpart B. Operators should keep the same information in a permanent file, along with the original zone approval (including any grant restrictions specified by the FTZB) and boundary modifications as noted in Section 4.4(b) FTZM and 15 CFR 400.46(c).

Chapter 5

ZONE STATUS

- 5.1 General "Zone status," means the legal status of merchandise, which has been admitted to a zone, thereby becoming subject to the provisions of the FTZ Act. Customs regulations have established four (4) kinds of zone status under the FTZ Act, which determines how the merchandise will be handled in the zone and treated upon its transfer to the Customs territory. The choice, of which zone status is applicable to merchandise, is normally at the option of the applicant for admission or the owner of merchandise in the zone. However, in some cases, the status is dictated by law because of the definition of the status in the regulations, the operation of other laws in conjunction with the FTZ Act, or special conditions in the grant.
- 5.2 Merchandise Not in Zone Status Not all merchandise in a zone has zone status. Examples of merchandise without zone status include, but are not limited to:
 - (a) merchandise, when a question exists whether it may be prohibited, pending a final determination of its status (19 CFR 146.31(a));
 - (b) merchandise temporarily deposited for manipulation under 19 U.S.C. §1562 and 19 CFR 19.11 (19 CFR 146.33);
 - (c) merchandise transiting a zone (19 CFR 146.34);
 - (d) merchandise temporarily deposited with incomplete documentation (19 CFR 146.35);
 - (e) merchandise retained at a zone pending examination and subsequent approval of an application for admission as specified in 19 CFR 146.36;
 - (f) merchandise received under direct delivery procedures pending admission (19 CFR 146.40(a));
 - (g) merchandise which has been constructively transferred to the Customs territory, but remains in the activated area of the zone; (19 CFR 146.61 and 146.71(c) and C.S.D. 79-249):
 - (h) excess merchandise (overages) which have not yet been admitted to a zone; (C.S.D. 81-70);
 - (i) articles used in the zone (e.g. office supplies) for purposes not specified in the Act (15 CFR 400.1(c)).

- 5.3 Application for Zone Status or Status Change An applicant for admission shall select a zone status for merchandise or the application will be returned on the grounds that it was prepared improperly or incompletely. Multiple lines of merchandise can have different zone statuses and they may be admitted under a single admission. Where permissible under the law and regulations, a User may apply for a change in zone status for merchandise already admitted to the zone. The description of the merchandise must be sufficiently clear so as to leave no doubt to CBP which merchandise is covered by the application.
- 5.4 Foreign Status Merchandise (FSM) FSM is imported merchandise which has not been properly released from CBP custody in the Customs territory (19 CFR 146.1(b)). Domestically-produced merchandise which has been exported and re-imported into the United States, but not properly released from CBP custody prior to entry, is foreign merchandise.
- Privileged Foreign Status (PFS) Under this provision, an importer chooses to have the merchandise treated, for tariff purposes, in its condition at the time of admission to the zone, although the choice of that status need not be exercised at the time of admission. The election of PFS may be made after merchandise has been admitted to the zone provided that the merchandise has not been manufactured or manipulated in any way which may have effected a change in tariff classification. PFS merchandise is subject to tariff classification according to its character, condition, and quantity at the rate of duty and tax in force on the date of filing in complete and proper form, the application for privileged status (19CFR 146.65(a)(1)). However, note the provision for TTB excise taxes at Section 11.7(k) FTZM and floor stocks taxes in Section 11.6(d)(5) FTZM. Privileged foreign status is not in effect until the merchandise is received in zone status in a zone. Whenever the privilege has been requested and there has been no manipulation, manufacture or processing in the zone effecting a change in tariff classification, the Port Director shall take under supervision any zone lot or part of a zone lot, including inventory categories and Unique Identifier Number (UIN's) of foreign merchandise in a zone and cause it to be appraised and taxes determined and duties liquidated thereon. 19 U.S.C. §81c is the legal authority for privileged foreign status. Specific applications of this provision to waste and scrap and for valuation purposes are dealt with in Section 9.7(g)(1) and (h)(1) FTZM.

Notwithstanding this provision, TTB excise taxes will be paid at the rate in effect at the time of transfer from the zone for consumption regardless of the zone status; further TTB may impose a floor stocks tax on non-tax paid distilled spirits, tobacco, and tobacco products in a zone (P.L. 101-508, 26 U.S.C. §5001 note, see Section 11.6(d)(5) FTZM).

Applicants usually choose PFS when the rate of duty on a product manufactured in a zone is higher than the rate of duty on some or all of the components as introduced into the zone. Special conditions have been placed in some zone grants to require Applicants to choose privileged foreign status, in order to preserve the protective effect of a rate of duty for a particular component or to identify components so as to impose antidumping or countervailing duties on them. The rates of antidumping or countervailing duties to be deposited are those in

effect at the time of entry of the merchandise into the commerce of the United States. (See Section 11.5(f) FTZM)

- (a) Application Requirements Application for privileged foreign status shall be made on CBPF 214 or electronically at the time of filing the application for admission of the merchandise into a zone or at any time thereafter before the merchandise has been processed or manufactured in the zone in a manner which has effected a change in tariff classification. Each Applicant for this status shall submit to the Port Director with the application, an invoice notated as set forth in 19 CFR 141.90, i.e. showing the claimed rate of duty and HTS number applicable to the merchandise in PF status (19 CFR 146.41(a), (b) and (c)). Note: 19 CFR 141.92 provides for a waiver of the invoice requirements). An application for PF status must be made for a specific kind and quantity of merchandise, and not, for example, for "sufficient parts to manufacture 400 trucks". The specification of the kind and number of parts must be made before the merchandise is manufactured (See C.S.D. 81-192).
- (b) Determination of Duties and Taxes Upon receipt of the application and accompanying invoice, the Port Director may examine the merchandise to determine whether to approve the application. The merchandise will be subject to tariff classification according to its character, condition, and quantity at the rate of duty and tax in force on the date of filing in complete and proper form the application for privileged status (19 CFR 146.41(d) and 146.65(a)(1)).
- (c) Privileged Foreign Status Binding PF status cannot be abandoned and remains applicable to the merchandise even if changed in form by manufacture, processing or manipulation except in the case of recoverable waste (19 CFR 146.42(b)), as long as the merchandise remains within the purview of the FTZ Act. It may not be entered for warehouse or under a TIB (19 CFR 146.64 (a) and C.S.D. 81-213). However, privileged foreign status merchandise may be exported or withdrawn for supplies, equipment, or repair material of vessels or aircraft without the payment of taxes and duties, in accordance with (19 CFR 146.67 and 146.69). Recoverable waste remaining from merchandise in PF status as the result of a casualty or an Act of God (not from manufacture or manipulation in a zone) retains its PF status (See C.S.D. 86-7).
- Non-privileged Foreign Status (NPF) By selecting NPF status, an importer chooses to have the merchandise treated, for tariff purposes, in its condition and quantity as it is constructively transferred to the Customs territory and entered for consumption (19 CFR 146.65(a)(2)). Restricted merchandise in NPF status, which cannot be entered for consumption, may be transformed in a zone into merchandise which may be entered for consumption into the commerce of the United States unless prohibited under the operation of another law (For examples, see C.S.D. 79-41 and C.S.D. 79-471). NPF status merchandise may also be transferred to the Customs territory for warehousing, exportation, vessel or aircraft supply use, temporary importation bond, or transfer to another zone or port. After the merchandise is admitted to a zone, NPF may be changed to PF or ZR status at the option of the zone User, if the merchandise is legally entitled to receive that status. For example, merchandise may not be

changed to privileged foreign status if it has already been manufactured, processed or manipulated so as to change its tariff classification (19 U.S.C. §81(c)); 19 CFR 146.41(b) and see Section 5.5(a) FTZM). All of the following will have the status of nonprivileged foreign merchandise, if requested in proper form by the Applicant for admission:

- (a) Foreign Merchandise Foreign merchandise properly in a zone which does not have the status of privileged foreign merchandise or of zone-restricted merchandise;
- (b) Waste Waste recovered from any manipulation or manufacture of privileged (or non-privileged) foreign status merchandise in a zone; and
- (c) Certain Domestic Merchandise Domestic merchandise in a zone which by reason of noncompliance with the regulations has lost its identity as domestic status merchandise will be treated as non-privileged foreign merchandise. Any domestic merchandise will be considered to have lost its identity if the Port Director determines that it cannot be identified positively by CBP personnel as domestic merchandise on the basis of an examination of the articles or consideration of any proof that may be submitted promptly by a party-in-interest. CBP may consider domestic status identity maintained when it can be established through inventory control and accounting procedures approved by CBP (19 CFR 146.42 and see Section 5.7(d) FTZM).
- 5.7 Domestic Status Domestic merchandise may be taken into a zone, placed under the supervision of CBP and, whether or not it has been combined with or made part of other articles in the zone, may be brought back to the Customs territory free of duty, or tax 19 U.S.C. §81c, is the legal authority for domestic status in zones. An importer or other User does not choose domestic status, as he or she may choose privileged foreign, non-privileged foreign or zone-restricted status. Rather, domestic status is determined by the qualifications of the merchandise as noted below.
 - (a) Merchandise Qualifying for Domestic Status Domestic status will be granted to merchandise which is:
 - (1) the growth, product, or manufacture of the U.S. on which all Internal Revenue Service taxes, if applicable, have been paid;
 - (2) previously imported and on which duty and tax has been paid; or
 - (3) previously imported and entered free of duty and tax (19 CFR 146.43(a)).

Domestic merchandise on which not all internal revenue taxes have been paid is not eligible for domestic status. Neither is it eligible for privileged or non-privileged foreign status. It may be admitted to a zone only in zone-restricted status (C.S.D. 82-112).

(b) Admission Requirements - No application or permit is required for the admission of domestic status merchandise, including domestic packing and repair material, to a

zone, except upon order of the Commissioner. No application or permit is required for the manipulation, manufacture, exhibition, destruction, or transfer to the Customs territory of domestic status merchandise, including packing and repair materials, except: (1) when it is mixed or combined with merchandise in another zone status, or (2) upon order of the Commissioner. When the Commissioner orders a permit to be required for domestic status merchandise, he/she may also order the procedures, forms, and terms under which the permit will be received and processed (19 CFR 146.43(b) and Section 5.7(e) FTZM).

- (c) Recordkeeping Requirements All domestic status merchandise in a zone must be accounted for under the procedures of 19 CFR 146 Subpart B (19 CFR 146.21(a)(1) and see Section 7.8(c)(4) FTZM).
- (d) Loss of Identity as Domestic Status Merchandise When identity of domestic status merchandise has been lost due to noncompliance with CBP Regulations, it shall be treated upon entry to the Customs territory as foreign merchandise under the provisions of the tariff and Internal Revenue Service laws in force at that time. CBP may consider domestic status identity maintained when it can be established through inventory control and accounting procedures approved by CBP (C.S.D. 81-67). Domestic status merchandise shall not be considered to have lost its identity if the zone inventory control and recordkeeping system accurately identifies both domestic and foreign status fungible merchandise. In a UIN system involving fungible merchandise of multiple statuses, the zone inventory control and recordkeeping system shall be sufficient for these purposes (See also Section 7.8(c)(4) FTZM on maintaining the record identity of domestic status merchandise).
- (e) Commissioner's Order to Require Permits CBP will not require permits for domestic status merchandise which has not been manufactured, manipulated or otherwise processed with other status merchandise unless and until an order of the Commissioner has been placed in effect, and then only to the extent of the provisions of the order (19 CFR 146.43(b)).
- (f) Return of Domestic Status Merchandise to Customs territory Upon compliance with the regulation in 19 CFR 146.43(c), any of the merchandise specified in Sections 5.7(a) FTZM remaining in domestic status may subsequently be returned to the Customs territory free of duty, or tax (19 U.S.C. §81(a) and 19 CFR 146.43(c) and Section 9.3 FTZM).
- Zone-Restricted Status Articles which have been taken into a zone from the Customs territory for the purpose of exportation, destruction (except destruction of distilled spirits, wines, and fermented malt liquors), or storage, shall be considered to be exported for the purposes of drawback, warehousing, bonding and other purposes of the Internal Revenue Service laws. Such a transfer may also be considered an exportation for the purpose of other Federal laws insofar as Federal agencies charged with enforcement of those laws deem it advisable. 19 U.S.C. §81c, is the legal authority for ZR status.

- Application for Zone-Restricted Status Merchandise taken into a zone for the (a) sole purpose of exportation, destruction (except destruction of distilled spirits, wines, and fermented malt liquors), or storage will be given ZR status on proper application. That status may be requested at any time the merchandise is located in a zone, even after it has been manufactured, but cannot be abandoned once granted, except by order of the Board. Applications for zone-restricted status will be made via e-214 or on a CBPF 214 (19 CFR 146.44(a), (b)). ZR status may be approved for either foreign or domestic merchandise. Applications for ZR status are normally at the option of the party in interest. However, it may be mandatory for certain merchandise, such as merchandise transferred from a bonded warehouse or non-tax paid alcoholic beverages or tobacco products (See Section 5.8(b)(2) and (3) FTZM below), or merchandise whose entry into U.S. commerce has been rejected by a Federal agency. Merchandise which arrives at a port on an entry for exportation or transportation and exportation (IE or T&E) will usually be admitted in zone-restricted status; however, the Port Director may approve admission in another status if he or she is satisfied that exportation is not required by any Federal law or regulation.
- (b) Merchandise Considered Exported Merchandise may be considered exported, for CBP or other purposes, upon its admission to a zone in zone-restricted status.
 - (1) For CBP Purposes If the Applicant desires ZR status in order that the merchandise may be considered exported for the purpose of any CBP law or regulation, all pertinent CBP requirements relating to an actual exportation shall be complied with as though the admission of the merchandise into the zone constituted a lading on an exporting carrier at a port of final exit from the U.S. Any declaration or form required for actual exportation will be modified to show the merchandise has been deposited in a zone in lieu of actual exportation and a copy of the approved CBPF 214 or a copy of the approved electronic version may be accepted in lieu of any proof of shipment required in cases of actual exportation (19 CFR 146.44(c)(1)). ZR status is used most often, for CBP purposes, to serve as a notice of exportation for drawback purposes or to close out a temporary importation bond. Procedures for satisfying these requirements are described in Sections 6.7(g)(1) and 6.13 FTZM.
 - (2) For Other Purposes If the merchandise is to be considered exported for the purpose of any Federal law other than CBP laws, the Port Director shall be satisfied that all pertinent laws, regulations, and rules administered by the Federal agency concerned have been complied with before the e-214 or CBPF 214 is approved (19 CFR 146.44(c)(2)). A certificate of arrival shall be executed by the Port Director and/or the Operator as specified in 19 CFR 146.38 and see Section 6.11 FTZM. TTB is the most frequent User of this provision, to satisfy the requirement for exportation of non-tax paid alcoholic beverages and tobacco products. Procedures for satisfying those requirements are specified in 27 CFR 252 and 19 CFR 290 and described in Section 6.12 of this manual. Non-tax paid

alcoholic beverages and tobacco products may be admitted to a zone only in zone-restricted status (C.S.D. 82-112). Whether ZR status may be used to satisfy exportation conditions or requirements of state or local governments is dependent on state or local laws.

- (3) Admission for Manufacturing Merchandise in a status other than zone-restricted, cannot be considered to be exported pursuant to the FTZ ACT when it is sent to a zone for the purpose of manufacturing. (TD 89-4, and Chrysler Motor Corporation v. United States, Slip Op 90-130, (CIT 1990) (755 F. Supp. 388 (1990), aff'f, 945F. 2d 1187(Fed. Cir. 1991)).
- (c) Merchandise Entered Into a Bonded Warehouse Merchandise entered into a bonded warehouse for storage and then transferred to a zone other than temporarily for manipulation as set forth in 19 CFR 146.33, will have ZR status when admitted into the zone. The application for admission for merchandise originally entered for warehouse must request ZR status. Otherwise, CBP will return the application for admission (19 CFR 146.44(d), 19 CFR 146.11(d) and 19 CFR 144.37(g)).

Merchandise which has been placed in a bonded warehouse for a purpose other than entry for warehouse, such as general order or under seizure, may be admitted to a zone in any eligible status.

- (d) Destruction Destruction requires that the merchandise, and its residue from the destruction, have been rendered valueless (C.S.D. 80-67); otherwise, the process will be treated as a manipulation. However, partial destruction, combined with exportation and/or storage, meets the requirements of the FTZ Act and regulations for treatment of merchandise in zone-restricted status. Any remaining valuable residue may be removed from the zone for entry for consumption only with the approval of the Board (C.S.D. 80-67). Alcoholic beverages may not be admitted to a zone for the purpose of destruction, but if they have been admitted and subsequently become unfit for export, they may be destroyed with the permission of TTB) under the procedures in 27 CFR 252.35 through 27 CFR 252.38.
- (e) Prohibition Against Manufacture, Processing or Manipulation Since zone-restricted status is granted for the sole purpose of storage, exportation, or destruction, merchandise in zone-restricted status may not be manufactured or processed in the zone. Merchandise may not be manipulated except to the extent necessary for its exportation, destruction, or storage, i.e., except for packing, unpacking, repacking, testing or similar operation incidental to exportation or destruction. After merchandise in non-privileged foreign or domestic status has been manufactured, processed or manipulated in the zone, the Port Director may approve a change to zone-restricted status (See Section 5.8(a) FTZM).
- (f) Transfer to Customs Territory Merchandise in ZR status may not be returned to Customs territory for domestic consumption except when the Board deems such return to

benefit the public interest, in which event the articles shall be subject to the provisions of Chapter 98, Subchapter I, Harmonized Tariff Schedules 19 U.S.C. §81c (a), 19 CFR 146.70(b). The provisions for transfer of ZR status merchandise to Customs territory are described in 19 CFR 146.70 and section 9.15 FTZM.

Chapter 6

ADMISSION OF MERCHANDISE

- 6.1 General Foreign and domestic merchandise of every description, except such as is prohibited by law, may, without being subject to CBP laws of the United States, except as otherwise set forth in the FTZ Act, be brought into a zone (19 U.S.C. §81c(a)). To "admit" merchandise means to bring it into a zone with zone status (19 CFR 146.1(b)).
- 6.2 Prohibited Merchandise The importation of merchandise which is prohibited by any law of the United States, or merchandise which is excluded from a zone by Board Order as set forth in 19 U.S.C. §810(c) (19 CFR 146.1(b) and see Section 11.2 FTZM). If a Port Director has any reason to suspect that an admission of merchandise into a zone might raise a public interest or scope of authority issue, the Port Director has the authority to request that the importer seek an opinion or decision from the Board prior to such admission (Arbor Foods, Incorporated v. United States 97F.3d 534 (Fed. Cir.1996)). If there is a question as to whether merchandise is prohibited, Port Director s may permit the temporary deposit of the merchandise in a zone pending a final determination of its status. Any prohibited merchandise which is found within a zone will be disposed of in the manner set forth in the laws and regulations applicable to that merchandise (19 CFR 146.31(a)). Prohibited merchandise which is found in a zone is not exempt from CBP laws, including those providing for seizure of the merchandise (C.S.D. 82-16).
- 6.3 Conditionally Admissible Merchandise Conditionally admissible merchandise which may be imported into the U.S. under certain conditions, such as merchandise which is subject to permits or licenses, or which may be reconditioned to bring it into compliance with the laws administered by various Federal agencies (19 CFR 146.1(b)). The admission of this merchandise into a zone is subject to the regulations of the Federal agency concerned, which could include a prohibition on its admission to a zone (19 CFR 146.31(b)). An example of conditionally admissible merchandise is a substance subject to the Toxic Substances Control Act (15 U.S.C. §2601 et seq.) which has not received approval by EPA for use in the United States (19 CFR 12.118-12.127).
 - (a) Conditionally Admissible to Zone Merchandise sent to a zone is conditionally admissible, since it may be admitted only if the administrative requirements are met, e.g. completely and properly filled out application, signature of authorized persons on form, etc. However, certain specific merchandise or kinds of shipments may be admitted to a zone only under certain conditions (See Sections 6.7(g) and 11.5 FTZM).
- 6.4 Time of Admission Merchandise is admitted to a zone via concurrence of the e-214 or when (1) the Port Director or his designee has properly approved the application for admission; and (2) the Operator has signed for receipt of the merchandise into the activated area of the zone with an accompanying date acknowledging receipt of the merchandise into the zone inventory control and recordkeeping system.

- (a) Domestic Status Merchandise Domestic status merchandise for which no permit is required (19 CFR 146.43(b)) is deemed admitted when it is (1) recorded in a receiving report or document as set forth in 19 CFR 146.22(a); and (2) physically brought into the activated area of a zone.
- 6.5 Non-Zone Status Transactions As noted in Section 5.2 FTZM, there are a number of circumstances where merchandise may be received in a zone, but not have zone status. Procedures for some of these transactions are covered by this section. Procedures for direct delivery under 19 CFR 146.39 and 146.40 are covered in Section 6.10 FTZM.
- (a) Temporary Deposit for Manipulation Imported merchandise for which an entry has been made and which has remained in continuous CBP custody may be brought temporarily into a zone for manipulation and returned to the Customs territory under CBP supervision, pursuant to 19 U.S.C. §1562. That merchandise will not be considered within the purview of the FTZ Act, but will be treated as though remaining in Customs territory. No zone form or procedure will be considered applicable, but the merchandise will remain subject to any requirements necessary for the enforcement of 19 U.S.C. §562 and other CBP laws while in a zone (19 CFR 146.33). Such manipulations are at the risk and expense of the importer under the importer's importation and entry bond (19 CFR 113.62(g)), not under the Operator's bond. Physical supervision of the manipulation may be conducted by CBP as set forth in 19 CFR 101.2(c).
- (b) Merchandise Transiting Zone The following procedure is applicable when merchandise is to be unladen from any carrier in the zone for immediate transfer to Customs territory, or when it is to be transferred from Customs territory through the zone for immediate lading on any carrier in the zone (19 CFR 146.34).
 - (1) Application Application for permission to lade or unlade will be filed with the Port Director on a CBPF 3171 (Application-Permit-Special License-Unlading Overtime Services) prior to transfer of the merchandise into the zone.
 - (2) Permit The Port Director shall permit the transfer unless he has reason to believe that the merchandise will not be moved promptly from the zone or will be made the subject of an application for admission in accordance with 19 CFR 146.32(a).
 - (3) Treatment of merchandise Upon the issuance of a permit to lade or unlade, the merchandise will be treated as though the lading or unlading were in the Customs territory.
 - (4) Delay in zone transit Merchandise delayed while transiting a zone shall be made the subject of an application for admission in accordance with 19 CFR 146.32 or it will be removed from the zone (19 CFR 146.34).

- (c) Temporary Deposit to Complete Documentation Temporary deposit of merchandise in a zone is allowed in circumstances where the information or documentation necessary to complete the e-214/CBPF 214 is not available at the time of arrival of merchandise within the jurisdiction of the port. The merchandise will be subject to examination as set forth in 19 CFR 146.35(a) and 146.36 (See 19 CFR 146.35).
 - (1) Application An application for temporary deposit will be made to the Port Director on a properly transmitted/signed and uniquely numbered e-214/CBPF 214, annotated clearly "Temporary Deposit in a Zone". For paper CBPF 214 applications, Data Elements 1 through 10, 15, 16, and 24 through 30 must be filled out, if applicable. The description in Element 16 must be at least as complete as that on the manifest bringing the merchandise to the port. The other elements should be filled out to the extent information is available to the Applicant (See Section 7.8(b)(5) FTZM).
 - (2) Supervision The Operator will open and maintain a file by zone admission number or in-bond manifest number of all applications for temporary deposit in a zone as set forth in 19 CFR 146.35, and shall see that such files are properly maintained until a zone admission, entry, or transfer to general order occurs. Merchandise temporarily deposited in a zone will be subject to periodic compliance reviews by CBP to assure compliance with the regulations.
 - (3) Conditions Merchandise temporarily deposited as set forth in 19 CFR 146.35 has no zone status and is considered to be in Customs territory. It will be
 - (i) physically segregated from all other zone merchandise;
 - (ii) held under the bond and at the risk of the Operator; and
 - (iii) manipulated only to the extent necessary to obtain sufficient information about the merchandise to file the appropriate e-214/CBPF 214 admission or entry documentation (19 CFR 146.35(c)).
 - (4) Approval The Port Director will approve the application for temporary deposit if the provisions of Section 6.5(c)(3)(i) through (iii) FTZM above are met (19 CFR 146.35(d)).
 - (5) Admission to Zone A complete and accurate e-214/CBPF 214 will be submitted as set forth in 19 CFR 146.32 within 5 working days, plus any extension granted by the Port Director or the merchandise shall be placed in general order (19 CFR 146.35(e)).
 - (6) Difference from Suspense Account Requirement Temporary deposit to complete documentation is distinct from the suspense account requirement of 19

CFR 146.22(c) (See Section 7.8(b)(1) FTZM.) Merchandise received without complete CBP documentation or which is unacceptable to the inventory control and recordkeeping system, will be recorded in an admission suspense account or record until documentation is complete or the system is capable of accepting the information, at which time it will be formally admitted to the zone under 19 CFR 146.32 or 19 CFR 146.40. The receiving report or document will provide sufficient information to identify the merchandise and distinguish it from other merchandise. The suspense account or record will be completely documented for CBP review to explain the differences noted and corrections made.

- 6.6 Arrival and Retention of Merchandise Merchandise may be admitted directly to a zone from any place within or outside the Customs territory. Foreign merchandise destined to a zone and transported in-bond through the Customs territory will be subject to the laws and regulations applicable to other merchandise transported in-bond between two places in the Customs territory (19 CFR 146.11(a), (b)). Except for direct delivery procedures set forth in 19 CFR 146.39, all merchandise covered by an e-214/CBPF 214 may be retained for CBP examination, pending approval of admission, at the place of unlading, the zone, or other location, as designated by the Port Director (19 CFR 146.32(c)(3), 146.36). Port Directors are reminded that CBP has the right to inspect any shipment for enforcement purposes.
 - (a) Movements Within a Customs Port of Entry When authorized by the Port Director, direct delivery procedures as set forth in 19 CFR 146.39 are available for merchandise arriving from a foreign country to be delivered directly to the zone within the same Customs port of entry boundaries (19 CFR 146.66(a)). Either a bonded carrier must be used or the Operator moves the merchandise under its bond.
- 6.7 Application for Admission and Permit Merchandise does not achieve zone status until concurrence via the e-214 or authorization is given by the Port Director on the CBPF 214 for its admission (except in the case of domestic status merchandise for which no permit is required), and the Operator signs for receipt of the merchandise into the zone with an accompanying date acknowledging receipt of the merchandise in the inventory control and recordkeeping system (See Section 6.4(a) FTZM).
 - (a) Application Merchandise may be admitted into a zone only upon application on a uniquely and sequentially numbered e-214/CBPF 214 and upon concurrence by the Port Director on the CBPF 214. If filing a paper CBPF 214, the Applicant for admission shall present the application to the Port Director and shall include, a statistical copy on CBPF 214A for transmittal to Census. The form shall bear the signature of approval of the Operator in the appropriate block, unless a separate individual or blanket approval has been given (19 CFR 146.9 and 146.32(a)). For applications submitted and concurred via e-214, data is provided to Census through the electronic interface and submission of a statistical hard copy CBPF 214A is not required.
 - (1) Who May Make Application Application for admission may be made only by the person with the right to make entry (19 CFR 146.32(b)(2)). Right to

make entry shall be determined according to the provisions of CD 099 3530-002, and other pertinent service-wide instructions. However, a Customs broker or Operator may prepare and/or file the application on behalf of the person with the right to make entry, if a proper power of attorney is on file as set forth in 19 CFR 141, Subpart C, for the precise acts authorized (See C.S.D. 84-23).

- (2) Printing and Preparation of e-214/CBPF 214 A blank CBPF 214 may be printed and distributed as specified in Section 2.8 FTZM. The form shall be prepared by the Applicant according to the instructions set forth in the Appendix to this manual
- (3) Place of Filing As noted in 19 CFR 146.37 (d) and set out in Section 2.9 FTZM, the e214/CBPF 214 will be filed electronically or presented to the location within a port of entry designated by the Port Director with jurisdiction over the zone.
- (4) Assignment of Sequential Zone Admission Number Operator s will establish and assign the sequential zone admission numbers on the CBPF 214 pursuant to 19 CFR 146.32(a), so long as the header information identifies the zone to which admission has been requested. The standard format for the e-214 admission number consists of three fields; (1) the Zone ID, (2) Calendar Year, and (3) Control Number. The Zone ID is composed of the Zone Number (3 numeric), the Subzone ID (2 alpha numeric), and the Site ID (2 numeric). The Calendar Year is in a "YY" format (2 numeric) and the Control Number is an open format controlled by the Operator (8 alpha-numeric).
- (b) Statistical Reporting Information reported on a e214/CBPF 214 is very important in preparing national trade statistics, including those for zones. The information is gathered by the Foreign Trade Division, Census, from the e214 transmission or statistical copy of the CBPF 214 required under 19 CFR 146.32(a). The statistical copy, designated as CBPF 214A, must be identified as "Statistical Copy." Forms printed on any other stock will be rejected by CBP and Census.
 - (1) When Statistical CBPF 214A is not Required The statistical copy of a CBPF 214A will not be required, or accepted by CBP, for the admission of:
 - (i) Merchandise in domestic status, whether or not a permit is required by the Commissioner; (19 CFR 146.43)
 - (ii) Merchandise of U.S. origin, not previously exported, which is admitted in ZR status;
 - (iii) Merchandise of foreign origin which was entered for consumption prior to admission in ZR status;

- (iv) Merchandise of any status which has been transferred by a zone to zone transfer from another zone where it had been admitted and so reported for statistical purposes.
- (2) When Statistical Reporting is required Providing statistical information is mandatory, by order of the Secretaries of Commerce and Treasury as set forth in 13 U.S.C. §302. The responsibility for obtaining and providing the information rests with the User for admission to the zone.
 - (i) Statistical Form (paper) required The statistical copy of a CBPF 214A shall include the data required in Items 1 through 8, 14 through 21, 23, 47, and 49 of the form, before it is accepted by CBP for transmittal to Census.
 - Review and Verification CBP personnel shall (a) ensure that the foregoing information is provided on CBPF 214A before transmittal to Census. CBP personnel shall not, however, attempt to verify the correctness of the information, except to the extent correct information is required for CBP purposes or to correct obvious errors. When the information is missing, incomplete, or illegible, the form shall be returned to the Applicant for addition or correction. The admission of goods to the zone shall not be delayed pending receipt of the additional or corrected information, unless it is necessary for proper CBP supervision of the zone or a determination of the admissibility of the goods to the zone. When Census discovers errors or anomalies in statistical reports, the appropriate Census personnel will review the transaction directly with the Applicant for zone admission.
 - (ii) Enforcement of Requirements Penalty procedures shall not be invoked against any person who in good faith attempts to comply with statistical instructions for CBPF 214A. However, violations of the instructions may be punishable by a fine imposed by the Secretary of Commerce as set forth in 13 U.S.C. §305. If an Applicant refuses to comply with the instructions, the ort director shall deny the application for admission on the grounds it was improperly executed. False or fraudulent statistical information on a CBPF 214A may also subject the Applicant to penalties as set forth in 19 U.S.C. §1592.
 - (iii) Correction or Completion of Information When goods are admitted to the zone with missing or incomplete statistical information, the Applicant must provide the missing or corrected information by the fifth day of the month following that in which the goods were admitted to

the zone. CBP personnel shall maintain controls appropriate to see that all statistical copies on a CBPF 214A are provided to Census for each admission to a zone, except for those admissions where the CBPF 214A is not required.

(iv) Transmittal to Census – The CBPF 214A shall be gathered by CBP personnel and transmitted to Census as soon as possible but not later than ten days after the merchandise has been admitted to the zone (after the Operator signs the CBPF 214A in Block 47). Box 14 of CBPF 214A must be marked as "No." These forms must be mailed to the following address:

Puerto Rico Trade Data Staff GSA Center 651 Federal Drive Suite 325-08 Guaynabo, P.R. 00965-5703

- (v) Electronic Submission For admission applications submitted and concurred via e-214, Census information is gathered electronically through concurrence and no hard copy documentation is required to be presented to Census for statistical purposes. If a "Docs Required" message is received on transmission of the e-214 that requires filing of a paper CBPF 214 with local Customs, Box 14 must be marked "Yes" for no further Census processing. Even though Customs requires the paper CBPF 214, the e-214 data will be provided to Census from Customs system in the normal manner upon concurrence of the e-214.
- (c) Supporting Documents The following documentation shall be submitted with or in support of e214/CBPF 214, to the extent indicated.
 - (1) Commercial Invoice The Applicant shall submit with the application two copies of an examination invoice meeting the requirements of 19 CFR 141, Subpart F, for any merchandise, other than domestic status merchandise for which no permit is required, to be admitted into a zone. The notation on the invoice of tariff classification and value required by 19 CFR 141.90 need not be made, unless the merchandise is to be admitted in privileged foreign status (19 CFR 146.32(b)(1)).
 - (2) Evidence of Right to Make Entry The Applicant for admission shall submit with the application a document similar to that which would be required as evidence of the right to make entry for merchandise in Customs territory as set forth in 19 U.S.C. §1484 (19 CFR 146.32(b)(2)).

(3) Release Order - CBP personnel shall not authorize any merchandise for delivery to a zone until a release order has been executed by the carrier which brought the merchandise to the port, unless the merchandise is released back to that same carrier for delivery to the zone (See 19 U.S.C. §1484). When a release order is required, it shall be made on any of the forms specified in 19 CFR 141.11 or by the following statement attached to the CBPF 214:

application to:	landise described in this
Name of Carrier	
Signature and title of carrier representative	

Authority is haraby given to release the marchandise described in this

A blanket or qualified release order may be authorized for the transfer of merchandise to a zone as set forth in 19 CFR 141.11 and 146.32(b)(3).

- (4) Application to Unlade For merchandise unladen in the zone directly from the importing carrier, the application on e-214/CBPF 214 shall be supported by an application to unlade on CBPF 3171 (19 CFR 146.32(b)(4)).
- (5) Other Documentation The Port Director may require additional information or documentation as needed to conduct an examination of merchandise under CBP selective processing criteria, or to determine whether the merchandise is admissible to the zone (19 CFR 146.32(b)(5) and Section 6.3(a) FTZM). This includes documentation such as export certificates for certain steel products and machine tools under voluntary restraint agreements (see Sections 11.5 and 11.6 FTZM) and information needed for selectivity processing such as importers' and manufacturers' numbers.
- (d) Selectivity Processing, Document Review, and Examination Cargo selectivity will be performed on all e-214 FTZ Admission Data. Automated Targeting System and Bio-Terrorism Act (BTA) processing will be performed. Prior Notice must be filed at the bill of lading level for FTZ and be in the system at the time of submission of the e-214 data. CBP will review the CBPF 214 application and supporting documentation for completeness and to determine whether the application may be approved without physical examination of the merchandise. CBP approves permits of admission for most low-risk shipments without examination, if the conditions for issuance of a permit under 19 CFR 146.32(c) otherwise exist. The decision whether to examine the merchandise is made according to various commodity and enforcement criteria maintained by CBP. Documentation will be reviewed and examinations conducted only at locations within a port of entry except as specified in Section 2.9 FTZM.

- (1) Authority for Examination The Port Director may cause any merchandise to be examined before or at the time of admission to a zone, or at any time thereafter, if the examination is considered necessary to facilitate the proper administration of any law, regulation, or instruction which CBP is authorized to enforce (19 CFR 146.10). The Port Director may authorize release of the merchandise without examination, as set forth in 19 CFR 151.2. If a physical examination is conducted, CBP personnel shall note the results of the examination on the examination invoices (19 CFR 146.36).
- (2) Purposes of Examination The overall objectives of CBP examination of merchandise to be admitted to a zone are to:
 - (i) determine if the goods are admissible to the zone;
 - (ii) determine the true liability of the Operator for merchandise received in a zone under its bond;
 - (iii) reduce the need for further examination of the merchandise if it is later transferred to the Customs territory, in the same condition, for consumption or warehouse; and
 - (iv) ensure full compliance with all applicable laws and regulations.
- (3) Centralized Examination Station (CES) Most examinations are done at a CES. These facilities allow CBP to position its personnel at fewer locations, where they can perform quicker and better examinations. The procedure for establishing and operating a CES (some of which may be located at foreign-trade zones) under a custodial bond (19 CFR 113.63) is set forth in 19 CFR 118. If CBP determines that examination at a CES is necessary, transfer to a CES may be authorized by CBP by a stamp on the CBPF 214 or other document requesting transfer of the merchandise to the zone (19 CFR 151.15). Transfer is done under the bond of a licensed cartman, importing carrier, bonded carrier, or container freight station, depending on the situation or under the Operator's bond from the CES. Payment of transfer costs to and from the CES or other place of examination is the responsibility of the Applicant for admission, or consignee, not the Operator.
- (4) Examination at Other Locations At ports where there is no CES, the merchandise will be examined at the place of unlading or arrival of the merchandise at the port, unless the Port Director requires or allows the merchandise to be examined at another place. The Applicant for admission may request examination at another place including a zone. However, CBP will not examine any merchandise at a location outside a port of entry, except as set forth in Section 2.9(a) FTZM.

- (5) Report of Examination Reports of examination will be made by CBP examining personnel on both examination invoices in the same manner as for merchandise entered for consumption. Approval of admission will not be denied or withheld for minor discrepancies found upon examination, unless there is an apparent intent to circumvent or violate a U.S. law or regulation. Such minor discrepancies will be faithfully reported on the invoices. If the discrepancies are not minor or there is an apparent intent to circumvent or violate a U.S. law or regulation, admission will be denied on the grounds that an application does not cover the merchandise, and the admission documentation may be referred to the Enforcement Evaluation Team for further action. The Operator shall, upon request of the Applicant for admission or the Port Director, provide a copy of the examination report to the person making entry to transfer the merchandise from the zone (19 CFR 146.37(b) and see Section 9.7(a)(5) FTZM).
- (e) Issuance of Permit for Admission After document review and physical examination, as appropriate under the foregoing provisions, the Port Director shall issue a permit for admission of merchandise to a zone when:
 - (1) the application is properly executed (and includes the zone status desired for the merchandise);
 - (2) the Operators approval appears either on the application or in a separate specific individual or blanket approval;
 - (3) the merchandise has been retained for examination at the place of unlading, the zone, or other location designated by the Port Director, except for merchandise for direct delivery to a zone under 19 CFR 146.39 and 146.40; and
 - (4) all requirements have been fulfilled (19 CFR 146.32(c)).
- (f) Blanket Application for Standard Admission Multiple shipments of merchandise to the zone, destined for one customer may be admitted to a zone under a blanket application. In order to utilize blanket admission procedures, the admissions must be filed via the e-214, in order that Census will receive the statistical information electronically. A blanket application for admission may be approved for:
 - (1) shipments which arrive under one transportation entry as described in 19 CFR 141.55; or
 - (2) shipments which are destined to the same Applicant on a single business day.
- (g) Admission of Certain Kinds of Shipments Some kinds of shipments presented for admission to a zone require special treatment or consideration.

- (1) Temporary Importations Under Bond - Merchandise entered under 19 CFR 10.31 as a Temporary Importation Under Bond (TIB) may be admitted to a zone in zone-restricted status for cancellation of bond charges under 19 CFR 10.39. This is one of the purposes the Fourth Proviso, 19 U.S.C. §81c(a) was meant to serve. Upon receipt of an e-214/CBPF 214 for admission in zone-restricted status, the appropriate CBP personnel shall examine the merchandise and execute CBPF 3495 as set forth in HB 35300-03, 1995 and CD 099 3240-062. A certificate of arrival in zone-restricted status on an e-214/CBPF 214 provided by the Operator under the provisions of 19 CFR 146.38 and set forth in Section 6.11 FTZM shall constitute evidence of exportation sufficient for cancellation of TIB bond charges under 19 CFR 10.39. TIB bond charges may not be canceled upon admission to a zone in any status other than zone-restricted (See Section 5. 8(b)(3) FTZM) However, admission in zone-restricted status is at the option of the Applicant for admission. The merchandise may be admitted in a foreign status, but the merchandise will not be considered exported. The TIB bond charges will not be canceled upon admission under these circumstances. Unless the bond charges are otherwise canceled by actual exportation or destruction as set forth in 19 CFR 10.39, the importer may be assessed liquidated damages after the merchandise has been admitted;
- (2) Transfer from a Bonded Warehouse Merchandise entered for warehouse and transferred to a zone, other than temporarily for the purpose of manipulation, may be admitted only in ZR status, as set forth in 19 CFR 146.44(d) and set out in Section 5.8(c) FTZM;
- (3) Transfer from a TTB Bonded or Controlled Premises Non-tax-paid alcoholic beverages or tobacco products transferred from TTB-bonded or controlled premises may be admitted only in ZR status, as determined in C.S.D. 82-112 (See Section 6.12 and 11.5(d) and (e) FTZM);
- (4) Trade Fair Entries Any merchandise entered for a trade fair under 19 CFR 147 may be transferred to a zone at any time before, or within 3 months after, the closing date of the fair (19 CFR 147.42(a)). Such merchandise may be admitted in PF, NPF or ZR, at the option of the Applicant for admission, unless otherwise restricted by law;
- (5) Admission of Merchandise from Another Zone Merchandise transferred from another zone at the same or a different port will be admitted under the procedures in 19 CFR 146.32 and 19 CFR 146.66(d) (and see Section 6.7 FTZM). However, no statistical copy on CBPF 214A shall be provided, and the Port Director may waive the presentation of a commercial invoice and the examination of the merchandise. The merchandise will not be admitted to the zone until a CBP permit is issued, except in the case of direct deliveries under 19 CFR 146.39 and 146.40. Merchandise must retain the zone status from the transferring zone

upon the admission to the new zone. Any subsequent application to change status designation must consider all activity since the first zone admission.

- (h) e-214 Processing Alternately to the preparation of a paper copy CBPF 214, the e-214 is a voluntary program for participation by the trade for the electronic filing of zone admissions and statistical information. Admissions are submitted via the Automated Broker Interface (ABI). For regular admissions, e-214 data shall be submitted to CBP prior to delivery of the merchandise to the zone. For zones authorized for direct delivery processing, the cargo will be allowed to move into the zone from the CBP port via in-bond authorization for inter-port movements or Permit to Transfer (PTT) authorization for intra-port movements. The e-214 shall be submitted within 24 hours of receipt of the merchandise into the inventory control and recordkeeping system. Cargo selectivity will be performed on all admissions data.
- 6.8 In-Bond Transfer to Zone - The cartage of merchandise for admission to a zone shall be done by a Bonded Carrier (Custodial Bond) or Licensed Cartman (within port only) (See 19 The importer shall select the bonded carrier or licensed cartman (within-port U.S.C. §1565). only) who will deliver the merchandise to the zone, subject to checks by the Port Director to see that the bonded carrier's Customs Custodial Bond is on file or the cartman and its employees are properly licensed under the procedures in 19 CFR 112, Subparts C and D. An Operator may engage in cartage or lighterage under his bond only for merchandise destined for his zone and may also transport merchandise to his zone from anywhere within the district in which the zone is located (19 CFR 112.2(b)(1) and (2)). Operator s may be designated private bonded carriers for the transportation in-bond and cartage of their own merchandise (19 CFR 112.12(b)(3)). Any Operator may be licensed as a cartman under the procedures in 19 CFR 112, Subparts C and D. Bonded carriers are authorized for within port cartage to zones (T.D. 86-16). Payment of cartage costs is the responsibility of the importer, not the Operator. The procedures for transferring merchandise from one zone and receipt into another zone are explained in Sections 9.11 and 6.7(g)(5) FTZM.
 - (a) Intra-Port Transfer Procedure In-bond merchandise shall be transferred to a zone from the place of unlading, CES, or other location upon Customs approval of the e-PTT via the e-214, on a CBPF 6043, or the "Permit to Transfer" section of CBPF 214 (19 CFR 125.31). The cartman or bonded carrier shall sign the form upon pickup, acknowledging receipt of the merchandise (19 CFR 125.34). Generally, merchandise will not be transferred to a zone until it has been released from the place of unlading, CES, or other location. However, the Port Director may authorize transfer to a zone for the purpose of examination, as set forth in 19 CFR 151.6 and 19 CFR 151.7. In that case, the merchandise will be released to the Operator after examination at the zone.
 - (b) Inter-Port Transfer Procedure In-bond merchandise shall be transferred to the zone from the place of unlading or other location upon Customs authorization of the in-bond move via QP/WP (vessel) QX/WX (air), AMS "V" Number, or CBPF 7512. The filer of the QP or QX transmission, if other than the bonded carrier, must have a Power of Attorney on file in order to obligate the Custodial bond of the bonded carrier.

- 6.9 Receipt into Zone The Operator shall accept for admission only merchandise for which an admission permit has been given by the Port Director (19 CFR 146.32(a)). Acceptable evidence of the permit is an electronic authorization from CBP for submissions on an e-214, a signature by authorized CBP personnel in Block 31 of CBPF 214 with an accompanying date in Block 33, or approval for direct delivery procedures.
 - (a) Supervision and Receipt Procedures The Port Director may authorize the receipt of zone status merchandise at a zone without physical supervision by CBP personnel. (19 CFR 146.37(d)) In that case, the Operator shall supervise the receipt of merchandise into the zone and report the receipt and condition of the merchandise (19 CFR 146.4(a) and 19 CFR 146.37(d)). Upon delivery of the merchandise to the zone, the Operator shall:
 - (1) check to see that the appropriate signature and date for permit of admission appear on the CBPF 214, unless under direct delivery procedures;
 - (2) check the description of the merchandise on the CBPF 214 and the commercial invoice against the merchandise, unless under direct delivery procedures;
 - (3) check the quantity in the shipment against the quantity in the cartage documentation on the CBPF 6043, CBPF 214, CBPF 7512, or other supporting documentation. This is important, since the cartage transaction may cover only a part of the merchandise approved for admission, or there may have been a quantity discrepancy upon pickup by the cartman;
 - (4) where applicable, check any and all of the seals affixed to the conveyance, request the concurrence of the carrier as to their condition, and report any broken, missing, or replaced seals to the Port Director. If there is a seal or conveyance discrepancy (i.e., any condition that would allow the removal or switching of merchandise without breaking the seal), the Operator shall hold the conveyance and its contents intact pending the advice of the Port Director. The Port Director may choose to physically supervise the unlading of the conveyance. Otherwise, the Port Director will authorize the Operator to break the seal. If there is no seal discrepancy, the Operator may break the seal unless specific instructions have been issued in advance by the Port Director to hold the conveyance intact (19 CFR 146.8 and Sections 10.4(c), (d), and (e) FTZM);
 - (5) have the merchandise unladen and received directly into the activated area (the merchandise should not be stored, even temporarily, outside the activated area);
 - (6) report to the Port Director any discrepancy in the quantity or condition of the merchandise found upon receipt in the zone; (19 CFR 146.37(c) and see Section 10.4 FTZM);

- (7) in a zone lot number management system, mark the packages with the unique zone lot number so that the merchandise can be traced to a particular e-214/CBPF 214. (This marking requirement shall be deemed met if the packages, unitized or palletized groups of packages, are marked at sufficient intervals of space to allow CBP to distinguish one zone lot from another during compliance reviews and audits);
- (8) adequately identify merchandise that is accounted for under a CBP-authorized inventory method other than specific identification, so that CBP can conduct an inventory count (the merchandise need not be marked with a zone lot number or a CBPF 214 number) (19 CFR 146.37(d)).
- (b) Liability for Merchandise The Operator shall be held liable under its bond for the receipt of merchandise admitted in the quantity and condition as described on e-214/CBPF 214, except as modified by a discrepancy report:
 - (1) signed jointly by the Operator and carrier on the CBPF 214 or other approved form within 15 days after admission of the merchandise, and reported to the Port Director within 2 working days thereafter, or;
 - (2) submitted to CBP and notified to the carrier through completion of the concurrence reporting processing on an e-214 transmission, or;
 - (3) submitted on a Manifest Discrepancy Report (MDR) under the provisions of 19 CFR 158 Subpart A, and CD 099 3240-067A, within 20 days after admission of the merchandise. The Operator may file an MDR on behalf of the person who applied for admission of merchandise to the zone (19 CFR 146.37(c)). Liability and discrepancy reporting are discussed in more detail in Section 10.4 FTZM.
- (c) Recordkeeping upon Admission Upon receipt of merchandise which has been admitted to the zone, the Operator shall:
 - (1) open a sequential zone admission number file marked with the sequential zone admission number of the e-214/CBPF 214 as the file number, and place the CBPF 214, the examination invoice, the cartage documentation, and any other required documentation in the zone admission number file folder (19 CFR 146.37(a)(1));
 - (2) where a CBP-approved inventory method other than a zone lot system is used, file the CBPF 214 in sequential order by the zone admission number (19 CFR 146.37(a)(2));

- (3) record the merchandise in a receiving report using the zone lot number or unique identifier (for merchandise under an authorized inventory method), so that the merchandise is traceable or attributable to a specific e-214/CBPF 214 (19 CFR 146.22(a));
- (4) reconcile the quantities received in the zone to the receiving report, such as the invoice or similar document (19 CFR 146.22(b));
- (5) record merchandise received without complete CBP documentation or which is unacceptable to the inventory control and recordkeeping system, in an admission suspense account or record until documentation is complete or the system is capable of accepting the information, at which time it will be formally admitted to the zone under 19 CFR 146.32 or 19 CFR 146.40. The receiving report or document will provide sufficient information to identify the merchandise and distinguish it from other merchandise. The suspense account or record will be completely documented for CBP review to explain the differences noted and corrections made (19 CFR 146.22(c)). (The difference between an admission suspense account and a temporary deposit to complete documentation is explained in Section 6.5(c)(6) FTZM); and,
- (6) record the receipt of the merchandise in the inventory system records from the receiving report, using the zone lot number or unique identifier for traceability or attribution, stating the quantity and date admitted, cost or value where applicable, zone status, and description of the merchandise, including any part or stock number (19 CFR 146.22(d)).

Further information on recordkeeping upon admission is set forth in Section 7.8(b) FTZM.

- (d) Receipt Admission of Domestic Status Merchandise The Operator shall supervise the admission of domestic status merchandise for which no permit is required, and shall account for its admission in its normal inventory control and recordkeeping system (19 CFR 146.4(a) and 19 CFR 146.21(a)(1)). If merchandise is formally admitted to the zone by being placed on an e-214/CBPF 214, then upon delivery of the merchandise to the zone, the Operator shall:
 - (1) record the merchandise in a receiving report using a zone lot number or unique identifier for traceability or attribution;
 - (2) reconcile quantities actually received to a receiving report or document, such as an invoice;
 - (3) record in a suspense account any merchandise received without complete documentation or which is unacceptable to the inventory system; and,

(4) record the merchandise receipt in the inventory system from the receiving report using the zone lot number or unique identifier for traceability or attribution; and record the quantity and date admitted, cost or value where appropriate, zone status, and description of the merchandise including any part or stock number (19 CFR 146.22(d))

In addition, the Operator should:

- (1) check the merchandise against the carrier's documentation, acknowledge receipt by signing and dating the receipt document, noting any exceptions in quantity or condition of the merchandise in the required spaces, and accept a copy of the receipt document;
- (2) check any seals, if used, and note any seal irregularities in the receipt document; and
- (3) have the merchandise unladen and promptly received into the activated area of the zone.
- 6.10 Direct Delivery of Merchandise This procedure is for delivery of merchandise to a zone without prior application and approval on e-214/CBPF 214 (19 CFR 146.39(a)). It is designed for low-risk, repetitive shipments for which the Operator is the owner or purchaser of the goods.
 - (a) Application for Direct Delivery Procedure An Operator meeting the criteria of 19 CFR 146.39(c) as explained by Section 6.10(a)(1) FTZM shall file a written application with the Port Director at least 30 days before the direct delivery procedure is to become effective. The application will describe the merchandise to be handled or processed and the kind of operation which it will undergo in the zone (19 CFR 146.39(b)). The application may cover all or a part of the merchandise expected to be admitted to the zone. The merchandise covered by the application shall be described in sufficient detail to enable CBP personnel to determine whether the merchandise covered by the application is eligible for direct delivery procedures.
 - (1) Criteria for Approval The Port Director will approve the application if all of the following criteria are met:
 - (i) The merchandise is not restricted or of a type which requires CBP examination or documentation review before or upon its arrival at the zone (19 CFR 146.39(c)(1)).

Examples Which Do Not Meet This Criterion

Merchandise subject to quota or visa requirements;

Merchandise which the Port Director has reason to believe has been, or will be, described inaccurately or incompletely in admission documentation presented by the Operator as set forth in 19 CFR 146.40.

- (ii) The merchandise to be admitted to the zone, and the operations to be conducted therein, are known well in advance, are predictable and stable over the long term, and are relatively fixed in variety by the nature of the business conducted at the site (19 CFR 146.39(c)(2)). Any new class or kind of merchandise must first be approved in writing by the Board. This criterion excludes from consideration any merchandise not within the scope of authority as set out in the application to the Board and approved by the Board Order. CBP must have an opportunity to examine and become familiar with any such merchandise before it may be included in an application for direct delivery. Supplemental applications for direct delivery procedures may be filed for any merchandise meeting the requirements and not included in the original application.
- (iii) The Operator is the owner or purchaser of the goods (19 CFR 146.39(c)(3)). A finding as to whether the Operator is the owner or purchaser of the goods is determined by CBP interpretation as to what constitutes the "owner or purchaser" under 19 U.S.C. §1484(a)(2)(B) and CD 099 3530-002. The Port Director may demand documentation to ascertain the Operator is the owner or purchaser of the goods. There may be more than one Operator of a general purpose zone.
- (2) Decision on Application The Port Director shall approve the application for direct delivery procedures if the criteria in 19 CFR 146.39(c) are met. The Port Director shall promptly notify the Operator in writing of CBP's decision on the application. If the application is denied, the Port Director shall specify the reason for denial in his reply. The Port Director's decision will constitute the final CBP administrative determination concerning the application (19 CFR 146.39(d)).
- (3) Revocation of Approval The Port Director may revoke the approval given under this section if it becomes necessary for CBP to routinely examine the merchandise or documentation before or upon admission to the zone (19 CFR 146. 39(e)). The revocation may cover all or part of the merchandise for which approval for direct delivery had previously been given. Upon revocation, the merchandise may be admitted to the zone only under a prior permit issued under 19 CFR 146.32(c).
- (b) Direct Delivery Procedures The primary responsibility for direct delivery procedures rests with the Operator, as a means of expediting the delivery of merchandise to the zone without having to await a prior approval for admission by CBP. If a shipment to be admitted to a zone includes both merchandise approved for direct delivery procedures and merchandise not approved for direct delivery procedures, none of the

shipment will be authorized for direct delivery to the zone unless the Applicant segregates the packages containing merchandise not approved for direct delivery from the rest of the shipment, and holds the segregated portion intact at a secure location designated by the Port Director pending approval of a separate application for admission. Operators are advised to have a secure area of their zone approved by the Port Director **to** accommodate this situation prior to application for direct delivery. If the segregation and separate admission requirements are met, the portion of the shipment approved for direct delivery may be delivered to the zone pursuant to the provisions of 19 CFR 146.40 and CD 099-3210-015.

- (1) Arrival of Conveyance Upon arrival at a subzone or zone site of a conveyance containing foreign merchandise for direct delivery, the Operator shall:
 - (i) Collect in-bond or cartage documentation from the carrier;
 - (ii) Check the condition of any seal affixed to the conveyance and, if broken, missing, or improperly affixed, so notify the Port Director and receive instructions before unloading the merchandise;
 - (iii) Check each incoming in-bond and cartage shipment to determine if the manifested quantity or the quantity on the cartage document agrees with the quantity actually received;
 - (iv) Sign and date the in-bond or cartage documentation to accept responsibility for the merchandise under the Operators Bond and to relieve the carrier of responsibility;
 - (v) Forward the in-bond or cartage documentation so as to reach the Port Director within 2 working days after the date of arrival of the conveyance at the subzone or zone site;
 - (vi) Maintain a file of open in-bond manifests in chronological order of date of conveyance arrival to identify shipments that have arrived, but the entire contents of which have not been admitted to the subzone or zone site; and,
 - (vii) Notify the Port Director, by annotation on CBPF 214, when the entire contents of a shipment have been admitted (19 CFR 146.40(a)). The Port Director should cooperate with the Operator in seeing that FTZ personnel are properly trained to obtain in-bond and cartage documentation from the carrier and transmit it to CBP as specified in 19 CFR 146.40(a). When the documentation is received by CBP, the appropriate in-bond and cartage transactions under CBP control will be closed out. Merchandise which has been received by the Operator may be

examined by CBP personnel at any time on the basis of comparison with in-bond, cartage, bill of lading, or manifest documentation, or information obtained independently by CBP.

For in-bond merchandise arriving at the zone that will be admitted using an e-214, completion of the arrival via the e-214 indicates the Operators acceptance of responsibility for the merchandise under the Operators bond, effectively relieving the carrier of responsibility. Forwarding of the accompanying in-bond documentation to the Port Director is not required if the carrier posts the arrival. If CBP is requested to post the arrival, then the CBPF 7512 shall be presented.

- (2) Admission to Zone There are two methods of reporting admission of direct delivered merchandise depending on whether the applicant provides statistical reports on the admitted merchandise directly (electronically) to Census.
 - (i) e-214 Options for single, daily, and cumulative reporting are available with the e214. Statistical information is obtained by Census from the Customs systems. The e214 is filed the first working day after the merchandise is received into the inventory control system.
 - (ii) Individual CBPF 214 If an e-214 is not submitted, the Operator shall file with the Port Director, an individual CBPF 214 with a statistical copy on a CBPF 214A, for each shipment recorded into the inventory control and recordkeeping system during the previous business day (19 CFR 146.40(c)(2)).

This provision also states that the CBPF 214A shall be submitted by CBP to Census no later than ten (10) days after the end of the month in which the merchandise was received in the zone and no extension beyond that time frame will be approved by the Port Director.

- (3) Merchandise not Admitted Merchandise which is not admitted to a subzone or zone site as set forth in this section within 15 calendar days after its arrival there, will be sent to general order. It is the Operator's responsibility to notify both CBP and a Class 11 (General Order Merchandise) warehouse of the non-admitted merchandise. The importer of record may also file an appropriate CBP entry for the merchandise and remove it from the zone premises.
- (4) Inventory and Recordkeeping System The Operator shall establish and maintain a continuing input quality control program to ensure that information concerning merchandise in admission documents, verified or corrected by counts and checks, is accurately recorded in the inventory control and recordkeeping system. Quantities recorded in the system, after allowance by the Port Director for any discrepancies, will be the quantities of merchandise for which the Operator shall be held liable under its bond for admission to the subzone or zone

site. A discrepancy involving a within case shortage (or overage) need not be reported on an MDR under CD 099 3240-067A, if the Operator is able to report that information in another manner so that the Port Director can determine whether there is liability for the discrepancy under the bond of any party to the importation (19 CFR 146.40(c)(4)).

- (5) Liability for Merchandise The Operator is responsible under its bond for merchandise for which it signs for receipt under 19 CFR 146.40(c)(4), whether or not the merchandise is admitted to the zone. The Operator is responsible for the merchandise in its condition and quantity as described in the cartage or in-bond documentation, except as modified by a discrepancy report set forth in 19 CFR 146.37(c) and described in Section 10.4 of this manual. This responsibility pertains to the merchandise in its packed condition as delivered. Responsibility for the contents of the package is assumed upon admission as specified in 19 CFR 146.40(c)(4) and see Section 10.4(a) of this manual.
- 6.11 Certificate of Arrival Whenever a certificate prepared by CBP as to the arrival of any merchandise in a zone is required by a Federal agency, the Port Director shall issue the document certifying only that authorization to deliver the merchandise to a zone has been made. The Operator shall issue a certificate of arrival of merchandise at a zone (19 CFR 146.38). Such certifications will usually be presented on a form of the concerned Federal agency, as described in Sections 6.12 and 11.5 of this manual. If no specific form is presented or required by the concerned agency, the Operator may present a copy of the e-214/CBPF 214 for admission as the certificate of arrival. Federal law and regulations do not require the Operator to provide a certificate of arrival to a state or local government agency, but they do not prohibit it either.
- 6.12 TTB Requirements The forms in this section are presented by persons exporting distilled spirits, wines, beer, tobacco products, or cigarette papers and tubes, without payment of internal revenue taxes, under the regulations of the TTB set forth in 27 CFR 28 (exportation of alcohol) and 27 CFR 44 (exportation of tobacco products and cigarette papers and tubes.) These regulations include provisions for the withdrawal of the merchandise without payment of tax from TTB-bonded or controlled premises for transfer to a zone. The Operator shall execute the certificates in these forms upon admission in zone-restricted status. The merchandise is then considered by TTB to have been exported for the purposes of 27 CFR 44.61a (Deliveries to foreign-trade zones—export status) and 27 CFR 28.30 (Export status) any TTB bond liability of the applicant or claimant is thereby relieved. The original of the forms must be forwarded by the CBP personnel in charge to the Director, TTB National Revenue Center, unless otherwise indicated in the form instructions. A copy must be kept by CBP.

FORM REGULATIONS

TTB Form 5100.11, Withdrawal of Spirits, Specially Denatured Spirits or Wines for Exportation.

27 CFR 28.91(a)(3), 28.92(a), 28.103, 28.123, 28.154, 28.121(c), 28.122(a), 28.151(b), 28.152, 28.245,

28.250, 28.291 and 28.290

TTB Form 5110.30, Drawback on Distilled Spirits Exported. 27 CFR 28.171(c), 28.190,

28.193, 28.245, 28.250, 28.291, and 28.290

TTB Form 5120.24, Drawback on Wines Exported. 27 CFR 28.211(c), 28.212,

28.214, 28.216, 28.245, 28.250 and 28.291

TTB Form 5130.6, Drawback on Beer Exported. 27 CFR 28.221(c), 28.222,

28.223, 28.225(c), and

28.290

TTB Form 5130.12, Beer for Exportation. 27 CFR 28.141, 28.142,

28.143, and 28.290

TTB Form 5200.14, (Taxable Articles Without Payment of

Tax Tobacco products or cigarette papers, or tubes.

27 CFR 44.61(b), 44.61a, 44.185, 44.196a, 44.198, 44.207a, 44.252, 44.256, 44.257, and 44.264a

A CBPF 6001 is specified for use in gauging spirits or wines as set forth in 27 CFR 28.290 – see 27 CFR 28.291.

- 6.13 Drawback upon Admission Drawback of duties and IRS paid to CBP is authorized under 19 U.S.C. §1313 and 19 CFR 191. The subpart dealing specifically with drawback upon transfer and admission to a zone is 19 CFR 191, Subpart R. Drawback is authorized upon admission only in zone-restricted status (19 CFR 191.182 and Section 5.8(b)(3) FTZM). Merchandise admitted in zone-restricted status for the purposes of drawback must be maintained under a zone lot system. Under certain circumstances drawback may be authorized on merchandise admitted in other statuses, but only upon actual exportation from the zone and Customs territory, or upon destruction, as set forth in 19 U.S.C. §1313, and 19 CFR 191, and see Section 9.18 FTZM.
 - (a) Manufacturing Drawback Except for the evidence of exportation procedure, the procedures for manufacturing drawback generally prescribed in 19 CFR 191 shall be

followed. This applies to merchandise admitted in zone-restricted status, on articles manufactured or produced in the U.S. with the use of imported or substituted merchandise, and on flavoring extracts or medicinal or toilet preparations (including perfumery) manufactured or produced with the use of domestic tax-paid alcohol (19 CFR 191.183(a)).

- (1) Notice of Transfer (Zone Admission) The notice of zone transfer (zone admission) on a e-214/CBPF 214 shall be used in place of the documents under 19 CFR 191 Subpart G, to establish the exportation effected by the admission in zone-restricted status. The notice of transfer shall be filed in triplicate with the Drawback Office where the zone is located prior to the transfer of the articles to the zone, or within 3 years after the transfer of the articles to the zone. A notice filed after the transfer shall state the zone admission number and the zone lot number (19 CFR 191.183(b)(2)). Each notice of transfer shall show the:
 - (i) Number and location of the foreign-trade zone;
 - (ii) Number and kind of packages and their marks and numbers;
 - (iii) Description of the articles including weight (gross and net), gauge, measure, or number; and
 - (iv) Name of the transferor, (19 CFR 191.183(b)).
- (2) Action by Operator After the articles have been admitted into the zone, the Operator shall certify on a copy of the notice of transfer the admission of the articles (see certification in 19 CFR 191.184(d)(2)), and forward the notice to the transferor or the person designated by the transferor, unless the export summary procedure set forth in 19 CFR 191.73 is used. The transferor shall verify that the notice has been certified before filing it with the drawback claim (19 CFR 191.183(c)).
- (3) Drawback Entries Drawback entries shall be filed on the form authorized by CBP to indicate that the merchandise was transferred to a zone (19 CFR 191.183(d)). The "Declaration of Exportation" on the appropriate form shall be modified as follows:

DECLARATION OF TRANSFER TO FOREIGN-TRADE ZONE

I, (member of firm, personnel representing corporation, agent, or attorney), of (Name of Company), declare that to the best of my knowledge and belief the particulars of transfer stated in this entry, the notices of transfer, and receipts are correct, and that the merchandise was transferred to a foreign-trade zone for the sole purpose of exportation, destruction, or storage, not to be removed from the foreign-trade zone for domestic consumption.

Dated:		
	Transferor or agent	

- (b) Drawback from Continuous CBP Custody The procedure described in 19 CFR 191 Subpart O shall be followed as applicable to drawback on merchandise transferred to a zone from continuous CBP custody (19 U.S.C. §1557(a)), 19 CFR 191.184).
 - (1) Drawback Entry Before the transfer of merchandise from continuous CBP custody to a zone for admission in zone-restricted status, the importer or a person designated in writing by the importer for that purpose shall file with the drawback office a direct export drawback entry on CBPF 7551 in duplicate. CBPF 7551 shall indicate that the merchandise is to be transferred to a zone. The drawback office shall forward one copy of CBPF 7551 to the Operator (19 CFR 191.184(b))
 - (2) Certifications of Operator and Transferor After the merchandise has been admitted in the zone, the Operator shall certify as an attachment to the copy of CBPF 7551 the receipt of the merchandise (see sample in the Appendix to the FTZM) and forward the form to the transferor or the person designated by the transferor, unless the export summary procedure in 19 CFR 191.73 is used. After executing the declaration as shown in the Appendix to the FTZM, the transferor shall resubmit CBPF 7551 to the drawback office in place of the bill of lading required by 19 CFR 191.156 (19 CFR 191.184(c)).
- (c) Unused or Rejected Merchandise The procedures described in 19 CFR 191, Subpart C relating to unused merchandise, and 19 CFR 191, Subpart D relating to rejected merchandise, shall be followed (19 CFR 191.185(a); see 19 U.S.C. §1313(c)(j)).
 - (1) Drawback Entry Before transfer of the merchandise to a zone for admission in zone-restricted status, the importer or a person designated in writing by the importer for that purpose shall file with drawback office an entry on a a CBPF 7551 in duplicate. A CBPF 7551 shall indicate that the merchandise is to be transferred to a zone. The drawback office shall forward one copy of the CBPF 7551 to the Operator (19 CFR 191.185(b)).
 - (2) Certifications of Operator and Transferor After the merchandise has been admitted to the zone in zone-restricted status, the Operator shall certify as an attachment to the copy of the CBPF 7551 the receipt of the merchandise (see sample in the Appendix to the FTZM) and forward the form to the transferor or the person designated by the transferor, unless the export summary procedure set forth in 19 CFR 191.73 is used. After executing the certification provided for in the Appendix to the FTZM, the transferor shall resubmit CBPF 7551 to the

drawback office in place of the bill of lading required by 19 CFR 191.156 (19 CFR 191.185(c)).

- (d) Person Entitled to Claim Drawback The person named in the Operator's certification on the notice of transfer or the drawback entry, as applicable, shall be considered to be the transferor. Drawback may be claimed by, and paid to the transferor (19 CFR 191.186).
- 6.14 Harbor Maintenance Fee When imported cargo is unloaded from a commercial vessel at a U.S. port and admitted into a zone, the Applicant for admission of that cargo into the zone may be subject to the harbor maintenance fee as set forth in 19 CFR 24.24 and 19 CFR 146.22(e) and see Section 3.11(c) FTZM. It is important to note that the zone status of the merchandise is irrelevant to the application of the HMF.

Chapter 7

OPERATOR RECORDKEEPING AND REPORTING RESPONSIBILITIES

- 7.1 General Recordkeeping is an important aspect of zone supervision because it allows CBP to check the representation made in the records against the physical reality found in merchandise in the zone and against independent records of other parties, such as CBP, carriers, sellers to the zone, and buyers from the zone. When audit-inspection supervision was initiated in zones in 1986, CBP ceased to maintain inventory records in zones, turning responsibility over to zone Operators, subject to CBP compliance reviews, audits, and liquidated damages for Operator failure to properly maintain required records.
- 7.2 Authority for Operator Recordkeeping Pursuant to the general authority to establish regulations necessary for the protection of the revenue and to ensure compliance with the FTZ Act, regulations have been established in 19 CFR 146, requiring Operators to maintain certain records and make certain reports, as will be described in this part of the manual (See 19 U.S.C. §66, 81h, 81p, and 1624).
- 7.3 Authority for Importer Recordkeeping Any owner, importer, consignee, or agent thereof who imports or knowingly causes to be imported any merchandise into the United States shall make, keep, and render for examination and inspection such records which (1) pertain to any importation, or the information contained in documents required in connection with the entry of the merchandise, and (2) are normally kept in the ordinary course of business (19 U.S.C. §1508). This requirement is further described and interpreted in 19 CFR 163.
- 7.4 CBP Authority to Examine Zone Records The Port Director may direct CBP personnel to supervise any transaction or procedure at a zone. Supervision may be performed through, among other methods, a periodic audit of the Operator's records or a review of recordkeeping in the zone (19 CFR 146.3(b)). The Operator shall permit representatives of the Board access to the zone (19 CFR 146.4(b)), and Section 3.3(c) FTZM). Although CBP laws do not, under certain circumstances, apply in zones, when merchandise is transferred from a zone to Customs territory or is otherwise not exempt from CBP laws under the FTZ Act, importers become subject to the records inspection requirement of Title 19 U.S.C. §1508 and to the examination of books and witnesses under Title 19 §1509 whether or not the books or records are physically kept in the zone activated area.
- 7.5 Foreign-Trade Zone Board Authority to Examine Zone Records The Board has authority to inspect the premises, operations and accounts of Grantees and Operators (15 CFR 400.11(a)(6)).
- 7.6 Accessibility and Retention of Records Records shall be readily available for CBP review (19 CFR 146.4(d), 15 CFR 400.28(a)(7), 15 CFR 400.41). UIN records, open zone lot records, and zone admission files shall be considered readily available for review if they are located within the activated area, or close enough to the activated area so as not to impede, delay, or create confusion during a CBP compliance review or audit. The records should be capable of

being produced for CBP review in the zone by a CBPO within a reasonable amount of time. The definition of "readily available" is not defined in the regulations and may vary depending on the nature of the records requested. Automated records will be considered readily available if they are accessible through a video terminal which is located in or close to the activated area, with a capability for printing a hard copy. Records may be stored in hard copy, electronic tape/diskette/CD ROM media microfiche or as determined by company procedures. If the original records were electronic, they may be maintained in an electronic mode; however, if the original records were hard copies, they must be maintained as such unless the recordkeeper has utilized the alternative storage procedures as set forth in 19 CFR 163.5. Closed lot records need not be retained in or adjacent to the zone, but must be retained within the area of jurisdiction of the CBP port responsible for supervision of the zone. However, if closed lot records are retained in or close to the activated area, they should be segregated from open lot records.

- (a) Records Retention Period The Operator shall maintain all records, as defined in 19 CFR 163.4, and required under 19 CFR 146 pertaining to zone merchandise for 5 years after the merchandise is removed from the zone (19 CFR 146.4(d)). In the case of merchandise which has been transferred from a zone to the Customs territory or is otherwise not exempt from CBP laws, records pertaining to importation which are required or made under 19 CFR 163.3 shall be kept for 5 years after the date of entry of the merchandise (19 CFR 163.4(a)). Note that since the five (5) year requirement begins with transfer from the zone, records must be maintained for longer than five (5) years since an entry must be traceable back to the related admission.
- 7.7 Safekeeping of Records and Proprietary Information The Operator is responsible for safekeeping of records concerning merchandise in a zone. The Operator, at its liability, may allow a User to store, safeguard, and otherwise maintain inventory records pertaining to their own merchandise (19 CFR 146.4(c)). Only authorized personnel should have access to the records concerning zone merchandise. Operator s shall protect proprietary information in its custody from unauthorized disclosure (19 CFR 146.4(d)). Generally, "proprietary information" is confidential trade or business information which would be exempt from disclosure by the Government under 5 U.S.C. §552(b)(4), such as value information on invoices or CBP forms. The disclosure of confidential information by CBP is punishable by dismissal, suspension, or other disciplinary action, and if done for a valuable consideration, by criminal prosecution (19 CFR 103.34(a)).
- 7.8 Inventory Control and Recordkeeping System An Operator shall maintain the inventory control and recordkeeping system of the zone in accordance with the provisions of 19 CFR 146 (19 CFR 113.73(a)(2), 19 CFR 146.4(d)). An Operator shall maintain either manual or automated inventory control and recordkeeping systems or a combination of manual and automated systems capable of:
 - (1) Accounting for all merchandise, including domestic status merchandise, temporarily deposited, admitted, granted a zone status or status changed, stored, exhibited, manipulated, manufactured, destroyed, transferred and/or removed from a zone;

- (2) Producing accurate and timely reports and documents as required by regulation;
- (3) Identifying overages and shortages of merchandise in a zone in sufficient detail to determine the quantity, description, tariff classification, zone status, and value of the missing or excess merchandise;
- (4) Providing all the information necessary to make entry for merchandise being transferred to Customs territory including, but not limited to, quantities, descriptions, values, HTS classification, and status; and,
- (5) Providing an audit trail to CBP of CBP forms from admission, through storage, manipulation, manufacture, destruction, exhibition and or transfer of merchandise from a zone either by a CBP-approved inventory method, i.e., zone lot number or unique identifier number (19 CFR 146.21(a)).

The Operator may authorize a User to maintain its individual inventory control and recordkeeping system (19 CFR 146.21(b)(3)). However, the Operator remains liable for complying with all inventory control and recordkeeping requirements of 19 CFR 146, including 19 CFR 146.21(c).

- (a) Operator's Procedures Manual The Operator shall provide the Port Director with an English language copy of its written inventory control and recordkeeping systems procedures manual in accordance with the requirements of 19 CFR 146.6(b)(4) and 146.21(b). The manual shall describe, in detail, the methodology to be used in recording transactions in the Operator's inventory and recordkeeping system. The purpose of the manual is to facilitate CBP compliance reviews and audits by explaining how the recordkeeping system works; however, Operator s and Users are responsible for further explaining their system and its operation to CBP officials upon request. The Operator shall keep current its procedures manual and shall submit to the Port Director any change at the time of its implementation (19 CFR 146.21(b)(2)).
 - (1) Maintenance by Zone User An Operator may authorize a User to maintain its individual inventory control and recordkeeping system and procedures manual. An Operator shall furnish CBP a copy of the User's procedures manual, including any subsequent changes. However, an Operator will remain responsible to CBP and liable under its bond for supervision, defects in, or failure of a system (19 CFR 146.21(b)(3)).
 - (2) Significance of CBP Receipt of Manual The Operator's procedures manual and subsequent changes will be furnished to the Port Director for

information purposes only. CBP receipt of a manual does not indicate approval or rejection of a system (19 CFR 146.21(b)(4) and see Section 4.7(a)(2) FTZM).

(b) Receipt and Admission to Zone - All merchandise will be recorded in a receiving report or document using a zone lot number or unique identifier. All merchandise, except domestic status merchandise for which no permit for admission is required pursuant to 19 CFR 146.43, will be traceable or attributable to an e-214/CBPF 214 and accompanying documentation. Quantities received will be reconciled to a receiving report or document such as an invoice with any discrepancy reported to the Port Director (19 CFR 146.22(a), (b)).

The "unique identifier" specified in 19 CFR 146, Subpart B, refers to the unique numerals, letters, or other characters used to identify a specific inventory category, including fungible merchandise. It is not to be confused with the sequential zone admission number of the e-214/CBPF 214 (19 CFR 146.32(a)) which is used to admit all merchandise to a zone, other than domestic status merchandise for which no permit is required.

- (1) Suspense File Merchandise received without complete documentation or which is unacceptable to the inventory control and recordkeeping system will be recorded in an admission suspense account or record until documentation is complete or the system is capable of accepting the information, at which time it will be formally admitted to the zone under 19 CFR 146.32 or 146.40. The receiving report or document will provide sufficient information to identify the merchandise and distinguish it from other merchandise. The admission suspense account or record will be completely documented for CBP review to explain the differences noted and corrections made (19 CFR 146.22(c)). The admission suspense file should not be confused with a temporary deposit due to incomplete documentation under 19 CFR 146.35 (See Section 6.5(c)(6) FTZM).
- (2) Recording into Inventory Record System Merchandise received will be accurately recorded in the inventory system records from the receiving report or document using the zone lot number or unique identifier number for traceability or attribution. The inventory records will state the quantity and date admitted, cost or value where applicable, zone status, and description of the merchandise, including any part or stock number (19 CFR 146.22(d)).
- (3) Zone Admission Files The Customs Modernization Act authorizes CBP recognition of computer records. In those situations where the zone lot records are maintained by use of a computer, there is no requirement that the Operator maintain a hard copy zone admission file. Instead the files authorized for a UIN system at Section 7.8(b)(4) of this manual are permissible. Except as provided in Section 7.8(b)(4) of this manual, upon receipt of merchandise in the zone, the Operator shall open and maintain a zone admission file containing a copy of the e-214/CBPF 214, the examination invoice, and all other documentation necessary

to account for the merchandise covered by each e-214/CBPF 214. The zone admission file shall be maintained in sequential order by using the number assigned to each e-214/CBPF 214 as the file reference numbers (19 CFR 146.37(a)(1)). Under a lot number inventory system, in the instance where more than one zone lot number is received under one zone admission number, the file is still maintained in the same order. The packages of merchandise shall be marked with the zone lot number, so the merchandise can be traced back to a particular e-214/CBPF 214 that these zone lot numbers were received under (19 CFR 146.37(d), see Section 6.9(a)(7) and 8.2(a)(1) FTZM).

- (4) Authorized Inventory Method Where a CBP authorized inventory method other than a zone lot number system (specific identification of merchandise) is used, e.g. First-in, First-Out (FIFO), the Operator shall maintain a file of all CBPF 214s in sequential order (19 CFR 146.37(a)(2)). If an Operator wishes to identify merchandise using a system other than zone lot or FIFO they should request approval through the internal advice procedures in 19 CFR 177.11(b)(1)(ii). The request should state the sources of the merchandise to be identified and the dutiable status of that merchandise, as well as describing the zone operation to which the merchandise is subject.
- (5) Temporary Deposit The Operator shall maintain an inventory and recordkeeping system capable of accounting for all merchandise temporarily deposited in a zone, whether or not it is later admitted to the zone (19 CFR 146.21(a)(1)). Applications for temporary deposit shall be made on a sequentially numbered CBPF 214, clearly annotated "Temporary Deposit in Zone" (19 CFR 146.35(b); (see Section 6.5(c) FTZM). The unique number shall be distinct in series, suffix, prefix, or other indicator from those assigned to merchandise under admittance to a zone. A file shall be assigned a temporary deposit number. The file shall be closed when the merchandise is removed from the temporary deposit status.
- (6) Direct Delivery 19 CFR 146.39, certain criteria must be met to allow direct delivery to a zone without prior permit.
- (c) Accountability for Zone Merchandise The Operator shall have an inventory control and recordkeeping system(s) capable of accounting for all merchandise stored, exhibited, manipulated, manufactured, processed, destroyed, or granted a zone status change (19 CFR 146.21 (a)(1)). Merchandise admitted to a zone can be identified by either a zone lot number inventory control method (ZLN) or a unique identification number inventory method (UIN).

Zone Lot Number (ZLN) - A collection of merchandise maintained under an inventory control method based on specific identification of merchandise admitted to a zone by lot number. Use of this method requires maintenance of zone lot records, and physical segregation and marking of zone lot numbers.

Unique Identifier Number (UIN) - This inventory method controls merchandise in a zone by cumulative identification or record identity, i.e., by unique numbers and/or letters that identify merchandise admitted to a zone. Inventory levels are adjusted on a first in first out (FIFO) basis or foreign first (FOFI) basis pursuant to generally accepted accounting principles. Merchandise need not be segregated.

- (1) Inventory Records The inventory records will specify by zone lot number or unique identifier number:
 - (i) location of merchandise;
 - (ii) zone status;
 - (iii) cost or value, unless Operator 's or User's financial records maintain cost or value and the records are made available for CBP review;
 - (iv) beginning balance, cumulative receipts and removals, adjustments, and current balance on hand by date and quantity;
 - (v) destruction of merchandise; and
 - (vi) scrap, waste, and by-products (19 CFR 146.23(b)).

A zone lot number or unique identifier number will be used to identify and trace merchandise in a zone (19 CFR 146.23(a)(1)).

(2) Lot Number System (Specific Identification) - Under a zone lot number system of specific identification for merchandise received and admitted to a zone, the zone lot number must be assigned at admission. An Operator may choose whatever lot number he requires. The zone lot number may be a consecutive number assigned to the merchandise; it may be the lot number already used by the company for other purposes, or the sequential number assigned by the Operator to the e-214/CBPF 214 upon admission. If there is more than one zone lot number assigned to merchandise placed on an e-214/CBPF 214, the zone admission number should not be used as a zone lot number. In an instance where every admission of merchandise is only one zone lot, the sequential zone admission number assigned to the e-214/CBPF 214 may be used as the zone lot number. The zone lot number must be displayed on the merchandise cartons, boxes, etc. (19 CFR 146.37(d)) (also Section 6.9(a)(7) of this manual). The merchandise shall be physically segregated to distinguish one zone lot from another. Each zone lot number shall be identified in the operator's inventory recordkeeping system. The zone admission file will be maintained as described in Section 7.8 of this manual.

When zone lot merchandise is manipulated by combining merchandise from different zone lots, a new manufacturing or finished product lot number will be established in the record to show, in addition to the information specified in 19 CFR 146.23(b); and Section 7.8(c)(1) of this manual, the original zone lot numbers from which the merchandise has been transferred.

EXAMPLE:

- Fishing poles admitted under zone lot #860813;
- Fishing reels admitted under zone lot #860814;
- Fishing kits (manipulated from zone lot #'s 860813 and 860814). New finished product lot #860815A must be traceable back to zone lot numbers 860813 and 860814.

Note: The zone's Grant of Authority must authorize the activity taking place in the zone.

(3) Authorized Inventory Method - An Operator may request, under the procedures of 19 CFR 177, approval from CBP for an authorized inventory method in lieu of the zone lot number system (19 CFR 146.23(a)(2) and 19 CFR 146.37 (a)(2)). CBP has approved as inventory methods FIFO under C.S.D. 81-62 and FOFI under C.S.D. 83-30. An Operator need not request approval for his zone of an inventory method already approved for general use by CBP. Aside from this method, the FTZ Act specifically authorizes Industry Standards of Potential Production on a Practical Operating Basis (known as producibility) as verified and adopted by the Secretary of the Treasury as an inventory method for feedstocks for petroleum refineries operating in zones (19 U.S.C. §81c(d); See Section 11.6(j) of this manual).

FIFO as approved in C.S.D. 81-62 applies to fungible merchandise, i.e. merchandise which for commercial purposes is identical and interchangeable in all situations (19 CFR 146.1(b)(12)).

(i) First In First Out (FIFO) - Under the FIFO method, any merchandise withdrawn from stock is assumed, in accordance with good business practice, to be withdrawn from the oldest stock first. In applying FIFO procedures, it is important to note that each admission is considered a layer and the Operator is responsible for identifying the appropriate sequence of transfers from the zone to assure the proper layer was decremented. This procedure becomes critical where different costs or values and/or different zone statuses are involved in the fungible merchandise. Since fungible merchandise, is, by definition, commercially interchangeable in all situations, it should be stored together whenever practical. Identification shall be maintained in the inventory control system records, generally, by description and part or stock number. A

history file, reflecting the layers by date of admission, shall be maintained by the Operator to provide the necessary details for CBP compliance reviews and audits.

- (ii) Foreign First (FOFI) An accounting method based on the assumption that foreign-status merchandise is decremented first (C.S.D.-83-30).
- (4) Domestic Status Merchandise - The Operator shall maintain an inventory control and recordkeeping system(s) capable of accounting for all merchandise in the zone, including domestic status merchandise. Domestic status merchandise which has lost its identity as domestic merchandise will be treated as foreign merchandise in non-privileged foreign status. It will be considered to have lost its identity as domestic merchandise if the Port Director determines that it cannot be identified positively as domestic merchandise on the basis of an examination of the merchandise or consideration of any proof that may be submitted by a party in interest. Acceptable proof includes, but is not limited to a recordkeeping system(s) (19 CFR 146.42(c); see Section 5.7(a)(3) and 5.7(d) of this manual). There is no authority under the FTZ Act permitting the substitution in a zone of domestic and imported merchandise, similar to the authorization found in 19 U.S.C. §1313(b) relating to drawback. Merchandise manufactured in a zone under such conditions is subject to full duty if sent into the Customs territory (TD 54873(2) and see comment and explanation in C.S.D. 82-152). methods involving record identification and attribution, such as FIFO or FOFI, are not substitution as understood in TD 54873(2) and C.S.D. 82-152.

Products derived in part from domestic status merchandise and in part from foreign status merchandise may retain their respective zone statuses when the quantities derived from the merchandise from which they were manufactured can be established through inventory control and accounting procedures approved by CBP (C.S.D. 81-67). For example, a known volume of one liquid in domestic status may be mixed with a known volume of another liquid in foreign status and retain its identity as domestic status merchandise when the proportion of domestic and foreign components of the mixture have been determined by independent volume measurements of each domestic and foreign component prior to their being mixed, assuming that it is possible to accurately calculate the quantity of domestic components in the mixture (C.S.D. 82-152).

Physical shortages identified in fungible merchandise consisting of a combination of foreign and domestic status will be treated as a shortage of foreign status merchandise unless it can be identified positively as domestic status merchandise (19 CFR 146.42(c)). Such identification would include any evidence that status has been properly maintained under an inventory method authorized by CBP.

Importations for which a formal entry would not be required may be admitted to a zone only in domestic or zone-restricted status after having been entered into Customs territory with or without payment of duty, as appropriate.

- (5) Petroleum Refinery Inventory Accounting Definition and attribution of products to feedstocks for petroleum manufacturing may be either in accordance with Industry Standards of Potential Production on a Practical Operating Basis as verified and adopted by the Secretary of the Treasury (known as producibility) or such other inventory control method as approved by the Secretary of the Treasury that protects the revenue (19 U.S.C. §81c(d), 19 CFR 146, Subpart H; See Section 11.6(j) FTZM).
- (d) Transfer of Merchandise from Zone All zone status merchandise transferred from a zone will be accurately recorded within the inventory control and recordkeeping systems. The inventory control and recordkeeping systems for merchandise transfers must have the capability to trace all transfers back to a zone admission under a CBP authorized inventory method or the zone lot system (19 CFR 146.24(a)(1), (2)). This recordkeeping responsibility applies to the transfer of merchandise which has already been constructively transferred to Customs territory and entered for consumption, although it is no longer in zone status (See Section 9.16(d)(2) FTZM).

The inventory control and recordkeeping systems must be capable of providing all information necessary to make entry for transfer of merchandise from the zone (19 CFR 146.21(a)(4) and 146.24(b)).

- (e) Physical Inventory The Operator shall take at least an annual physical inventory of all merchandise in the zone (unless continuous cycle counts are taken as part of an on-going inventory control program) with prior notification of the date(s) given to CBP for any supervision of the inventory deemed necessary. The Operator shall notify the Port Director of any discrepancies in accordance with 19 CFR 146.53, 19 CFR 146.23(c). The physical inventory may, but need not be, conducted in conjunction with the annual reconciliation report under 19 CFR 146.25. Identification of overages and shortages in the inventory system are required whether or not they are recorded as an adjustment to inventory. Port Directors shall query each zone to determine their physical inventory schedule and to assure that prior notification is given of each inventory.
- (f) Annual Reconciliation The Operator shall prepare a reconciliation report within 90 days after the end of the zone/subzone year unless the Port Director authorizes an extension for reasonable cause. The Operator shall retain that annual reconciliation report for compliance reviews or audit by CBP and need not furnish it to CBP unless requested. There is no form specified for the preparation of the report. The report must contain a description of merchandise, for each zone lot number or unique identifier number, showing zone status, quantity on hand at the beginning of the year, cumulative receipts and transfers (by unit), quantity on hand at the end of the year, and cumulative

positive and negative adjustments (by unit) made during the year (19 CFR 146.25(a), (b)).

- (1) Certification to CBP The Operator shall submit to the Port Director within 10 working days after the annual reconciliation report is complete, a letter signed by the Operator certifying that the annual reconciliation report has been prepared, is available for review, and is accurate. The certification letter must contain the name and street address of the Operator, where the required records are available for review, and the name, title and telephone number of the person having custody of the records. Port Directors shall maintain a suspense file to assure that the certification of the annual reconciliation is timely received from each Operator. Reporting of shortages and overages based on the annual reconciliation will be made in accordance with 19 CFR 146.53; see Section 10.5 FTZM). These reports must accompany the certification letter (19 CFR 146.25(c)).
- (g) Annual System Review The Operator shall perform an annual internal review of the inventory control and recordkeeping system(s) and shall report to the Port Director any deficiency discovered and corrective action taken, to ensure that the system meets the requirements of 19 CFR 146.26.
- (h) Sanctions for Recordkeeping Violations The Operator is subject to assessment of liquidated damages by the Port Director for violations involving the inventory control and recordkeeping system(s) under 19 CFR 113.73(a)(2). In addition, the activated status of the zone may be suspended by the Port Director for cause on the grounds that the inventory control and recordkeeping system is impaired to the point where the identity of merchandise in zone status has been lost and cannot be reestablished without a suspension of zone operations (19 CFR 146.82(a)(8)).

7.9 Zone Admission File Maintenance -

(a) Lot System - If the merchandise is accounted for under specific identification, and not under another authorized inventory method, the Operator shall maintain a zone admission file containing a copy of e-214/CBPF 214, the examination invoice, and all other documentation necessary to account for the merchandise covered by each e-214/CBPF 214. There may be multiple zone lot numbers received under one e-214/CBPF 214 sequential admission number. As previously noted in Section 7.8(b)(3) FTZM, if lot records are maintained in a computer system, there is no requirement that an Operator maintain a hard copy file of each individual zone lot number. Instead, files maintained by a sequential e-214/CBPF 214 number are permitted. The zone admission file will be maintained in sequential order by using the sequential number assigned by the Operator to each e-214/CBPF 214 as the reference number (19 CFR 146.37(a)(1)). In those instances where computerized lot records are not maintained, then a zone admission file will be maintained in sequential e-

214/CBPF 214 admission number order. The Operator shall give a copy of the examination invoice to the person making entry to transfer the merchandise covered by the invoice from the zone upon demand of that person or the Port Director (19 CFR 146.37(b) and Section 9.7(a)(5) FTZM). In the zone admission file will be copies of all permits by CBP to admit, exhibit, manipulate, manufacture, process, destroy, or transfer zone merchandise, and any other documents necessary to account for any change in the condition or quantity of merchandise while it is in the zone. If a yearly blanket CBPF 216 has been approved, it is not necessary to keep a copy in each file, but should be maintained separately for reference. Each permit or other document should be recorded in the inventory and recordkeeping system within two working days after the event occurs so the file will be current when reviewed by CBP personnel.

Zone Admission File Contents - CBP forms and other documents that may be expected to be maintained in the zone admission file include, but are not limited to:

FORM NUMBER & NAME

USE

1. Deposit in Zone

e-214/CBPF 214, Application for Foreign-
Trade Zone Admission and/or Designation

CBPF 6043, Delivery Ticket

e-PTT Authorization Message

CBPF 7512, Transportation Entry and Manifest of Goods Subject to CBP Inspection and Permit

Manifest Discrepancy Report (MDR)

CBPF 6423, Notice of Damage, Shortage, or Sample

CBPFs 7514, 7512, or 7551

TTBFs 5100.11, 1689, 5110.30, 1582-A,

Admission to zone and selection of zone status

Transfer to zone pursuant to permit on e-214/CBPF 214

Transfer to zone pursuant to permit on e-214/CBPF 214, or direct delivery to zone with concurrence of carrier

Quantity discrepancy found before or upon admission to zone. Free form letter with supporting documentation prepared by the carrier or Operator

Report of within case damage or shortage found by CBP personnel, or report of sample taken by CBP personnel

Claim of drawback on merchandise admitted to zone

Transfer of alcoholic beverages to zone

and 1582-B	without payment of Internal Revenue tax		
TTBF 5200.14	Transfer of tobacco products to zone without payment of Internal Revenue tax		
2. Operational Activities			
CBPF 216, Foreign-Trade Zone Activity Permit	Exhibition, manipulation, manufacture, processing, or destruction of zone merchandise; temporary removal and return of zone merchandise		
e-214/CBPF 214 or CBPF 7501	Report overage found in zone		
CBPF 7501, or Manifest Discrepancy Report (MDR)	Report shortages found in zone		
Operator 's form or written statement	Used for reporting shortages of less than 1% and/or \$100.00 in duty		
CBPF 214	Request for zone status change for goods in zone		
CBPF 4315, Application for Allowance in Duties	Report of merchandise lost through casualty or other qualifying causes		
CBPF 5955A, Notice of Penalty or Liquidated Damages Incurred and Demand for Payment	Default involving merchandise		
3.Transfers from Zone			
CBPF 3461/Estimated CBPF 3461(Immediate Delivery)	Entry for consumption		
CBPF 7501, Entry Summary	Entry for consumption		
CBPF 7501, Entry Summary consumption	"Live" entry for consumption		
CBPF 368, or 368A	Informal Entry		
CBPF 7512, Estimated CBPF 7512 Transportation Entry and Manifest of Goods Subject to CBP Inspection and	Entry for transportation, exportation, transportation and exportation, or vessel or aircraft supply and certain other goods		

Permit

CBPF 5125, Application for Withdrawal of Bonded Stores for Fishing Vessels and Certificate of Use Entry for supplies for fishing vessels

CBPF 6043, Delivery Ticket, CBPF 7512

Removal for transfer to another zone, or transfer to a bonded warehouse, or dock or airport for vessel or aircraft supply or for exportation

CBPF 216, Foreign-trade zone Activity Permit

Temporary removal and return of zone merchandise; removal of zone status merchandise for which no entry is required

- (b) UIN or Other CBP Authorized Systems In a UIN or other CBP Authorized inventory system, the zone admission file consists of the e-214/CPBF 214 and supporting documentation. The zone admission file will be maintained in a sequential order based on the zone admission number. Documentation related to other zone activities and/or removal of the merchandise is kept as part of the inventory control system(s) and need not be kept in the zone admission file (19 CFR 146.37 (a)(2)).
- 7.10 Transfer of Merchandise History to Another Zone When merchandise is transferred from one zone to another, including at the same CBP port, the Operator of the transferring zone shall provide the Operator of the destination zone with the documented history of the merchandise being transferred. The documentation to be forwarded is set forth in 19 CFR 146.66(c) and described in Section 9.11(b) FTZM and may be transmitted by electronic means.
- 7.11 Record of Foreign-Trade Zone Activity Under CBP Permit The Operator shall record on a CBPF 216 the results of an approved manipulation, processing, manufacture, exhibition, temporary removal or certification of destruction, other than by a blanket application, unless the Port Director chooses to physically supervise the operation (19 CFR 146.52(d),(e); see Section 8.3(d) and 8.4(e) FTZM).
 - (a) Blanket Application The Operator shall maintain a record of an approved manipulation, manufacture, process, exhibition or certification of destruction in its inventory control and recordkeeping system so as to provide an accounting and audit trail of the merchandise through the approved operation. A temporary removal from a zone must be reported on an individual CBPF 216 unless otherwise approved by CBP (19 CFR 146.52(d)(2)).
- 7.12 Reporting of Merchandise Discrepancies Discrepancies in the quantity or condition of merchandise admitted to, within, or transferred from the zone shall be reported by the Operator to the Port Director as set forth in 19 CFR 146.37(c), 146.53(a), and 146.71(b). See Sections 10.4 through 10.6 FTZM.

- 7.13 Grantee Responsibilities The Grantee has certain recordkeeping and reporting responsibilities under the FTZ Act, Board regulations, and instructions of the Executive Secretary.
 - (a) Accounts and Records The form and manner of keeping accounts on each zone shall be prescribed by the Board (19 U.S.C. §81p). Grantees shall maintain their accounts in accordance with generally accepted principles of accounting, and in compliance with any requirements of Federal, State, or local governmental agencies having appropriate jurisdiction over the Grantee (15 CFR 400.46).
 - (b) Annual Reports Each Grantee shall make available to the Board annually, and at such other times as the Board may prescribe, reports containing a full statement of the operations, receipts, expenditures, and such other information as the Board may prescribe (19 U.S.C. §81p(b); 15 CFR400.46 (d)).
 - (c) Operator Responsibilities to Grantee The Operator is responsible for maintaining records and making reports to the extent provided for in the operating agreement between the Operator and the Grantee.

Chapter 8

OPERATIONS IN ZONES

- 8.1 General Foreign and domestic merchandise of every description, except such as is prohibited by law, may, without being subject to CBP laws of the United States, except as otherwise set forth in the FTZ Act, be brought into a zone and stored, sold, exhibited, broken up, repacked, assembled, distributed, sorted, graded, cleaned, mixed with foreign or domestic merchandise or otherwise manipulated, or be manufactured except as otherwise set forth in the FTZ Act, and be exported, destroyed, or sent into the Customs territory of the United States therefrom in the original package or otherwise (19 U.S.C. §81c). Operations prohibited in a zone are described in Chapter 11 FTZM.
 - (a) Operations Involving Nonexempt Merchandise Importations brought into a zone for a purpose other than those listed in 19 U.S.C. §81c, are not exempt from duty under the FTZ Act (Section 2.6 FTZM). Such merchandise is not subject to any restriction under the FTZ Act, and any operation may be performed on them that would be permitted for merchandise not in a zone and not in CBP custody.
 - (b) Supervision by Operators The Operator shall supervise all manipulations, manufacturing, destruction, exhibition, physical and procedural security, and conditions of storage in the zone as required by law and regulations. Supervision by the Operator shall be that which a prudent manager of a storage, manipulation, or manufacturing facility would be expected to exercise, and may take into account the degree of supervision exercised by the User having physical possession of zone merchandise (19 CFR 146.4(a)).
 - (c) Safekeeping of Merchandise and Records The Operator, at its liability, may allow the zone importer or owner of the goods to store, safeguard, and otherwise maintain or handle the goods (19 CFR 146.4(c)).
 - (d) Transfer of Title The Operator shall maintain records of all transfers of title between or among Users or Operators. Transfers of title do not constitute a retail sale as prohibited by the 19 U.S.C. §810(d). While the transfer of title within a zone is permissible and there is no CBP form that is required to authorize such a transfer, records must be maintained in order for CBP to evaluate the information to ensure compliance with all necessary laws. All CBP entries must be made by the party that has the right to make entry (19 U.S.C. §1484).
 - (e) Third Party/Toll Processing The Operator or the User may cause to be admitted to the zone and processed and manufactured therein merchandise which the Operator and/or User does not own or have title to. As long as the processing/manufacturing activities are authorized by the Board, the title to merchandise is not relevant for this purpose. However, at the time of entry, entries may only be made by the party who has the right to make entry (19 U.S.C. §1484).

- (f) Storage Merchandise that is not owned by the Operator may be stored in a zone or subzone. At the time of entry, entries may only be made by the party who has the right to make entry. Any transfer to another zone or export may be made by the Operator.
- 8.2 Storage Conditions Generally, merchandise should be stored in a zone in such a manner that (1) the zone will be safe for personnel working therein; (2) the merchandise will be protected from theft or other loss, damage, and deterioration; (3) checks, inventories, and audits can be efficiently conducted by CBP and zone personnel; and (4) the zone will not endanger the public health or safety.
 - (a) Safety and Sanitary Conditions Grantees and Operator s shall operate zones and buildings in a safe, sanitary manner in compliance with all applicable local, state and federal regulations (15 CFR 400.28 (a)(4)). The Port Director may suspend the activated status of the zone or the receipt or delivery of merchandise until such a deficiency is corrected (19 CFR 146.82 (a)(2)). The Operator should consult with the state and local officials to ascertain or clarify local safety and sanitary requirements, such as fire codes, zoning laws and similar provisions.
 - (b) Separate Storage The Port Director may require segregation of any zone status merchandise whenever necessary to protect the revenue or properly administer U.S. laws or regulations (19 CFR 146.51). Port Directors may require separate storage of zone-restricted merchandise if there is the possibility that the merchandise could become combined with or confused with other merchandise of the same type. The Port Director should determine the type of storage required upon presentation of the CBPF 214 requesting zone-restricted status. The key consideration here is to ensure that the revenue is protected and restricted merchandise programs are protected and the feasibility of compliance reviews. A compliance review must be feasible, as opposed to a full audit. CBP must be able to perform a compliance review and an audit to prevent the merchandise from entering the U.S. Examples are fungible merchandise such as oil or grain. There is a clear obligation that the merchandise must be identifiable.

Merchandise which has been constructively transferred to Customs territory and is awaiting removal from the zone, will not be further manufactured or manipulated in the zone, but will be segregated or otherwise identified by the Operator as constructively-transferred merchandise (19 CFR 146.71(c)). The Operator shall segregate, mark, or otherwise secure damaged merchandise to preserve its identity as damaged merchandise (19 CFR 146.53(e)). Merchandise which is not in zone status for any of the reasons stated in Section 5.2 FTZM shall be segregated from zone status merchandise.

Manner of Storage – Aisles will be established and maintained, and doors and entrances left unblocked, for access by CBP personnel and other persons in the performance of their official duties (See 19 CFR 146.4(f)). All merchandise stored in a zone shall be identified by a merchandise location system that clearly identifies the location of all UIN

or zone lot number merchandise. There is no legal requirement for physical segregation of foreign and domestic merchandise, nor to limit storage to a single location.

- (1) For merchandise maintained under a zone lot number system only, packages shall be marked with the zone lot number so that the merchandise can be traced to a particular e-214/CBPF 214 (19 CFR 146.37). Generally, these marks shall be on all packages, but the Port Director may waive this requirement if the packages, or unitized or palletized units of packages, are marked at sufficient intervals of space to allow a proper spot check, quantity count, or audit.
- (2) Packages that are accounted for under a CBP-authorized inventory method other than specific identification must be adequately identified by product markings, bin markings, the unique character of the merchandise, or other means so CBP and the Operator can conduct an inventory count.
- (c) Storage Outside Activated Area Except merchandise covered by a temporary removal permit under Section 9.2 FTZM, no foreign status merchandise shall be stored, even temporarily, outside the activated area. Except in the case of domestic status merchandise for which no permit is required, such storage will be treated by CBP as a transfer from the zone without a permit. Similarly, except in the case of domestic status merchandise for which no permit is required, no merchandise shall be stored inside the activated area without a CBP permit of admission, temporary deposit, direct delivery or similar permit. Whenever domestic status merchandise is placed inside the activated area, the inventory records shall be adjusted as set forth in 19 CFR 146.22. Upon the removal of domestic status merchandise for storage outside the activated area, the inventory records shall be adjusted as set forth for in 19 CFR 146.24. Since CBP is not responsible for the administration of state ad valorem tax laws, the Operator should obtain the requirements for storage of merchandise for such benefit from the state tax authority where such legislation exists.
- 8.3 Permit for Manipulation, Processing, or Manufacture No merchandise, other than domestic status merchandise for which no permit is required under 19 CFR 146.43(a), will be manipulated, manufactured, exhibited or destroyed in a zone in any manner or for any purpose except under CBP permit as set forth in 19 CFR 146, Subpart E. A permit will be required for the exhibition, manufacture, or manipulation of domestic status merchandise, when (1) it is required by the Commissioner, or (2) when it is to be mixed or combined with merchandise of another zone status (19 CFR 146.43(b)). For the purposes of filling out the CBPF 216, "manufacturing" means any manufacture or processing as defined in 15 CFR 400.2 (g), (k) which include activities that result in a substantial transformation of a foreign article to a new and different article having a different name, character, and use, or which causes a change in its HTSUS classification of the merchandise or in its eligibility for entry for consumption (15 CFR 400.2 (g), (k)).

"Exhibition" means the display of merchandise in the zone to prospective buyers or other interested parties. Board approval must be received prior to changing previously approved processing or manufacturing operations to make new products, significantly expand production, change sourcing of materials to be used or source materials that will be subject to inverted tariffs (15 CFR 400.28(a)(2)(3)). The request will be submitted according to procedures shown in 15 CFR 400.32(b) and evaluated on the criteria shown in 15 CFR 400.31.

(a) Application - Prior to any action, the Operator shall file with the Port Director an application (or blanket application) on CBPF 216 for permission to manipulate, manufacture, destroy or exhibit merchandise in a zone (19 CFR 146.52(a), 146.9). For instructions in preparing CBPF 216, see Appendix FTZM. The description of the intended manipulation or manufacture should be detailed enough to enable the Port Director to determine whether a permit should be given. The Port Director may return the application to the Applicant for a more precise description, if needed to administer any law, regulation, or policy for which CBP is responsible.

A blanket application may be made for a period of up to one (1) year for a continuous or repetitive operation. If the operation is changed during the period so that the application as approved no longer describes the operation, a new application shall be filed (19 CFR 146.52(b)(2)).

- (1) Exhibition and Examination The User may, with the concurrence of the Operator and upon application approved by the Port Director on a CBPF 216, exhibit and examine merchandise in the zone, or allow prospective purchasers to examine merchandise in the zone.
- (2) Samples The User may, with the concurrence of the Operator and upon application approved by the Port Director on a CBPF 216, sample merchandise in the zone or allow perspective purchasers to sample merchandise and remove samples from the zone. Samples taken under the authority of this permit may be removed from the zone without payment of duty if (1) the merchandise is temporarily removed from the zone under the authorized temporary removal procedures; (2) sampling can be done without the need to repack the merchandise; and (3) the merchandise is not restricted merchandise.

Additionally samples may be removed from the zone under HS 9811.00.60. An entry must be filed for samples taken from the zone under this provision; all requirements of 9811.00.60 must be adhered to and each sample taken from the zone under this provision must be presented to an import specialist for approval of the mutilation.

Otherwise, the User must file a duty-paid consumption entry to obtain a permit to remove the sample.

(3) Temporary Removal - See Section 9.2 FTZM.

- (4) Concurrence of Executive Secretary for Manufacturing or Processing An application for a new manufacturing or processing operation must have the concurrence of the Board. The purpose of the concurrence is to assure that there are no public interest concerns in the new operation (15 CFR 400.28(a)(2), (3)).
- (b) Decision on Application A Port Director will approve the application unless (1) the proposed operation would be in violation of law or regulation, (2) the place designated for its performance is not suitable for preventing loss of identity or status of the merchandise, or (3) the Board has not granted approval of a new manufacturing operation. The Port Director may disapprove any application, or may deny a blanket application and require instead an individual application (19 CFR 146.52(b)(1), (2)).
- (c) Revocation of Approval A Port Director may revoke approval of any application for cause as defined in approval criteria at 19 CFR 146.52 (b), or may require the Operator to file an individual application in lieu of an existing blanket application. In such a case, the Applicant may appeal the adverse ruling pursuant to the hearing provisions of 19 CFR 146.82(b) (2). The revocation shall remain in effect pending the decision on the appeal (19 CFR 146.52(c)).
- (d) Recordkeeping In the case of a blanket application, the Operator shall maintain a record of an approved manipulation, manufacture, or exhibition in a separate yearly blanket CBPF 216 file or in its inventory control and recordkeeping system so as to provide an accounting and audit trail of the merchandise through the approved operation (19 CFR 146.52(d)(2)). If it is an individual CBPF 216 that relates to a zone lot number, the CBPF 216 will be filed in that specific zone admission CBPF 216 file. If the zone is operating under any other CBP authorized inventory system, the CBPF 216 is filed with the FTZ inventory records (19 CFR 146.52(a)). In its supervision of the operation, the Operator shall record on CBPF 216 the results of an approved manipulation, manufacture or exhibition (other than by a blanket application) unless a Port Director chooses physically to supervise the operation.
 - (1) Sending of Record to Port Director Where necessary the Port Director may request from the Operator a copy of CBPF 216 showing the results of the manipulation recorded by the Operator, before accepting or liquidating any entry.
- 8.4 Destruction Under Permit The Operator is not responsible under its Operator's bond for any loss of merchandise or any merchandise which cannot be located in the zone, if the Port Director is satisfied that the merchandise was never received in the zone, was not removed without proper permit from the zone, or that it was lost or destroyed in the zone through fire or other casualty, evaporation, spillage, leakage, absorption, or similar cause, and did not enter the commerce of the U.S. (19 CFR 146.53(c)(1)). The Port Director may issue a permit for the destruction of merchandise in any zone status. Any residue of destruction which has no commercial value may be transferred to Customs territory for disposal (19 CFR 146.52(e)). It

will be so transferred without entry and free of duty and treated as a non-importation under 19 CFR 158.11.

- (a) Alcoholic Beverages Distilled spirits, wines, and fermented malt liquors may not be admitted in zone-restricted status for the sole purpose of destruction (19 CFR 146.44 (a)). Such merchandise admitted to the zone in zone-restricted status for other purposes or in any other zone status may be destroyed in the zone only with the approval of the Regional Director, Compliance, TTB (Fourth Proviso, 19 U.S.C. §81c and Section 11.5(d)(4) FTZM). The procedure for obtaining approval by the TTB for destruction is set forth in 27 CFR 252.35 through 252.38, 19 CFR 146.44(c)(2) and see Section 11.5(d)(4) FTZM.
- (b) Application for Destruction Prior to any action to destroy merchandise in a zone, the Operator shall file with the Port Director an application on CBPF 216 for permission to destroy the merchandise. A yearly blanket CBPF 216 may be authorized. The proposed method of destruction shall be stated in the application, and should be sufficient to assure that the merchandise will be completely destroyed. A sample of a properly filled out CBPF 216 for destruction is shown in the Appendix to the FTZM.
 - (1) Method of Destruction The method of destruction shall be appropriate to the nature of the merchandise so as to assure its complete destruction and to assure that there is no recoverable residue of any commercial value. If destruction is not complete or there is recoverable residue having commercial value, the transaction will be treated as a manipulation and so recorded in the inventory and recordkeeping system of the Operator.

Incomplete destruction does not meet the "destruction" condition of zone-restricted status. However, partial destruction, coupled with exportation or storage of the valuable waste, may meet the condition of "exportation, destruction, or storage" in the Fourth Proviso, 19 U.S.C. §81c.

Residue or scrap from destruction of zone-restricted merchandise may not be entered for consumption, unless the Board finds it to be in the public interest. (See Section 11.3 FTZM).

Destruction shall be carried out in a safe and sanitary manner, and any residue disposed of in accordance with the applicable State and Federal laws.

- (c) Decision on Application The Port Director shall approve the application unless
 - (1) the proposed destruction would be in violation of law or regulation,
 - (2) the place designated for its performance is not suitable for preventing the loss of the identity or status of the merchandise, or for safeguarding the revenue, or
 - (3) the Port Director is not satisfied that destruction will be effective, complete, and carried out in a safe and sanitary manner.

The Port Director may disapprove or revoke approval of any application (19 CFR 146.52(b)). However, if an approved application is subsequently rescinded by the Port Director for any reason, the Applicant may appeal the adverse ruling pursuant to the hearing provisions of 19 CFR 146.82(b)(2). The rescission shall remain in effect pending the decision on the appeal (19 CFR 146.52(c))

- (d) Supervision Since destruction, by its nature, leaves little or no physical evidence of the activity, the Port Director may choose to physically supervise the destruction operations, particularly destructions of restricted merchandise. At the discretion of the Port Director, the Operator may be allowed to supervise the destruction. If the Operator supervises destruction, it shall be done as set forth in 19 CFR 146.4(a).
- (e) Removal from Zone for Destruction The Port Director may permit destruction to be done outside the zone, in whole or in part, at the risk and expense of the Applicant and under such conditions as are necessary to protect the revenue, if proper destruction cannot be accomplished within the zone (19 CFR 146.52(e)). In such cases, destruction may be supervised by CBP as set forth in 19 CFR 158.43 (d) and 19 CFR 101.2 (c), in which case the certified copy of CBPF 216 shall be returned to the Operator for retention in the recordkeeping system. Transfer of the merchandise to the destruction site shall be done under the procedures of a local control system.
- (f) Recordkeeping of Destruction If CBP supervises the destruction, the supervising CBP personnel shall certify the destruction on CBPF 216, and provide a copy of the form to the Operator for filing in its recordkeeping system (19 CFR 146.52(a)). If the Operator supervises the destruction, the Operator shall certify the destruction on that form (19 CFR 146.52(d)(1)). For each occurrence under an individual or yearly blanket CBPF 216 procedure, records must be maintained.
- 8.5 Merchandise Marking Requirements Every article of foreign origin or its container imported into the Customs territory of the United States shall be marked in a conspicuous place

and in such manner as to indicate to the ultimate purchaser in the United States the English name of the country of origin of the article (19 U.S.C. §1304 and 19 CFR 134.11). Goods not so marked may be brought into a zone to be properly marked under a permit to manipulate issued by the Port Director. Similarly, goods which are improperly or falsely marked may be brought into a zone under a permit to manipulate to correct or remove such marking so as to comply with the laws and regulations (19 CFR 134.13(b)).

- (a) U.S. Origin Marking for Goods Manufactured in a Zone If further work or material has been added to an imported article in a zone which has effected a substantial transformation of the article into a new and different article of commerce, it may be considered to be no longer an article of foreign origin, for marking purposes, when it is transferred to the Customs territory (See 19 CFR 134.1(b)). The United States becomes the country of origin of such articles for marking purposes. Such articles need not be marked, under 19 U.S.C. §1304, to show the United States as the country of origin, and CBP has no authority to approve or deny the marking "Made in U.S.A." However, goods of U.S. origin in a zone shall not be marked to show a country of origin other than the United States. Also, goods of U.S. origin manufactured in a zone may be subject to marking requirements administered by the Federal Trade Commission (FTC). Note that the FTC has jurisdiction in this area.
- (b) False Country of Origin Marking CBP personnel shall not approve an application on a CBPF 216 to mark merchandise with a false and/or deceptive country of origin marking, even if the merchandise is destined for export from the U.S (OCOD 89-4, CBP Bulletin, Vol. 23, No. 43, October 25, 1989 and see Section 11.6(u) FTZM).
- (c) Copyrighted Articles Copyrighted articles made without the authorization of the copyright owner are deemed to be infringements and their importation into the U.S. is prohibited by CBP (19 CFR 133.42). Articles made abroad displaying a trademark identical to one owned by a U.S. citizen, corporation or association is subject to seizure as a prohibited importation.
- 8.6 Abandonment, Bankruptcy, and Absconding 19 CFR 146.9 requires the written concurrence of the Operator on applications for permission to admit, manufacture, manipulate, exhibit, or destroy zone merchandise. 19 CFR 146.71(a) requires the Port Director to authorize delivery of merchandise only to the Operator, who may release the merchandise from the zone. These provisions place on the Operator the primary responsibility for disposition of merchandise which has been abandoned in a zone or whose owner is insolvent, has absconded, or is in arrears in payments owed to the Operator. No entry or permit to transfer or constructively transfer such merchandise from the zone for the purpose of immediate abandonment to the Government shall be accepted or approved by CBP.
 - (a) Abandonment If merchandise is abandoned, in writing, by its owner to the Operator, the Operator has the right to authorize the exportation, destruction, or sale of the merchandise. The Operator may dispose of the merchandise in its own name on the proper form in accordance with pertinent laws and regulations. If the merchandise is to

be entered for consumption the owner or purchaser will be held liable for any duties, taxes, and deficiencies due. Auction sales will be conducted by the Operator, or representative thereof, and not by CBP or a CBP contractor.

- (b) Insolvency, Absconding, or Arrearage If merchandise has not been abandoned in writing to the Operator, but the owner or the person in whose account the merchandise is held in the zone apparently has absconded, is insolvent, or is in serious arrears in payments owed to the Operator, the Operator must take legal action under the laws of the state in which the zone is located to dispose of the merchandise. If the Operator is authorized to sell the merchandise to recover a debt, the buyer will have title to the goods to dispose of them at his or her option.
- (c) Bankruptcy If the owner or person in whose account the merchandise is held in the zone declares bankruptcy, the merchandise will be disposed of according to the applicable bankruptcy law.
- 8.7 Security of Zone Each Grantee, Operator, or their designee shall provide and maintain in connection with the zone adequate enclosures to segregate the zone from the Customs territory for the protection of the revenue, together with suitable provisions for ingress and egress of persons, conveyances, vessels, and merchandise (19 U.S.C. §811(f)). In 15 CFR 400.41, the Board delegates to Port Directors the enforcement responsibility for physical security and access requirements for zone projects activated in their ports. To achieve the necessary physical security of the zone, recommendations set forth in TD 72-56, Standards for Cargo Security will serve as guidelines only, because significant improvements in security methods have been developed since 1972.
 - (a) CBP Security Requirements The Operator shall maintain the zone and establish procedures adequate to ensure the security of merchandise located in the zone in accordance with applicable CBP security standards and specifications (19 CFR 146.4(e)). The Operator is responsible for safekeeping of merchandise and records concerning merchandise admitted to a zone. The Operator , at its liability, may allow the User or the owner of the goods to store, safeguard and otherwise maintain or handle the goods and the inventory records pertaining to them (19 CFR 146.4(c)).
 - (b) Guard Service 19 U.S.C. §81d, directs the Secretary of the Treasury to assign to the zone the CBP personnel necessary to protect the revenue. However, in view of the Grantee's responsibility in 19 U.S.C. §81f, the Operator through its contract with the Grantee, is authorized to provide guards or contract for guard service to safeguard the merchandise and ensure the security of the zone. This authorization does not limit the authority of the Port Director to assign CBP personnel to protect the revenue under 19 U.S.C. §81d (19 CFR 146.4(g)).
 - (c) Cargo Security Standards and Specifications The standards of security to be maintained by the Operator are the same as those used by CBP in the security survey required for approval of the application to activate the zone. These standards, set forth in

TD 72-56, "Standards for Cargo Security," serve as a guide, and each of the standards includes specifications which suggest means by which standards may be met. Copies of TD 72-56 should be obtained from the Port Director.

CBP Security Standards for Buildings permit the walls of the building itself to be considered the perimeter of the activated area, and that fencing need not be required for the facility to meet the Security Standards.

- High-Security Storage for Title II Firearms This description does not refer to pistols or revolvers having a rifled bore, or rifled bores, or weapons designed, made, or intended to be fired from the shoulder. Imported machine guns, destructive devices, certain other firearms, or any other weapon capable of being concealed on a person as covered by 26 U.S.C. §5845 may be imported for storage in a zone under a conditional import permit of TTB (TTB Form 6) for subsequent sale to military and law enforcement agencies or other use specified in 26 U.S.C. §5844 and 26 U.S.C. §5845 (See Sections 11.5(e) and 11.6(m) FTZM). These firearms may also be imported for re-exportation from a zone under the provisions of the International Traffic in Arms Regulations, 22 CFR 120. TTB will not issue an import permit for these firearms for storage in a foreign-trade zone until a special survey has been conducted by CBP to determine whether the facility meets certain standards of security that are higher than those for other zones (TD 81-69). CBP personnel shall conduct such surveys only upon request of the TTB. The standards for such high-security warehouses are shown in the Appendix to the FTZM. The surveying CBP personnel shall advise the Operator of the purpose of the survey and provide consultations concerning its progress and findings. If the Operator is able to make changes necessary to bring the facility into compliance with the standards within 1 month after the survey, the final report to the TTB will be delayed until the changes are made and verified by the surveying personnel.
- (e) Compliance Reviews of Cargo Security After the initial cargo security survey and follow-up conducted upon activation of the zone, CBP personnel shall monitor the Operator's maintenance of security standards, specifications, and requirements through compliance reviews and audits.
- (f) Full Security Surveys Full cargo security surveys may be carried out periodically by CBP in selected zones. Such a survey can be expected when:
 - (1) A serious breach of security has been identified or a major theft has occurred at a zone; or
 - (2) The Port Director contemplates suspension of activation or revocation of a zone permit, or recommendation for revocation of a zone grant because the Operator does not provide secured facilities or properly safeguard merchandise in the zone (19 CFR 146.82(a)(5)); or

- (3) At periodic intervals in the case of zone facilities for the high-security storage of Title II firearms. (See Section 8.7(d) FTZM).
- (g) Action on Cargo Security Deficiencies The cargo security standards and specifications in TD 72-56 are guidelines, rather than regulations, for acceptable physical and procedural security in international commerce. Therefore, factors outside the standards and specifications may be considered in making a judgment as to whether there is a deficiency which constitutes a default in the Operator's bond. For instance, a deficiency in one specification which is counterbalanced by another security measure so as to preserve overall security of the facility may not be a default. Similarly, a deficiency which is in keeping with the nature of the merchandise (e.g., a 6-foot fence surrounding an activated area containing low-value, high bulk merchandise with low theft potential) may not be a default.
 - (1) Deficiencies in zone security are defaults not involving merchandise subject to liquidated damages of \$1,000 for each default (See Section 12.7 FTZM).
 - (2) If merchandise is removed from a zone without permit, or cannot be accounted for, through a deficiency in security, the loss of the merchandise breach is a default involving merchandise, and the security deficiency will be considered an aggravating factor in the consideration of any petition for relief from liquidated damages (See Section 12.14(a) FTZM).
 - (3) Action will ordinarily be taken by CBP to suspend a zone permit or activated status on the grounds set forth in 19 CFR 146.82(a) only for a major deficiency or deficiencies, pending the correction of the deficiencies. Action to revoke the activated status will ordinarily be taken only in the case of chronic or uncorrected deficiencies which have resulted in a major loss or losses of merchandise (See Sections 13.8-13.11 FTZM).
- (h) Theft Reporting The Operator is required to immediately report to the Port Director any theft or suspected theft from the zone, including thefts or suspected thefts of domestic status merchandise (19 CFR 146.53(a) and (b)). Theft of merchandise from a zone is usually considered a theft in interstate or foreign commerce, a felony under 18 U.S.C. §549 punishable by a fine of up to \$250,000 or imprisonment for not more than 10 years, or both (18 U.S.C. §549, 3571).
- (j) List of Persons The Port Director may at any time make a written demand upon the zone Operator to submit, within 30 days, a written list of the names, addresses, social security numbers, and dates and places of birth of personnel or key persons having a direct or indirect financial interest in the zone operation, and of persons employed in the carriage, receipt or delivery of merchandise in zone status, whether employed by the Operator or the User (19 CFR 146.7(g)). The Port Director may request a background investigation to be

made of any of the persons on the list, if necessary, to protect the revenue. Operators are encouraged to obtain Privacy Act waivers from all employees

Chapter 9

TRANSFER OF MERCHANDISE FROM ZONE

- 9.1 General Foreign and domestic merchandise which has been brought into a zone for certain purposes may thereafter be exported, destroyed, or sent into the Customs territory in the original package or otherwise. When foreign merchandise is sent from a zone into the Customs territory of the United States, or is exported to Canada or Mexico subject to the NAFTA Duty Deferral found in 19 CFR 181.53, it shall be subject to the laws and regulations of the United States affecting imported merchandise (19 U.S.C. §81c(a)). Domestic status merchandise, whether or not it has been combined with or made part of merchandise in any other status, may be brought back into the Customs territory free of duty, or tax (Second Proviso, 19 §81c(a)). "Transfer" means to take merchandise with zone status from a zone for consumption, transportation, exportation, warehousing, cartage or lighterage, vessel supplies and equipment, admission to another zone, and like purposes (19 CFR 146.1(b)).
- 9.2 Temporary Removal Upon permit by the Port Director, zone status merchandise may be temporarily removed to the Customs territory from the zone for up to 120 days for repair, restoration, or incidental operation. "Incidental operation" means inspection, analysis, testing, calibration, measurement, sampling, photography, cleaning, repacking, or similar process which would not constitute "manufacturing or production" under drawback law (19 U.S.C. §1313(a) and HQ Ltr 214189 dated August 31, 1982).
 - (a) Application Application for temporary removal shall be made by the owner or purchaser on CBPF 216, appropriately modified. Each application must be made on an individual basis and shall not be authorized under a yearly blanket CBPF 216. The operation the merchandise is to undergo outside the zone must be described in sufficient detail to enable the Port Director to determine whether the application should be approved. If the application meets the conditions set forth below and is administratively acceptable, it will be approved by the ort director. A sample of a properly filled out CBPF 216 is shown in the Appendix to the FTZM. Transfer of merchandise may be under a local control system (19 CFR 146.66).
 - (b) Conditions The temporary removal transaction covered by the application must meet all of the following conditions:
 - (1) The merchandise may not be removed from the zone before the application is approved;
 - (2) Except for repairs, no other merchandise may be added to, combined with, or incorporated in the removed merchandise, and no value may be added to it;
 - (3) The procedure may not be used to circumvent the FTZ Act through subjecting the merchandise to any operation or transaction which would not be

authorized while in the zone, such as an excluded process of treatment or retail trade under 19 §810;

- (4) The merchandise must be returned to the same zone within 120 days and so reported to the Port Director. No extensions beyond the 120 days will be granted. The merchandise may be sent to any location within Customs territory so long as it is returned to the same zone and reported within the 120 day period. The merchandise remains the responsibility of the Operator under the Operator's bond while outside the zone. The merchandise may not be entered for consumption while it is temporarily removed from the zone (See Section 11.7(h).
- (5) Quantities removed and returned must be balanced on a transaction-bytransaction basis. However, fungible merchandise removed to a particular destination under different transactions and thereafter returned may be tracked using the FIFO inventory method.
- (c) Return to Zone When all of the merchandise covered by an application has been returned to the zone, the Applicant shall certify the return on CBPF 216 and forward it to the Port Director to close out the temporary removal record. Upon return to the zone, the merchandise shall be received in the same zone status it had upon temporary removal.
- (d) Violation of Conditions If the merchandise is not timely returned to the zone or otherwise fails to meet the conditions of temporary removal, the merchandise shall be considered to have been transferred from the zone without a permit, as required by 19 CFR 146.51 and 19 CFR 146.71(a). The violation will be treated as a default involving merchandise under the Operator's bond (See Section 12.6 FTZM).
- 9.3 Domestic Status Merchandise Domestic status merchandise which has not been combined with or made part of merchandise in another zone status may be transferred from the zone without a CBP permit, unless a permit is ordered by the Commissioner (19 CFR 146.43(b)). The merchandise will be considered as having been transferred from the zone and as having lost zone status when it is: (1.) physically transferred from the zone, as evidenced by a receipt for removal by a carrier, owner, purchaser, or other appropriate party; and (2.) recorded in the inventory control and recordkeeping system(s) as transferred from the zone (19 CFR 146.24(a) and see Section 9.5 and 9.16(e) FTZM) for comparison with treatment of foreign status merchandise.
- 9.4 Articles Exempt from Entry Requirements Merchandise in foreign status may be manufactured in a zone into an article which is exempt from entry requirements. An example of such an article is electricity produced in the zone (C.S.D. 83-49 and see General Headnotes of the HTS). In such cases the merchandise may be transferred from the zone only upon issuance of a permit by the Port Director on a CBPF 216, appropriately modified. The Description of Activity in the form shall contain a citation of the authority to exempt the article(s) from entry.

- 9.5 Constructive Transfer "Constructive transfer" is a legal fiction which permits acceptance of a entry for merchandise in a zone before its physical transfer to the Customs territory (19 CFR 146.1(b)). The legal fiction is necessary because of the entry requirement and procedures imposed pursuant to 19 U.S.C. §1484. Zone status merchandise is exempt, for certain purposes, from CBP laws, pursuant to 19 U.S.C. §81c. The law would require that the merchandise be removed from the zone and held pending examination and release under 19 U.S.C. §§1484 and 1499. Under constructive transfer, this delay and double handling is eliminated. The merchandise loses its zone status, and is deemed to be no longer subject to, or covered by, the FTZ Act upon constructive transfer (C.S.D. 79-249). However, the Operator is still responsible for the merchandise until its physical transfer from the zone (19 CFR 146.4(a), and see Section 9.16(e) FTZM).
 - (a) Procedure The Port Director shall accept receipt of any entry in proper form as set forth in 19 CFR 146 Subpart F, and the merchandise described therein will be considered to have been constructively transferred to the Customs territory at that time, even though the merchandise remains physically in the zone. If the entry is thereafter rejected or canceled, the merchandise will be considered at that time to be constructively transferred back into the zone in its previous zone status (19 CFR 146.61). This procedure replaces a procedure in effect prior to 1986 where the Operator was required to apply for constructive transfer. Since there was rarely any reason to deny constructive transfer that could not also be satisfied by rejecting or canceling the entry, constructive transfer was made automatic under 19 CFR 146.61 upon receipt of the entry.
- 9.6 Entry Procedures An entry is that documentation required by 19 CFR 142.3 to be filed with the appropriate CBP personnel to secure the physical transfer of foreign or zone-restricted status merchandise from the zone (19 CFR 146, Subpart F). This section covers entry procedures peculiar to zones, but does not attempt to cover all entry procedures pursuant to 19 CFR 141 19 CFR 144 that may be applicable to entries for zone status merchandise.
 - (a) Place of Filing Entry documentation shall be presented to the appropriate CBP office at the location designated by the Port Director, as set forth in 19 CFR 141.62(a). An entry for zone status merchandise must be presented to CBP personnel within a port of entry, even if the zone is distant from the port, except as set forth in Section 2.9 FTZM.
 - (b) Administrative Requirements All entry forms shall be filled out completely, accurately, and legibly. All signatures required on the form and supporting documents shall be valid signatures. All supporting documentation shall be filed with the entry or otherwise timely filed. An entry will not be accepted nor a permit for transfer from the zone issued unless and until all administrative requirements are met.
 - (c) Penalties for False or Inaccurate Information If the information in any entry document is not true or accurate, the importer or agent may be subject to civil penalties under 19 U.S.C. §1592 for fraudulent, negligent, or grossly negligent acts or omissions in connection with the entry. In addition the importer or agent may be subject to a fine or

imprisonment, or both, under 18 U.S.C. §541, 18 U.S.C. §542, 18 U.S.C. §543, 18 U.S.C. §545, or other provisions of Title 18, U.S.C. (See Section 13.15 and 13.16 FTZM).

- 9.7 Entry for Consumption In order to transfer zone merchandise into the Customs territory for consumption, the importer of record shall file a consumption entry under the procedures of 19 CFR 141 and 19 CFR 142. In addition, an appraisement, informal, or electronic entry for consumption may be filed as set forth in 19 CFR 143. An informal consumption entry may be filed for samples that do not qualify for zone removal (19 CFR 10.151 and Section 8.3 FTZM). Also, an informal entry may be filed for payment of internal revenue tax on any sample of beer in excess of 8 ounces, wine in excess of 4 ounces, or distilled spirits in excess of 2 ounces, taken for the purpose of soliciting orders of products of foreign countries (27 CFR 251.49)).
 - (a) Entry Filing Procedure In the case of formal entries, there are 3 basic procedures to initiate the entry filing process: entry, entry summary which acts as an entry/entry summary, and immediate delivery.
 - (1) Entry An entry is filed on CBPF 3461, Entry/Immediate Delivery (19 CFR 142.3(a)(1)). Instructions for preparing this form are found in CD 099 3550-029, and a properly filled-out sample is shown in the Appendix to the FTZM. The following special instructions should be noted in filling out CBPF 3461.
 - 1. Block 1 The arrival date is the date of presentation of the entry.
 - 2. Block 3 Insert Entry Type Code 06.
 - 3. Block 14 Insert Facilities Information Resource Management Systems (FIRMS) code. This identifies the zone in which the merchandise is located. The FIRMS code is assigned by the ort director.
 - 4. Block 15 Insert "FTZ" with the three digit zone number and alpha character if applicable with no spaces (ex. FTZ123, FTZ123A, FTZ001, FTZ001A).
 - 5. Block 22 Show the zone status for each line.
 - 6. Block 23 Since there is no manifest, insert the number of packages. If loose freight, insert the number of pieces. If in bulk, insert the quantity in the units in which the merchandise was sold or shipped.
 - 7. Blocks 12, 13, 16, 17, 18, and 21 should not be filled out.

8. Block 26 - If the product is of foreign origin, or in privileged foreign status, use the MID of the foreign manufacturer. If the product is manufactured in the foreign-trade zone, use the following MID construction: CD 099 3550-055, November 24, 1986, "Instructions for Deriving Manufacturer/Shipper Identification Code".

Country of Origin: US

Manufacturer Name: FTZ

Street Address: Zone number (if subzone, Place subzone designation in City field in front of zone User's name)

City: Zone User's name

Example: Smith Manufacturing, Inc., located in subzone 215G. USFTZ215GSM

Example: Jones & Co., located in general-purpose zone 220. USFTZ220JON

The entry shall be accompanied by an invoice, packing list, or other documentation required to obtain a permit by the Port Director to transfer the merchandise from the zone. An entry summary shall be filed within 10 working days after time of entry, as set forth in 19 CFR 142.12(a) and 146.62(a).

- (2) Entry/Entry Summary At the option of the importer, an importer may file the entry summary documentation at the time of entry (19 CFR 142.12(a)). The Port Director may also require an importer to file the entry summary documentation at the time of entry under certain circumstances (19 CFR 142.13). Otherwise, the entry summary will be filed 10 working days after CBPF 3461, as set forth in 19 CFR 142.12 and 19 CFR 146.62(a). Instructions for preparing CBPF 7501 are found in CD 099 3550-061. A properly filled-out sample is shown in the Appendix to the FTZM. The following special instructions for zone status merchandise are noted in filling out CBPF 7501:
 - 1. Block 2 Insert entry type code 06 or 08 (NAFTA Duty Deferral). This entry type code must be used for all merchandise transferred from a zone for consumption; including visa, and AD/CVD merchandise.
 - 2. Block 10 Regardless of marking or other country of origin considerations, the country to be reported in block 15 is the

country of origin of the foreign status merchandise for which entry is required. If components of articles are from more than one country, report the foreign country of the components with the greatest value. When a single entry summary covers separately classifiable merchandise from more than one country, record the word "MULTI" in this block. To identify the country of origin in Block 28, directly below the line number, and prefixed with the letter "O", indicate a separate ISO (International Standard Country Code) for the country of origin corresponding for each line item. CD 099 3550-061, September 18, 1992, "Instructions for Preparation of CF 7501."

3. Block 13 - If the product is of foreign origin, or in privileged foreign status, use the MID of the foreign manufacturer. If the product is manufactured in the zone use the following MID construction: CD 099 3550-055, November 24, 1986, "Instructions for Deriving Manufacturer/Shipper Identification Code"

Country of Origin: US

Manufacturer Name: FTZ

Street Address: Zone number (if subzone, place subzone designation in City field in front of zone User's name)

Example: Smith Manufacturing, Inc., located in subzone 215G. USFTZ215GSM

Example: Jones & Co., located in general-purpose zone 220. USFTZ220JON

- 4. Block 8 Insert "FTZ" followed by zone or subzone number. (Also, for certain textiles and textile products, insert the zone admission number). (See Section 9.7(e) FTZM.)
- 5. Block 21 Insert Facilities Information Resource Management Systems (FIRMS) code. This identifies the zone where the merchandise is located. The FIRMS code is assigned by the Port Director.
- 6. Do not fill out Blocks 9, 11, 12, 14, 15, 16, 17, 19, and 20.
- (3) Immediate Delivery Port Directors shall not issue any Special Permit for Immediate Delivery from a zone unless authorized under 19 CFR 142.21. If such

- a Special Permit is issued, the procedures of 19 CFR 142.21 through 19 CFR 142.28 shall be followed.
- (4) Electronic Entry Filing Entry documentation may be transmitted electronically from a zone to a CBP facility for approval and signature by CBP and returned to the zone through use of a facsimile machine, provided one is available C.S.D. 87-2. Procedures for qualifying for the Automated Broker Interface (ABI) are specified in 19 CFR 143, Subpart A. Procedures for filing electronic entries and electronic entry summaries are found in 19 CFR 143, Subpart D. Such entries must be filed through ABI (TD 90-92). Specific procedures for filing electronic entries and entry summaries for zone merchandise were issued via administrative message #0696 dated 10/2/90, and became effective 6/30/91.
- (5) Supporting Documentation with Entry Summary CBPF 3461 or the entry/entry summary on CBPF 7501 will be accompanied by the entry documentation, including invoices, as set forth in 19 CFR 141 and 19 CFR 142. The person with the right to make entry shall submit any other supporting documents required by law or regulations that relate to the transferred merchandise and provide the information necessary to support the admissibility, declared values, quantity, and classification of the merchandise.
 - (i) Examination Invoice The Operator shall give a copy of the examination invoice filed upon admission of the merchandise to the zone to the person making entry to transfer the merchandise from the zone upon request of that person or the Port Director (19 CFR 146.37(b)).

The examination invoice presented by the importer upon admission of the goods to the zone may be used for making entry at the option of either the person making entry or the Port Director (19 CFR 146.32(b)). That is, the Port Director may demand that the examination invoice be used upon entry of the same merchandise. The examination invoice should be used when the merchandise has not been transformed in the zone into a new and different product. A copy of the invoice, modified to show the quantity and value of the merchandise being entered, may be accepted when merchandise covered by one examination invoice appears in two or more subsequent entries.

(ii) Waiver of Documentation - The Port Director may waive presentation of an invoice and supporting documentation required in 19 CFR 146.62(b)(1) with the entry or entry summary, if satisfied that presentation of those documents would be impractical, and the person making entry or the Operator either files invoices and supporting documentation with the Port Director or maintains and makes those records available for examination by CBP (19 CFR 146.62(c)). The Port

Director and the Operator or importer should develop a format of such documentation which is within the practical capability of the Operator or importer and which provides all the information required by CBP.

- (iii) Estimated Values If the declared values are predicated on estimates or estimated costs, that information must be clearly identified to CBP as required at the time an entry or entry summary is filed (19 CFR 146.62(b)(1)). Estimated values must be reconciled upon determination of the actual value using CBP reconciliation procedures.
- (6) Right to Make Entry Under 19 U.S.C. §484, only the "importer of record" has the right to make entry. The importer of record is the owner or purchaser of the goods, or when designated by the owner, purchaser, or consignee, a licensed broker. The owner or purchaser is any party with a financial interest in a transaction, including, but not limited to, the actual owner or purchaser of the goods, a buying or selling agent; or a person or firm which imports on consignment, or under loan or lease, or for exhibition at a trade fair, or for repair, alteration, or further fabrication of the goods (See CD 099 3530-002, June 29, 2001). Operator s may not make entry for zone merchandise unless they are the owner or purchaser or otherwise qualify as a party with a financial interest as noted above. Operators who do not otherwise qualify also may not designate a customs broker to make entry, since only an owner, purchaser, or consignee may make such a designation.
- (b) Entry Bond Merchandise will not be released for transfer to the Customs territory unless the importer of record has a single transaction or continuous bond on CBPF 301 Activity Code 1 Importer's bond, containing the bond conditions set forth in 19 CFR 113.62, on file with CBP (19 CFR 142.4(a)). A continuous bond may cover entries made at more than one port by the same principal. These types of bonds should not be confused with the Operator's bond (See Section 4.7(c)(2) FTZM).
 - (1) Bond Amount The amount of a consumption entry bond will be determined by the Port Director according to the guidelines in 19 CFR 113.13(b) and CD 099 3510-004. The minimum amount of a continuous bond is \$50,000. In the case of a continuous bond, the amount will be fixed in multiples of \$10,000 nearest to 10 percent of the duties paid by the importer during the previous calendar year, up to \$1,000,000 of the previous year's duties. If the amount paid during the previous calendar year was more than \$1,000,000, the amount will be fixed in multiples of \$100,000 nearest to 10 percent of the duties paid during the previous calendar year. If no imports were made during the previous year, the amount will be based on the importer's estimate of duties that will accrue on merchandise during the current calendar year. In no event shall the limit of liability amount of any continuous Activity 1 bond be less than \$50,000. In the case of a single transaction bond, generally, the bond amount will be the total value of the merchandise, plus the estimated amount of any duties, taxes, fees,

and other charges collectable by the Port Director. If restricted merchandise is being entered, the bond amount will be the entered value of the merchandise or value plus duty and tax, as appropriate. When merchandise is subject to other agency requirements, where failure to redeliver could pose a threat to public health and safety, the bond amount will be set at not less than 3 times the total entered value of the merchandise. These bond amounts will be periodically reviewed by the Port Director as set forth in 19 CFR 113.13(c), and increased security may be demanded as set forth in 19 CFR 113.13(c) or (d).

- (c) Deposit of Duties Estimated duties and taxes will be deposited by the importer of record with the CBP personnel designated to receive them, as set forth in 19 CFR 141, Subpart G.
- (d) Entry Processing and Release Consumption entries are processed by CBP on a selective basis according to the risk to the revenue and proper law enforcement represented by the merchandise or the transaction. At most ports, selective processing is done through the automated procedures of the Automated Commercial System (ACS). Key data elements are fed into the computer, which advises CBP personnel whether examination or further review by Import Specialists is needed before the goods are released for transfer from the zone. Because of the selective processing, the majority of entries are approved for release without examination within a day after the entry is filed. If the required duties and taxes have been paid, the merchandise is admissible into U.S. commerce, and CBP determines that no examination is necessary, a permit to transfer the merchandise into the Customs territory will be issued through an authorized signature on the CBPF 3461 and CBPF 7501, or other document authorized by the Port Director. In the case of electronic entry filing, a permit to transfer shall be transmitted electronically to the filer by CBP (19 CFR 143.34).
- (e) Admissibility and Restricted Merchandise "Restricted merchandise" as defined in CD 099 3250-005, is merchandise which may not be authorized for delivery from CBP custody without a special permit, or waiver thereof, by an agency of the U.S. Government. In the absence of the special permit, such merchandise is not admissible into the commerce of the U.S. Entry procedures for zone-restricted status merchandise are described in Section 9.15 FTZM.
- (f) Merchandise Examination Merchandise entered for consumption from a zone is examined by CBP much less frequently than other merchandise entered for consumption. This is because much of the merchandise (1) is in the same condition as it was when it was subject to examination upon admission (19 CFR 146.36), or (2) is the product of repetitive manufacturing or processing in the zone. Nevertheless, the Port Director is authorized to examine the merchandise upon entry for consumption as well as upon admission. If merchandise is ordered for examination, it may be required to be transferred to a CES or other location as designated by the Port Director (See Sections 6.7(d) and 2.9 FTZM). Upon completion of the examination, a permit to release shall be issued on CBP Form 3461, 7501, or other authorized form, or electronically in the case of

an entry filed electronically under the procedures of 19 CFR 143 Subpart D, if the merchandise is admissible and all requirements of law and regulation have been met.

- (g) Tariff Classification Zone merchandise which is entered for consumption is classified according to the Harmonized Tariff Schedule of the United States, with the General Rules of Interpretation taken in order.
 - (1) Privileged Foreign (PF) Status Merchandise Merchandise in PF status will be subject to tariff classification according to its character, condition, and quantity at the rate of duty and tax in force on the date of filing, in complete and proper form, of the application for admission to the zone and for privileged status designation (19 CFR 146.65(a)(1)). Merchandise will not be deemed to have any zone status, including privileged foreign status, unless and until it is admitted to a zone on an e-214/CBPF 214 (See Section 6.4 FTZM). Once privileged foreign status is selected, it can not be abandoned. Entries containing PF status merchandise must list the PF status merchandise on a separate line item(s) from any NPF status merchandise. (See Section 9.9(d) FTZM concerning the effect of privileged foreign status on U.S. Government importations of goods manufactured in a zone into other articles.).
 - (i) Waste Recoverable waste of privileged foreign status merchandise remaining after a casualty retains its privileged foreign status, and the merchandise is dutiable in its condition at the time privileged foreign status was requested (LD 86-0007, C.S.D. 86-7 and HQ 216240). It is distinct from recoverable waste of a manufacturing or manipulation operation under 19 CFR 146.42(b), which is changed to nonprivileged foreign status (See Section 9.7(g)(2) and 9.7(h)(1)(i) FTZM).
 - (iii) Duties Based on Relative Value Where two or more products result from the manipulation or manufacture of privileged foreign status merchandise in a zone, the duties and taxes applicable upon entry for consumption shall be distributed to the several products in accordance with their relative values at the time of separation, with due allowance for recoverable waste (First Proviso, 19 U.S.C. §81c (a)). In the calculation of relative values in the operations of a petroleum refinery in a zone, the time of separation is defined as the entire manufacturing period. The price of products required for computing relative values shall be the average per unit value of each product for the manufacturing period (19 U.S.C. §81c and see Section 11.6(j) FTZM).
 - (2) Nonprivileged Foreign Status Merchandise Nonprivileged foreign status merchandise set forth for in 19 CFR 146.42 will be subject to tariff classification in accordance with the character, condition and quantity of the actual product that is constructively transferred to the Customs territory at the time the entry or entry summary is filed with CBP (19 CFR 146.65(a)(2)).

Waste resulting from the manufacture or manipulation of merchandise in privileged foreign status, and domestic status merchandise which has lost its identity as such, lose their former statuses and become nonprivileged foreign status merchandise (19 CFR 146.42(b) and (c); see sections 5.7(d), 9.7(g)(1)(ii), and 9.7(h)(1)(i) FTZM).

- (3) Generalized System of Preferences To be eligible for duty-free treatment under the Generalized System of Preferences (19 CFR 10.171 through 10.178), several conditions must be met. 1) The merchandise must be imported directly from the Beneficiary Developing Country; (19 CFR 10.175); 2) nonprivileged foreign status merchandise entered for consumption must be in the same condition that it was in at the time of admission to the zone (C.S.D. 80-188).
- (4) Entireties Any reference in a HTS heading to an article shall include a reference to that article entered unassembled or disassembled. General Rule of Interpretation 2(a), HTS. This allows unassembled articles imported into the United States on the same conveyance on the same day to be treated as an entirety for tariff classification purposes. The FTZ Act creates an exception to this general rule governing entireties, with a choice of changing entireties being afforded a User. When nonprivileged foreign and privileged domestic status merchandise are attached or packaged together or withdrawn in an entirety in a zone, the nonprivileged foreign merchandise is classifiable at the rate of the entirety, even though the domestic portion is separately classifiable (C.S.D. 82-29). Election of privileged foreign status allows a User to have individual parts or components classified at the time privileged status is requested without reference to the classification of the entirety (C.S.D. 83-97).
- (h) Valuation Both privileged and nonprivileged foreign status merchandise are appraised in the same manner, as set forth in 19 CFR 146.65(b).
 - (1) Dutiable Value The dutiable value of merchandise transferred from a zone is the price actually paid or payable for the merchandise in the transaction that caused the merchandise to be admitted into the zone, plus the statutory additions contained in 19 U.S.C. §402(b)(1) less, if included, international shipment and insurance costs and U.S. inland freight costs. If there is no such price actually paid or payable, or no reasonable representation of that cost or of the statutory additions, the dutiable value may be determined by excluding from the total zone value any included zone costs of processing or fabrication, general expenses and profit, and the international shipment and insurance costs and U.S. inland freight costs related to the merchandise transferred from the zone. The dutiable value shall reflect the total value of the foreign merchandise used in the manipulation or manufacture of the entered merchandise. In order to arrive at the dutiable value, a deduction from the transaction value of the foreign merchandise or total zone value, as appropriate, for recoverable and irrecoverable waste or

scrap, generated as a result of the processing performed in that zone, will be permitted (19 CFR 146.65(b)(2)).

When items are combined in the zone, the dutiable value of the merchandise transferred from the zone shall be the sum of the value of the foreign status materials used in the manipulation or manufacture of the entered merchandise (HQ Ltrs 543048 dated June 17, 1983 and 543197 dated May 23, 1984).

- (i) Valuation of Waste or Scrap The dutiable value of recoverable waste or scrap from the manufacture or manipulation of foreign status merchandise will be the price actually paid or payable to the zone seller in the transaction that caused the recoverable waste or scrap to be transferred from the zone (19 CFR 146.42(b) and 19 CFR 146.65(b)(2)).
- (ii) Allowance for waste or scrap for privileged foreign merchandise The deduction for waste or scrap to be made from the transaction value of the foreign merchandise used in the manipulation or manufacture of the entered merchandise shall represent the difference between: 1) the market value of the foreign merchandise admitted to the zone (i.e., the quantity of scrap or waste multiplied by the value, as appropriately measured, of the foreign merchandise); and 2) the market value of the scrap or waste (i.e., the transaction value of the scrap or waste).
- (iii) Allowance for Damage An allowance in the dutiable value of zone merchandise may be made by the Port Director in accordance with the provisions of 19 CFR 158, Subparts B and C, for damage, deterioration, or casualty while the merchandise is in the zone (19 CFR 146.65(b)(3)).
- (2) Total Zone Value The total value of merchandise transferred from a zone will be determined in accordance with the principles of valuation contained in 19 U.S.C. §1401a and 19 U.S.C. §1500. The total value is that price actually paid or payable to the seller in the transaction that caused the merchandise to be transferred from the zone. Where there is no price actually paid or payable, the total zone value shall be the cost of all materials and zone processing costs related to the merchandise transferred from the zone (19 CFR 146.65(b)(1)).
- (i) Liquidation When the declared value or values of the merchandise are based on an estimate or estimates or standard cost, the person making entry may request an extension of liquidation pending the presentation of updated or actual cost data. A request for an extension may be granted at the discretion of the Port Director (19 CFR 146.65(c)). If the declared values are predicated on estimates, estimated costs, or standard costs, that information must be clearly identified to CBP as required at the time an entry or entry summary is filed (19 CFR 146.62(b)(1)). Estimated values must be reconciled upon determination of the actual value using CBP reconciliation procedures.

(j) Antidumping and Countervailing Duties - The Board requires that merchandise which is subject to antidumping or countervailing duties in its condition as imported be placed in privileged foreign status (15 CFR 400.33(b)). Upon entry for consumption such items shall be subject to bonding or cash deposit requirements under the AD/CVD order or the suspension of liquidation, as appropriate, under 19 CFR 351. HTSUS duty rates are established for privileged foreign status merchandise based on the date of admission to the zone. However, AD/CVD rates are assessed based on the rates applicable at the time of entry from the zone.

An AD/CVD investigation or order may be initiated on merchandise that has been previously admitted to a zone in nonprivileged foreign status. When this merchandise is entered for consumption, a determination is made as to whether the merchandise is within the scope of an AD/CDV investigation or order.

9.8 Weekly Entries for other than Quota Class Merchandise - The Trade and Development Act of 2000, Pub. L. 106-200, amended CBP entry statute, 19 U.S.C. §1484, to extend the use of weekly consumption entries to include all zones instead of limiting the use to only manufacturing operations. The week is defined as any consecutive 7-day period. The use of the weekly consumption entry is at the option of the Operator or User.

The CBPF 3461 (estimate), filed on or before the first day of the 7-day period in which the merchandise is to be withdrawn from the zone, must be accompanied by a proforma invoice or schedule showing the number of units of each type of merchandise anticipated to be transferred during the week and their total zone or dutiable values. For special instructions on filling out the Weekly Estimated CBPF 3461, please refer to Section 9.7((a)(1). Notwithstanding that a weekly estimated entry may be allowed, all merchandise will be dutiable as set forth in 19 CFR 146.65; see sections 9.7(g) and (h) FTZM. If estimated removals exceed actual removals, that excess quantity will not be considered to have been entered or constructively transferred to the Customs territory (19 CFR 146.63(c)). If the entry summary presented to CBP per Section 9.8(c) FTZM shows merchandise transferred from the zone in excess of the quantity entered on the weekly entry (plus any supplemental entry), the excess will be considered to have been transferred to the Customs territory without a permit, a breach of the Operator's bond (19 CFR 146.71(a); see section 12.6 FTZM).

The estimated entry or release shall be treated as a single entry and a single release of merchandise. All fee exclusions and limitations of Section 13031(a)(9)(A) of the Consolidated Omnibus Budget Reconciliation Act of 1985 (19 U.S.C. §58c(a)(9)(a)) shall apply, including the maximum and minimum fee amounts set forth for under subsection (b)(8)(a)(i) of such Section.

The Secretary of the Treasury may require the Operator or User to use an electronic data interchange approved by CBP (such as its Automated Broker Interface) to file the weekly consumption entry and to pay the applicable duties, fees, and taxes with respect to the entries. The Operator or User may also be required to satisfy CBP that accounting, transportation, and

other controls over the merchandise are adequate to protect the revenue and meet the requirements of other Federal agencies.

- (a) Purpose of Weekly Entries Weekly entries are authorized under The Trade and Development Act of 2000, Title IV, Section 410, (amending 19 U.S.C. §1484; see also 19 CFR 146.63(c)), for rapid removal of merchandise from zones. If weekly entries were not allowed, it would be very difficult for the importer to make entry for the zone merchandise without unacceptable delays in the manufacturing process or in the distribution of goods. It would also be difficult for CBP to review and accept the entry and examine the merchandise in its form as entered.
- (b) Specific Estimate of Quantities and Values Estimated weekly entries shall contain a specific estimate of the quantity of each HTS unit and total value to be entered during the covered 7-day period. This is required whether the merchandise is in privileged or nonprivileged foreign status. Weekly entries of parts shall contain a specific estimate of the quantity of each HTS unit to be shipped, and the total value. (19 CFR 146.63(c)).
- (c) Filing of Entry Summary The follow-up entry summary for a weekly entry shall be filed within the time limit specified in 19 CFR 142.12(b). The entry summary on CBPF 7501 must show in item 29 the quantity and value of the merchandise by HTS number, actually entered during the covered 7-day period. An entry summary shall not be accepted if:
 - (1) The quantity of HTS numbers on the entry summary exceeds the corresponding quantity on the weekly entry; or
 - (2) It does not include the quantity, value, as specified in this paragraph.

If a supplementary entry is filed to cover extra estimated entered merchandise during the 7-day period, a separate entry summary must be filed to cover that entry. The above quantity, value, requirements apply also to supplementary entries and their follow-up entry summaries. If there are no shipments of non-duty paid merchandise made during the week, the Operator requests, by letter, that the Port Director cancel the estimated CBPF 3461 entry. The entry number assigned to the weekly estimated CBPF 3461 is cancelled and cannot be re-used for the next shipping period.

(d) Limitations and Prohibitions -

- (1) Sugar-Processing Operations A weekly entry for estimated production shall not be accepted for any sugar or sugar-containing product which is subject to an absolute quota (See Section 11.6(i) FTZM).
- (2) "Blue Sky" Estimates Port Director s shall not accept weekly entries for estimated quantities that are grossly in excess of past amounts covered by entry

summaries unless the estimate is accompanied by a reasonable explanation for the disparity.

- (e) Exceptions to Other Provisions The following exceptions to other provisions of the Customs regulations are noted:
 - (1) The restriction in 19 CFR 146.71(c) against further manufacture or manipulation will not be applied by CBP to weekly entries of estimated production.
 - (2) Weekly entry estimates may be accepted for merchandise which has not yet been imported or admitted to the zone, as long as that merchandise is subsequently admitted, and is included in the follow up entry summary.
- (f) Inventory Relief Point Where the Operator of a zone relieves its inventory at or before the inventory relief point (usually the end of the final assembly line), the ort director may accept the weekly entry on the basis of the quantity and value of merchandise passing through the inventory relief point. In this case, the follow-up entry summary shall be made on the quantity and value of the merchandise passing the same inventory relief point. Extension of the time limit for physical removal from the zone pursuant to 19 CFR 146.71(c) for a period of up to 60 days shall not be deemed excessive for such operations.
- 9.9 9.9 Procedures for FDA Weekly Entry Filing (WEF) - Filers of entries containing FDA regulated products wishing to participate in CBP Weekly Entry Filing procedures must apply in writing to both the local FDA office and the local CBP Port Director under special procedures developed jointly by CBP and FDA. A copy of the letter of application should also be sent to the FDA Division of Import Operations and Policy (DIOP). The application must contain specific product and firm information depending on the actions taken within the zone. For goods that are manufactured in a FTZ, the following information is required: Commercial product name and a description, Harmonized Tariff Schedule (HTS) codes, FDA Product code, and Name, Address and MID of the site specific manufacturer of the highest valued, non-US, FDA regulated ingredient or component. For goods stored in an FTZ, the following information is required: The name, address, and MID of the site specific foreign manufacturer, FDA Country of origin, name, address and MID of the FDA shipper (i.e. the FTZ site from which the product is offered for entry into the US, and any FDA required registrations, listings, approvals, and/or notifications (FCE# FDA 2877, etc.). This information is needed to determine if the product is suitable for WEF. In order to qualify for FDA WEF the product must be "low risk" and therefore, not subject to an import alert.

The FDA District office will determine the compliance history of both the products and the firms (both manufacturer and filer). The Application will be forwarded to the DIOP and the FDA Center responsible for the commodities in the application to determine the level of risk to the public health posed by the product. Once the application is completed by DIOP's Operations Branch and the appropriate center and found acceptable, DIOP's System's Branch will set

screening criteria into OASIS. FDA and CBP will then inform the filer, in writing, that they have been accepted for WEF. Entries filed under the WEF will be issued FDA MAY PROCEED status through ACS/OASIS and will not be subject to redelivery for sampling and examination. WEF are subject to reevaluation and withdrawal at any time by DIOP, the Centers, or the local District Office. If at any time a decision is made that a filer should be removed from the WEF, the Division of Import Operations and Policy will be notified in writing. DIOP's System's Branch will ensure that all subsequent entries will undergo normal OASIS screening, and will be subject to redelivery for sampling and examination if required by FDA. The FTZ sites identified in the application will be added to the district OEI (if not already there), and appropriate inspectional coverage will be initiated. Any violative products collected or violative conditions observed, during inspection of the FTZ site will cause a review of the WEF application. FDA data for WEF entry processing will be submitted via ACS/OASIS. Products entered through WEF matching the elements submitted in WEF Application will receive a "MAY PROCEED" through ACS/OASIS, and will not normally be subject to redelivery.

- 9.10 Entries for Bonded Warehouse, TIB, Trade Fair, or U.S. Government Importations The first proviso of 19 U.S.C. §81c, provides that when foreign merchandise in a zone has been given privileged status, the merchandise may be sent into Customs territory on payment of duties and taxes. CBP interprets this proviso to forbid any delay in the payment of duties upon transfer to Customs territory (19 CFR 146.62). A similar kind of restriction is applicable to entries of merchandise imported by the U.S. Government which have been manufactured in a zone. It must also be noted that entries for quota-class merchandise under TIB must also be charged to the appropriate quota. If the quota is closed, the TIB is not admissible.
 - (a) Bonded Warehouse Entries Most of the procedures and requirements of Section 9.7 FTZM, apply also to entries for warehouse. However, the following provisions apply to entries for warehouse, but not to entries for consumption:
 - (1) Privileged Foreign Status Merchandise Merchandise in privileged foreign status or composed in part of merchandise in privileged foreign status may not be entered for warehouse from a zone (C.S.D. 81-8 and 19 CFR 146.64(a)).
 - (2) Nonprivileged Foreign Status Merchandise Merchandise in nonprivileged foreign status containing no components in privileged foreign status may be entered for warehouse in the same or at a different port, unless it was imported more than 5 years before the warehouse entry was filed (19 CFR 146.64(a), (d)). If such merchandise was entered for warehouse within the 5-year period, it may not remain in a bonded warehouse longer than 5 years from the date of importation of the merchandise. For example, if merchandise has been in a zone for a period of three (3) years from the date of original unlading at the first port of arrival in the U.S., the merchandise could only remain in the bonded warehouse for the remaining two (2) years. (See 19 CFR 146.64(d)).

- (3) Zone-Restricted Status Merchandise Foreign or domestic merchandise in zone-restricted status may be entered for warehouse in the same or a different port only for storage pending exportation, unless the Board has approved another disposition (19 CFR 146.64(b), 146.70(c) and HQ 224147).
- (4) Domestic or Previously-Entered Merchandise Per bonded warehouse regulations, domestic status, or previously entered merchandise, may not be entered into a bonded warehouse (See 19 U.S.C. §1555 and 19 U.S.C. §1557; 19 CFR 19).
- (5) Textile Articles Textiles and textile products, including quota class merchandise, which have been changed as set forth in 19 CFR 146.63(d) may be entered for warehouse only if the entry is endorsed by the Port Director to show that the merchandise may not be withdrawn for consumption (19 CFR 146.64(c) and See Section 11.6(h) FTZM).
- (b) Entry under Temporary Importation Bond -
 - (1) Privileged Foreign Status Privileged Foreign status merchandise or composed in part of Privileged Foreign status merchandise may not be entered on a TIB (19 CFR 10.31 and C.S.D. 81-213).
 - (2) Zone-Restricted Status Zone-restricted status merchandise may be entered for TIB only if the Board has ruled that it is in the public interest and has not specified that an entry other than a consumption entry shall be made (C.S.D. 81-119).
 - (3) Non-Privileged Foreign Status Nonprivileged foreign status merchandise may be entered on a TIB unless:
 - (i) It contains Privileged Foreign status merchandise; or
 - (ii) It has been in the zone for longer than one year (C.S.D. 79-454); or
 - (iii) If quota merchandise, the quota is closed
- (c) Trade Fair Entries -
 - (1) Privileged Foreign Status Merchandise Cannot be subject to a Trade Fair Entry.
 - (2) Zone-Restricted Status Merchandise –May be transferred to a trade fair for exhibition and subsequently entered for consumption if the Board has approved it as being in the public interest (19 CFR 147.45).

- (3) Non-Privileged Foreign Status Merchandise Merchandise in NPF status not composed in part of merchandise in privileged foreign status may be transferred to a trade fair, from which it may be granted any disposition set forth in 19 CFR 147.42.
- (d) Entries of U.S. Government Importations Importations, by or for the account of the U.S. Government, are subject to the usual CBP entry and examination requirements. In the absence of express exemptions from duty, such as those contained in Chapter 98, Subchapter VIII, HTS, they are subject to duty (19 CFR 10.100). Procedures for entering such merchandise are set forth in 19 CFR 10.100-10.104. Merchandise in privileged foreign status which is transformed in a zone into merchandise which is covered by an HTS subheading in Chapter 98, Subchapter VII, HTS, as a U.S. Government importation is not eligible for the duty-free exemption because the duties were fixed on the merchandise at the time privileged foreign status was requested (HQ 222452 dated August 15, 1990 and See Section 5.5 FTZM).
- 9.11 Entries for Diplomatic and Foreign Military Use Entries for diplomatic use are, for CBP purposes, entries for consumption, filed as set forth in Section 9.7(a) FTZM but with special control requirements of the Operator. Upon the request of the Department of State, merchandise may be transferred from a zone without the payment of duty and tax for the personal use of diplomatic, consular, and other privileged personnel. It cannot be used as an accommodation for others or for sale or other commercial use (19 CFR 148.85). There is no authority for a blanket transfer permit from a zone for such merchandise similar to the blanket withdrawal permit authorized under 19 CFR 19.6(d) for bonded warehouse operations.
 - (a) Regulatory Provisions for Exemption from Duties and Taxes Diplomatic, consular and other privileged personnel are referenced in 19 CFR 148.85(a)-(c). In addition, 19 CFR 148.87 identifies public international organizations whose officers and employees, and representatives are entitled to such free entry privileges. 19 CFR 148.88 specifies certain representatives to, and personnel of, the United Nations and the Organization of American States entitled to the same privileges. Property of designated international organizations listed in 19 CFR 148.87 or of foreign governments will be admitted free of duty and internal revenue taxes imposed upon or by reason of importation under Title 22 U.S.C. §288, but such exemption will be granted only upon the receipt, in each instance, of instructions from CBP issued at the request of the Department of State (19 CFR 148.87(a)). Foreign military personnel are entitled to exemption from duty and internal revenue taxes in accordance with 19 CFR 148.90.
 - (1) Procedure for Diplomatic Exemptions Permission for transfer to the Customs territory of zone status merchandise for diplomatic, consular, or public international organization use is requested by a representative of such an entity from the Department of State on DS Form 1504. If the Department of State finds that U.S. personnel in those countries are offered reciprocal privileges and approval of the request is otherwise warranted, it will so advise the appropriate

Port Director. However, requests from the United Nations or foreign country missions to the United Nations in New York for exemptions shall be made directly to the New York Area Port Director. Requests from consulates in New York other than those associated with the United Nations shall be made to the Department of State.

- (2) Procedures for Military Exemptions A request for exemption from duties and taxes on merchandise withdrawn for the official or personal use of members of the armed forces of a foreign country in the United States, but not as an accommodation to others or for sale or other commercial use, shall be made directly to the Port Director having jurisdiction over the zone and not, through the State Department (19 CFR 148.90(a)). The Port Director will accord duty and tax-free treatment only to the extent to which the foreign government accords similar treatment to members of U.S. armed forces in that country, as advised by CBP Headquarters Diplomatic Privileges Office (19 CFR 148.90(b)). The withdrawal of alcoholic beverages for the personal or family use of foreign military personnel is limited to one case per month, except in exceptional circumstances (19 CFR 148.90(d)).
- (b) Zone Admission File Records Entries under this section shall be made on CBPF 7501, accompanied by a copy of DS Form 1504. The Operator shall place a copy of the entry and the foregoing support documents in the zone admission file folder.
- (c) Transfer to Other Ports after Entry An entry made under this section may be made at a port other than the one where the embassy, consulate, public international organization, or foreign military unit is located. Merchandise which has been entered for such diplomatic use may be transported to the location where it will be used for diplomatic purposes, but shall not be transported under CBP bond.
- (d) Refused Merchandise If merchandise entered for diplomatic use is rejected by the diplomatic agency due to a language misunderstanding as to the quantity or kind of merchandise ordered, the refused delivery may be treated as a mistake of fact pursuant to 19 §1520(c)(1). The refused merchandise may be returned to the zone in its last zone status and the entry amended to reflect the quantity of merchandise accepted upon delivery (LD 80-197 and HQ.713144 dated 9/12/80).
- 9.12 Transfer to Another Zone or Port Merchandise may be transferred from one zone to another in the same or a different port, or may be entered for transportation in-bond to another port for entry for consumption or warehouse as permitted by law (See Section 6.7 FTZM). Zone to zone transfers may be made on an individual basis or on a weekly basis using the procedures in 19 CFR 146.66, 19 CFR 146.68, 26 U.S.C. §5061(d) and 26 U.S.C. §5703(b)(2).

CBPF 7512 for merchandise to be transferred to another port or zone shall state the following:

(1) That the merchandise is zone merchandise;

- (2) Give the number of the zone from which it was transferred; and
- (3) State the zone status of the merchandise.
- When applicable, it shall bear the notation or endorsement set forth in 19 CFR 146.64(c) (textile articles), 146.66(b) (transfer to a zone at a different port), 146.70(c) (zone-restricted status merchandise for entry for warehouse), and 146.70(d) (zone-restricted status merchandise for other purposes) (19 CFR 146.62(b)(2)). It shall also bear, when applicable, the notation specified in 26 U.S.C. §5061(d), 26 U.S.C. §5703(b)(2), 26 U.S.C. §5702(k), and 26 U.S.C. §5703(b)(ii) (Internal Revenue Service non-diversion statement).
- Procedures for QP/WP and QX/WX (CBP In-Bond System) The QP/WP (a) electronic module is an Automated Broker Interface (ABI) in-bond processing system that allows ABI filers to create and initiate in-bond shipments. QX/WX is a new functionality in ACS to give air carriers the same functionality as QP/WP, so that intermodal in-bonds may be more easily conducted. It also allows the arrival and export of material at destination. QP/WP and QX/WX requires transport by CBP bonded carriers and the printing of a hard copy of the CBPF 7512 that must travel with the in-bond cargo. The QP/WP and QX/WX in-bond system has been modified to allow Zone Operator s to initiate in-bond withdrawals from zones. This eliminates the need for the weekly CBPF 7512 procedure under 19 CFR 146.68. Any operational ABI entry summary filer may participate and anyone who does not file entry summaries and is not an ocean, rail, or air carrier may apply to become a stand alone participant in QP/QX in-bond. Sea carriers, air carriers, and railroads have AMS (Automated Manifest System) modules available and will not be allowed as direct participants in the QP/WP or QX/WX programs. QP/WP and QX/WX do not currently eliminate the need to have a hard copy of in-bond documentation. After receiving approval from CBP through the ABI transmission, the CBPF 7512 must be printed out and travel with the bonded driver to be presented to CBP at the destination port. CBP personnel at the port of destination will perform the arrival and exportation update functions after the CBPF 7512 is presented, or the Zone Operator, ABI filer or bonded carrier may perform the arrival and exportation update if they have the proper functionality (WP/WX). The QP/QX application allows in-bond filing at less than full bill of lading quantity; multiple QP/QX in-bonds can be filed against the same import bill.
- (b) Transfer to Another Zone in Same Port A transfer of merchandise to another zone with a different Operator at the same port (including a consolidated port) will be by a licensed cartman or lighterman, a bonded carrier, or the Operator when the merchandise is destined for his zone (19 CFR 112.2(b)) under an entry for immediate transportation on CBPF 7512 or other appropriate form under the procedures of 19 CFR 118, with a e-214/CBPF 214 filed at the destination zone. (19 CFR 146.66(a)). The CBPF 7512 must include the statements required by 19 CFR 146.62(b)(1), 19 CFR 146.66(b), and 19 CFR 146.70(d) when applicable. The purpose of using CBPF 7512, rather than CBPF 6043, within-port transfers, is to assure that the Operator of the destination receiving zone has

the necessary information to exercise proper control over the merchandise and to file proper admission documentation.

- (1) The Operator of the sending zone shall also transmit the merchandise history to the Operator of the destination zone, as set forth in 19 CFR 146.66(c) (see also section 9.11(b) FTZM. If the Port Director requires, a copy of a CBPF 7512 shall be transmitted to CBP by the destination zone Operator upon receipt of the shipment. Upon arrival at the destination zone, the merchandise shall be admitted (See Section 6.7(g)(5) FTZM).
- (c) Transfer to Zone Site with Same Operator A transfer of merchandise between zone sites at the same port having the same Operator may be made under permit on CBPF 6043 or by the zone Operator under a local control system approved by the Port Director wherein any loss of merchandise between sites will be treated as if the loss occurred in the zone. The statements specified in 19 CFR 146.62(b)(1), 19 CFR 146.66(b), and 19 CFR 146.70(d) shall not be required for such transfers.
- (d) Transfer to Zone at Another Port A transfer of merchandise from a zone at one port of entry to a zone at another port will be by bonded carrier under an entry for immediate transportation on QP/QX 7512. All copies of the entry must bear a notation that the merchandise is being transferred to another zone designated by its number, as well as the statements required by 19 CFR 146.62(b)(1) and 19 CFR 146.70(d), when applicable (19 CFR 146.66(b)). A sample of a properly filled-out CBPF 7512 for transfer to a zone at another port is shown in the Appendix to the FTZM.
 - (1) Forwarding of Merchandise History When merchandise is transferred under the provisions of 19 CFR 146.66(b), the Operator of the transferring zone shall provide the Operator of the destination zone with the documented history of the merchandise being transferred.
 - (i) The following documentation must accompany merchandise maintained under a lot inventory control system:
 - (1) A copy of the original e-214/CBPF 214 with accompanying invoices for admission of the merchandise and all components thereof;
 - (2) A copy of any e-214/CBPF 214 filed subsequent to admission to change the status of the merchandise or its components; and
 - (3) A copy of any CBPF 216 to manipulate or manufacture the merchandise.

- (ii) The following documentation must accompany merchandise not under a lot system, and not manufactured in a zone:
 - (1) A copy of the original e-214/CBPF 214 with accompanying invoices for admission of the merchandise as attributed under the particular zone inventory method;
 - (2) A copy of any e-214/CBPF 214 filed subsequent to admission to change the status of the merchandise as attributed under the particular zone inventory method; and
 - (3) A copy of any CBPF 216 to manipulate the merchandise as attributed under the particular zone inventory method.
- (iii) If the documents specified in Section 9.11(b)(1) (i) and (ii) FTZM are not presented, the Operator of the transferring zone shall submit the following:
 - (1) A statement of the zone value, dutiable value, quantity, description, unique identifier, and zone status (showing any changes of status after admission and whether the merchandise was manipulated so as to change its tariff classification) of all the merchandise in the shipment covered by the transportation entry; and
 - (2) A certification that the above statement is true and that the information therein is contained in the inventory control and recordkeeping system of the transferring zone.
- (iv) The following documentation must accompany merchandise not under a lot system but manufactured in a zone:
 - (1) A statement by the transferring Operator of the zone value, dutiable value, quantity, description, unique identifier, and zone status of all the merchandise (and components thereof, where applicable) covered by the transportation entry. The statement will also show any change in zone status in the transferring zone and state whether the merchandise has been manufactured or manipulated in the zone so as to change its tariff classification; and
 - (2) A certification by the Operator of the transferring zone that the above statement is true and the information therein is contained in the inventory control and recordkeeping system of the zone.

- (v) In circumstances where a lot inventory control system is used, but the forwarding of the documentation called for in Section (i) would cause a significant paperwork burden, the provision of paragraph (iii) and paragraph (iv) allowing a statement and certification by the Operator , shall be permissible (Procedural accommodation for USCS)
- (vi) The Operator of the transferring zone shall transmit the historical documentation of the merchandise to the receiving zone within 10 working days after the merchandise has been delivered to the bonded carrier for transportation. The documentation will be referenced to the I.T. number covering the merchandise (19 CFR 146.66(c)).
- (2) Arrival at Destination Zone Upon arrival of the merchandise at the destination zone, it will be admitted under the procedure set forth in 19 CFR 146.32, except that invoicing or CBP examination requirements may be waived by the Port Director. Merchandise must retain the zone status from the transferring zone upon admission to the new zone. Any subsequent application to change the zone status must consider all activity since the first zone admission. (See Section 6.7(g)(5) FTZM). When the historical documentation is received, the Operator of the destination zone shall associate it with the e-214/CBPF 214 for admission of the merchandise and incorporate that information into the zone inventory control and recordkeeping system (19 CFR 146.66(d)).
- (3) Records of Transferred Merchandise The Operator of the sending zone remains responsible for maintaining a complete record of all goods that have been transferred to another zone for a period of 5 years from the date of transfer (19 CFR 146.4(d)). The CBPF 214 and the CBPF 216 to be sent to the receiving zone should be copies of the original forms required to be maintained by the Operator of the sending zone.
- (4) Valuation Option For confidentiality purposes, with the agreement of the Port Director, a firm in a zone may use total zone value for zone-to-zone transfer purposes. In those instances where the merchandise in question has a specific rate of duty, agreement may be reached by the parties on a standard valuation method using public sources, as value would not be a requirement for CBP admissions.
- (e) Transfer to Another Port for Entry for Consumption or Warehouse Merchandise may be entered for transportation to another port under the procedures in 19 CFR 118, and there entered for consumption or warehouse, if such entry is not prohibited by law or regulation. The statements on CBPF 7512 required as set forth in 19 CFR 146.62(b)(2), 146.64(c), and 146.70(d) alert the consignee and the Port Director at the destination as to the proper treatment of the merchandise. The responsibility will be on the owner, purchaser, or other eligible person to file a proper entry for the zone merchandise, as specified in 19 141, Subparts B and C. An owner or purchaser who has a bond on file containing the conditions in 19 CFR 113.62 and desires to export the merchandise (rather

than enter it for consumption or warehouse) shall advise the Port Director in writing. The Port Director shall permit the exportation under CBP supervision under an entry for direct or indirect exportation as set forth in 19 CFR 18.25 or 19 CFR 18.26, as applicable, if such a permit may be lawfully issued.

- (f) Liability and Payment of Internal Revenue Tax In the case of any alcohol or tobacco product imported, entered for warehouse, or brought into the United States or a foreign-trade zone after December 15, 1986, the Internal Revenue tax, but not the duty, becomes due and payable the 14th day after the last day of the semimonthly period during which the article is removed from the first zone into which it is admitted (26 U.S.C. §5061(d) and 26 U.S.C. §5703(b)(2).
 - (1) Exception for Export Merchandise Internal Revenue tax is not due or payable on any alcohol or tobacco product which the Port Director is satisfied is destined for export. The burden of proof for furnishing evidence of export destiny lies with the person filing the application for transfer from the zone. Acceptable evidence includes entry for transfer of zone-restricted status merchandise to another zone or a bonded warehouse at the same or a different port, or entry of foreign or zone-restricted status merchandise for exportation or transportation and exportation with a notation on CBPF 7512 which reads: "THIS MERCHANDISE IS DESTINED ONLY FOR EXPORTATION, AND MAY NOT BE WITHDRAWN OR DIVERTED FOR CONSUMPTION".
 - (2) Payment Procedure When alcohol or tobacco products are to be removed for transfer to another zone, for transportation to another port, or entry for warehouse, and are not destined for export, the application for transfer or the entry on CBPF 7512 shall be accompanied by CBPF 7501, in the number of copies required by the Port Director, and the taxes due on the merchandise. The following statement shall be contained in the body of the form: "FOR COLLECTION OF INTERNAL REVENUE TAXES ONLY". A receipted copy of CBPF 7501 should accompany the merchandise to the destination, port, warehouse, or zone as evidence of proof of Internal Revenue tax paid status. The burden will be on the subsequent person filing an entry or withdrawal to provide satisfactory evidence to CBP of tax paid status.
- 9.13 Transfer from Zone for Exportation Foreign and domestic merchandise may be brought into a zone and be exported therefrom (19 U.S.C. §81c(a)). "Exportation" is defined in 19 CFR 101.1 as a severance of goods from the mass of things belonging to this country with the intention of uniting them to the mass of things belonging to another country. For statistical reporting purposes, zone exporters must follow the procedures set forth in Census's Foreign Trade Statistics Regulations (FTSR), in 19 CFR 30, for all merchandise removed from a zone for direct export out of the United States.

The North American Free Trade Agreement (NAFTA) Duty Deferral provision, found in 19 CFR 181.53, applies to foreign merchandise imported under a duty deferral program that is

manufactured or changed in condition and subsequently exported to Canada or Mexico. (See Section 9.19) Under these provisions, withdrawals for exportation to Canada or Mexico are treated as if the merchandise was entered for consumption. The NAFTA Duty Deferral entry (Type 08) may be filed for these withdrawals or when appropriate, the foreign parts in the shipment can be reported on an individual or weekly entry (Type 06) from the zone. The collection of regular duties and fees due on a Type 08 entry must be deposited within 60 days of the date of export. The collection of regular duties and fees due on a Type 06 entry must be deposited within 10 days of the date of entry unless the importer is paying duties and fees under the Periodic Monthly Statement (PMS). In that case, the duties and fees will be paid by the 15th working day of the month following the month of the date of entry. Foreign goods manufactured in a zone and exported to Canada and Mexico are subject to NAFTA Duty Deferral and requires the filing of entry summary documentation within 10 working days from the date of exportation, unless filing on an individual or weekly entry (Type 06) is appropriate.

General provisions for the statistical reporting of exports from a zone are also contained in (a) and (b) below.

- (a) Export Statistical Reporting All goods subject to statistical reporting requirements, foreign or domestic, removed from the zone for export to a foreign country must be reported at the CBP port of export via an Electronic Export Information (EEI) transmission through the Automated Export System (AES). The use of the Automated Export System (AES) is required for all goods foreign or domestic subject to statistical reporting requirements which are removed from the zone for export to a foreign country. When reporting shipper's export data, the proper exemption statements must be included on the bill of lading, airway bill, or other manifest document. For detailed information on the mandatory electronic filing of the EEI, please refer to www.aesdirect.gov.
- (b) EEI/AES Filing Requirements An EEI/AES record is required for all export shipments valued over \$2,500 from the United States, Puerto Rico, or the U.S. Virgin Islands to a foreign country, or for all licensed shipments regardless of value (See 15 CFR 30.01 through 30.16 for EEI exporter's general filing requirements). AES filing requirements are contained in, Subpart E-Electronic Filing Requirements-Shipper's Export Information, 15 CFR 30.60-30.66. An EEI/AES filing is not required for exports to Canada, except when a license or other report is required, because of the data exchange program the United States has with Canada (See 15 CFR 30.58). For other items exempt from EEI/AES filing requirements, refer to the FTSR, subpart D-Exemptions from the Requirements for the Filing of Shipper's Export Declarations, 15 CFR 30.50-30.58. For specific AES exemption provisions, see 15 CFR 30.65.

September 30, 2008 was the deadline for mandatory electronic filing of export information. Census issued a final rule making it mandatory to file export information electronically through the Automated Export System ("AES") or AESDirect, for all shipments where a Shipper's Export Declaration ("SED") is now required. AES is the electronic filing system for submitting export information to the U.S. Customs and Border Protection, and AESDirect is Census's free Internet filing system for submitting

export information to the AES. Information previously included on a paper SED is now referred to under the final rule as Electronic Export Information ("EEI"). Enforcement of this requirement began on October 1, 2008 when all filings of EEI must be done electronically. Parties submitting paper SEDs on or after October 1 will be considered in violation of the Foreign Trade Regulations ("FTR") and subject to the penalties.

- (c) Exportation for Purposes of Other Agency Laws Exports from foreign-trade zones are subject to the requirements of various agencies charged with the administration and enforcement of export control laws and regulations, such as the Department of Commerce, Export Administration Regulations, 15 CFR 730-774; Department of State, 22 CFR 121-130; and The Drug Enforcement Administration, 21 CFR 1312. The requirements of these agencies shall be met only upon actual exportation, not upon admission to the zone in zone-restricted status.
- (d) Shipments to Insular Possessions Shipments to insular possessions of the United States are not exported for the purposes of CBP laws, and thus do not meet any condition or requirement of CBP laws for exportation, including that for zone-restricted status merchandise. Shipments transferred from a zone to an insular possession must be duty paid (HQ Ruling 223828 dated 7/1/92). Such a shipment may be deemed an exportation for the purposes of other Federal laws if the Federal agencies administering those laws so deem it. The principal insular possessions include the U.S. Virgin Islands, American Samoa, Guam, Wake Island, Midway Islands, and Johnston Atoll (19 CFR 7.2(a)). Puerto Rico is part of the Customs territory of the United States (19 CFR 101.1). Therefore, an entry for merchandise destined to Puerto Rico shall be made for transportation only (I.T.) under the procedures of 19 CFR 18 and Section 9.11 FTZM. Please see chart below for filing requirements for EEI. (For export statistical reporting, follow procedures specified in 15 CFR Part 30).

EEI required

From the United States or Puerto Rico to the Virgin Islands From Puerto Rico to the United States From the United States to Puerto Rico

EEI not required

From United States or Puerto Rico to Insular possessions (except Virgin Islands) From Insular possessions to other possessions

(e) Direct Exportation - Any merchandise in a zone may be exported directly therefrom (without transfer into the Customs territory) in the same manner as an immediate exportation under Section 9.12(c) FTZM, except that the merchandise need not be transported to the exporting carrier by licensed cartman under the procedures of 19 CFR 125 (19 CFR 146.67(a)). Direct exportation occurs when the merchandise is laden directly on an exporting carrier (e.g. vessel or aircraft) within a zone, and no cartage or

transportation in-bond through the Customs territory is required to deliver it to the carrier. (For export statistical reporting, follow procedures specified in 15 CFR 30).

- (f) Immediate Exportation Each transfer of merchandise to the Customs territory for exportation at the port where the zone is located will be made under an entry for immediate exportation on CBPF 7512. The person making entry shall furnish an importation and entry bond guaranteeing export on CBPF 301 containing the bond conditions set forth in 19 CFR 113.62 and 19 CFR 146.67(b). The merchandise will be transported by licensed cartman to the place of lading under the procedures in 19 CFR 125, and exported under the procedures in 19 CFR 18.25. Care should be taken by CBP personnel and bonded cartmen to execute both the front and the back (Record of Cartage or Lighterage) of CBPF 7512. (For export statistical reporting, follow procedures specified in 15 CFR 30).
- (g) Indirect Exportation Each transfer of merchandise to the Customs territory for transportation to and exportation from a different port will be made under an entry for transportation and exportation on CBPF 7512. The bonded carrier will be responsible for exportation of the merchandise in accordance with 19 CFR 18.26 and 146.67(c). (For export statistical reporting, follow procedures specified in 15 CFR 30).
 - (1) Splitting of Shipments and Changes of Destination If any part of a shipment is not exported or if a shipment is divided at the port of exportation, extracts in duplicate from the manifest on file in the Customhouse shall be made on CBPF 7512 for each portion. The splitting up for exportation of shipments arriving under entries from zones for indirect exportation shall be permitted only when various portions of a shipment are destined to different destinations, when the export vessel cannot properly accommodate the entire quantity, or in other similar circumstances. The provisions of 19 CFR 18.23 and 19 CFR 18.24 concerning change of destination or retention of merchandise on the dock shall also be followed in applicable cases.
 - (2) Special Conditions Special Conditions set forth in CBP ruling 223268 dated 10/15/91, allow PF status aircraft turbine fuel to be transferred from a zone, using in-bond procedures to destination and temporarily stored in segregated non-bonded storage prior to exportation as fuel under 19 U.S.C. §1309.
 - (3) Diversions at Destination An entry from a zone for indirect exportation may be converted to an entry for consumption upon a request to the Port Director at the port where the entry for indirect exportation was made, unless such diversion is prohibited by law or regulation, e.g., ZR status merchandise. The potential for diversions for consumption at the port of exportation is one of the reasons for the statement requirements in 19 CFR 146.62(b)(2) and 19 CFR 146.70(d); see section 9.11(d)(1) FTZM.

- (4) Indirect Exportation Pursuant to 19 CFR 18.24, merchandise may be retained in a non-bonded area (CBP approved storerooms-CASR) for a period not to exceed 90 days. Extensions may be granted, but such extension(s) shall not exceed 1 year. Merchandise retained at an airport for export through in-flight sales aboard aircraft is subject to special procedures under CD 099-3280-008.
- (h) Filing of Entry- An entry for direct, immediate, or indirect exportation shall be filed on CBPF 7512 in five copies or electronically using QP/WP or QX/WX. The Port Director may require an extra copy or copies of CBPF 7512 for use in connection with the delivery of the merchandise to the carrier. One copy of the form shall be used as the permit copy to be retained by the Operator. One copy shall be used to initiate the inbond movement under the Automated Commercial System. The other copies shall be returned to the Operator or User for transmittal to the bonded carrier or licensed cartman. The same form and procedure will be used for shipments to insular possessions, but not to Puerto Rico, although such shipments are not deemed exportations under CBP laws. For export statistical reporting, follow procedures specified in 15 CFR 30.
 - (1) Statements on CBPF 7512 Whether the entry is for direct, immediate, or indirect exportation, the person making entry shall include on CBPF 7512 the statements required under 19 CFR 146.62(b)(2); and, when applicable, the endorsement in 19 CFR 146.70(c)(d) for zone-restricted status merchandise and the statement in Section 9.11(d)(1) FTZM for alcohol and tobacco products. A sample of a filled-out CBPF 7512 for immediate exportation from a zone is shown in the Appendix to the FTZM.
 - (2) Bond Liability For direct or immediate exportation, a bond on CBPF 301 shall be on file with the Port Director, containing the bond conditions set forth in 19 CFR 113.62. The principal named in the bond is the person responsible for exportation of the merchandise. When merchandise is accepted by a bonded carrier for transportation and exportation, the bonded carrier becomes responsible for causing the merchandise to be exported and providing proof of exportation (19 CFR 18.26(d)). If a bonded carrier is not reasonably available, a bond containing the conditions set forth in 19 CFR 113.62 may be used for indirect importations upon approval of the local Port Director as set forth in 19 U.S.C. §1553. A licensed cartman is not responsible for exportation, but rather only for cartage to the place of lading for exportation.
- (i) Exportation by Mail Merchandise may be transferred from a zone for exportation by mail in accordance with the provisions of 19 CFR 145 Subpart, F. The zone exporter shall endorse on each mail article a waiver of the right to withdraw the merchandise from the mails (19 CFR 145.71(b)). Mail export shipments will be transported by licensed cartman to the postal facility under the procedures of 19 CFR 125. For export statistical reporting, follow procedures specified in 15 CFR 30.

- (j) Supervision and Control of Exports Exportation will be generally supervised under CBP procedures for export supervision. CBP personnel or an authorized representative of the exporting carrier will certify lading for exportation on CBPF 7512. Responsibility for furnishing proof of exportation, if demanded by CBP, is on the bonded carrier named on the CBPF 7512 (19 CFR 113.62(h) (3) or 19 CFR 18.26(d)). Conditions for cancellation of a bond to assure exportation by CBP are set forth in 19 CFR 113.55. Merchandise exported by mail may be deemed exported when an authorized representative of the Post Office signs for receipt of the mail package. If satisfactory proof of exportation is not furnished, the Port Director may issue a claim for liquidated damages against the principal named in the importation and entry bond or the custodial bond of the bonded carrier, as applicable. For export statistical reporting, follow procedures specified in the 15 CFR 30.
- (k) Merchandise Not Laden Merchandise entered for exportation from a zone, but not laden, shall be sent to general order unless other disposition is prescribed by the Port Director (19 CFR 144.37(f)). However, if the merchandise is not accepted by the exporting carrier, the merchandise may be returned to the zone in its previous zone status. If the zone is in the same port, the merchandise not laden may be returned to the zone and the entry on CBPF 7512 canceled. If the zone is located in a different port, a CBPF 7512 (IT) is submitted to return the merchandise not laden to the zone. Upon written application by the person making entry for exportation, and with the consent of the owner of the dock or airline, the Port Director may allow merchandise so entered for exportation to remain at the dock or airport for up to 90 days, with extensions for up to 1 year (19 CFR 18.24(a)). Merchandise retained at an airport for export through in-flight sales aboard aircraft is subject to special procedures under CD 099 3280-008.
- (l) Relanding of Goods If any merchandise entered or withdrawn for exportation without payment of duties is re-landed at any place in the United States without an entry having been made, the merchandise will be considered to have been imported contrary to law, and each person concerned will be subject to fine or imprisonment or both, and the merchandise will be forfeited (18 U.S.C. §544). Civil penalties may also be applicable if false or negligent drawback claims have been filed in connection with such merchandise pursuant to 19 U.S.C. §1593a.
- (m) Reimportation of Textile Articles Changed in Zone Textiles and textile products which have been changed as set forth in 19 CFR 146.63(d), may be exported and returned to the Customs territory for warehousing, provided the entry for warehouse is endorsed by the Port Director to show that the merchandise may not be withdrawn for consumption (19 CFR 146.67(d)).
- (n) Reimportation of Merchandise Manufactured in Zone Articles produced or manufactured in a zone, and exported therefrom without the payment of duties on the foreign content, shall on subsequent importation into the Customs territory of the United States, be subject to the import laws applicable to like articles manufactured in a foreign country (HQ 225903 dated January 9, 1995). The domestic content of zone-manufactured

articles, as well as the foreign content, become subject to duty and tax, to the same extent as like articles manufactured or produced in a foreign country, upon reimportation of the merchandise. Articles produced or manufactured in a zone exclusively with the use of domestic merchandise, the identity of which has been maintained in accordance with the Second Proviso, 19 U.S.C. §81c (a) may, on such reimportation, be entered under the provisions of Chapter 98, Sub-Chapter I, HTS. Sixth Proviso, 19 U.S.C. §81c(a), 19 CFR 146.67(e).

- 9.14 Weekly Entry for Transportation or Exportation The Operator or User may elect to file an application for a weekly permit to enter and release merchandise during any seven-day period for exportation, transportation, or transportation and exportation (19 §1448, as amended by The Trade and Development Act of 2000, Title IV, Section 410; see 19 CFR 146.68(a)). The application shall be on CBPF 7512 stating at the top the words "Application for Weekly Zone Permit" in capital letters, and shall be filed with the Port Director. The application shall be accompanied by a pro forma invoice or schedule like that required in 19 CFR 146.63(c)(1). If actual transfers will exceed the estimate for the week, the person with the right to make entry shall file a supplemental CBPF 7512 to cover the additional merchandise to be transferred from the zone or subzone site. No merchandise covered by the weekly permit shall be transferred from the zone before approval of the application by the Port Director (19 CFR 146.68(a)).
 - (a) Purpose of Weekly Permit The principal purpose of the weekly permit is to enable CBP to issue authorization to the Operator to execute the individual CBPF 7512 on Customs behalf for the merchandise estimated to be removed from the zone for transportation and/or exportation during a seven day period. Upon approval of the estimated CBPF 7512, there is no need for the Applicant to secure Customs signature for release on every shipment from the zone during the seven day period covered. However, unlike the weekly procedure for entry for consumption, entry for transportation and/or exportation does not occur until the carrier signs for receipt of the merchandise covered by an individual entry filed under the weekly permit (19 CFR 146.68(b)).
 - (b) Preparation of Application The Application for Weekly Permit, appropriately marked as required under 19 CFR 146.68(a), need only be filled out so as to show the description of the merchandise for which the permit is sought and the seven day period during which it is to be removed from the zone. The application shall not be accepted without the required invoice or schedule of the number of units of each model or type of merchandise to be removed, and their dutiable values (See Section 9.7(h)(1) FTZM). Merchandise covered by the application may be removed from the zone only when the Port Director has approved the weekly permit. The permit shall be signed by the Port Director, if appropriate, and returned to the Applicant before the beginning of the covered seven-day period. If actual removals will exceed the estimate for the week, the person filing the weekly application shall file a supplemental application on CBPF 7512 to cover the additional removals before CBP will permit their removal from the zone.
 - (c) Individual Entries on CBPF 7512 After approval of the application for a weekly permit by the Port Director, the person making entry will be authorized to execute

individual CBPF 7512 for exportation, transportation, or transportation and exportation of the merchandise covered by the permit. Upon transfer of the merchandise, the Operator shall obtain a receipt from the carrier on CBPF 7512 to ensure transfer of liability to the custodial bond of the carrier or cartman. CBP will consider the time of entry to be when the removing carrier signs the receipt for the merchandise. The Operator shall give the bonded carrier a copy of the individual CBPF 7512, as set forth in 19 CFR 18.2(c). The Operator also shall ensure that the Port Director receives a copy of the CBPF 7512 by the end of the next working day after the carrier has receipted for the merchandise (19 CFR 146.68(b)).

- (1) Statements on Form and CBP Assistance Individual entries shall contain the statements required in 19 CFR 146.62(b)(1) and, when applicable, 19 CFR 146.66(b). The Port Director will see that the Operator and/or the Applicant is properly trained to assure that original in-bond documentation is properly prepared and transmitted to the carrier for transportation in-bond, as provided in 19 CFR 18.2(c).
- (2) CBP Processing of the CF 7512 Upon receipt, CBP will process the CBPF 7512 through the automated system according to current procedures. Timely entry of the information into the automated system is critical for the accurate tracking of the movement of the cargo between ports.
- (d) Weekly Statement of Merchandise Entered The person making entry for merchandise under an approved weekly permit shall file with the Port Director, by the close of business on the second working day of the week following the end of the seven day period, a statement of the merchandise entered under that permit. The statement must list each CBPF 7512 by its unique in-bond number, and will provide a reconciliation of the quantities on the individual CBPF 7512 submitted to CBP to the quantity estimated for the week, as well as an explanation of any discrepancy (19 CFR 146.68(c)).
 - (1) Processing of Weekly Statement When the statement of merchandise entered under a weekly permit is received, it shall be checked in a timely manner by CBP against the weekly permit and copies of CBPF 7512 received from the Operator. If the total quantity of merchandise entered on CBPF 7512 for exportation, transportation, or transportation and exportation is in excess of the amount covered by the weekly permit, the excess will be considered a default involving merchandise under the Operator 's bond (19 CFR 113.73(a)(2)). If the weekly permit covers merchandise which was not entered on CBPF 7512 during the seven-day period, the permit for removal will be deemed to have lapsed at the end of the seven-day period (See Section 9.8 FTZM).
- 9.15 Transfers for Vessel or Aircraft Supply Articles in zone status may be transferred from a foreign-trade zone for supplies and equipment of various aircraft and vessels actually engaged in foreign-trade or engaged in certain trades among various portions of the United States or

between the United States and its insular possessions (19 U.S.C. §1309 and 19 U.S.C. §1317). Complete guidelines and instructions for the application of those laws is beyond the scope of this manual, since they touch also on bonded warehouses, other places in continuous CBP custody, and facilities under bond of the Internal Revenue laws. The principal regulations dealing with such articles are 19 CFR 10.59 through 19 CFR 10.65 and 19 CFR 146.69(a).

- (a) Entry Procedures Each application for transfer of merchandise from a zone for delivery to a qualified vessel or aircraft at the same or a different port shall be made on CBPF 7512 as provided in 19 CFR 10.59(e), 19 CFR 10.60, and 19 CFR 146.69(a). Each application shall contain the statements required in 19 CFR 146.62(b)(2) and, when applicable, 19 CFR 146.70(d) and see Section 9.11(d)(1) FTZM. The person making entry shall furnish a bond on CBPF 301 containing the bond conditions provided for in 19 CFR 113.62, whether the merchandise is to be laden as supplies at the same or a different port (19 CFR 146.69(a)).
- (b) Eligibility for Duty-Free Entry Eligibility for duty-free entry of vessel and aircraft supplies and equipment is limited by 19 U.S.C. §1309, 19 U.S.C. §1317 and/or CBP regulations.
 - (1) Actually Engaged in Foreign Trade - Generally, supplies for vessels and aircraft which are actually engaged in foreign trade are entitled to the duty-free withdrawal privilege. However, duty-free treatment does not extend to supplies for vessels or aircraft on trial or test trips, or being ferried or delivered to a new owner or home port, preparatory to being actually engaged in foreign trade (19 CFR 10.59(b)). However, vessels or aircraft departing without passengers or cargo for the purpose of taking on cargo or passengers at another port may be considered to be engaged in foreign trade (19 CFR 10.59(a)(3)). Vessels or aircraft on the domestic leg of a voyage or flight proceeding to, or arriving from, a foreign port or place are considered engaged in foreign trade. However, on inward voyages or flights, duty-free entry privileges do not extend to quantities of supplies in excess of actual needs for the remainder of the voyage or flight, unless the vessel or aircraft is proceeding to a foreign port or place immediately after the discharge of cargo or passengers at the termination point of the voyage or flight (CD 099 3240-028 dated May 27, 1999).
 - (2) Other Eligible Trades Generally, supplies for vessels or aircraft in coastwise trade are not entitled to duty-free entry privileges. However, duty-free entries are permitted for supplies for (1) voyages or flights between the U.S. and any of its possessions, (2) voyages or flights between Hawaii or Alaska and the rest of the United States, or (3) vessel voyages (but not aircraft flights) between the Atlantic and Pacific coasts of the United States. Duty-free entry privileges do not extend, however, to petroleum products for vessel voyages or aircraft flights exclusively between Hawaii or Alaska and any airport or Pacific Coast seaport of the United States (19 U.S.C. §1309(a)).

- (3) U.S. Flag vs. Foreign Flag Duty-free entry privileges for U.S. flag vessels (including fishing vessels) and aircraft is limited to supplies, i.e. consumables such as food, beverages, films, paint, rope, fuel, etc. Foreign-flag vessels (including fishing vessels) may enter free of duty not only supplies, but also vessel equipment and articles and materials needed to repair the vessel. Foreign-flag aircraft may withdraw free of duty supplies, equipment (including ground equipment), and articles and material for aircraft maintenance and repair, but only if substantially reciprocal privileges are accorded by that country to aircraft registered in the U.S. (19 U.S.C. §1309(d)). A list of countries which currently accord such reciprocal treatment, with certain exceptions, is provided in 19 CFR 10.59(f). Entries of duty-free supplies and equipment for U.S. flag vessels may not qualify as exportations for the purposes of some laws. (See Section 9.14(c) FTZM).
- (4) Fisheries Vessels Supplies, equipment, and repair articles may be entered free of duty and tax for U.S. or foreign fishing vessels to the extent noted in Section 9.14(b)(3) FTZM above. However, duty and tax-free entry of distilled spirits, wines, and beer is limited to (1) U.S. flag fishing vessels which are documented and have a fisheries license endorsement, or (2) foreign-flag vessels of 5 net tons or more. Furthermore, the Port Director may deny duty and tax-free entry of alcoholic beverages if he or she is satisfied that (1) the vessel is not engaged in substantially continuous fishing activities and (2) the vessel's complement and expected time at sea are not appropriate for the quantities to be withdrawn (19 CFR 10.59(e). Special reporting requirements apply for fishing vessels (See Section 9.14(g) FTZM).

As a condition to the lading of domestic tobacco products, CBP may, in order to protect the revenue, require assurances from the vessel master of the receiving vessel that quantities to be laden are reasonable considering the number of persons to be carried, the vessel's itinerary, the duration of its intended voyage, etc., and that such articles are to be used exclusively as supplies on the voyage. Tobacco products are not authorized to be laden free of duty and taxes for foreign fishing vessels less than 5 net tons (See 19 CFR 10.65(a) and 27 CFR 290.62).

(5) Government and Military Vessels and Aircraft - Duty-free entry privileges extend to supplies for vessels or aircraft operated by the United States and vessels of war of any foreign flag. In addition, many foreign governments own vessels and aircraft which actually engage in foreign trade and for that reason their supplies are entitled to duty-free entry privileges. There is no provision in 19 U.S.C. §1309 or 19 U.S.C. §1317 for duty-free entry of supplies for foreign military aircraft. However, supplies for such aircraft may be entitled to duty-free entry privileges under the provisions of 19 CFR 148.90 and HTS Subheading 9808.00.30. And see Section 9.10(c) FTZM.

- (6) Cigars, Cigarettes, and Tobacco Products Cigars, cigarettes, and other tobacco products may be shipped or delivered for consumption free of duty and Internal Revenue tax only for consumption beyond the jurisdiction of the United States, i.e., beyond the 3-mile limit or the international boundary (19 U.S.C. §1317). Foreign tobacco products are not authorized to be entered free of duty and tax for fishing vessels. The privilege is also not granted to vessels stationed in American waters for an indefinite period without a sailing schedule (19 CFR 10.65(b)).
- (c) Supply Use vs. Exportation Requirements The shipment or delivery of any merchandise for use as supplies, or in the maintenance or repair of any vessel or aircraft, or as ground equipment for any aircraft under 19 U.S.C. §1309(a)(2) or (3), i.e., for foreign flag vessels or aircraft, will be deemed an exportation within the meaning of CBP and Internal Revenue laws applicable to the exportation of such merchandise without the payment of duty or Internal Revenue tax. Thus, section 1317(b) does not provide authority for the entry of such articles for use on U.S. flag vessels or aircraft to be deemed exported within the meaning of CBP and Internal Revenue laws. Therefore, there is no authority for the transfer to U.S. flag vessels of zone-restricted status merchandise, alcoholic beverages or tobacco products specified in Section 9.11(d) FTZM, or any other merchandise where exportation is a condition or requirement under CBP or Internal Revenue laws.
- (d) Filing of Withdrawal Entries for aircraft or vessel supplies shall be filed on CBPF 7512 in 5 copies. The Port Director may require an extra copy or copies of CBPF 7512 for use in connection with the delivery of the merchandise to the carrier. One copy of the form will be used as the permit copy to be retained by the Operator. One copy of the form will be used to initiate the in-bond movement under the Automated Commercial System. The other copies will be returned to the Operator or User for transmittal to the bonded carrier or licensed cartman.
- (e) Bond Requirements A bond on CBPF 301 containing the conditions set forth in 19 CFR 113.62 shall be required for any entry for vessel or aircraft use, including entries for fishing vessels (19 CFR 10.60(c) and 19 CFR 146.69(a)). However, no bond shall be required of vessels of war (19 CFR 10.60(e)) or for lading and use of tobacco products on vessels operated by the U.S. Government (19 CFR 10.65(c)(1)). The conditions in 19 CFR 113.62(h) holds the withdrawer responsible for lading and proper use as vessel or aircraft supplies. There is no authority to hold a bonded carrier responsible for vessel or aircraft supply use under an entry for transportation and lading at another port. When merchandise has been laden for vessel or aircraft supply use, the carrier becomes responsible for the safekeeping and disposition of the merchandise under its international carriers bond (19 CFR 113.64(b)).
- (f) Delivery to and Retention on Dock Upon acceptance of the entry and bond, the merchandise will be released to the Operator for delivery directly to the vessel or aircraft, if for lading on a vessel or aircraft within the zone. If the merchandise is not to be laden

on a vessel or aircraft within the zone, it will be released to the Operator for delivery to a licensed cartman, or bonded carrier for transportation to the place of lading (19 CFR 146.69(b) and (c)).

However, upon application of the person filing the entry, and with the consent of the owner of the dock or of the airline, merchandise entered for lading as vessel or aircraft supplies may be retained at the dock or airport for up to 90 days, with extensions of up to 1 year (19 CFR 18.24(a)).

- (g) Special Permit and Procedure for Fishing Vessels Before a permit is given for entry for delivery of merchandise to fishing vessels for their use under 19 U.S.C. §1309, the Port Director shall have approved a special permit on CBPF 5125, supported by a bond on CBPF 301, containing the conditions set forth in 19 CFR 113.62. The vessel master shall present the original and triplicate copy of CBPF 5125 to CBP within 24 hours after each successive arrival at a CBP port or station for an accounting for the entry until the Port Director is satisfied that the merchandise was consumed on board or properly relanded under CBP supervision (19 CFR 10.59(e) and 19 CFR 113.62(h)(2)). Failure to comply with the conditions of the entry will subject the total quantity of merchandise entered to the assessment and collection of duties and taxes equal to the amount that would have been collected if they had been regularly entered or withdrawn for consumption (19 CFR 10.59(e)).
- (h) Special Procedures for Cigars and Cigarettes The duty-free entry of cigars and cigarettes under Section 1317 requires that they be consumed only beyond the 3-mile limit or international boundary. The provisions of this subsection apply to such articles.
 - (1) Packing in Shipping Cases When a shipping case of cigars and cigarettes is made up of a number of units, each in a separate package, such units may be entered separately, provided each is numbered and marked for identification and contains not less than 250 cigars or 1,000 cigarettes (19 CFR 10.65(c)).
 - (2) Declarations of Use The Port Director may require a declaration of use from a person having knowledge of the facts that consumption of the cigars and cigarettes did not begin until the aircraft or vessel had proceeded beyond the 3-mile limit or international boundary (19 CFR 10.65(c)(5)).
- (i) Supervision and Cancellation of Bonds CBP personnel do not routinely conduct physical supervision of lading of vessel or aircraft supplies. General supervision is maintained through certificates of lading as aircraft or vessel supplies signed by an authorized representative of the using carrier. A copy of the certification on CBPF 7512 may be associated by CBP with the outward manifest of the vessel or aircraft. The entry bond is usually credited or canceled upon filing the appropriate declaration with the director of the port of withdrawal (19 CFR 10.64(a)).

- 9.16 Transfer of Zone-Restricted Status Merchandise ZR status merchandise may be transferred to the Customs territory only for entry for exportation, for entry for transportation and exportation, for warehousing pending exportation, for destruction (except destruction of distilled spirits, wines and fermented malt liquors), for transfer from one zone to another, or for delivery to a qualified aircraft under sections 1309 or 1317, unless the Board has ruled that the return of the merchandise to the Customs territory for domestic consumption is in the public interest. With Board approval, the merchandise may be entered for consumption, for warehousing, for immediate transportation without appraisement, or under any other provision of CBP laws, unless the Board has specified the form of entry to be made (19 CFR 146.70(a)).
 - (a) Entry for Consumption If the Board has ruled that the return of ZR status merchandise to the Customs territory for consumption is in the public interest, the entry will be endorsed by the Port Director to show the authority under which it was made, and that the merchandise is subject to the provisions of Chapter 98, Sub-Chapter I, HTS (19 CFR 146.70(b)). Thus, if ZR status merchandise is of domestic origin or was previously entered for consumption, it shall nevertheless be entered for consumption and is subject to duty unless exempt therefrom under the HTS or other law.
 - (b) Entry for Warehouse Pending Exportation ZR status merchandise may be transferred from a zone to a bonded warehouse for storage pending exportation. The CBPF 7501 shall be endorsed by the Port Director to show that the merchandise may not be withdrawn for consumption. In the case of ZR status merchandise transported in-bond to another port for warehousing and exportation, CBPF 7512 shall be endorsed by the Port Director to show that the merchandise is in ZR status, which shall be entered for warehouse with proper endorsement on CBPF 7501, and which may not be withdrawn for consumption. ZR status merchandise transferred from a zone to a bonded warehouse may not be manipulated, except for packing or unpacking incidental to exportation (19 CFR 146.70(c)).
 - (c) Transfer to Customs Territory ZR status merchandise may be returned to Customs territory only when the Board determines it is in the public interest. A request for authority to return merchandise must be made in letter form to the Executive Secretary of the Board. The Executive Secretary may act for the Board in cases involving merchandise valued at \$500,000.00 or less (15 CFR 400.44). In all cases the Port Director must endorse the entry (19 CFR 146.70).
 - (d) Transfer for Other Purposes Upon acceptance of an entry or withdrawal for zone-restricted status merchandise for any purpose other than that described in a FTZB order, the entry shall be endorsed by the person making entry to show that actual exportation of the merchandise is required by the Fourth Proviso, 19 U.S.C. §81c(a), or the entry endorsed to require delivery to a qualified vessel or aircraft, under 19 U.S.C. §1309 or 19 U.S.C. §1317 (19 CFR 146.70(d)). For the treatment of zone-restricted status merchandise transferred to Customs territory for Temporary Importation Under Bond (TIB) or exhibition at a trade fair. (See Sections 9.9(b) and 9.9(c) FTZM). Zone-restricted status merchandise may not be transferred from a zone for shipment to

insular possessions of the U.S. (See Section 9.12(c) FTZM). For use on a U.S. flag vessel see Section 9.14(c) FTZM.

- 9.17 Release and Transfer from Zone Except in the case of domestic status merchandise for which no permit is required under 19 CFR 146.43, no merchandise shall be transferred from a zone without a CBP permit on the appropriate entry form or other document as required in 19 CFR 146 and this manual. The Port Director may authorize transfer from a zone without physical supervision or examination by CBP personnel. Upon issuance of a permit, the Port Director will authorize delivery of the merchandise only to the Operator, who then may release the merchandise to the importer or carrier (19 CFR 146.71(a)). No permit of the Operator under 19 CFR 146.9 is required for release from a zone. If the Operator refuses release of the merchandise, the entry shall be canceled and the merchandise will be deemed constructively transferred back to the zone in its previous zone status (19 CFR 146.61). The Operator shall supervise all transfers from the zone as required by law and regulation (19 CFR 146.4(a)).
 - (a) Liability for Discrepancies When a transfer is not physically supervised by CBP personnel, the Operator will be relieved of responsibility for the merchandise in a zone only in the condition and quantity as shown on the entry or other appropriate form. The Operator will be relieved of responsibility only if it receives the signed receipt on the document of the importer or the carrier named in that document. The responsibility of the Operator may be adjusted by any discrepancy report made as described in Section 10.6 FTZM.
 - (b) Sealing of Conveyances The Operator is authorized to affix a seal to a conveyance or intermodal container at the order of the Port Director (19 CFR 146.8). Operators are authorized under 19 CFR 24.13(c) to purchase such seals, which must meet the standards and specifications of 19 CFR 24.13(a) to be accepted by CBP. Seals obtained by the Operator must be handled as provided in the standard and specifications in TD 72-56.
 - (c) Recordkeeping and Accounting for Transfers The Operator must have an inventory control and recordkeeping system(s) capable of accounting for all merchandise transferred or removed from the zone (19 CFR 146.21(a)(1)). The Operator shall record in that system(s), all zone status merchandise transferred from the zone, and the system(s) must have the capability to trace or attribute all transfers back to a zone admission under a CBP authorized inventory method (19 CFR 146.24(a)).
 - (d) Time Limit for Removal from Zone Except in the case of articles for use in a zone (see Section 2.6 FTZM), merchandise for which a CBP permit for transfer to the Customs territory has been issued must be physically removed from the zone within 5 working days of issuance of that permit. The Port Director, upon request of the Operator, may extend that period for good cause. Merchandise awaiting removal within the required time limit will not be further manipulated or manufactured in the zone, but will be segregated or otherwise identified by the Operator as merchandise that has been constructively transferred to the Customs territory (19 CFR 146.71(c)).

Although merchandise which has been constructively transferred may not be manufactured or manipulated, and shall be removed from the zone within a time limit, CBP nevertheless allows some administrative exceptions to these requirements, as noted in subparagraphs below:

- (1) Merchandise Covered by Weekly Entry or Permit The restriction against manufacture or manipulation of merchandise which has been constructively transferred is not applicable to merchandise covered by weekly entries for consumption or weekly permits for transportation and/or exportation under 19 CFR 146.63(c) and 19 CFR 146.68. A reasonable period of time may be allowed for the removal from the zone of certain merchandise covered by a weekly entry or permit (See Section 9.8(e) FTZM).
- (2) Non-Processing Stipulation The Port Director may, for good cause at the request of the Operator, allow merchandise which has been constructively transferred to remain in the zone indefinitely if the importer of record provides a written stipulation with, or in connection with, the entry that:
 - (i) the merchandise will not be further manufactured, manipulated, or otherwise changed in the zone so as to change its tariff classification or to avoid any restriction on its entry into U.S. commerce; and
 - (ii) the Operator will provide a proper audit trail and recordkeeping for Customs to verify whether or not the merchandise was so manufactured, manipulated or changed after entry but while it was still in the zone.

If the terms of the stipulation are violated or not followed, the provisions of 19 CFR 146.71(d)(1) will be followed. In addition, the importer of record may be subject to penalty action under 19 U.S.C. §1592.

- (e) Time of Transfer from Zone Foreign and ZR status merchandise, and merchandise which has been constructively transferred under 19 CFR 146.61, shall be considered to be transferred from the zone, and the Operator relieved of responsibility for the merchandise when (1) a proper permit has been issued by CBP for its transfer from the zone, and (2) the carrier, importer, or other responsible party has signed to physically remove the merchandise from the activated area of the zone. Domestic status merchandise is considered to be transferred from the zone when it has been physically removed from the zone and recorded in the inventory control and recordkeeping system(s).
- 9.18 Retention or Return of Zone Merchandise The Port Director shall cancel any entry for consumption where (1) the merchandise is not transferred from the zone within the period specified in 19 CFR 146.71(c), or (2) the merchandise was transferred from the zone but did not enter the commerce of the U.S. in the Customs territory, and was subsequently readmitted to a

zone in domestic status. If the Port Director has reason to believe any new entry would be canceled under the provisions of this section, he or she will reject the entry or demand a written stipulation, as a condition of entry acceptance that the merchandise will not be returned to a zone in domestic status. Merchandise covered by an entry which has been canceled under this section shall be restored to its last foreign status (19 CFR 146.71(d)(1)).

- (a) Purposes The purposes of this provision are to (1) provide a means for dealing with merchandise which remains in the zone beyond the specified time limit, (2) affirm the legal status of constructively-transferred merchandise as being in the Customs territory and no longer in zone status, (3) prevent the filing of entries for consumption when the merchandise does not enter the commerce of U.S. in its form as entered, and (4) to avoid import laws by entering partially-finished merchandise at a low rate of duty, and leaving it or returning it to the zone for further manufacturing or manipulation into a product which would have had a high rate of duty or would have been inadmissible into U.S. commerce. It was to avoid the latter situation that CBP issued LD 79-0229, C.S.D. 79-378 and HQ 210108 concerning such a practice.
- (b) Exceptions Entries of the following categories of merchandise will not be canceled, even though the merchandise may be retained in the zone:
 - (1) Articles for use in a zone, such as production equipment, construction materials, and articles to be consumed in the zone;
 - (2) Merchandise which is readmitted in zone-restricted status; and
 - (3) Merchandise which is exempted from the time limit requirement under the conditions set forth in 19 CFR 146.71(d)(1).

Also, merchandise and articles which are exempt from entry requirements, such as domestic status merchandise, intangibles under General Headnote 4 HTS, and articles exempted under 19 CFR 141.4 are unaffected by entry cancellation.

- (c) Entry into Commerce A determination whether merchandise has or has not entered United States commerce shall be made by the Port Director on a case-by-case basis considering factors such as:
 - (1) Length of time merchandise was outside the zone before its return to a zone;
 - (2) Whether return of the merchandise was made by the importer of record or his agent, or by a person acting in collusion with the importer of record;
 - (3) Credible evidence that there was intent by the importer or others, at the time of entry, to return the merchandise to the zone;

- (4) The merchandise would avoid a higher rate of duty or an import restriction by being returned in domestic status;
- (5) The merchandise was not used nor was not the subject of a bona fide sale by the importer after entry;
- (6) The merchandise was not further processed or manufactured outside the zone, or such processing or manufacture was minimal or cosmetic in nature.

An administrative ruling on whether merchandise, in any particular fact situation, has or has not entered U.S. commerce, including prospective transactions, may be obtained by interested parties or CBP personnel as specified in 19 CFR 177 (TD 86-16, 51 FR 5040).

- (d) Actions in Lieu of Entry Cancellation If the merchandise which was retained or returned constitutes only a part of the merchandise in an entry, the entry shall be liquidated or reliquidated as authorized by law and regulations, without payment of duty for the retained or returned part and an appropriate refund of estimated duties made to the importer of record. If the original consumption entry has already been liquidated and cannot be reliquidated under the law and CBP regulations, merchandise which is returned to the zone will nevertheless be returned to its previous zone status. The merchandise will become subject to entry once again when it is later constructively transferred to Customs territory.
- (e) Erroneous Entry A component of merchandise which has been entered, but not physically removed from a zone, shall be restored to its last zone status, provided the Port Director determines that the component was included in the entry through clerical error, mistake of fact, or other inadvertence not amounting to an error in the construction of the laws. Such an error, including that in appraisement of any entry or liquidation due to the above circumstances, may be corrected pursuant to 19 U.S.C. §1520(c)(1), in accordance with the procedures described in 19 CFR 173. If the Port Director decides there has been no error, mistake, or inadvertence, or that the information was not timely provided, the component will be considered an overage and subject to the provisions of 19 CFR 146.53(d) and 19 CFR 146.71(d)(2) and TD 90-32.
- (f) Readmission in Domestic Status When merchandise which has been entered for consumption is subsequently returned to a zone for a reason other than that specified in 19 CFR 146.71(d)(1), it shall be admitted in domestic status (19 CFR 146.71(d)(3)). This provision is applicable when merchandise which was entered for consumption has, to the satisfaction of the Port Director, entered U.S. commerce and is subsequently returned to the zone, without having been exported.
- 9.19 Drawback on Zone Merchandise Drawback procedures in this section are distinct from drawback paid upon admission to a zone. This section deals with merchandise which is entered for consumption from a zone and subsequently exported. Drawback may not be allowed on shipments from zones to insular possessions, which are not considered exportations (19 CFR

- 191.5 and see Section 9.12(c) FTZM). Following are legal positions taken by CBP on drawback of merchandise transferred from zones for entry for consumption:
 - (a) Manufacturing drawback under 19 U.S.C. §1313 (a) and (b) will not be allowed unless the completed article is exported within 5 years from the date of importation of the imported merchandise (19 U.S.C. §1313(i)). In 19 U.S.C. §1313 (a) and (b), unused merchandise drawback will not be allowed for merchandise unless it is exported or destroyed within 3 years from the date of importation of the imported merchandise. The date of importation is the date defined in 19 CFR 101.1. It is not the date of admission to the zone or of entry for consumption (See C.S.D. 79-19).
 - (b) Manufacturing in a zone is manufacturing in the United States for the purposes of drawback law (C.S.D. 81-44).
 - (c) Privileged foreign status merchandise which is used to make articles that are to be entered for consumption from a zone and then exported is eligible for manufacturing drawback under 19 U.S.C. §1313 (a). The fact that duty on privileged foreign status merchandise used in the manufacture of an article in a zone is not paid before the manufacture occurs does not preclude drawback eligibility (C.S.D. 83-85).
 - (d) An article which is manufactured in a zone with the use of nonprivileged foreign status merchandise, and which is withdrawn from the zone with the payment of duty and thereafter exported, is eligible for direct or substitution manufacturing drawback under 19 U.S.C. §1313 (a) and (b) (C.S.D. 85-33, modifying C.S.D. 83-85).
 - (e) An article manufactured from privileged foreign or nonprivileged foreign status merchandise exported, having not been entered for consumption, cannot be the subject of a claim for drawback after exportation (C.S.D. 85-49).
 - (f) Manufacturing and unused merchandise drawbacks are not complementary provisions. An operation that fails to qualify for less than one of the conditions does not automatically qualify under the other condition (C.S.D.-91-18 and HQ 225985 dated November 30, 1995).
- 9.20 NAFTA Duty Deferral Requirements—While a zone is not within the U.S. Customs territory, it is within the NAFTA territory (19 U.S.C. §3332(p)(31) (citing the North America Free Trade Agreement Annex 201.1 definition of U.S. territory including foreign-trade zones. Goods manufactured in a zone may be NAFTA qualified for shipments to Canada and Mexico.

Foreign merchandise admitted to the zone and subsequently shipped to Canada and/or Mexico in the same condition as admitted to the zone has no special NAFTA Duty Deferral procedures required.

(a) MANUFACTURING OPERATIONS—Merchandise manufactured in a zone is subject to NAFTA Duty Deferral requirements. If the finished product is duty-free in the

- U.S., Canada, or Mexico, the special requirements do not apply. CBP and the Census Bureau have signed an Agreement that allows foreign goods manufactured or changed in condition and shipped from the zone to Canada or Mexico to be filed using two options at the filer's discretion: 1) including the merchandise in an "06" (FTZ) entry or 2) filing a separate "08" (NAFTA Duty Deferral) entry. In either case, such goods will be subject to U.S. CBP duties, applicable antidumping/countervailing duties, merchandise processing fees and other potentially applicable CBP fees. The duties are calculated pursuant to 19 CFR 181.53(e)(1) and (2).
 - (1) MANUFACTURING OPERATIONS—"08" ENTRY—PART I AND PART II—When duties are owed in the U.S. and Canada/Mexico and the company wishes to defer, reduce, or waive the U.S. duties owed, an "08" entry is prepared. The entry consists of two parts: Part I (NAFTA Duty Deferral Entry Summary) and Part II (NAFTA Duty Deferral Claim).
 - (i) A CBPF 7501 NAFTA Duty Deferral Entry Summary summarizing all of the shipments to Canada/Mexico during the previously business week is prepared. A CBPF 3461 is not prepared. The CBPF 7501 is completed according to the instructions provided by the Executive Director, Trade Programs, U.S. CBP and Border Protection Memorandum dated December 22, 2000 (attached). Entry type "08" is indicated in Block 3. Part I is notated as a NAFTA Duty Deferral Claim setting forth the company's obligation of U.S. duties due and payable. No merchandise processing fees are paid at this time. Part I of the NAFTA Duty Deferral Entry Summary may be filed electronically.
 - (ii) CBP holds Part I of the entry for sixty (60) calendar days from the export date, which is the Canadian/Mexican import (arrival) date. Within the sixty-day period, the Operator secures proof of export, which includes Canadian/Mexican entry number; Canadian/Mexican Tariff classification number; Canadian/Mexican rate of duty; Canadian/Mexican duties paid; stated in both local currency and U.S. dollars; and Canadian/Mexican date of import. Proof of payment of Canadian/Mexican duties must be secured, which includes: Canadian entry documents, including both Canada Customs invoice (B-3) and either the K-84 or detailed coding statement; final customs duty determination of Canada; Mexican entry documents (pedimento); or an affidavit concerning same.
 - (iii) Thereafter, Part II of CBPF 7501 NAFTA Duty Deferral Entry using Part I of CBPF 7501 entry filed with the full U.S. duty liability, the amount of Canadian/Mexican duties paid in U.S. dollars, and the adjusted U.S. dollar amount of duty liability. Files supporting the entry must be maintained for a period of five years from the date of exportation. All adjusted duties and merchandise processing fees are paid with the

- submission of Part II. Part II, NAFTA Duty Deferral Entry can be filed electronically.
- (2) MANUFACTURING OPERATIONS "06" ENTRY Agreement At the filer's discretion, if there is no opportunity to reduce or waive the U.S. duties owed, the manufactured merchandise being shipped to Canada or Mexico may be included on the individual or weekly "06" Entry.
 - (i) Merchandise placed on the "06" Entry does not require the filing of CBPF 7512 or QP 7512 on these export moves.
 - (ii) Shipments to Canada no EEI/AES record is required, except for U.S. Munitions List (USML) or Commerce Control List (CCL) Articles. For these articles, the EEI/AES record must be filed with CBP prior to exportation.
 - (iii) Shipments to Mexico the export must be filed in AES or AESDirect, with the correct FTZ number and a new export information code "ZD" which is now assigned for this purpose.

Chapter 10

DISCREPANCY REPORTING AND LIABILITY

- 10.1 General The responsibility of Operators for merchandise and discrepancies is covered partially in Chapters 6, 8, 9, and 12 FTZM. The purpose of this Chapter is to describe the overall responsibility of Operator s for merchandise discrepancies, overages, shortages, and damage and to relate their responsibilities to those of the cartmen, carriers, and other parties who deliver merchandise to and from the zone. CBP general responsibility is to see that the proper duties and taxes are assessed, or exempted from assessment, according to law, on all merchandise imported into the U.S., as well as to ensure compliance with all laws and regulations concerning the importation of merchandise.
- 10.2 Foreign-Trade Zone Operator Responsibility The Operator is responsible for supervising all receipts, admissions, transfers, recordkeeping, manipulation, manufacture, processing, destruction, exhibition, physical and procedural security, and conditions of storage in a zone as required by law and regulations (19 CFR 146.4(a)). This includes prompt verification of the quantity and condition of merchandise upon admission to the zone when the Port Director has authorized receipt without CBP supervision (19 CFR 146.37(d)). The Operator is also responsible for conducting a physical inventory at least annually, unless it conducts continuous cycle counts as part of an ongoing inventory control program, and for notifying the Port Director of any discrepancies found during the annual inventory or cycle count (19 CFR 146.23(c)).
- 10.3 Sealing of Containers and Conveyances With the approval of the Port Director, an Operator may break or affix a CBP seal and such breaking or affixing will be considered to have been done under CBP supervision, as provided in 19 CFR 146.8. The Operator shall inspect CBP seals on arriving shipments, including those received under procedures for temporary deposit (19 CFR 146.35) and direct delivery (19 CFR 146.40(a)). Any seal found to be broken, missing, improperly affixed, improperly replaced with another seal, or defeated (the same seal opened and re-affixed with little or no evidence of tampering) shall be reported immediately to the Port Director and the shipment held intact pending CBP instructions. If the Operator does not receive the concurrence of the carrier as to the condition of the seal, the Port Director will deem the seal to be intact (19 CFR 146.8).
- 10.4 Discrepancies Found upon Receipt The treatment of discrepancies found upon receipt in a zone is described generally in 19 CFR 146.37(c); 19 CFR 158, Subpart A, and CBP Directive 099 3240-067 A.
 - (a) When Operator Becomes Responsible The Operator becomes responsible for merchandise under the Operator's bond when he signs for receipt in the zone of merchandise for which the Port Director has given a prior permit for admission under (19 CFR 146.32(c)), with the following exceptions:
 - (1) The Operator becomes responsible for the merchandise when he signs for

receipt of merchandise for temporary deposit (19 CFR 146.35(c)(2)) or for direct delivery (19 CFR 146.40(a)(4)).

- (2) When merchandise is otherwise delivered to the zone under proper permit to transfer, it remains under the carrier's bond, (e.g., for CBP examination prior to admission). The Operator although having signed for receipt of the merchandise does not become responsible for the merchandise under their bond until a CBP permit for admission is received.
- (3) When an overage is discovered of merchandise already in the zone, the Operator becomes responsible when he receives a CBP permit for admission.
- (b) Manifest Discrepancy Reporting 19 U.S.C. §1431 requires every vessel making entry or obtaining clearance to have a manifest in compliance with the statute. Any manifest discrepancy discovered upon the arrival of the merchandise at foreign-trade zones shall be reported by a Manifest Discrepancy Report. A Manifest Discrepancy Report must be filed to assure the appropriate correction of the manifest and to determine the proper responsibility for the discrepancy. Manifest discrepancies may be reported using the CBPF 214 or if transmitted electronically, through the concurrence functions of the e-214. CBP general procedures for filing Manifest Discrepancy Reports are found in CBP Directive 099 3240-067A. The procedures in this section incorporate the procedures of that directive.
- (c) Breakbulk, Conveyance Not Sealed CBP policy is that a discrepancy (overage or shortage, including an unconcealed within-case shortage) in loose freight or in merchandise in an unsealed conveyance or a conveyance whose seal(s) is not intact, found upon receipt in a zone shall be reported by the Operator on CBPF 6043, 7512, 214, or other approved form signed jointly within 2 business days by the Operator and the delivering carrier on the discrepancy (19 CFR 146.37(c)(1)). Since CBPF 5931 has been abolished, it will no longer be accepted as a discrepancy report. The report shall be jointly signed by the parties on the approved form within 15 days after receipt in the zone. The Manifest Discrepancy Report shall include a clear and concise statement for the discrepancy and include the following statement signed by both the Operator and carrier:

"We hereby declare that the information contained in this document is true and correct to the best of our knowledge and believe that the discrepancy described herein occurred for the reasons stated."

An Operator need not file a Manifest Discrepancy Report when the marks and numbers on merchandise packages are different from the marks and numbers listed on the manifest or cartage ticket if the quantity and description of the merchandise is accurate.

- (d) Concealed Within-Case Shortage or Discrepancy in Sealed Conveyance When an Operator discovers, within 20 days after admission of the merchandise in the zone a concealed within-case discrepancy (shortage or overage) in quantities in a container or conveyance which was received with seals intact, the discrepancy shall be reported to CBP as a Manifest Discrepancy Report on the Operator 's letterhead, with the explanation and declaration specified in Section 10.4(d) (1) FTZM and 19 CFR 146.37(c)(2). The report should be accompanied by a dock receipt or other evidence of nonreceipt or nonimportation of the merchandise, and in the case of containerized merchandise, evidence that the seal(s) was intact and that the seal(s) was affixed before importation into the U.S. If a discrepancy is found after 20 days, the procedures set forth in 19 CFR 146.53(a) should be followed.
 - (1) "Seals Intact" "Seals intact" means a conveyance or intermodal container containing imported merchandise and the seals which were reported to have been affixed to it meet all of the following conditions:
 - (i) All of the seals are of a type which will not permit removal without breaking or leaving evidence of tampering, cannot be used more than once, and cannot be easily counterfeited;
 - (ii) None of the seals which were affixed have been broken, removed, replaced, or tampered with in any manner which indicates it was removed and replaced;
 - (iii) The unique numbers of all the seals are the same as the seal numbers reported on the manifest or cartage ticket under which the conveyance was accepted for transportation to the zone;
 - (iv) The condition of the conveyance or intermodal container is such that merchandise can be removed therefrom only through sealed doors or other authorized openings; and
 - (v) The Operator and delivering carrier have executed a written concurrence that the seals are intact, or CBP has certified that the seals are intact.
- (e) Conveyance With Seals Not Intact When an Operator receives a conveyance or intermodal container whose seals are not intact according to the definition in Section 10.4(d)(1) FTZM, and the Port Director allows unlading without CBP supervision, any discrepancies between the received and manifested quantities shall be reported by the Operator as set forth in Section 10.4(c) FTZM. If CBP personnel conduct physical supervision, any discrepancies will be reported by the personnel on the permit copy of CBPF 214 and the Operator's responsibility for the merchandise will be adjusted accordingly.

- (f) Bulk Merchandise A discrepancy (overage or shortage) between the manifested amount of bulk cargo and the quantity received in a zone need not be reported as a Manifest Discrepancy Report (MDR) under Section 10.4(c) or (d) FTZM if the ort director is satisfied that the discrepancy is an ordinary and usual difference attributable to absorption of moisture, temperature difference, faulty weighing or gauging at the port of lading, or similar cause. However, any discrepancy in quantities of petroleum or petroleum products imported in bulk must be reported as a Manifest Discrepancy Report if the discrepancy exceeds 1 percent of the manifested quantity (19 CFR 4.12(c)). When a Manifest Discrepancy Report is required for a shortage or overage of bulk merchandise, it shall be made as provided in Section 10.4(c) or (d) FTZM, as appropriate.
 - (1) Weigher, Gauger, or Measurer's Report In the case of bulk merchandise, any discrepancy reported by an independent weigher, gauger, or measurer and signed by an authorized representative of the company employing such weigher, gauger, or measurer may be accepted by the Port Director to amend the quantities for which the Operator is liable.
- (g) Overages upon Admission The Operator is responsible for reporting any merchandise in excess of that reported on e-214/CBPF 214 under the procedures of Section 10.4(c) or (d) FTZM. If the e-214/CBPF 214 or a CBP entry is not filed within 15 days after discovery of the overage, the merchandise shall be sent to general order.
- (h) Damage upon Admission When merchandise is found to be damaged upon admission, the Operator may file CBPF 4315 under the procedures set forth in 19 CFR 158, Subpart B. Any allowance granted by the Port Director will be deemed to reduce accordingly the responsibility of the Operator for the merchandise received into the zone.
- (i) Temporary Deposit and Direct Delivery Procedures The Operator is responsible under its bond for the quantity of merchandise received under temporary deposit (19 CFR 146.35(c)(2)) and direct delivery procedures (19 CFR 146.40(a)(4)). Any discrepancies in the quantity or condition of the merchandise actually received shall be reported to the Port Director as provided in Sections 10.4(c) through (h) FTZM, as applicable. However, the Operator will be responsible for the quantity admitted after temporary deposit as provided in 19 CFR 146.37(c), and for the quantity admitted after direct delivery as provided in 19 CFR 146.40(c)(4).

In the case of merchandise delivered under direct delivery procedures, within-case shortages or overages found during the input quality control program of the Operator need not be reported on an MDR, if the Operator is able to report the information in another manner so that the Port Director can determine whether there is liability for the discrepancy under the bond of any party to the importation (19 CFR 146.40(c)(4)).

10.5 Discrepancies Found After Admission - After merchandise has been admitted to a zone and adjustments have been made for any discrepancies reported under 19 CFR 146.37, any

discrepancy found by a Operator before transfer from the zone will be reported and treated according to the provisions of 19 CFR 146.53 and this Section of the FTZM.

- (a) Operator Responsibility After Admission The Operator is responsible under its Operator's Bond for any loss of merchandise or for any merchandise which cannot be located or otherwise accounted for (except domestic status merchandise for which no permit is required), unless the Port Director is satisfied that the merchandise was (1) never received in the zone; (2) removed from the zone under proper permit; (3) not removed from the zone; or (4) lost or destroyed in the zone through fire or other casualty, evaporation, spillage, leakage, absorption, or similar cause and did not enter the United States commerce (19 CFR 146.53(c)(1)).
 - (1) Liability for Duty and Taxes Upon demand of the Port Director, the Operator shall pay duties and taxes applicable to merchandise which is missing or otherwise not accounted for (19 CFR 113.73(b)).
- (b) Discrepancy Reporting and Recording Requirements The Operator shall report in writing to the Port Director upon identification as such, of any (1) theft or suspected theft of merchandise; (2) merchandise not properly admitted to the zone; or (3) shortage of one percent (1%) or more of the quantity of merchandise in a lot or covered by a unique identifier, if the missing merchandise would have been subject to duties and taxes of \$100 or more upon entry into the Customs territory.

The Operator shall record upon identification all shortages and overages, whether or not they are required to be reported to the Port Director at that time, in its inventory control and recordkeeping system. The Operator shall record all shortages and overages as required in the annual reconciliation report under 19 CFR 146.25 and 19 CFR 146.53(a).

Except in the case of theft or suspected theft, the Operator need not file a report with the Port Director or note in the annual reconciliation report, any shortage or overage concerning domestic status merchandise for which no permit is required (19 CFR 146.53(b)).

- (c) Identification of Discrepancy "Upon identification" of a discrepancy under 19 CFR 146.53(a) or Section 10.5(b) FTZM means upon discovery and determination of the nature and extent of the discrepancy. A discrepancy will be considered "identified" if the Operator or zone User demonstrates that it believes the discrepancy sufficiently authoritative to take a management action to correct it. A written report within 5 working days after identification will be considered to meet the requirement of reporting "upon identification"; however, any theft or suspected theft of merchandise, including domestic status merchandise must be reported to the Port Director immediately by telephone or fax, so a prompt investigation may be initiated, if warranted, by the ICE.
- (d) Netting Policy It is CBP policy not to accept "net" reports of discrepancies except as noted below. "Netting" means to balance off shortages against overages, within

the same or different inventory categories, e.g. to balance off a shortage found during one cycle count against an overage found during the next cycle count. Each identified discrepancy must be recorded and treated separately.

- (e) Offsetting Error If the Operator can show to the satisfaction of the Port Director that an overage in one or more lots or inventory categories, counterbalanced by a shortage in one or more other lots or inventory categories, occurred through a clerical error or other inadvertence in order picking, records posting, or similar cause, the Port Director may concur with no further necessary action. In such cases, the inventory records may be so adjusted by a report to the Port Director of the error, without requiring a report of the shortage or entry or admission of the overage under 19 CFR 146.53. In any case involving an entry for consumption, the provisions of 19 U.S.C. §1520 and 19 CFR 173.4 or 19 CFR 173.4a, as appropriate, shall be followed.
- (f) Shortages Over 1% and \$100 Duty The Operator shall report to the Port Director, in writing, any shortage, including any found during the annual physical inventory (19 CFR 146.23(c)), of 1% or more of the quantity of merchandise in a lot or covered by a unique identifier, and if the missing merchandise would have been subject to duties and taxes of \$100 or more (19 CFR 146.53(a)(3)). That is, the shortage must meet both the 1% and \$100 criteria in order for the Operator to be required to report it to the Port Director in writing upon identification of the shortage. This limitation is intended to avoid the paperwork burden on both CBP and Operator's of the repetitive reporting of minor losses which must, in any event, be reported at the end of the Operator's business year.
 - (1) Negligent or Deliberate Shortages The 1%/\$100 requirement is a reporting requirement for unaccountable losses, and is not to be construed as a pretext for any deliberate, fraudulent, or negligent removal of merchandise from the zone without a proper permit (19 CFR 146.51 or 146.71(a)). Any shortage, regardless of its quantity, value, or duty liability, will be subject to liquidated damages if it occurred through a deliberate or fraudulent act or omission or through negligence.
 - (2) Cumulative Effect of Shortages Both the 1% and the \$100 criteria are cumulative during the Operator's business year. That is, whenever a series of minor shortages accumulate to exceed the 1% and \$100 requirements, the entire shortage must be reported to the Port Director. Minor losses that occur after such a report during the same business year will continue to be treated as cumulative until they reach a subsequent 1%/\$100 level. Since all shortages must be reported at the end of the Operator's business year, minor losses will be considered newly cumulative at the beginning of the next business year.
 - (3) Statutorily Duty-Free Merchandise Operators should report to the Port Director any shortage in excess of the 1% level of statutorily duty-free merchandise in foreign or zone-restricted status valued at \$1,000 or more. Port

Director s will treat any shortage above the 1%/\$1,000 value level as a transfer from the zone without proper permit for which liquidated damages may be assessed.

- (4) Method of Making Required Reports A shortage, including a theft, which is required to be reported in writing under 19 CFR 146.53(a) may be reported either on a Manifest Discrepancy Report (see Section 10.4(b) FTZM) or by filing a consumption entry under the procedures of 19 CFR 141 and 19 CFR 142. Ordinarily, the Operator will file a Manifest Discrepancy Report if it believes that it is not responsible for the loss under 19 CFR 146.53(c)(1), e.g., a concealed within-case shortage. However, if the Port Director finds the Operator responsible, duties and taxes will be demanded under 19 CFR 146.53(c)(2) and 113.73(b).
- (g) Shortages Not Over 1% and \$100 Duty Any and all shortages and other discrepancies, regardless of quantity value, or duty liability, must be recorded in the Operator's inventory records (19 CFR 146.23(b)) and/or lot file folder (19 CFR 146.37(a)(1)), where applicable, and reported in its annual reconciliation report (19 CFR 146.25). The Operator shall instruct the importer with the right to make entry to file, with the certification letter for the annual reconciliation report, a consumption entry or entries covering any and all shortages which were not previously reported to the Port Director upon identification. It is the Operator's responsibility to ensure that the discrepancy is reported to CBP.
- (h) Overages Found in Zone The person with the right to make entry shall file, within 5 days after identification of an overage, an application for admission of the merchandise to the zone on e-214/CBPF 214 or file a CBP entry for the merchandise. If an e-214/CBPF 214 or a CBP entry is not timely filed the merchandise shall be sent to general order (19 CFR 146.53(d)). A Manifest Discrepancy Report will not be accepted for overages found after admission to a zone.
- (i) Casualty and Similar Losses The Operator is not responsible for merchandise destroyed in the zone through an act of God, explosion, or fire (C.S.D. 86-7). Evaporation, spillage, leakage, absorption, or similar causes are not subject to the assessment of CBP duties (See 19 CFR 146.53(c)(1)(iv)). Volumetric contraction or gauging differences which cause apparent losses, are also not subject to the assessment of CBP duties because there is no actual change in the volume (19 CFR 146.53(c)(1)(iv)). Such losses must, however, be reported upon identification as specified in 19 CFR 146.53 (a)(3) or with the annual reconciliation, as appropriate. However, if the loss exceeds the 1%/\$100 duty criteria of 19 CFR 146.53(a)(3), it shall be reported to the Port Director on CBPF 4315 by using the procedures of 19 CFR 158, Subpart C. Additional evidence of injury or destruction by fire or other casualty as specified in 19 CFR 158.27 should be fully documented and justify such loss in order for the Operator to be relieved of liability for the merchandise. In the case of petroleum refineries operating under subpart H of Part 146 of CBP Regulations, the refiner is required to correct the volume for

temperature.(See 19 CFR 146.94 (d)(e)). The Operator and the Port Director are encouraged to discuss potential losses and gains specific to each commodity prior to admittance to the zone. Commodity variances may be detailed in the procedures manual. CBP Headquarters may authorize an allowance for variances in volume caused by nitrogen entrained in bulk concentrated orange juice stored in stainless steel tanks (See Section 11.10(b) FTZM).

- (j) Damage to Merchandise in Zone The liability of the Operator under its bond may be adjusted for the loss of value resulting from damage to merchandise occurring in the zone. The Operator shall segregate, mark, and otherwise secure damaged merchandise to preserve its identity as damaged merchandise (19 CFR 146.53(e)). There is no requirement that such a loss of value be reported upon identification. However, zone Operators who wish to document such damage may do so by filing CBPF 4315 using the procedures of 19 CFR 158, Subpart C. Damaged merchandise which is entered for consumption will be granted an allowance in value as provided in 19 CFR 146.65(b)(3).
- 10.6 Discrepancies Found Upon Transfer to the Customs Territory When a transfer is not physically supervised by CBP personnel, the Operator will be relieved of responsibility only for the merchandise transferred from the zone in the condition and quantity as shown on the entry, transfer, or other appropriate form. The Operator will be relieved of responsibility only if it receives the signed receipt on the document of the importer or the carrier named in that document. The responsibility of the Operator may be adjusted by any discrepancy report made jointly by the Operator and the bonded cartman, lighterman, or carrier, or the importer, and signed by the above or an authorized representative within 15 days after transfer of the merchandise from the zone. Any adjustment must be noted on the permit copy of the entry, withdrawal, or other appropriate form or document. A copy of any joint report or discrepancy must be submitted to the Port Director within 10 working days of signing by the parties (19 CFR 146.71(b)).
 - (a) Seal Concurrence If the cartman, carrier, or importer does not obtain the written concurrence of the Operator in the condition of any CBP seal affixed by the Operator, the seal will be deemed intact.
 - (b) CBP Physical Supervision If the Port Director requires lading and/or transfer from the zone under CBP physical supervision, any discrepancies will be reported by the appropriate CBP personnel on the permit copy of the entry document. The Operator's responsibility will be adjusted to the extent of any quantity discrepancy reported by CBP personnel.
 - (c) When Operator Is Relieved of Responsibility Merchandise is considered transferred from a zone, and the Operator thereby relieved from liability, when:
 - (1) The Port Director has properly issued a permit for transfer from the zone under 19 CFR 146.71(a); and

- (2) The Operator receives the signed receipt on the document of the importer or carrier named in the document for removal of the merchandise from the zone under 19 CFR 146.71(b) (See Section 9.16(e) FTZM).
- 10.7 Discrepancy Recordkeeping Any discrepancy identified under CBP Regulations must be accurately recorded in the inventory records of the Operator and fully reported in the annual reconciliation report. Copies of any documents establishing or providing evidence of a discrepancy must be retained by the Operator for 5 years after the merchandise is removed from the zone (19 CFR 146.4(d)). When any claim for a discrepancy allowance must be approved or substantiated by CBP, the Port Director shall provide a report to the Operator on the resolution of the claim so the Operator is fully aware of its responsibility and obligations under the Operator's bond.

Chapter 11

PROHIBITIONS, RESTRICTIONS, AND SPECIAL MERCHANDISE PROVISIONS

- 11.1 General - Foreign and domestic merchandise of every description, except such as is prohibited by law, may, without being subject to CBP laws of the United States, be brought into a zone and may be stored, sold, exhibited, broken up, repacked, assembled, distributed, sorted, graded, cleaned, mixed with foreign or domestic merchandise, or otherwise manipulated, or be manufactured except as otherwise prohibited by the Foreign Trade Zone Act of 1934 as amended (19 U.S.C. §81c(a)). Thus, although many kinds of merchandise may be admitted to a zone and many kinds of activities undertaken in a zone, there are prohibitions and/or restrictions on a few kinds of merchandise and activities. On the other hand, there are provisions of law that provide a benefit to some merchandise, but not others, in a zone. Prohibitions and restrictions may be specifically set forth in the law, or authorized by the law but implemented through regulations or specific Federal agency directives of Federal Government agencies. The prohibition or restriction may bar or limit the admission of the merchandise to a zone or, if allowed to be admitted, restrict what may be done with it in the zone or how it may be disposed of from the zone.
- 11.2 Prohibited Merchandise No merchandise or article whose importation is prohibited may be admitted to a zone (19 CFR 146.31(b)). If found in a zone, prohibited merchandise may be seized by a Federal agency under any pertinent statute, including CBP laws since it is not eligible for admission or exemption from CBP laws (C.S.D. 82-16; July 9, 1981). If, on the other hand, it is the entry into commerce which is prohibited, the merchandise may be admitted to a zone with the approval of the agency responsible for administering the pertinent law. While in the zone it may be permitted to be changed or otherwise brought into compliance for subsequent entry into commerce (See Sections 6.2 and 6.3 FTZM).
- 11.3 Public Interest Exclusion The Board may at any time order the exclusion from the zone of any goods or process of treatment that in its judgment, after due investigation, is detrimental to the public interest, health, or safety (19 U.S.C. §81o(c); 15 CFR 400.43). The Board's action could result in a total prohibition of certain merchandise or activities from a zone. However, the exclusion is usually a restriction or condition placed in a grant or other approval by the Board. Grant restrictions may limit the zone status allowed, the kind of operation on the merchandise in a zone, the entry of the merchandise into the commerce, the life of the grant, or similar transactions or activities (15 CFR 400.31, 15 CFR 400.32, and 15 CFR 400.33).
- 11.4 Limitations to Scope of Authority Components, products and activities related to zone authorized manufacturing are limited to the scope of authority described in the application and approved by the Board. This information can be found in the application notice and the official Board order as published in the Federal Register. This shall not be interpreted to mean that the manufacturing is limited only to those products listed within the scope. In general, the Grantee or Operator must notify the Board of changes in sourcing for authorized manufacturing or processing activity. Other products may be manufactured using wholly domestic status components.

- 11.5 Prohibitions and Restrictions on Admission In addition to prohibited importations or public interest exclusions, any merchandise may be denied admission for failure to meet the administrative requirements of admission, such as failure to properly prepare and present the admission documentation or to obtain a CBP permit (19 CFR 146.32).
 - (a) Bonded Warehouse Shipments Merchandise that has been entered for bonded warehouse may be admitted to a zone only in zone-restricted status (19 CFR 146.11(d), 19 CFR 146.44(d), and 19 CFR 144.37(g). (See Sections 5.8(c) and 6.7(g)(3) FTZM).
 - (b) Grant Restricted Steel Products Grant restrictions of some zones specify that steel products imported for use in the manufacturing of vessels and certain other products are subject to duties in accordance with applicable law if the same item is also being produced by a domestic steel mill (19 U.S.C. §1626 and FTZO 234 and FTZO 321). CBP assumes, when such steel products are considered for admission that a domestic mill is producing the same item and will demand an entry for consumption prior to delivery to the zone, unless written notification has been furnished to the Port Director by the Board that the same item is not manufactured by a domestic mill. If there are questions concerning grant restrictions, contact the Board.
 - (c) Motor Vehicles for Environmental Protection Agency (EPA) /Department of Transportation (DOT) Conversion in Zone Application for admission may not be made by motor vehicle dealers or brokers, or by EPA Independent Commercial Importers (ICI's) or DOT Registered Importers (RI's) on behalf of individual actual owners of motor vehicles who wish to have the vehicles repaired or processed for the purpose of bringing them into compliance with the regulatory requirements of the EPA under 40 CFR 85 and 86 Environmental Protection Agency regulations (EPAR), 19 CFR 12.73, The National Highway Traffic Safety Administration, DOT under 49 CFR 555, 49 CFR 567, 49 CFR 568, 49 CFR 571; 19 CFR 12.80, and see Sections 11.6(c) and 11.7(h) FTZM. Such applications must be made by a person with the right to make entry, such as the owner or purchaser (19 CFR 146.32(b)(2)). However, such applications may be made by Customs brokers or Operator s on behalf of the person with the right to make entry, if a proper power of attorney is on file (See Section 6.7(a)(1) FTZM).
 - (d) Non tax-paid Alcohol and Tobacco Products (1) Domestic non tax-paid alcohol and tobacco products may be admitted to zones only in zone-restricted status (C.S.D. 82-112; March 26, 1982). These products are deemed exported upon admission to the zone. As of January 1, 2000, only manufacturers and export warehouse proprietors with a valid permit from Alcohol & Tobacco Tax and Trade Bureau (TTB) may import tobacco products in-bond, including into zones (27 CFR 41). Export Warehouse Proprietors with a permit from the TTB may import tobacco products in-bond including into a zone.

Tobacco products previously exported from the United States may be imported or brought into the United States only under a license provided by the TTB. Because this

legislation refers to merchandise brought into the United States, as opposed to noting a "Customs territory" designation, zones would be included, as they are clearly a part of the United States. Merchandise brought into a zone would be subject to the relevant provisions of "The Balanced Budget Act of 1997" (P.L. 105-33; 111 Stat. 251) regarding the reimportation of tobacco products.

- (e) Firearms and Other Weapons Firearms generally require TTB permits and high security storage. Any admission and other manipulation activity involving firearms must be conducted in accordance with TTB regulations and with the written concurrence of the Board. Firearms and other articles on the U.S. Munitions Import List may be admitted to a zone only if a properly executed import permit on TTBF 6 is presented to the Port Director—with, or in conjunction with, e-214/CBPF 214 (See 27 CFR 47.41, 27 CFR 47.42). Articles on the U.S. Munitions List shown in 22 CFR, 121 (distinct from the more limited U.S. Munitions Import List shown in 27 CFR 47) which have been imported for re-export may be admitted to a zone only if a State Department In-Transit license on DSPF 61 is presented to the Port Director with or in conjunction with e-214/CBPF 214 (22 CFR 123.3). TTB will not grant a conditional import permit for certain weapons covered by the Gun Control Act of 1968 unless the zone facility in which they are to be stored meets CBP high-security storage conditions (See Section 8.7(d) FTZM).
- (f) Merchandise Subject to Antidumping or Countervailing Duties The FTZB's policy is that zone procedures shall not be used to circumvent ADD/CVD orders under 19 CFR 353 & 19 CFR 355; 19 CFR 351.. Items subject to ADD/CVD orders will only be admitted to the zone in privileged foreign status. When these items are entered into the Customs territory for consumption, the items shall be subject to ADD/CVD procedures as appropriate (15 CFR 400.33(b)(2)) (<u>Timken Co. v. United States</u>, 865 F. Supp. 413 (CIT. 1994).
- (g) Other Foreign-Trade Zone Grant Restrictions Public interest concerns as defined in the Board regulations as detrimental to public interest, health or safety have been associated with several industries. In these instances, the Board has issued grant restrictions such as time limits, required privileged foreign status, or export only grants of authority. Interested parties should review grants of authority and/or contact the Board for additional information.
- 11.6 Prohibitions and Restrictions on Zone Activities Certain activities prohibited or restricted in zones are found in both the FTZ Act and Board orders. Any application for an activity in a zone for which a permit is required may be denied for failure to meet the administrative requirements for the permit. All new manufacturing or processing activities in a zone require approval by the Board (15 CFR 400.28(a)(2)). Approval may be given by the Board for a new manufacturing or processing activity through an administrative procedure, a production authorization notice in the Federal Register, or a formal application and review. Such authorization may be denied, and have been denied in some circumstances on the grounds that the proposed operation would be detrimental to the public interest under Section 81o(c).

- (a) Residence in Zone No person shall be allowed to reside within an activated area of a zone except Federal, state, or municipal personnel whose residence is deemed necessary by the FTZB (19 U.S.C. §81o(a)).
- (b) Retail Trade Retail trade without special permit is prohibited within activated portions of foreign-trade zones (15 CFR 400.45).

Definition of "Retail Trade" - "Retail trade" means, generally, sales or offers to sell goods or services in small quantities directly to consumers or to individuals for personal use. If there are any questions on retail trade issues, the Port Director should contact the Board. Consistent with the foregoing definition, the following activities are considered retail trade:

- (i) Sales or offers to sell goods at retail in a zone;
- (ii) Sales or offers to sell services at retail in a zone (e.g., airline ticket offices, travel agencies, automobile repair shops, household or personal effects storage); or
- (iii) Sales or offers to sell services at retail made outside a zone of goods held in or services rendered in a zone.

This restriction applies only to activated areas of the zone and retail trade can be conducted without restriction from any place outside an activated zone area. A violation would occur if retail trade was conducted in an activated area of the zone without Board approval. Retail trade can be conducted from an area within the zone that has been excluded or removed from activation

- (1) Penalties for Violations Violations of the retail trade prohibition may be subject to a fine by the Port Director under 19 U.S.C. §81s and 19 CFR 146.81. Liquidated damages, where applicable, may be imposed in addition to any fines in appropriate cases (Section 13.4 FTZM). The fine may be assessed against the Grantee, or any agent, Operator, officer, or employee thereof, whether or not they are located in the zone, which has responsibility for, or has permitted the violation. All such fines pursuant to 19 CFR 146.81 are subject to review by CBP (19 CFR 146.81(b)). Petitions for relief from the fine may be filed with the Port Director under the provisions of 19 CFR Part 171.
- (c) Motor Vehicles Owned by Individuals Motor vehicles owned by individuals may not be stored, repaired, or sold in a zone in violation of the prohibition against retail trade. Neither may the vehicles owned by individuals be stored, worked on, or processed pending bringing them into compliance with the regulatory requirements of EPA under 40 CFR 85 and 40 CFR 86, 19 CFR 12.73 and the National Traffic Safety Administration (49 CFR Part 555, 49 CFR 567, 49 CFR 568, and 49 CFR 571), if such activities constitute a violation of the prohibition against retail trade. Such activities may be

conducted by registered importers (RIs) and independent commercial importers (ICIs), motor vehicle manufacturers, motor vehicle wholesalers, and similar parties, so long as the activities do not constitute prohibited retail trade (Telex 10481 dated October 17, 1984, and 2533 dated March 15, 1985, and see Sections 11.5(c) and 11.7(h) FTZM).

- (d) Articles Subject to Internal Revenue Service Code The FTZ Act prohibits the manufacture of alcohol, tobacco and firearms in zones. No operation involving any foreign or domestic merchandise brought into a zone, which operation would be subject to the IRS Code the Customs territory, shall be performed in a zone except those operations (other than rectification of distilled spirits and wines or the manufacture or production of alcohol products unfit for beverage purposes) that were permissible under the FTZ Act before July 1, 1949 (Fifth Proviso, 19 U.S.C. §81c(a)). The FTZ Act has been amended to allow the use of distilled spirits and denatured distilled spirits in certain specified processes (19 §81c (a)).
 - (1) Internal Revenue Code Provisions Now in Effect The provisions cited in the Fifth Proviso which are now in effect are those involving:

Original Provision		Current Provision
-Tobacco Products -Firearms	Chapter 15 Chapter 25	Title 26 U.S.C. §5701 et seq. Title 26 U.S.C. §4181, 26 U.S.C. §4182, and 26 U.S.C. §5801 et seq.
-Alcohol Products	Chapter 26	Title 26 U.S.C. 5001 et seq.

All other provisions cited in the Fifth Proviso are repealed or suspended.

- (2) Nature of Restriction Generally, the purpose of the Fifth Proviso is to prohibit the circumvention of any taxes or controls on merchandise imposed by the Internal Revenue Code through activities in a zone, unless the activity was permitted before July 1, 1949. However, the rectification of distilled spirits and wines, and the manufacture of alcohol products unfit for beverage use are prohibited even if they were permitted before July 1, 1949. Since no manufacturing was permitted in zones before July 1, 1949, manufacturing of alcohol products, tobacco products, or firearms is prohibited in zones.
 - (i) Permitted Activities TTB generally has jurisdiction in zones and should be notified regarding all operations in zones involving these products, especially those involving any type of manipulation. Though manufacturing is prohibited by statute, certain manipulations may be permitted subject to TTB written approval and the Board concurrence that the activity is a manipulation and not a manufacture.

- (ii) Some examples of the activities concerning alcohol products, tobacco products, and firearms that have been determined not to be prohibited by the Fifth Proviso include:
 - (1) storage, bottling, and reduction in proof of distilled spirits and wines; mixing and blending of wines not amounting to rectification:
 - (2) mixing of two (2) alcohol products unfit for beverage purposes;
 - (3) storage, repacking, sorting, and similar manipulation of firearms (C.S.D. 84-4; November 11, 1983).
- (iii) Prohibited Activities Some examples of the activities that have been determined to be prohibited include:
 - (1) reconstitution of imported beer concentrate;
 - (2) production of industrial ethyl alcohol from propane and other substances;
 - (3) production of anhydrous ethanol; (HQ 213747 dated November 30, 1981);
 - (4) assembly or manufacture of firearms; (C.S.D. 84-4);
 - (5) blending of ethanol and gasoline to produce gasohol. (HQ 218119 dated July 23, 1985).
- (3) Denatured Distilled Spirits The provisions of the Fifth Proviso notwithstanding, any articles within the meaning of 26 U.S.C. §5002(a)(14) (denatured distilled spirits) which have been withdrawn free of tax from a TTB distilled spirits plant may be used in the manufacture or production in a zone of medicines, medicinal preparations, food products, flavors, or flavoring extracts which are unfit for beverage purposes. Similarly, denatured distilled spirits which have been withdrawn from a distilled spirits plant upon payment or determination of tax may be used for manufacturing or production in zones, and such products will be eligible for drawback under IRS laws under the same conditions applicable to similar manufacturing or production operations occurring in the Customs territory (see 19 U.S.C. §81c(c). In essence, this provision allows either domestic or foreign denatured distilled spirits to be used in zone manufacturing processes, so long as they have previously been sent to and withdrawn from a TTB distilled spirits plant. Procedures for their withdrawal from a TTB distilled

spirits plant and transfer to a zone for manufacturing purposes are found in 27 CFR 19.536 through 19 CFR 19.541 and 27 CFR Part 20.

- (4) Destruction of Alcohol Products Distilled spirits, wines, and fermented malt liquors may be brought into a zone for the purpose of destruction only with the approval of TTB (See Fourth Proviso, 19 U.S.C. §81c). However, such products which have been admitted to a zone for other purposes may, with the approval of TTB, be destroyed if they become unmerchantable. The procedure for doing this is found in 27 CFR 252.35 through 27 CFR 252.38 (See Section 8.4(a) FTZM).
- (5) Floor Stocks Taxes CBP has ruled that privileged foreign status may be granted to non tax-paid distilled spirits in a foreign-trade zone so as to "lock in" the existing tax rate in the face of an impending increase in the tax (C.S.D. 86-4; December 31, 1985).
- (e) Steel Products Grants of a few zones and subzones prohibit any manufacturing in the zone using foreign steel products except for export. Individual grants of authority should be reviewed.

Imported steel may be warehoused in a zone. Any processing of steel products must be approved by the Board. Generally the Board has not approved inverted tariff and scrap-related savings for steel products in recent years. In the 1980s the Board approved such savings for certain steel products on public interest grounds. Other operations have been approved primarily for export only; any other non-export activity has been subject to significant restrictions by the Board.

(f) Textiles and Textile Products – The warehousing of textile and apparel products is permissible in a zone.

The purpose of the regulation is to prevent a circumvention of the quota, visa, and/or export license agreements negotiated multilaterally with foreign countries. 19 CFR 146.63(d) is not an outright prohibition on manufacturing textile products in a zone, since it does not prohibit manufacturing of textile products for exportation.

However, the Board has restricted, on public interest grounds, textile manufacturing operations in some zones whether or not they are barred by 19 CFR 146.63(d), by limiting such operations to export only (e.g., FTZ Order 680 for FTZ 129) or by prohibiting any operation that changes the tariff classification or country of origin of the merchandise, whether or not the goods are subject to quota, visa, or export license requirements (e.g., FTZ Order 257 for FTZ 99A).

(g) Sugar and Sugar Products - The Board has limited a number of ongoing zone operations using sugar imported into zones outside the sugar quota by limiting the operation to a specified annual volume of production. CBP enforces this production limit

by restricting the volume of sugar-containing products that may be entered for consumption or otherwise transferred to the Customs territory to the specified amount. When this limit is reached, CBP will not accept any more entries for consumption of sugar-containing products until the beginning of the next calendar year, except upon Board order, as long as the quota is open. These limits were established under the public interest authority of 19 U.S.C. §810(c) to restrict the circumvention of the sugar quota. These were already established operations that were "grandfathered" in 1986. Since that time, the Board has denied clearances for most new manufacturing and blending operations in zones using sugar imported outside the quota. The Board has approved certain operations for export only.

A separate application on CBPF 216 is required for each sugar-containing product to be manufactured in a zone. A copy of the approved CBPF 216 shall be sent by the approving CBP personnel to the import specialist handling the product to ensure that no entry for consumption is approved for any product not approved for manufacture in the zone.

Under the Trade Act of 1990, blended syrups are exempted from the sugar import quota of zero for that product, if admitted to a foreign-trade zone, to the extent that the amount of those syrups entered into the Customs territory does not contain an amount of sugar of non-domestic origin greater than that authorized by the Board for processing in that zone during Calendar Year 1985 (GN 15(d), HTSUS (2008)).

- (h) Petroleum Products Restrictions have been placed on the grants of some zones to prohibit the blending of petroleum products which would circumvent the duty rate on gasoline, except such operations on a product destined for exportation. Also, privileged foreign status shall be elected prior to any blending or processing operations in those zones. Restrictions have been placed in the grants of subzones for oil refineries requiring entry to be made on petroleum products produced and consumed in the zone in the operation of the refinery (also see Sections 11.7(j) and 11.8 FTZM).
- (i) Alcohol/Gasohol The production and blending of alcohol/gasohol in a zone has been restricted by the Board (See CBP Ruling 217421 (Dec. 7, 1984) 216937 (July 14, 1984), 217851 (May 2, 1985), and 213747 (Nov. 30, 1981) and See 19 U.S.C. §81c; Ruling 21741 (Dec. 7, 1984); 216937 (July 14, 1984); 217851 (May 2, 1985); and 213747 (Nov. 30, 1981).
- (j) Explosives TTB has determined that explosives admitted to a zone are subject to the storage and security regulations of 27 CFR 55 even though the TTB does not consider its regulation of the flow of interstate and foreign commerce in explosives to be otherwise applicable until the explosives enter the Customs territory.
- (k) Firearms Firearms generally require TTB permits and high security storage. Any admission and other manipulation activity involving firearms must be conducted in accordance with TTB regulations and the written concurrence of the Board.

- (l) Meat Processing Processing in a zone can not be used to circumvent any meat quotas. The Board under its public interest standard, 19 U.S.C. §81o(c) has held that certain meat processing operations utilizing imported meat cannot be accommodated in zones except for export. (See Board investigation docket No. 6-76, 41 Fed. Reg. 31618 (July 29, 1976)/41 Fed. Reg. 41478 (Sept 22, 1976). For further information, contact the Board.
- (m) Pigments & Printers Ink The Board has restricted the use of certain foreign made pigments used to make printers ink.
- (n) Tires The Board has restricted certain inverted duty savings opportunities for the manufacturer of tires.
- (o) Chain Saws The Board has restricted certain inverted duty savings opportunities for the manufacturer of chainsaws.
- (p) Golf Carts The Board has restricted certain inverted duty savings opportunities for the manufacturer of golf carts.
- (q) Television Tubes The Board has in the past restrained any manufacturing operations that seek to change the HTS classification and rate of duty of foreign made television tubes used to manufacture televisions.
- (r) Pistachios/Nuts Nut processing in a zone that would change the HTS classification has not been authorized on "public interest" considerations
- (s) False Marking of Merchandise - No imported merchandise may bear a name or mark calculated to induce the public to believe that the merchandise was manufactured in the United States, or that it was manufactured in any foreign country or locality other than the one in which it was in fact manufactured (15 U.S.C. §1124). It is unlawful to import gold or silver articles bearing a mark indicating a greater degree of fineness or quality than the actual degree of fineness or quality (15 U.S.C. §294). Port Directors shall not approve any application to mark merchandise or containers in a zone in violation of these statutes, including the false marking of merchandise intended solely for exportation. If such marking occurred in a zone without a permit, or not in accordance with a permit, liquidated damages may be assessed as provided in Section 12.6 FTZM (O.C.O.D. 89-4, CBP Bulletin, Vol. 23, No. 43, October 25, 1989, p. 35). However, Port Director s may approve an application for admission and manipulation of falsely marked merchandise that is expressly intended to be brought into a zone for the purpose of obliterating and/or correcting the false marking (See Section 8.5(b) FTZM). Admission of infringing trademark goods in a zone brings them under the jurisdiction of the Lanham Act (15 U.S.C. §1127).

- (t) Zone-Restricted Status Merchandise Merchandise admitted to the zone in ZR status may only be for exportation, destruction (except destruction of distilled spirits, wines, or fermented malt liquors), for storage (19 CFR 146.44). Such merchandise may be manipulated only to the extent necessary for its exportation, destruction, or storage. (Section 81c(a) Fourth Proviso, 19 U.S.C. §81c(a); TD 89-4 and see Section 5.8 (e) FTZM).
- 11.7 Restrictions on Transfer to Customs Territory No merchandise whose entry into commerce is prohibited may be transferred to the Customs territory for consumption. A permit for transfer to the Customs territory for any purpose may be denied or withheld by CBP for failure to meet the administrative requirements of the application.
 - (a) Entry for Warehouse Merchandise in privileged foreign status or composed in part of privileged foreign status merchandise may not be entered for warehouse from a zone. Foreign merchandise in ZR status may be entered for warehouse only for storage pending exportation, unless the Board has approved another disposition. Any such entry must be endorsed as specified in 19 CFR 146.70(c). Non-privileged foreign merchandise may be entered for warehouse at any time before 5 years from the date of importation of the merchandise (19 CFR 146.64(d)). Domestic status merchandise or domestic or imported duty-paid merchandise in ZR status may not be entered for warehouse (19 U.S.C. §1557). Textile and textile products changed (as provided in 19 CFR 146.63(d) and 11.6(h) FTZM) may be entered for warehouse only if the entry is endorsed by the Port Director to show that the merchandise may not be withdrawn for consumption (19 CFR 146.64(c)).
 - (b) Entry under Temporary Importation Bond PF status merchandise or composed in part of PF status merchandise may not be entered on a TIB (19 CFR 10.31 and C.S.D. 81-213; May 5, 1981). Merchandise in zone-restricted status may be entered under a temporary importation bond if the Board has deemed it in the public interest. Such merchandise must have been in the zone for less than 1 year from the date of importation and any requirements for quota class merchandise must be met. The TIB period commences with the original date of importation. The maximum time period for a TIB is three years from the original date of importation (one year plus two one-year extensions as approved by CBP). (Note: TIB period does not begin with the date of entry from the zone). (19 CFR 10.31 and C.S.D. 79-454 and see Section 9.9(b) FTZM).
 - (c) Zone-Restricted Status Merchandise No entry that would allow domestic consumption will be accepted by CBP for merchandise in ZR status, unless the Board has approved such an entry. No entry for warehouse of merchandise in ZR status for a purpose other than storage pending exportation will be accepted by CBP, unless the Board has approved such an entry (19 CFR 146.70 and see Section 9.15 FTZM).
 - (d) Textiles and Textile Products No entry for consumption will be accepted by CBP for textiles or textile products changed in a zone (as provided in 19 CFR 146.63(d) and Section 11.6(h) FTZM).

- (e) Export-Only Restrictions When any Federal law, regulation, or directive, including a zone grant, requires exportation of merchandise from a zone, no entry for consumption will be accepted by CBP for the merchandise. An entry for transportation to another zone would be accepted. Any other kind of entry (including shipment to an insular possession or use on a U.S. flag vessel, (as noted in Sections 9.12(c) and 9.14(c) FTZM)) may be accepted provided it satisfies the nature and wording of the law, regulation, or directive requiring exportation.
- (f) Meat Subject to Quota Processing in a zone can not be used to circumvent any meat quotas. The Board has specifically authorized meat processing in foreign-trade zones for export only.
- (g) Toxic Substances When toxic substances are imported into a foreign-trade zone for manufacture into another substance, and both the imported substance and the product of the zone are subject to the Toxic Substances Control Act (15 U.S.C. §2601 et seq), EPA must be notified for clearance on both the importation of the imported substance and on the entry for consumption of the substance produced in the zone (19 CFR 12.118 through 19 CFR 12.127).
- (h) Motor Vehicles Pending EPA/National Highway Transportation Safety Administration (NHTSA) Approval An importer who has obtained a conditional release of a vehicle which was not manufactured in compliance with NHTSA safety standards must certify that the vehicle will not be sold or offered for sale until the Administrator, NHTSA, issues an approval letter to the Port Director (19 CFR 12.80(e), 12.80(b)(1)(iii)). This requirement cannot be waived by the NHTSA. A vehicle pending modification to meet EPA requirements may not be sold until it is granted final admission by the EPA; unless the importer is granted a waiver from emission control requirements (40 CFR 85.1504, 40 CFR 85.1507 and see Section 11.6(c) FTZM. Thus, while sale at retail of vehicles is prohibited under 19 U.S.C. §810(d), a sale at any level, i.e. wholesale or retail, while the nonconforming vehicle is in the zone or after its release from the zone is also restricted by EPA and NHTSA regulatory requirements. General procedures for the entry of vehicles subject to EPA and DOT/NHTSA regulations and requirements are set forth in Telex 310 dated January 27, 1990.
 - (1) Temporary Removal Motor vehicles which have been properly admitted to a zone may be temporarily removed under the procedure in Section 9.2 FTZM for testing and adjustment to bring them into compliance with EPA and DOT requirements and returned to the zone, if the transaction otherwise complies with the conditions for temporary removal (HQ 217378 B dated October 5, 1984). They may not be entered for consumption while outside the zone.
- (i) Sugar and Sugar-Containing Products A weekly entry for estimated production under 19 CFR 146.63(c) shall not be accepted by CBP for any sugar or sugar-containing

product subject to a quota, since the merchandise must be in its form as entered when a charge is made against the quota (19 CFR 132.11a(a) and 9.8(d)(2) FTZM).

- (j) Petroleum Products Petroleum products refined from crude oil in privileged foreign status may not be entered free of duty under HTS 9808.00.30 for sale to U.S. military departments when the certification required under that provision is not for the crude oil but rather for the petroleum product manufactured in the zone. (see Section 9.9(d) FTZM, 19 CFR 146, and T.D. 95-35).
- (k) Alcohol/Gasohol Any imported alcohol fuel that is denatured prior to entry into the United States must be transferred to a TTB distilled spirits plant pursuant to 26 U.S.C. §5232 to avoid paying the excise taxes on the spirits contained in the product. The alcohol fuel may not be entered directly for consumption from a zone without payment of these taxes. If determined to be classifiable as an "article" under 26 U.S.C. §5002(a)(14), it may be removed from the distilled spirits plant, subsequent to transfer from the zone, without payment of tax pursuant to 26 U.S.C. §5214(a)(11).
- 11.8 Time Limits on Grant Some grants are subject to conditions limiting their legal effect to a specific period of time. If this time limit is not extended, or the conditions removed by the Board, no merchandise may be admitted to or processed in the zone after the time limit, and all foreign or zone-restricted status merchandise in the zone shall be transferred to another zone or to a bonded warehouse, or entered for consumption, exported, or otherwise removed from the zone and disposed of according to the law, as provided in Section 13.13 and Section 4.3 FTZM.
- 11.9 Space Limits on Grants Grants can contain Board limits on physical size of activated areas. Certain grants require Board concurrence before the activated area may exceed a specific acreage. Zone grants may contain a stipulation that only a designated amount of the total acreage granted in the Board order may be activated.
- 11.10 Special Merchandise Provisions The provisions covered by this section are not prohibitions or restrictions, but rather special provisions of law or regulation dealing with certain categories of merchandise not dealt with elsewhere in this manual.
 - (a) Semi-conductors Semi-conductors may be transferred to the zone for temporary deposit in commingled standard, substandard, and defective units. While in the zone the standard, substandard, and defective units may be segregated. The standard units and some of the substandard units may be entered for consumption at the option of the importer. The defective and the remainder of the substandard units may be admitted to a zone in zone-restricted status and destroyed in the zone. The appraised value of the units entered for consumption does not include the cost of producing the defective and substandard units destroyed in the zone. If the destruction process does not result in complete destruction because a valuable waste containing precious metals remains, the Board may allow the transfer of the remaining zone-restricted status waste to the Customs territory for consumption where the precious metals are salvaged from the

waste. The Board has in the past issued the order because salvaging the precious metal in the United States, rather than requiring exportation, was found to be in the public interest.

(b) Bulk Concentrated Orange Juice - Bulk concentrated orange juice admitted to a zone that, for a variety of reasons, causes variances and apparent discrepancies in the volume of juice in the storage tanks may receive special allowances. CBP may authorize an allowance for relief from the Operator's liability for these apparent discrepancies, under the authority of 19 CFR 146.53(c), in circumstances where the juice cannot be weighed both upon receipt into and removal from the tank.

CHAPTER 12

BONDS AND LIQUIDATED DAMAGES

12.1 General - In any case in which a bond or other security is not specifically required by law, the Secretary of the Treasury and the Secretary of Homeland Security may by regulation or specific instruction require, or authorize CBP personnel to require such bonds or other security as may be deemed necessary for the protection of the revenue or to assure compliance with any pertinent law, regulation, or instruction (19 U.S.C. §66, 19 CFR 0.1 and 0.2. A surety bond is a contract whereby one party, the surety, guarantees the performance of a second party, the principal, for the benefit of a third party, the obligee (the Federal government, in the case of CBP bonds). Should the principal fail to perform his agreement with the obligee, the surety will be required to pay liquidated damages and will have the right to obtain reimbursement from the defaulting principal. The Secretary of the Treasury is authorized to permit the deposit of money or obligations of the United States in lieu of surety on any bond required or authorized by law, regulation, or instruction of the Secretary of the Treasury or CBP (19U.S.C. §1623(e)).

The general regulations concerning bond operations are found in 19 CFR 113. Through the general rulemaking authority of 10 U.S.C. §81h, Operator's are required, under 19 CFR 146.6(d), to execute an Operator's bond on CBPF 301, (Activity Code 4), containing the bond conditions set forth in 19 CFR 113.73. A general-purpose zone may be structured in such a manner that more than one Operator authorized by the Grantee may be permitted to conduct business under the same zone grant. The principal bonds covered in this part of the manual are the importation and entry bond, as it relates to Users, and the Operator's bond. On March 1, 2010, CBP centralized the continuous bond program into the Revenue Division, Office of Administration, CBP. Pursuant to this centralization, continuous bonds must be filed at the Revenue Division, via mail, fax, or in an electronic format, and the Revenue Division will ass the bond functions previously performed at the port level. The authority to approve single transaction bonds remains with the Port Director.

Liquidated damages are amounts of money which the principal agrees to pay to compensate for damages, often difficult to determine, to the obligee arising from a default by the principal in his performance under the bond. As defined in the regulations in 19 CFR 146.1 (b)(8), "default" means an action or omission that will result in a claim for duties, taxes, charges, or liquidated damages under the Operator Bond. The purpose of a liquidated damages provision in a bond is to avoid the necessity for litigation to determine actual damages for each and every bond default. If the principal fails to pay liquidated damages provided for in the bond, the surety is obligated to pay the liquidated damages. Generally, the principal and surety are jointly and severally liable under a CBP bond.

12.2 Bond Administration – The face amount of the bond generally forms the obligor's limit of liability (C.S.D. 86-17). Determination of the face amount is at the discretion of the Port Director, or the Director, Revenue Division, considering the guidelines specified in 19 CFR 113.13(b) and CD3510-004. Also, Port Directors are directed to periodically review each bond filed in their port to determine whether they are adequate to protect the revenue and insure

compliance with the law and regulations (19 CFR 113.13(c)). In the case of the Operator's bonds, this review shall be done at least once each year. The initial standard should be based upon CBP duties and fees owed on the average value of foreign status non-duty paid merchandise held in the zone. Furthermore, Port Directors require additional security when they believe acceptance of a transaction secured by a continuous bond would place the revenue in jeopardy (19 CFR 113.13(d)). The authority to CBP for requiring bonds and setting their amount is strongly stated and comprehensive in 19 U.S.C. §623. The right of a party to keep doing business at the same bond premium cost is not paramount to the Government's right to protect the revenue (Hera Shipping, Inc. v. Carnes, 10 CIT 493, 640 F. Supp. 266 (1986)).

- 12.3 Agreements of User in Importation and Entry Bond Users share with other importers for consumption the agreements of the importation and entry bond under 19 CFR 113.62. The principal agreements directly affecting zone Users in particular are covered below:
 - (a) Agreement to Pay Duties, Taxes, and Charges If merchandise is imported and released from CBP custody for consumption in the United States, the principal and surety agree:
 - (1) to deposit within the prescribed time, any duties, taxes, or charges imposed or estimated to be due at the time of entry;
 - (2) to pay, as demanded by CBP, all additional duties, taxes, and charges subsequently found to be due on any entry secured by the bond (19 CFR 113.62(a)(1)).
 - (b) Reimbursement and Exoneration of the United States Principal and surety agree to exonerate the U.S. from any risk, loss, or expense arising out of the principal's importation, entry, or withdrawal of the merchandise (19 CFR 113.62(g)(2)).
 - (c) Agreement on Duty-Free Entries or Withdrawals If the principal enters or withdraws any merchandise without payment of duty and tax, or at a reduced rate of duty and tax, as permitted by law, the principal agrees:
 - (1) To use and handle the merchandise in the manner and for the purpose entitling it to duty-free treatment;
 - (2) If a fishing vessel, to present the original approved application to CBP with 24 hours on each arrival in the Customs territory from a fishing voyage;
 - (3) To furnish timely proof to CBP that any merchandise withdrawn under any law permitting duty-free treatment was used in accordance with that law; and
 - (4) To keep safely all withdrawn beverages remaining on board while the vessel is in port, as may be required by CBP (19 CFR 113.62(h)).

- (d) Consequences of Default Liquidated damages are not assessed for failure to comply with the agreements in 19 CFR 113.62(a) or 19 CFR 113.62(g), since the actual damages are known and easy to determine. The principal and surety are obliged only to pay the applicable duty, tax, charge, or other cost itself. Liquidated damages are assessed, however, for a default in the agreement in 19 CFR 113.62(h), (19 CFR 113.62(m)).
- 12.4 Agreements of Operator in Operator's bond Under the terms of this bond, the Operator, as principal, and surety agree to the following conditions:
 - (a) Agreement Concerning Receipt Handling and Disposition of Merchandise The principal and surety agree to comply with:
 - (1) The law and CBP regulations relating to the receipt, admission, status, handling, transfer, and removal of merchandise from the zone or subzone; and
 - (2) CBP regulations concerning the maintenance of the inventory control and recordkeeping system covering merchandise in the zone or subzone (19 CFR 113.73(a)).
 - (b) Agreement to Pay Duties, Taxes, and Charges The principal and surety agree to pay any duties, taxes, and charges found to be due on any merchandise, properly admitted to the zone, which is found to be missing from the zone or cannot be accounted for in the zone (19 CFR 113.73(b)).
 - (c) Reimbursement and Exoneration of the United States The principal and surety agree to exonerate the U.S. and its personnel from any risk, loss, or expense arising from the principal's operation of the foreign-trade zone or subzone (19 CFR 113.73(d)(1)).
 - (d) Consequences of Default Liquidated damages are not assessed for failure to comply with the agreements in 19 CFR 113.73(b) or (c), since the actual damages are known and easy to determine. The principal and surety are obliged only to pay the applicable expense, charge, or cost itself. However, liquidated damages are assessed for a default in the agreement in 19 CFR 113.73(a) under the Operator's bond.

12.5 Consequences of Default

(a) Consequences of Merchandise Default - The principal and surety on both the importation and entry bond (19 CFR 113.62(h)) and the Operator 's bond (19 CFR 113.73(a)(2) agree to pay, in the event of default by the principal in an agreement, liquidated damages equal to the value of the merchandise involved in the default or three times the value of the merchandise if the merchandise is restricted merchandise or alcoholic beverages, or such other amount as may be authorized by law or regulation.

- (b) Consequences of Non-merchandise Default If the principal defaults in an agreement and the default does not involve merchandise, the principal and surety agree to pay liquidated damages of \$1,000 for each default or such other amount as may be authorized by law or regulation (19 CFR 113.62(m)(3)).
- (c) CBP Determination of Nature of Default Principal and surety agree that whether or not a default involves merchandise is determined by CBP. Furthermore, the amount to be collected for defaults involving merchandise is based on the quantity and value of the merchandise as determined by CBP (19 CFR 113.62(m)(2), 19 CFR 113.62(m)(3) and 19 CFR 113.73(a)(2)).

The presence of domestic or duty-paid articles in the zone without CBP permit, such as personal items, equipment, construction material, and other articles which are not stock in trade, is not a default involving merchandise. However, the presence of such articles could be a non-merchandise default if it constitutes a violation of the regulations (e.g., if aisles are not properly maintained as required under 19 CFR 146.4(f)).

- 12.6 What Constitutes Merchandise Default Merchandise involved in the default (19 CFR 113.73(a)(2)), as it applies to the Operator's bond, can be merchandise which:
 - (a) Cannot be located or accounted for in the activated area of a zone;
 - (b) Has been removed from the activated area of a zone without a proper CBP permit; or
 - (c) Has been admitted, manipulated, manufactured, exhibited, or destroyed in the activated area of a zone:
 - (1) Without proper CBP permit; or
 - (2) Not in accordance with the description of the activity in the CBP permit.
- 12.7 What Constitutes Non-merchandise Default A non-merchandise default is any default which does not meet the definition of a merchandise default. Generally, it is a default in the regulations concerning recordkeeping or merchandise handling, where there is no evidence of a merchandise default in connection with the same event or transaction.
 - (a) Application and Examples of Non-merchandise Defaults The same act shall not be regarded as both a default involving merchandise and a default not involving merchandise.
 - (1) Defaults not involving merchandise are to be treated as distinct from one another and shall not be grouped into one default. (Example: If merchandise is covered by 2 lots or 2 unique identifiers and is stored in the same unsanitary manner, there are 2 defaults 19 CFR 146.4(f)).

- (2) A default involving one zone lot or unique identifier may not be combined with a default under another lot or unique identifier. (Example: If merchandise covered by 2 applications for admission on two CBPF 214s and neither are recorded in the receiving report, there are 2 defaults (19 CFR 146.22(a))
- (3) Where there is a default in the timeliness of an action, each business day of default constitutes a separate default. (Example: If in-bond documents are not forwarded by a subzone under the direct delivery procedure to the Port Director until 4 working days after the date of arrival of the conveyance, there are 2 defaults (2 days late) (19 CFR 146.40(a)(5)).
- (4) When one irregularity leads to one or more other irregularities, only the original one constitutes a default. (Examples: If 1 keying error in recording merchandise into the recordkeeping system causes an incorrect current balance and an incorrect reconciliation statement at the business year end, there is only 1 default (19 CFR 146.23(b)(4) and 146.25(b)).
- If 1 clerical error causes an in-bond document to be forwarded under the direct delivery procedure to the Port Director 4 days late, there is only 1 default (19 CFR 146.40(a)(5)).
- 12.8 What Constitutes Restricted Merchandise Restricted merchandise, for the purpose of determining liquidated damages for defaults involving merchandise, means merchandise that may not be authorized for delivery from CBP custody without a special permit, or a waiver thereof, by an agency of the U.S. Government; or merchandise that does not meet the restrictions specified by law or regulation (A.N. Deringer, Inc., v. United States, 84 Cust. Ct. 196, C.D. 4858 (1980); see, e.g., 19 CFR Parts 11 and 12, and 19 CFR Part 133, Subpart C). It includes all quota merchandise, whether the quota is administered by CBP or another agency. Such merchandise is prohibited from entry for consumption in the United States in the absence of the special permit, and actual damages to the government resulting from its diversion into the commerce without a permit are greater than the mere nonpayment of duties and taxes on the merchandise.
- 12.9 Value of Merchandise The value of merchandise used in determining the amount of liquidated damages for defaults involving merchandise shall be the dutiable value as provided for in 19 CFR 146.65(b)(2), excluding the value of merchandise or component materials in domestic status (unless a CBP permit is required) and any allowance granted under 19 CFR 146.65(b)(3).

- 12.10 Payment of Duties by Operator on Missing Merchandise Under 19 CFR 113.73(b), the Operator agrees to pay duties, taxes, and charges found to be due on any merchandise which is found to be missing or cannot be accounted for in a zone. Port Directors may accept entries for consumption from Operator s to pay the duties and taxes on such merchandise, and the Operator will thereby be treated as the owner or purchaser of the merchandise for entry acceptance purposes. However, the actual owner or purchaser may, at its option, file entry for such merchandise and thereby relieve the Operator of the responsibility to pay duties and taxes on the merchandise. If merchandise is removed from a zone without CBP permit, but returned to the zone before the duties and taxes are paid thereon, the Port Director may waive a demand to comply with the agreement to pay the duties and taxes.
- 12.11 Issuance of Claims of Liquidated Damages Not every default found by CBP results in a claim for liquidated damages. CBP personnel will not issue such claims without a consideration as to whether (a) CBP has suffered actual or potential damages; (b) compliance with the laws and regulations can be achieved without liquidated damages; (c) liquidated damages can and will bring about compliance; and (d) the Operator has cooperated with CBP in disclosing defaults and promptly corrected them.

Many minor defaults will be handled by CBP through oral or written warnings. (See Section 13.2 FTZM). If CBP decides to issue a claim, it will be issued on CBPF 5955A, Notice of Penalty or Liquidated Damages Incurred and Demand for Payment.

- (a) Liquidated Damage Claims Note: 19 CFR 171 and 19 CFR 172 include petition processing in seizures and unsecured penalty cases under 19 CFR 171 and liquidated damages and secured penalty petition processing under 19 CFR 172. General procedures for these claims are set forth in 19 CFR 172, and the Fines, Penalties and Forfeitures (FP&F) Handbook, HB 4400-01. When there is a failure to meet the conditions of a bond posted with CBP, or when a violation occurs which results in an assessment of a penalty secured by a bond, the principal will be notified in writing of any liability for liquidated damages or penalty incurred, and a demand will be made for payment. Sureties on such bond will also be notified in writing of any such liability at the same time (19 CFR 172.1(a)). If the principal fails to pay or make arrangements to pay the liquidated damages, or fails to timely file a petition for relief, CBP will make a demand for payment on the surety (19 CFR 172.4).
- (b) Petitions for Relief A petition for relief must be filed with the FP&F within 60 days from the date of mailing to the bond principal the notice of claim for liquidated damages or penalty secured by a bond (19 CFR 172.3(b)). The surety has 60 days from the date of the demand upon it, to file a petition for relief (19 CFR 172.4). If the statute of limitations will run in less than 180 days from the date of liquidated damages or penalty notice, a shorter time period for the filing of a petition for relief may be specified. If a petition is not filed within the time specified, the matter will be transmitted promptly to the appropriate Office of the Chief Counsel for referral to the Department of Justice (19 CFR 172.3(e)). The petition need not be in any particular form, but shall set forth the facts relied on by the petitioner to justify cancellation of the claim for liquidated damages

without any payment or upon payment of a lesser amount than stated in the notice of liquidated damages. The petition must set forth the date and place of the violation, and the facts and circumstances relied on by the petitioner to justify the cancellation, remission or mitigation of the claim for liquidated damages (19 CFR 172.2(c)). The petition must be signed by an authorized person as specified in 19 CFR 172.2(b).

- (c) Where Filed In accordance with the appropriate delegation of authority, the petition shall be filed with the appropriate FP&F office. FP&F will either act upon, or refer the petition to the Office of International Trade (19 CFR 172.1 and 19 CFR 172.12).
- (d) Supplemental Petitions for Relief If the petitioner is not satisfied with the decision of FP&F, a supplemental petition may be filed under the guidelines of 19 CFR Part 172, Subpart E.
- 12.12 Waiver or Cancellation of Claim Without Payment FP&F may, consistent with delegated authority, either waive issuance of a claim for liquidated damages, or if such a claim has already been issued, may cancel the claim without payment (19 CFR 172.11 (a)).
 - (a) Mitigation or Cancellation FP&F is empowered to cancel any case without payment of the mitigated or cancellation amount it is definitely determined that the act or omission forming the basis of any claim of penalty or claim for liquidated damages did not occur (19 CFR 172.11 (b)). Otherwise, FP&F is empowered to mitigate or cancel any claim in accordance with appropriate delegations of authority (19 CFR 172.11 (a)).
 - (b) Mandatory Cancellation FP&F shall waive or cancel any claim for liquidated damages if it is determined that the claimed default did not actually occur (19 CFR 172.11(b)).
 - (c) Discretionary Cancellation for Merchandise Default FP&F may waive the imposition of a claim for liquidated damages for defaults in the Operator 's bond involving merchandise, or if a claim has already been made, may cancel the claim without payment, upon showing of proper evidence that:
 - (1) The default was reported by the Operator or by the User, or the Operator or User has a demonstrated history of reporting defaults (both defaults involving merchandise and those not involving merchandise) to CBP; and,
 - (2) The default does not involve restricted merchandise; and,
 - (3) There was no negligent, deliberate, or fraudulent act or omission by the Operator in connection with the default.
 - (d) Discretionary Waiver for Minor Losses In the case of any shortage which does not meet the 1 percent and \$100 criteria as provided in 19 CFR 146.53(a)(3), a claim for

liquidated damages may be waived or canceled by FP&F upon the showing of proper evidence that:

- (1) The shortage was recorded in the Operator's inventory control and recordkeeping system (19 CFR 146.21 and 146.23); and,
- (2) The shortage does not involve restricted merchandise; and,
- (3) There was no negligent, deliberate, or fraudulent act or omission by the Operator, or its personnel, employees, or agents, in connection with the default.
- (e) Discretionary Cancellation for Non-merchandise Default FP&F may waive the imposition of a claim for liquidated damages for defaults not involving merchandise or if a claim has already been made, may cancel the claim without payment, upon the showing of proper evidence that:
 - (1) The regulatory provision is subject to interpretation and the default was inconsequential or did not affect CBP ability to accomplish the underlying objective of the provision.

Examples:

An aisle was narrow or partially or temporarily blocked but the Officer was not materially delayed or impeded in conducting a compliance review (19 CFR 146.4(f)).

Security was deficient in some CBP security specifications but was compensated for by other security features, and overall security was adequate (19 CFR 146.4(e)).

If there is no room for interpretation in the regulation, a claim for liquidated damages will be issued. A regulatory provision requiring action by a particular time or date, or on a particular CBP form, is considered to leave no room for interpretation.

(2) The Operator could not comply solely because of circumstances outside the control of the Operator or User.

Example:

The bonded carrier did not timely furnish in-bond documentation required to be submitted by an Operator within 2 business days under direct delivery procedures (19 CFR 146.40(a)(5)).

- (3) The default was voluntarily disclosed by the Operator or User, or the Operator or User has a demonstrated history of voluntarily disclosing defaults to CBP; and the Operator or User has a demonstrated history of promptly correcting defaults; and the default was the result of a clerical error or other inadvertence.
- (f) Discretionary Cancellation (Entry Bond) FP&F may waive the imposition of a claim for liquidated damages for defaults against the importation and entry bond of the zone User, or, if a claim has already been made, may cancel the claim without payment, upon the showing of proper evidence that the default resulted from a clerical error or other non-negligent, inadvertent mistake and that any duties and taxes due on the merchandise have been paid.
- (g) Effect of Discretionary Waiver or Cancellation A decision by FP&F not to issue a claim for liquidated damages does not mean that the default did not exist, nor that it does not need correction. FP&F shall bring the default to the attention of the zone Operator through a warning notice instead of a notice of claim of liquidated damages. If there is an apparent misunderstanding of regulatory requirements, FP&F will hold a meeting with the Operator to explain the requirements. If the default is not promptly corrected or if the same default is repeated, a claim for liquidated damages shall be issued. If FP&F issues such a claim because the Operator or User has failed to take corrective action, the Operator or User faces a substantial burden in justifying relief from the damages on the grounds of clerical error or other inadvertence.
- 12.13 Prior Disclosure/Voluntary Notification If a violator comes forward and discloses a violation to CBP prior to discovery of the violation by CBP, the claim for liquidated damages and penalties may be canceled upon payment of an amount equal to any duties, fees, taxes and charges due, plus \$50.
- 12.14 Cancellation upon Payment of Smaller Amount (Operators bond) If the Port Director or other reviewer does not believe a discretionary waiver or cancellation is warranted in the case of a default in the Operators bond, the petition will be treated according to the guidelines in this Section.
 - (a) Defaults Involving Merchandise Petitions for relief from liquidated damages for defaults involving merchandise will be processed according to the following guidelines:
 - (1) If the breach resulted from a clerical error or mistake (a non-negligent, inadvertent error), and the situation does not qualify for a waiver or cancellation, the obligation may be canceled upon payment of:
 - (i) 1 percent of the value of merchandise wholly in domestic status for which a CBP permit is required, not to exceed \$500 per claim; or
 - (ii) 10 percent of the value of restricted merchandise, with no maximum amount per claim; or

- (iii) three percent of the value of other merchandise involved in the default, not to exceed \$1,000 per claim.
- (2) If the breach was the result of negligence (an act which is more than a clerical error or mistake but not committed deliberately with the knowledge it would constitute a default in the bond), the obligation may be canceled upon payment of:
 - (i) not less than two percent nor more than five percent of the value of domestic status merchandise for which a CBP permit is required, not to exceed \$1,500 per claim; or
 - (ii) Not less than 40 percent nor more than 100 percent of the value of merchandise in the case of restricted merchandise; or
 - (iii) Not less than 20 percent nor more than 35 percent of the value in the case of other merchandise.

The sum to be collected shall be determined by the presence or absence of aggravating or extenuating factors, such as the following not all-inclusive factors:

- (i) Aggravating factors:
 - (1) Principal's failure or refusal to cooperate with CBP;
 - (2) Large number of violations in relation to number of zone admissions:
 - (3) Principal's long experience in handling zone transactions;
 - (4) Principal's carelessness or willful disregard toward its responsibility, including defaults which did not involve merchandise;
 - (5) Merchandise released without permit has not been returned to the zone or CBP custody.
- (ii) Extenuating factors
 - (1) Contributory error by CBP or by a party independent of the Operator;
 - (2) Small number of violations in relation to total number of zone transactions:

- (3) Remedial action taken by principal;
- (4) Principal's lack of experience in handling zone transactions;
- (5) Principal's cooperation with CBP;
- (6) Return of merchandise to zone or CBP custody.
- (3) If the breach resulted from an act that was intentional (an act which was done deliberately with the knowledge that it would breach the bond), there will be no relief from the damages.
- (b) Defaults Not Involving Merchandise Petitions for relief from liquidated damages for defaults that do not involve merchandise will be processed according to the following guidelines:
 - (1) If the default resulted from a clerical error and the situation does not qualify for a waiver or cancellation, the obligation may be canceled upon payment of \$50 per default.
 - (2) If the default resulted from negligence, the obligation may be canceled upon the payment of not less than \$100 nor more than \$250 per default. The sum to be collected will be determined by the presence or absence of the not all-inclusive aggravating and extenuating factors (See Section 12.14(a)(2) FTZM).
 - (3) If the default resulted from an act that was intentional, there will be no relief from liquidated damages.
- 12.15 Cancellation upon Payment of Smaller Amount (Entry Bond) If the Port Director or other reviewer does not believe a waiver or cancellation without payment is warranted under the importation and entry bond, the petition will be treated according to the guidelines in this Section
 - (a) Defaults Involving Merchandise Petitions for relief from liquidated damages for defaults involving merchandise will be processed according to the following guidelines:
 - (1) If the default was the result of negligence by the principal (an act which is more than a clerical error but not committed with knowledge that it would constitute a default in the bond), the obligation may be canceled upon payment of the duties that would have been due on the merchandise involved in the default, plus an amount between \$100 and \$500 depending on the presence of aggravating and extenuating factors such as the following not all-inclusive factors:
 - (i) Aggravating Factors:

- (1) Violator refuses to cooperate with CBP or acts to impede CBP action on the case;
- (2) Violator experienced in handling zone shipments of the type in question;
- (3) Large number of violations in relation to total number of the same type of zone shipments handled by the violator, i.e., a frequent violator; and
- (4) Default involves restricted merchandise.

(ii) Mitigating Factors:

- (1) Violator inexperienced in handling zone shipments of the type in question;
- (2) Small number of violations in relation to the total number of the same type of zone shipments handled by the violator, i.e., violations are infrequent;
- (3) Violator demonstrates remedial action to prevent future violations has been taken;
- (4) Evidence of contributory CBP error; and
- (5) Circumstances intervened that were beyond the principal's control. NOTE: Human failure by the principal or any agent of the principal will not be considered as a circumstance beyond the principal's control.

If the principal voluntarily files a duty-paid entry for the merchandise or returns the merchandise to CBP custody before the issuance of the claim, the claim may be canceled upon payment of between \$100 and \$500 depending on the presence of aggravating or extenuating factors.

- (2) If the default resulted from an act that was intentional (an act which was done deliberately with the knowledge it would breach the bond), there will be no relief from the damages.
- (b) Defaults Not Involving Merchandise Petitions for relief for defaults not involving merchandise will be treated according to the following guidelines:

- (1) If the default resulted from negligence, the obligation may be canceled upon payment of not less than \$100 but not more than \$500 per default. The sum to be collected will be determined by the presence of aggravating and extenuating factors (See section 12.15(a)(1) FTZM).
- (2) If the default resulted from an act that was intentional (an act which was done deliberately with the knowledge it would breach the bond), there will be no relief from liquidated damages.
- 12.16 Offers in Compromise An offer in compromise (OIC) is made by the violator to settle a claim at less than the amount demanded by the government. The OIC may be made at any point prior to initiation of a case or during the petition and collection processes. An OIC is not valid unless monies are tendered. Subject to the recommendation of the Chief Counsel, FP&F may accept an OIC except in the case of a show cause letter being issued by designated CBP Headquarters personnel. All such actions shall be in accordance with 19 CFR 172 Subpart D.
- 12.17 Suspension of Activation for Nonpayment of Claim The Port Director may take action to suspend for cause the activated status of a zone as set forth in 19 CFR 146.82 and Section 13.8-13.11 FTZM, if the Operator has refused or neglected to pay a claim for liquidated damages upon proper order of CBP. This is in addition to any action that may be taken by a court pursuant to a suit initiated by the U.S. Attorney to enforce collection of liquidated damages.

CHAPTER 13

FOREIGN-TRADE ZONE ENFORCEMENT ACTIONS

- 13.1 General There are ample means at CBP disposal to insure compliance with the laws and regulations it is responsible for administering. CBP will work with the trade through the informed compliance process to ensure that all laws and regulations dealing with foreign-trade zones are complied with prior to pursuing the enforced compliance approach. Enforcement measures range from simple warning notices to criminal sanctions with prison sentences. At the same time, CBP recognizes that zones provide an important service to the importing community and are deemed to operate to serve the convenience of commerce. CBP supervision and sanctions should therefore be fair and not unduly burdensome to Grantees, Operators, and Users in the lawful exercise of their trade, as well as effective in gaining compliance with the laws and regulations. Actions taken by Port Directors to correct violations of the laws and regulations should be commensurate with the seriousness of the violation. A heavier sanction should not be applied if a lighter one will achieve the purpose of gaining compliance. The enforcement actions in this Part are listed, generally, in increasing order of severity so as to deal with increasingly serious violations.
- 13.2 Warning Notices It is CBP policy that when there is reason to believe minor non-merchandise defaults can and will be promptly corrected upon simple notification of the Operator, the Port Director may issue an oral or written warning notice or hold an explanatory meeting with the Operator. Records shall be maintained by CBP of such notices and meetings, and they shall be followed up to see that the Operator has corrected the default. (See section 3.5 FTZM). If the Operator continues to be in noncompliance, one of the other actions in this Part shall be taken.
- 13.3 Assessment of Liquidated Damages If a warning notice is not appropriate and a breach of the conditions of the Operator's bond has occurred, CBPF 5955A, Notice of Penalty or Liquidated Damages Incurred and Demand for Payment, will usually be issued as described in Part 12 FTZM. However, if the Port Director has reason to believe the Operator is unable or unwilling to comply with the laws and regulations, other applicable provisions of this Part shall be followed in addition to the issuance of CBPF 5955A (19 CFR 146.81(a)).
- 13.4 Fines In the case of a violation of the FTZ Act or any regulation under the FTZ Act by the Grantee or by any officer, agent, or employee thereof, the person responsible for or permitting any such violation, shall be subject to a fine of not more than \$1,000. Each day during which a violation continues shall constitute a separate offense (19 U.S.C. §81s, 19 CFR 146.81(a)). Liquidated damages, where applicable, will be imposed in addition to the fine (19 CFR 146.81(a)). Fines assessed by the Port Director under this section shall be reviewed by the Assistant Commissioner, Office of International Trade, or his designee, Headquarters. Office of International Trade to determine whether further action against the Grantee or Operator, such as suspension of activation of the zone or recommendation for revocation of the zone grant, is warranted (19 CFR 146.81(b)). In addition, the Board has authority to impose fines for violations of the FTZ Act (15 CFR 400.11(a)(10)).

- (a) Notice of Fine Notice of such a fine will be issued on CBPF 5955A. The fine is subject to relief under the provisions of 19 CFR 171 and the FP&F Handbook, HB 4400-01. A copy of each such notice shall be forwarded to the Office of International Trade for review per 19 CFR 146.81(b).
- (b) When Fine Assessed The fine will ordinarily be assessed in situations where the violator is not the principal on a CBP bond, the violation does not result in damages to the Government, or where, for some other reason, liquidated damages for the violation would be inappropriate.

Examples:

- (1) The violation was committed by a Grantee which is not the zone Operator.
- (2) The violation was committed by an officer or employee of the Grantee or Operator, acting alone.
- (3) The violation was a retail trade violation (19 CFR 146.14 and see section 11.6(b) FTZM) and there was no apparent damage to the Government.

However, the Port Director has the discretion to assess liquidated damages against the Operator in addition to the fine, as specified in 19 CFR 146.81

- 13.5 Revocation of Blanket Permit If a blanket permit for admission (19 CFR 146.32(d)), blanket permit to manipulate or manufacture (19 CFR 146.52), or any other kind of blanket permit, including approval of direct delivery procedures under 19 CFR 146.39(d), is not deemed effective in protecting the revenue or enforcing any U.S. law or regulation, the Port Director may revoke the permit and demand instead that the interested party present individual applications for permits.
- 13.6 Physical Supervision If the Port Director has reason to believe there is an unacceptable risk that merchandise will be removed from a zone without proper permit, or that special security is demanded in connection with a CBP investigation, the zone, or a facility within the zone, may be locked with a CBP lock and physical supervision conducted by CBP of all admissions, transfers, manipulations, destruction, manufacturing, processing or other transactions requiring a permit in a zone. Because of CBP staffing limits, physical supervision is generally regarded as a temporary measure. Such supervision shall be terminated upon reestablishment of proper control by the Operator; when the presence of other factors indicates that the need for special security has ended; or upon suspension or revocation as specified in 19 CFR 146.82 and Sections 13.8 through 13.11 FTZM.
- 13.7 New Bond Requirement If a zone violation(s) or potential violation(s) is deemed a threat to the revenue or proper law enforcement, the Port Director may require an increased bond amount, as provided in 19 CFR 113.13(c) or (d). If an Operator does not furnish a new bond

within 10 days of written demand by the Port Director, no more goods will be received in the zone in zone status and those therein will be duty paid or removed at the expense of the Operator

- (19 CFR 146.7(d) and section 4.9 FTZM). Alternatively, the Port Director may allow a 30-day period to remedy the deficiency, as provided in 19 CFR 113.13(c). There is no provision in the CR for an administrative appeal against the demand under these provisions for a new bond.
- 13.8 Suspension Procedure The Port Director may suspend for cause the activated status of a zone or zone site, or the privilege to admit, manufacture, process, manipulate, exhibit, destroy, transfer or remove merchandise at a zone or zone site for a period not to exceed 90 days. Upon order of the Board, the suspension may be continued. If appropriate, the suspension may be limited to an individual User or Users and not to the zone or zone site as a whole, or may be limited to a particular activity of an Operator or User, such as suspension of the privilege to admit merchandise or the privilege to manufacture (19 CFR 146.82(a)).
 - (a) Grounds for Suspension An action to suspend will be taken in accordance with the procedure in 19 CFR 146.82(b) if:
 - (1) The approval of the application to activate the zone was obtained through fraud or the misstatement of a material fact:
 - (2) The Operator neglects or refuses to obey any proper order of CBP Officer or any CBP order, rule, or regulation relating to the operation or administration of a zone:
 - (3) The Operator, or any employee of a corporation which has been granted the right to operate a zone, is convicted of or has committed an act that would constitute a felony, or would constitute a misdemeanor involving theft, smuggling or a theft related crime;
 - (4) The Operator fails to furnish a current list of names, addresses, or other information as required by 19 CFR 146.7;
 - (5) The Operator does not provide a secure facility or properly safeguard merchandise within a zone;
 - (6) The Operator, or any personnel, agent, or employee of the Operator, discloses to an unauthorized person proprietary information contained on a CBP form or in the inventory control and recordkeeping system; or
 - (7) The inventory control and recordkeeping system is impaired to the point where the identity of merchandise in zone status has been lost and cannot be reestablished without a suspension of zone operations (19 CFR 146.82(a)).

The grounds for suspension are limited to those specified in 19 CFR 146.82(a). If there is any other reason for which suspension is sought by CBP, action for cause may be taken only indirectly. For example, the Operator could be instructed in writing to discontinue a practice which interferes with proper CBP administration of a law or regulation. If the practice continues, grounds may exist for suspension because of the Operator's refusal or neglect to obey a proper order of CBP personnel.

- (b) Show Cause Notice to Operator The Port Director may, at any time, serve notice in writing upon an Operator to show cause why its right to continue operation of a zone should not be suspended or why an individual User or individual activity should not be suspended, under the procedure provided for in 19 CFR 146.82(b). The notice shall advise the Operator of the grounds for the proposed action and will afford the Operator an opportunity to respond, in writing, within 15 days after receipt of the notice. Thereafter, the Port Director shall consider the allegations and any response made by the Operator and issue a decision unless the Operator requests a hearing in the matter (19 CFR 146.82(b)(1)). The notice to show cause why the suspension should not be carried out shall contain:
 - (1) The specific grounds for suspension;
 - (2) The specific facts which serve as the basis for the grounds; and
 - (3) The specific period of suspension.
- (c) Notice to Grantee If the Grantee of the zone is not the Operator, a copy of the notice to show cause shall be served upon the Grantee. The Grantee, as a party in interest, may join the Operator in any proceedings under this Section (19 CFR 146.82(b)(4)). A copy of the notice need not be sent to a User, and there is no authority for a User to join the Operator in any suspension proceedings (TD 86-16, 51 FR 5040).
- (d) Hearing If the Operator requests a hearing, it shall be held before a hearing officer designated by the Commissioner within 30 days following the Operator's request. The Operator may be represented by counsel at the hearing, and any evidence and testimony of witnesses in the proceeding, including substantiation of the allegations and the response thereto, shall be presented. The right of cross-examination shall be available to both parties. A stenographic record of the proceeding shall be made and a copy will be delivered to the Operator. At the conclusion of the hearing, the hearing officer shall transmit promptly all papers and the stenographic record of the hearing to the Assistant Commissioner, Office of Field Operations, together with a recommendation for final action (19 CFR 146.82(b)(2)).
- (e) Decision After Hearing Within 10 calendar days after delivery to the Operator of a copy of the stenographic record of the hearing, the Operator may submit in writing to the Assistant Commissioner, Office of Field Operations, any additional views or

arguments. The Assistant Commissioner or designee shall then render a written decision stating his reasons therefor. The decision shall be served on the Operator (and the Grantee, if applicable) and shall be considered the final CBP administrative action in the case (19 CFR 146.82(b)(3)).

- (f) Placing Suspension in Effect The suspension shall go into effect when the Port Director issues a decision in the matter, unless the Operator requests a hearing. If the Operator requests a hearing, the suspension shall be placed in effect when the Assistant Commissioner, Office of Field Operations, issues a decision, unless suspension is stayed by a court order.
- (g) Extension of Suspension by FTZ Board Suspensions shall be for a specified period of time up to 90 days. Any suspension beyond 90 days, or any revocation of activation, may be carried out only upon order of the Board. The 90-day limit is based on a presumption that continued operation of the zone serves the convenience of commerce, and that only the Board can affirm a conclusion that a longer suspension, or revocation of activation, is in the public interest.

If the Port Director believes the grounds for the suspension cannot or will not be corrected within 90 days from the beginning of the suspension, he or she should seek an extension by sending a written request to the Board, to exclude the affected goods or operation from the zone under the authority of 19 U.S.C. §81o(c), for a specified period of time recommended by the Port Director. This request may be made without any further administrative hearing, but shall be sent only through the Assistant Commissioner, Office of Field Operations. Such a request should be made as soon as the Port Director is aware of the need for an extension, to allow the Board order to be issued upon the expiration of the initial suspension without any break.

- 13.9 Suspension of Privilege As set forth in 19 CFR 146.82(a), the Port Director may suspend any zone privileges. Such a suspension may cover all or part of the merchandise in a zone, or all or part of the transactions for which a privilege was granted. It may be limited to one or a few Users, or to particular activities, or kinds of merchandise.
 - (a) Notice to Show Cause The notice to show cause why the suspension should not be carried out will contain, in addition to the information cited in 19 CFR 146.82(a) and Section 13.8(b) FTZM, specific information concerning which privileges are to be suspended. The specific privileges to be suspended should be (1) relevant to the grounds for suspension, and (2) of a type whose suspension would bring about compliance. For example, if the season is finished for seasonal goods to be admitted to a zone, it would be more appropriate to suspend privileges to transfer merchandise than to suspend privileges to admit merchandise (19 CFR 146.82(b)(1)).
 - (b) Effect of Suspension of Privilege When privileges are suspended, there is no requirement that the merchandise in the zone be transferred to the Customs territory or to another zone. After termination of the suspension, privileges may be granted upon

application in proper form. The privilege will be restored without an additional qualification requirement of the Operator.

- 13.10 Suspension of Activation The Port Director may suspend the activation of a zone on any of the grounds specified in 19 CFR 146.82(a). Suspensions are generally applicable to grounds directly attributed to the Operator, or grounds which are applicable to more than one or a few Users, activities, or kinds of merchandise. The show-cause notice shall contain a clear statement that the notice is for suspension of activated status. The suspension shall be for a specified period of time up to 90 days.
 - (a) Effect of Suspension of Activated Status When activation has been suspended, the merchandise must be either duty paid or removed from the zone without expense to the Government, as specified in 19 CFR 146.7(c) and 19 CFR 113.73(c)(1) and Section 13.13 FTZM. Upon termination of the suspension, activation of the zone will be restored without application by the Operator.
- 13.11 Revocation of Activated Status or Privilege If a Port Director has reason to believe that the grounds for suspension under 19 CFR 146.82(a) cannot or will not be corrected within at least six months after a suspension has been put into effect, or when less severe sanctions have been or would be ineffective in obtaining compliance by the Operator or permittee with the laws and regulations, he or she may seek revocation of privileges or of activation by sending a written request to the Board, to exclude permanently or indefinitely the affected goods or operation(s) from the zone. This request shall be made in the same manner as a request to extend a suspension under Section 13.8(g) FTZM. When activation has been revoked, the merchandise must be either duty paid or removed from the zone without expense to the Government, as specified in 19 CFR 146.7(c), 19 CFR 113.73(c)(1) and Section 13.13 FTZM.
 - (a) Restoration of Approval If activation is suspended or revoked it may be restored to the zone only upon Board Order and be reactivated by reapplication under 19 CFR 146.6. Generally, CBP will not recommend approval of a Board Order to reinstate activated status or permits if (1) the Order would be issued less than a year after revocation, (2) any charges due under the Operator's bond have not yet been paid, or (3) application is made under another company or corporate name involving principals of the Operator whose activated status or privilege was revoked, and the Port Director has reason to believe that the grounds for suspension or revocation still exist.
- 13.12 Revocation of Grant In the event of willful violations of any of the provisions of the FTZ Act by the Grantee, the Board may revoke the Grant after four months' notice to the Grantee and affording it an opportunity to be heard. The testimony taken before the Board shall be reduced to writing and filed in the records of the Board together with the decision reached thereon (19 U.S.C. §81r(a)).
 - (a) Recommendation of Port Director for revocation The Port Director may at any time recommend to the Board that the privilege of establishing, operating, and maintaining a zone or subzone under CBP jurisdiction be revoked for willful and

repeated violations of the FTZ Act. If the Port Director believes that a substantial question of law exists as to whether willful and repeated violations of the Act have occurred that officer may request internal advice under the provisions of 19 CFR 177 from the Executive Director, Regulations and Rulings, Office of International Trade, Headquarters. A recommendation to the Board that a zone or subzone Grant be revoked does not preclude, and may be in addition to, any liquidated damages, fine, or suspension for cause (19 CFR 146.83(a)).

This is the most serious civil enforcement action that may be taken for foreign-trade zone violations. If a Port Director believes such a recommendation is advisable, it shall be made without any further CBP administrative proceedings, since the necessary administrative actions will be carried out by the Board. However, such a recommendation shall be made to the Board through the Assistant Commissioner, Office of Field Operations, Headquarters.

- (b) Procedure and Decision The procedure for revocation of a grant, the decision of the Board, and appeal is covered by the provisions of 19 U.S.C. §81r, and 15 CFR 400.28(c). If a grant is revoked, any and all zone status merchandise (except domestic status merchandise for which no permit is required under 19 CFR 146.43(b)) in any activated area of that zone shall be either duty paid or removed therefrom without expense to the Government, as specified in 19 CFR 146.7(c), 19 CFR 113.73(a)(1) and Section 13.13 FTZM.
- 13.13 Removal of Merchandise When approval of activated status has been suspended under 19 CFR 146.82(b) and Section 13.8 FTZM, or the Grant is revoked under Section 13.12 FTZM, the Port Director shall require all goods in that area in zone status (other than domestic status for which no permit is required) to be transferred to another zone, a bonded warehouse, or other location where they may lawfully be stored, or transferred to the Customs territory, if the Port Director considers it advisable to protect the revenue or administer any Federal law or regulation (19 CFR 146.7(c)).
 - (a) Procedure Activated status is suspended or, Grant is revoked when the final administrative action for suspension or revocation has been completed. Port Director s shall notify all Users having merchandise in the zone of the action, and advise them that they shall transfer their merchandise to the Customs territory or to another zone within 30 days after notification. If the merchandise is not so transferred within that time, the Port Director shall order it to be transferred to a location as specified by the Port Director (19 CFR 146.7(c)). Transfer at the order of the Port Director shall be at the risk and expense of the Operator, as specified in 19 CFR 113.73(c)(1). The Operator shall remain responsible for the merchandise until the cartman, carrier, or other responsible party signs for the merchandise acknowledging its receipt for removal from the zone (See Sections 9.16(e) and 10.6(c) FTZM).
- 13.14 Responsibility of Operator after Removal The liability of the Operator under its bond is not cancelled by reason of Grant revocation or suspension of zone status. The Operator remains

liable for payment of liquidated damages assessed under the bond, reimbursement of costs incurred by the Government in connection with the suspension or revocation, and other responsibilities under the bond, as applicable.

- 13.15 Penalties for Fraud, Gross Negligence, or Negligence Without regard to whether the United States is or may be deprived of all or a portion of any lawful duty thereby, no person, by fraud, gross negligence, or negligence--
 - (a) May enter, introduce, or attempt to enter or introduce, any merchandise into the commerce of the United States by means of -
 - (1) Any document or electronically transmitted data or information, written or oral statement, or act which is material and false, or
 - (2) Any omissions which is material, or
 - (2) May aid or abet any other person to violate subparagraph (A) (19 U.S.C. §1592(a)).

The maximum penalties for such violations range from the domestic value of the merchandise to 20 percent of the dutiable value, depending on whether fraud, gross negligence, or negligence is involved (19 U.S.C. §1592(c)). These are civil penalties. 19 U.S.C. §1592 is a civil penalty statute and thus penalties do not require a criminal conviction. The criminal provision that covers similar fraud is 18 U.S.C. §542, a felony statute. CBP procedures for handling these civil penalties are found in 19 CFR 162. Procedures for petitioning for relief from the penalties are found in Appendix B to 19 CFR 171.

- (a) Application to Zones While 19 CFR 1592 is a CBP law, it is not applicable to merchandise in zone status, according to 19 U.S.C. §81c(a). However, it is applicable to merchandise and articles in a zone which do not have zone status, such as prohibited articles, constructively transferred merchandise, and other merchandise and articles described in Sections 2.6 and 5.2 FTZM. Furthermore, in respect to merchandise and articles in zone status, 19 U.S.C. §1592 is applicable to transactions, practices, documentation, acts, and omissions which occur or occurred in the zone, even while the merchandise was in zone status, if it is later transferred to the Customs territory. 19 U.S.C. §1592 penalties connected with a zone would usually involve a User, rather than an Operator , unless the Operator is also the importer or has aided or abetted a Section 1592 violation by an importer or other party.
- (b) Procedural Protections Persons accused of violations of Section 1592 have some important procedural safeguards to assure fairness in administration of penalties by CBP:
 - (1) CBP must provide a pre-penalty notice to the person concerned, and time for a response, as set forth in 19 U.S.C. §1592(b)(1), 19 CFR 162.77 and 19 CFR 162.78;

- (2) After considering any presentations made in response to the pre-penalty notice, CBP must determine whether any violation has occurred and promptly notify the person concerned, in writing, of its final determination as provided in 19 U.S.C. §1592(b)(2), 19 CFR 162.79, and 19 CFR 171, Appendix B.
- (3) Lower penalties are assessed when the person concerned has disclosed the circumstances of a violation before, or without knowledge of, a formal investigation of the violation, as provided for in 19 U.S.C. §1592(c)(4), 19 CFR 162.74, and 19 CFR 171, Appendix B.
- (4) Limitations are placed on seizure the of merchandise involved in the violation, as prescribed in 19 U.S.C. §1592(c)(6) and 19 CFR 162.75; and,
- (5) CBP must meet certain proof requirements in any action to recover monetary penalties before the Court of International Trade, as prescribed in 19 U.S.C. §1592(e).
- (c) Payment of Duties Notwithstanding the assessment or non-assessment of a monetary penalty, CBP shall demand payment of any loss of duties, taxes and fees resulting from a 19 U.S.C. §1592(b) violation, even if the entry covering the merchandise has already been liquidated (See 19 U.S.C. §1592(d), 19 CFR 162.79(b) and 19 CFR 162.80).
- 13.16 Criminal Violation Criminal violations are those for which an individual, including an officer or agent of a violating corporation, may be criminally fined and/or imprisoned. In light of the constitutional safeguards against improper prosecutions and convictions, criminal violations are usually investigated by ICE and sometimes in conjunction with special investigators of other Federal agencies. The principal criminal violations of relevance to zone Operator s and Users are shown below:

Authority	Nature of violation	Penalty
18 U.S.C. §111	Assaulting, resisting, or impeding officer in performance of their duties	Fine under Title 18 U.S.C. (\$250,000 or less) or not more than 8 years prison (if with deadly weapon, not more than 20 years) or both
18 U.S.C. §201	Bribery of public official	Whichever is greater, fine under Title 18 U.S.C. (\$250,000 or less) or not more than 3 times amount of bribe, or not more than 15 years prison, or both; and disqualified from any office of honor, trust, or profit with U.S.

18 U.S.C. §371	Conspiracy to commit offense or fraud against the United States	Fine under Title 18 U.S.C. (\$250,000 or less) or not more than 5 years prison or both. If the underlying offense is a misdemeanor, punishment not to exceed offense maximum.				
18 U.S.C. §1341	Mail fraud	Fine under Title 18 U.S.C. (\$250,000 or less), not more than 20 years prison, or both. If a violation is in relation to a presidentially declared major disaster or emergency or affects a financial institution, not more than \$1,000,000, not more than 30 yrs. in prison, or both.				
18 U.S.C. §1343	Wire, radio, or television fraud	Fine under Title 18 U.S.C. (\$250,000 or less), not more than five years prison, or both. If violation affects a financial institution, not more than \$1,000,000 fine, not more than 30 yrs. prison, or both.				

The following criminal violations are contained in CBP law. These laws apply to merchandise in a zone which is not in zone status, and cover acts and omissions that occur in a zone involving merchandise which was either introduced, or attempted to be introduced, into the Customs territory. These laws do not apply to zone status merchandise which has never been introduced, or attempted to be introduced into the Customs territory.

Authority	Nature of Violation	Penalty
18 U.S.C. §541	Entry of falsely classified goods	Fine under Title 18 U.S.C. (\$250,000 or less), not more than two years prison, or both.
18 U.S.C. §542	Entry of goods by false statement	Fine under Title 18 U.S.C. (\$250,000 or less), not more than two years prison, or both.
18 U.S.C. §544	Relanding of goods without payment of duty	Fine under Title 18 U.S.C. (\$250,000 or less), not more than two years prison, or both; plus forfeiture of goods.

18 U.S.C. §545	Smuggling, including through use of false documents	Fine under Title 18 U.S.C. (\$250,000 or less), not more than 20 years prison or both.			
18 U.S.C. §549	Removal from CBP custody or tampering with CBP seal without permission	Fine under Title 18 U.S.C. (\$250,000 or less), not more than 10 years prison or both.			
18 U.S.C. §550	False claim or refund of duties or duty exemption upon exportation of goods	Fine under Title 18 U.S.C. (\$250,000 or less), not more than two years prison, or both; plus forfeiture of goods or its value.			
18 U.S.C. §551	Concealing or destroying invoices or other papers	Fine under 18 U.S.C. (\$250,000 or less), not more than two years prison, or both.			
19 U.S.C. §1304	Improper alterations of country of origin marking	Fine under Title 18 U.S.C. not more than \$100,000 fine for the first conviction for violation and fine not more than \$250,000 for second or subsequent conviction for violation, not more than 1 year prison, or both.			
0	THER RELEVANT CRIMINAL PROV	ISIONS			
18 U.S.C. §496	Forgery, counterfeiting, or falsely altering entry or withdrawal documents	Fine as set forth in 18 U.S.C. (\$250,000 or less), not more than three years prison, or both.			
18 U.S.C. §659	Theft in international or interstate commerce	Fine as set forth in 18 U.S.C. (\$250,000 or less), not more than 10 years prison, or both, if theft value is \$1000 or more. If less than \$1000, fined under Title 18 U.S.C. (\$250,000 or less), not more than three years prison, or both.			
18 U.S.C. §1001	Falsifying, concealing, or covering up material within jurisdiction of U.S. Government by any trick, scheme, or device	Fine as set forth in 18 U.S.C. (\$250,000 or less), not more than five years prison, or both.			

Statutes administered by agencies other than CBP may be applicable for violations involving particular kinds of merchandise, such as arms and munitions, endangered plants and animals, toxic substances, and other prohibited or controlled articles or substances.

- 13.17 Search, Arrest, and Seizure The supervision authority of 19 U.S.C. §81d and 19 U.S.C. §81o(b) and 19 CFR 146.3 is sufficient to conduct compliance reviews and audits by CBP personnel. However, in regard to more serious law enforcement violations, Operators and Users may come into contact with ICE, Contraband Enforcement Teams, Detector Dog Teams and other CBP enforcement personnel conducting searches or investigations. To properly enforce the laws and regulations, CBP personnel are given considerable authority to conduct searches, arrest suspected violators, and seize merchandise and articles.
 - (a) Search Authority The principal authority for CBP searches in zones is through the authority of the Board and the Secretary of the Treasury to issue regulations to carry out the FTZ Act and to protect the revenue (19 U.S.C. §810(b)). Under that authority, the Port Director may cause any merchandise to be examined before or at the time of admission to a zone, or at any time thereafter, if the examination is considered necessary to facilitate the proper administration of any law, regulation, or instruction which CBP is authorized to enforce (19 CFR 146.10). Also, CBP personnel representing the Board are authorized to inspect and examine at such times as the Board may deem advisable the premises, operations, or accounts of the Grantee (15 CFR 400.11(a)(6)).

Finally, in the case of any merchandise which has been constructively transferred to the Customs territory, or is otherwise not in zone status, but is in CBP custody, CBP personnel may stop, search, and examine any vehicle or person on which they suspect there is merchandise which is subject to duty or was introduced into the United States in any manner contrary to law (19 U.S.C. §482).

- (b) Arrest Authority 19 U.S.C. §81c(a), exempts certain merchandise in zones from CBP laws, but does not exempt persons from CBP laws. CBP personnel may be authorized to make an arrest without a warrant for any offense against the United States committed in the personnel's presence, or for any felony against the United States committed outside the personnel's presence if the personnel has reasonable grounds to believe the person to be arrested has committed a felony (19 U.S.C. 1589a(3)). Also, incident to a border search (noting Section 13.17(a) FTZM above), CBP personnel are authorized to arrest any person who is liable to arrest, by virtue of any law respecting revenue, without, as well as within their respective port, and to use all necessary force to make the arrest (19 U.S.C. 1581(f)).
- (c) Seizure Authority Merchandise whose importation is prohibited may be seized by CBP personnel in a zone (C.S.D. 82-16). Merchandise may be seized by any CBP personnel who have reasonable cause to believe that any law or regulation enforced by CBP has been violated, by reason of which the merchandise has become subject to seizure or forfeiture. The personnel must provide a receipt for the seizure to the person

from whom the merchandise was seized (19 CFR 162.21(a)). CBP procedures for handling seizures are set forth in 19 U.S.C §1602 through 19 U.S.C. §1616 and 19 CFR 162. The most frequent authorities for seizures that may be encountered by Operator s and Users, aside from those mentioned in Sections 13.15 and 13.16 FTZM are shown below:

Authority	Nature of Seizure
19 U.S.C. §1305	Pornographic, immoral, treasonous material and lottery tickets
19 U.S.C. §1526	Merchandise imported without consent of trademark holder, or with counterfeit trademark
19 U.S.C. §1527	Importation of wild mammals, birds, or parts of same in violation of foreign laws
19 U.S.C. §1594	Conveyance with prohibited merchandise, or seized for payment of penalty for violation in which conveyance or owner or Operator thereof was involved
19 U.S.C. §1595a(a)	Any conveyance or other property used to effectuate an importation contrary to law
19 U.S.C. §1595a(c)	Merchandise introduced into the U.S. contrary to law.
19 CFR 162.45 and 19 CFR 162.45a and cited statutes of Title 21 U.S.C.	Narcotics and other controlled substances

APPENDIX

- A. CBPF 214 Application for FTZ admission and/or status designation
- B. Instructions for filling out CBPF 214 and 214A (salmon or pink copy)
- C. CBPF 216 Application for FTZ Activity Permit
- D. Instructions for filling out CBPF 216
- E. Instructions for filling out CBPF 216 for Temporary Removal
- F. Instructions for filling out CBPF 216 for Destruction
- G. CBPF 3461 Estimated Weekly Entry
- H. CBPF 7501 Entry for Consumption
- I. CBPF 7512 In-bond Document
- J. Instructions for completing CBPF 7512 for Zone-to-Zone transfers
- K. Instructions for completing CBPF 7512 for Immediate or Indirect Exportation or for Transportation and Exportation
- L. CBPF 349 Harbor Maintenance Fee Quarterly Report
- M. CBPF 350 Harbor Maintenance Fee Amended Quarterly Report
- N. Standards for Cargo Security

Note:

The Customs and Trade Automated Interface Requirements (CATAIR) documentation provides complete information describing how importers and/or their agents can become Automated Broker Interface (ABI) participants, as well as how they can provide electronic import information to ABI, and receive transmissions from ABI once they have become a participant.

The CATAIR begins with general information about ABI participation along with instructions on how to become an ABI participant. The main body of the CATAIR consists of the data record formats used for conducting electronic data interchange (EDI) between the participant's computer system and Customs ABI.

E214 admissions can be found in the CATAIR Chapter for Foreign Trade Zone. E-214 Error Codes can be found in Appendix P of the CATAIR.

The (CATAIR) for Entry Summary and related information filed through the Automated Commercial Systems (ACS) can be found at: http://www.cbp.gov/xp/cgov/trade/automated/automated_systems/abi/catair/

APPENDIX A

CENSUS L	DEPARTMENT OF HOMELAND SECURITY U.S. Customs and Border Protection APPLICATION FOR FOREIGN-TRADE ZONE ADMISSION AND/OR STATUS DESIGNATION 19 CFR 146.22, 146.32, 146.35-146.37, 146.39-146.41, 146.44, 146.53, 146.66								-0029 Exp. 01-31-2010 TION (Address)	
3. IMPORTING	VESSEL (& FL/	AG)/OTHER CARRIE	ER.	4. EXF	PORT DATE	5. IM	PORT DATE	6. ZONE	ADMIS	SION NO.
7. U.S. PORT O	F UNLADING		8. FOREIGN	8. FOREIGN PORT OF LADING			9. BILL OF LADING/AWB NO. 10. INWA			NWARD M'FEST NO.
11. INBOND CAR	RRIER		12. I.T. NO. AND DATE			13.	13. I.T. FROM (Port)			
14. STATISTICA	L INFORMATIO	N FURNISHED DIR	ECTLY TO BU	REAU OF CEN	SUS BY APPLIC	CANT?	YES	□ NO		
15. NO. OF PACKAGES AND COUNTR OF ORIGIN COI	Y	16. SCRIPTION OF ME	RCHANDISE	CHANDISE 17.			18. 19. QUANTITY GROSS (HTSUS) WEIGHT		- 1	20. SEPARATE VALUE & AGGR CHGS.
		n of the above mercing		Foreign-Trade		to the	best of my kn	owledge and b	eller th	
and section	146.31, Custon			within the mean	ing or section 3	or the i	Foreign-Trade	Zones Act of	1934, a	as amended,
	NONPRIVILE (19 CFR	designation indicate GED FOREIGN [146.42)	PRIVILEGE (19 CF	D FOREIGN R 146.41)	146.41)		ZONE RESTRICTED (19 CFR 146.44)			DOMESTIC (19 CFR 146.43)
24. APPLICANT	FIRM NAME		25. BY (Sig	5. BY (Signature) 26			26. TITLE			27. DATE
F.T.Z. AGREES TO RECEIVE MERCHANDISE INTO THE ZONE			28. FOR THE F.T.Z. OPERATOR (Signature)			re)	29. TITLE			30. DATE
PERMIT	Permission is transfer the into the Zone	s hereby granted t above merchandis	31. PORT 0	IRECTOR OF	CBP: BY (Signa	ture)	32. TITLE			33. DATE
PERMIT	The above merchandise has been granted the requested status. 34. PORT DIRECTOR OF CBP: BY (Signature) 35. TITLE							36. DATE		
PERMIT TO	37. The goods	described herein are	e authorized to	be transferred:	v	without	exception	exce	ept as r	noted below
TRANSFER	38. CBP OFF	ICER AT STATION ((Signature)		39. TITLE			40. STATION	'	41. DATE
	42. RECEIVE	RECEIVED FOR TRANSFER TO ZONE (Driver's Signature) 43. CARTMAN					44. CHL NO. 45. DATE			45. DATE
FTZ OPERATOR'S REPORT OF MERCHANDISE RECEIVED AT	E									
ZONE	47. FOR THE FTZ OPERATOR (Signature) 48. TITLE 49. DA							49. DATE		
(See page 2 for Paperwork Reduction Act Notice.) Previous Editions are Obsolete CBP Form 214 (02/96)										

APPENDIX B

Instructions for filling out CBPF 214 and 214A

Items

- 1. **Zone Number and Location (Address) -** Zone number designated in FTZ grant and delivery address. Do not report post office box number for address.
- 2. **Port Code** The Port Code where the zone is located as shown in Schedule D, Harmonized Tariff Schedule of the United States (HTS).
- 3. **Importing Vessel and Flag/Other Carrier -** For merchandise arriving in the United States by vessel, give the name of the vessel and the name (flag) of the country in which the vessel registered (or the country code as provided in Schedule C, HTS), which transported the merchandise from the last foreign port of lading to the first United States port of unlading prior to admission to the zone.

For merchandise unladen in the zone directly from the carrier which transported the merchandise from the last foreign port of lading as "mail", "railroad", "truck", "pipeline", "ferry", or "other", as appropriate.

Leave this block blank for goods in domestic status for domestic or duty-paid goods in zone-restricted status.

When merchandise is admitted, after transportation from another foreign-trade zone show, only "zone transfer" should be in this space.

4. **Export Date -** For merchandise exported by vessel, note the month, day and year on which the carrier departed the last port of the country of exportation. For merchandise exported by air, note the month and year in which the aircraft departed the last airport of the country of exportation. For overland shipments, including those where the port of lading is located outside the country of exportation (e.g., the goods are exported from Switzerland, but laden at Hamburg), note the month and year in which the carrier crossed the border of the country of exportation. For mail shipments, not the date of exportation in CBPF 3509, Notice of Addressee of Arrival of Mail Shipment.

Leave blank for goods in domestic status, domestic or duty-paid goods in zone-restricted status, or goods transferred from another zone.

5. **Import Date -** For merchandise arriving in the United States by vessel, note the month, day and year on which the importing vessel transporting the merchandise from the foreign country arrived within the limits of the U.S. port at which the merchandise was unladen. For merchandise arriving in the U.S. other than by vessel, note the month, day and year in which the merchandise arrived within the limits of the U.S. For mail shipments, note the date of the Notice of Addressee on CBPF 3509.

Leave blank for goods in domestic status, domestic or duty-paid goods in zone-restricted status, or goods transferred from another zone.

- 6. **Zone Admission No. -** Report unique and sequential number assigned for numbering shipments admitted to a zone under 19 CFR 146.32(a) whether or not the lot method is used for inventory control in the zone. The number should be assigned by the zone Operator or User, according to local practice.
- 7. **U.S. Port of Unlading -** For merchandise arriving in the United States by vessel or air, note the U.S. port (not port code), as provided in Schedule D, Harmonized Tariff Schedule of the United States, at which the merchandise was unladen from the importing vessel or aircraft, whether or not such a port is a CBP port of entry. For example, if CBPF 214/e-214 is filed at the Port of Los Angeles for merchandise unladen at Long Beach, California, show Long Beach as the port of unlading. The same principle applies when the goods are unladen at a smaller port within a consolidated port of entry, e.g., Stockton within the San Francisco consolidated port.

When merchandise is transported in-bond from the U.S. port where unladen from the importing vessel or carrier to another U.S. port to be admitted to a foreign-trade zone, show as the port of unlading the port or point where the merchandise was unladen from the importing carrier before transportation in-bond.

For imported mail shipments, the port of unlading is the port which issued CBPF 3509 to the importer. Leave blank for goods in domestic status, domestic or duty-paid goods in zone-restricted status and goods transferred from another zone.

8. **Foreign Port of Lading -** For merchandise arriving in the United States by vessel or air, report the name of the foreign port at which the merchandise was actually laden on the vessel or aircraft that carried the merchandise to the U.S. For shipments originating in either Canada or Mexico and arriving in the U.S. from Canada or Mexico by rail, truck, pipeline, or other non-vessel/no-air mode of transportation, supply the name of the province (Canada) or state (Mexico) where the merchandise was first laden for exportation to the U.S. For imported mail shipments, show the city listed under the address of the foreign shipper on CBPF 3509.

For merchandise transshipped abroad in the course of shipment to the U.S., whether or not covered by a through bill of lading, report the foreign port at which the merchandise was laden on the vessel, aircraft, or the carrier which transported it to the first U.S. port of unlading. Neither the foreign port of original lading nor any port of lading other than the last foreign port of lading shall be substituted.

When a single CBPF 214/e-214 covers merchandise laden at more than one foreign port, the foreign port of lading shall be indicated separately in Item 15, "Marks, Nos., and County of Origin", immediately below the country of origin designation and on the same line as the merchandise laden at each foreign port. In such cases, report "See Item 15" in Item 8.

Leave blank for goods in domestic status, domestic or duty-paid goods in zone-restricted status, or goods transferred from another zone.

- 9. **Bill of Lading/AWB No. -** Number of bill of lading or air waybill assigned to shipment, whether the merchandise is imported or not. Leave blank for mail shipments.
- 10. **Inward Foreign Manifest No. -** Number assigned by importing carrier to cargo declaration in which goods were included. Leave blank if carrier does not assign manifest numbers. Also leave blank for in-bond shipments, mail shipments, and shipments of goods in domestic status, domestic or duty-paid goods in zone-restricted status, and goods transferred from another zone.
- 11. **In-Bond Carrier -** Name of party responsible under custodial bond of bonded carrier for transportation in-bond or to port with jurisdiction over zone. Leave blank if goods were not transported in-bond from another port.
- 12 **I.T. Number and Date -**Type of in-bond shipment (e.g., IT, WDT&E) and nine-digit Immediate Transportation number assigned to the in-bond shipment and date the CBPF 7512 was prepared. Leave blank if goods were not transported in-bond.
- 13. **I.T. from (Port) -** Port of origin of in-bond shipment. If transported from a foreign-trade zone at another port, show also in parentheses "FTZ" and the foreign-trade zone number after the name of the port. Example Miami (FTZ 32) is transporting to another zone in the same port, report in parentheses only "FTZ" and the zone number. Otherwise, leave blank if goods were not transported in-bond.
- 14. **Statistical Information Furnished directly to the U.S. Census Bureau by Applicant** If the Applicant is a participant of the E-214 Program an electronic CBPF 214 is used and the statistical information will be abstracted from the E-214 program data and the "Yes" box is marked. If a hard copy CBPF 214 is used the statistical copy (214A) should be sent to CBP to be forwarded to the U.S. Census Bureau and the "No" box is marked. For use of this form for blanket admissions under 19 CFR 146.32(d), see Section 6.7(g) FTZM. For use and instruction of this form for direct delivery admissions see 19 CFR 146.40(b)(1) and Section 6.10 FTZM.
- Number of Packages & Country of Origin Quantity and kind of packages (cases, bundles, cartons, etc.) in shipment. For containerized merchandise, show quantity of packages with intermodal container(s); show also container number(s). For bulk shipments, show "Bulk."

Report country of origin code provided in Annex B, ISO code, Harmonized Tariff Schedule. The country of origin, in which, the product was mined, grown or manufactured. Further, labor work or material added to an article in another country must affect a substantial transformation in order for such other country to become the actual "country of origin."

When the merchandise is invoiced or exported from a country other than that in which it originated the actual country of origin shall be specified rather than the country of invoice or exportation.

When a single CBPF 214/e-214 covers merchandise from more than one country of origin the country of origin shall be indicated separately against each HTS subheading (or group of subheadings.)

Multiple foreign ports of lading may be reported here, as noted in the instructions in Item 8.

- 16. **Description of Merchandise** Detailed description of goods as shown in the commercial invoice or the documentation. If no commercial documentation is available, describe goods in sufficient detail that CBP personnel can ascertain whether the goods whose admission is sought are the goods actually received at the zone, and/or determine whether the goods are eligible for admission to the zone.
- 17. **HTS No. -** Report the appropriate subheading including statistical suffix under which the merchandise is classified in the Harmonized Tariff Schedule. For items admitted under the provisions of HTS numbers 9802.00.5010, 9802.00.6000, 9802.00.8015, 9802.00.8040, 9802.00.8055, 9802.00.8065, or 9802.00.9000 show the Chapter 98 number and the value on the first line and immediately below the Chapter 98 number enter the Chapter 1 through 97 HTS applicable for the specific commodity.

If more than one zone status is shown for different articles of merchandise reported on CBPF 214/e-214, identify each line item by noting "Privileged Foreign", "Non-Privileged Foreign", or "Zone-Restricted" below the appropriate HTS subheading in Item 18; or group merchandise in like statuses and divide them by a dotted line. In addition, the boxes for all of the applicable statuses in Item 23 shall be checked. For goods in domestic status or domestic or duty-paid goods in zone-restricted status, leave Item 18 blank.

18. **Quantity** (HTS) - When a unit of quantity is specified in the HTS for the subheading, report the net quantity in the specified unit, and show the unit after the net quantity figure. If two units of quantity are shown for the commodity in the HTS, report the net quantity in both units, with the unit indicated in each case. Insert the quantity in terms of the unit marked with a superior "V" on the line horizontal with the value of Item 21. Put the quantity in terms of any other unit below the first quantity and enclose it in parentheses.

If no unit of quantity is specified in the HTS or for goods in domestic status or domestic or dutypaid goods in zone-restricted status, report the quantity in parentheses, using the unit of quantity in which the goods were invoiced. If the goods are not invoiced the unit of quantity in which the goods are ordinarily purchased.

Report quantities in whole units, unless fractions of units are required by the Port Director for CBP control purposes.

For goods in domestic status or domestic or duty-paid goods in zone-restricted status, report in the invoiced unit of quantity or in the unit of quantity in which the goods are ordinarily invoiced.

19. **Gross Weight -** Report the gross shipping weight in kilograms for articles admitted to the zone. Supply separate gross weight information for each HTS subheading. If the gross weight is not available for each subheading, approximate shipping weight for each item shall be estimated and reported. The total of these estimated weights shall equal the actual gross shipping weight for all of the merchandise covered by all of the HTS subheadings on the form.

In the case of containerized cargo, the weight of the intermodal containers should not be included in the shipping weight of the merchandise covered by each HTS subheading.

20. **Separate Value and Aggregate Charges -** Report the purchase price in U.S. dollars or in the case of merchandise not acquired by purchase, the equivalent of such price. This value must be shown for each HTS number described separately in Item 17.

Report the aggregate cost in U.S. dollars of freight, insurance, and all other costs, charges and expenses incurred in bringing the merchandise from alongside the carrier at the foreign port of exportation in the country of exportation and placing it alongside the carrier at the first U.S. port of entry (unlading). This value shall be reported immediately beneath the entered value and identified with the letter "C."

In the case of overland shipments (i.e., merchandise transported to the U.S. by means other than vessel or air) originating in Canada or Mexico, foreign inland freight will be reported as charges.

Charges are required for each item valued over \$1,250, and in certain special cases for each line item valued over \$250.

21. **Harbor Maintenance Fee -** Commercial cargo loaded or unloaded from a commercial vessel is subject to a port User fee, XXXXX of its value. Users, not Operator s, are subject to the fee.

22. Not to be filled out

- 23. **Status designation -** Fill in appropriate box with "X" or check mark to show desired zone status. For "Privileged Foreign", note that a CBP entry is not required with the application. See the instructions in Item 18 when merchandise of more than one status is reported on a CBPF 214/e-214.
- 24. **Applicant Firm Name** Name of firm with right to make entry which applies for admission of goods to zone.
- 25. **By (Signature)** Signature of authorized representative of firm applying for admission of goods to zone.

- 26. **Title -** Title of authorized representative of applicant firm.
- 27. **Date** Date application is made and signed by authorized representative of applicant firm to admit goods to zone.
- 28 **For the FTZ (Signature)** Signature of authorized representative of zone Operator who agrees to receive merchandise into the zone.
- 29. **Title** Title of authorized representative of Operator.
- 30. **Date** Date authorized representative of Operator signed to agree to receive merchandise into zone.
- 31. **Port Director of CBP: By (Signature)** Signature of authorized representative of Port Director who approves admission of goods to zone.
- 32. **Title** Title of authorized representative of Port Director.
- 33. **Date** Date authorized representative of Port Director signed to approve admission of goods to zone.
- 34. **Port Director of CBP: By (Signature)** Signature of authorized representative of Port Director who approves application for zone status.
- 35. **Title** Title of authorized representative of Port Director.
- 36. **Date** Date authorized representative of Port Director signed to approve requested zone status.
- 37. **Permit to Transfer (PTT)** Fill in appropriate box with "X" or check mark. If box "Except as noted below" is filled in, described any shortage, overage, damage or other discrepancy between the goods described in Items 15 through 17 and those approved to be transported by cartman or other approved party to zone. If CBP personnel have not actually inspected merchandise, exception report may be made by cartman representative. If CBPF 214 is not used as a delivery ticket, Items 37-45 shall not be filled in.
- 38. **CBP Officer at Station "Port" (Signature)** Signature of CBP personnel at port from which goods are to be transported by cartman or other approved party to zone authorizing such cartage.
- 39. **Title** Title of CBP personnel authorizing cartage to zone.
- 40. **Station** Name of CBP port from which cartage to zone was authorized.
- 41. **Date** Date CBP personnel signed to authorize cartage to zone.

- 42. **Received for Transfer to Zone (Driver's Signature)** Signature of driver of cartage vehicle transporting goods to zone or other authorized representative of cartman firm.
- 43. **Cartman** Name of licensed cartman firm transporting goods to zone.
- 44. **CHL Number -** Number of Customhouse License of cartman firm transporting goods to zone.
- 45. **Date** Date cartman driver or their representative signed to receive goods for transfer to zone.

APPENDIX C

					Approved through	OMB No. 1651-0029	Exp. 01-31-2010		
DEPARTMENT OF HOMELAND SECURITY U.S.Customs and Border Protection			1. ZONE NO. AND LO	ZONE NO. AND LOCATION (Address)					
APPLICATION FOR FOREIGN-TRADE ZONE									
ACTIVITY PERMIT				2. ZONE ADMISSION	I NO.	3. APPLICATION D	3. APPLICATION DATE		
4. TYPE OF ACT		R 146.52, 146.66							
Manipulate	9	Manufacture		Exhibit	Destroy	Destroy Temporary Removal ration is to be performed and, in the case of a			
merchandise hav	ılng different zol	facture, a statemen ne status. If addition occurred in propose	nai space required, a	handise with one zone statu: ttach separate sheet. If first	s is to be packed, com application for manuf	imingled, or combine acturing of this kind, :	d with state whether		
6. ZONE LOT NO.	7.		8.		9.	10.	11.		
OR UNIQUE IDENTIFIER	MARKS AND NUMBERS	ľ	DESCRIPTION OF M	IERCHANDISE	QUANTITY	WEIGHTS, MEASURES	ZONE STATUS		
if any morehandia	se is to be man	ne al batelina	or manufactured. La	igree to maintain the records	s provided for in section	ns 145 21/2) 145 2	3 and 146 52(d)		
of the Customs R	Regulations and		able to CBP officers f		·	2116 140.21(a), 140.23	o, and 140.02(u)		
12. APPLICANT	FIRM NAME		13. BY (Signature)		14. TITLE	14. TITLE			
	D BY FORE		15. BY (Signature)		16. TITLE	16. TITLE			
				PERMIT	<u>'</u>				
The application requested, or	on made above n condition that	is hereby approved the applicable regul	and permission is gr ations are compiled v	ranted to manipulate, manufa with and the records required	acture, exhibit, destroy d to be maintained will	y, or temporarily remo be available for insp	oved, as ection.		
17. PORT DIRECTOR OF CBP: By (Signature)			18. TITLE		19. DATE				
			FTZ OPE	RATOR'S					
20. TO THE POR I certify that t			en disposed of as di	rected except as noted below	м.				
21. FOR THE FT	Z OPERATOR:	(Signature)		22. TITLE		23. DATE			
						CBP For	m 216 (01/01)		

214

Instructions for filling out CBPF 216

Items

- 1. **Zone Number and Location -**Zone number designated in foreign-trade zone grant and delivery address. Do not report postal box number for address.
- 2. **Zone Admission Number -** Show all zone admission numbers affected by activity. For blanket applications, show "Blanket" in this space.
- 3. **Application Date -** Show the date the form was prepared. This will also be considered by CBP to be the date the form was signed by the authorized representative of the applicant, unless a date of signature is shown in Item 13.
- 4. **Type of Activity for Which Permit is Requested -** Place "X" in appropriate box. Consult with the Port Director when unsure as to which box describes desired activity. More than one box may be marked, if appropriate to describe activity.
- 5. **Full Description of the Activity -** Show description of activity in sufficient detail to enable CBP and zone Operator to ascertain what applicant desires to be done with the merchandise, and to enable them to determine whether they should approve the application. Consult with the Port Director if unsure as to the degree of detail required in description. Show where operation is to be performed, and whether goods in one status will be combined with goods in another status. Show any new lot numbers or UINs to be created through the activity. Check Section 8.4, FTZM concerning description of activity in applications for destruction.

For blanket application, show also beginning and ending date of the period for which the permit is required.

- 6. **Zone Lot Number or Unique Identifier** Show both initial and new lot numbers, or UINs. For blanket applications, show "Blanket."
- 7. **Marks and Numbers** Show marks and numbers appearing on outside of packages of both initial and resulting goods. If not packed either before or after the activity, show "Unpacked." For blanket application, show "Blanket."
- 8. **Description of Merchandise** Show description of both initial and resulting merchandise in sufficient detail to enable CBP and zone Operator to ascertain which merchandise is covered by the activity.

- 9. **Quantity** Show quantity of goods to be affected by activity, in either metric or U.S. units of quantity. For blanket applications, show estimated quantity to be affected during period of permit.
- 10. **Weights and Measures** Show net weight of goods to be affected by activity, in either metric or U.S. units of quantity. For blanket applications, show estimated quantity to be affected during period of permit.
- 11. **Zone Status** Zone status of goods to be affected by request. If more than one status is represented in a lot, show all statuses represented. Do not show "Various", even for blanket applications.
- 12. **Applicant Firm Name -** Show name of firm requesting permit for activity.
- 13. **By** (**Signature**) Show signature of authorized representative filing application for applicant firm. Show date of signing if different from application date in Item 4.
- 14. **Title** Title of authorized representative of applicant firm.
- 15. **By (Signature)** Show signature of authorized representative who approved application for zone Operator.
- 16. **Title** Show title of authorized representative of zone Operator.
- 17. **Port Director of CBP: By (Signature)** Show signature of authorized representative who approves request for the Port Director.
- 18. **Title** Title of authorized representative of the Port Director.
- 19. **Date** Date of signing to show when application was approved by the Port Director.
- 20. **To the Port Director of CBP** Report of authorized representative of zone Operator certifying that permitted activity has been carried out as approved by the Port Director. If CBP provided physical supervision of the activity, the supervising CBP personnel will make this report. If there are exceptions, give a report in sufficient detail to inform the Port Director of what happened. If there are not exceptions, this space need not be filled out.
- 21. **CBP Officer Signature** Signature of authorized representative of the Operator or of the CBP personnel making the report of activity.
- 22. **Title** Title of authorized representative of the Operator or CBP personnel.
- 23. **Date** Date of signature of authorized representative of the Operator or CBP personnel.

NOTE: Full completion of Items 17 through 19 on CBPF 216 constitutes the permit for the applicant to initiate the desired activity. The activity shall not commence unless and until the permit has been both signed and dated by the Port Director. Items 20 through 23 shall not be completed for blanket applications, since the report must be carried in the Operator's inventory and recordkeeping system under 19 CFR 146.52(d)(2).

Instructions for filling out CBPF 216 For Temporary Removal

Instructions are the same as those for applications to manufacture except as noted below:

- Item 5 Describe the operation to be performed during temporary removal in sufficient detail to enable the Port Director to determine whether it meets the conditions of temporary removal. Also, show the location where the operation will be performed.
- Permit In the permit section type "temporary removal" under the word permit.
- Item 20 Make the change in the certification as noted in the example.

APPENDIX F

Instructions for filling out CBPF 216 For Destruction

Instructions are the same as those for applications to manipulate, except as noted below:

- Item 4 Place and "X" or check mark in the block for "Destroy."
- Item 5 The description must show the method of destruction in sufficient detail to enable the Port Director to determine whether the destruction will be complete, and can be done in a safe and sanitary manner.
- Item 20-23 These items will be filled out by CBP personnel when destruction is physically supervised by CBP.

APPENDIX G

DEPARTMENT OF HOMELAND SECURITY U.S. Customs and Border Protection

Form Approved OMB No. 1651-0024 Exp. 11-30-2008

ENTRY/IMMEDIATE DELIVERY

19 CFR 142.3, 142.16, 142.22, 142.24

1. ARRIVAL DATE		2. ELECTED ENTRY DATA	3. ENTRY TYPE CODE/	ENTRY NUMBER			
5. PORT 6		6. SINGLE TRANS. BOND	7. BROKER/IMPORTER	FILE NUMBER			
8		8. CONSIGNEE NUMBER			9. IMPORTER NUMBER		
10. ULTIMATE CO	INSIGNEE NAME		11. IMPORTER OF RECORD NAME				
12. CARRIER COL	DE	13. VOYAGE/FLIGHT/TRIP	14. LOCATION OF GOODS-CODE(S)/NAME(S)				
15. VESSEL CODE	E/NAME						
16. U.S. PORT OF	UNLADING	17. MANIFEST NUMBER	18. G.O. NUMBER	19. TOTAL VALUE			
20. DESCRIPTION	OF MERCHANDISE						
21. IT/BL/AWB CODE	22. IT/BL/AWB NO.	23. MANIFEST QUANTITY	24. H.S. NUMBER	25. COUNTRY OF ORIGIN	26. MANUFACTURER NO.		
	27. CERTIFICATION			28. CBP USE	ONLY		
Information is accu	olication for entry/immediate delivi urale, the bond is sufficient, valid, OCFR Part 142 have been met.	ery. I certify that the above and current, and that all	OTHER AGENCY ACTION REQUIRED, NAMELY:				
PHONE NO.		DATE	CBP EXAMINATION REQUIRED.				
29. BROKER OR OTHER GOVT. AGENCY USE			ENTRY REJECTED, BECAUSE:				
			DELIVERY AUTHORIZED: SIGNATI	URE	DATE		

PAPERWORK REDUCTION ACT NOTICE: This information is to determine the admissibility of imports into the United States and to provide the necessary information for the examination of the cargo and to establish the liability for payment of duties and taxes. Your response is necessary. The estimated average burden associated with this collection of information is 15 minutes per respondent depending on individual circumstances. Comments concerning the accuracy of this burden estimate and suggestions for reducing this burden should be directed to U.S. Customs and Border Protection, Information Services Branch, Washington, DC 20229, and to the Office of Management and Budget, Paperwork Reduction Project (1651-0024), Washington, DC 20503.

CBP Form 3461 (01/89)

APPENDIX H

Form Approved OMB No. 1651-0022 DEPARTMENT OF HOMELAND SECURITY Filer Code/Entry No. 3. Summary Date U.S. Customs and Border Protection 4. Surety No. 5. Bond Type 6. Port Code 7. Entry Date **ENTRY SUMMARY** 8. Importing Carrier 9. Mode of Transport 10. Country of Origin 11. Import Date 12. B/L or AWB No. 13. Manufacturer ID 14. Exporting Country Export Date 16. I.T. No. 17. I.T. Date 18. Missing Docs 19. Foreign Port of Lading 20. U.S. Port of Unlading 21. Location of Goods/G.O. No. 24. Reference No. 22. Consignee No. 23. Importer No. 25. Ultimate Consignee Name and Address 26. Importer of Record Name and Address City State Zip City State Zip 32 33 34 27. 28. Description of Merchandise A. HTSUS Rate Duty and I.R. Tax A. Entered Value B. ADA/CVD Rate C. IRC Rate A. HTSUS No. A. Grossweight Net Quantity in Dollars Line B. CHGS B. ADA/CVD No. B. Manifest Qty. HTSUS Units Nο C. Relationship D. Visa No. Other Fee Summary for Block 39 35. Total Entered Value CBP USE ONLY TOTALS A. LIQ CODE 37. Duty B. Ascertained Duty Total Other Fees REASON CODE C. Ascertained Tax 38. Tax 36. DECLARATION OF IMPORTER OF RECORD (OWNER OR PURCHASER) OR AUTHORIZED AGENT D. Ascertained Other 39. Other I declare that I am the ___ importer of record and that the actual owner, E. Ascertained Total 40. Total purchaser, or consignee for CBP purposes is as shown above, OR owner or purchaser or agent thereof. I further declare that the merchandise was obtained pursuant to a purchase or agreement to purchase and that the prices set forth in the invoices are true, OR was not obtained pursuant to a purchase or agreement to purchase and the statements in the invoices as to value or price are true to the best of my knowledge and belief. I also declare that the statements in the documents herein flied fully disclose to the best of my knowledge and belief the true prices, values, quantities, rebates, drawbacks, fees, commissions, and royalties and are true and correct, and that all goods or services provided to the seller of the merchandise either free or at reduced cost are fully disclosed. I will immediately furnish to the appropriate CBP officer any information showing a different statement of facts. 41. DECLARANT NAME SIGNATURE DATE 42. Broker/Filer Information (Name, address, phone number) 43. Broker/Importer File No. CBP Form 7501 (04/05)

APPENDIX I

		TDANSDODTATION D	NTDV AND MAN	IICCET _	OMB No. 1651	1-0003 Exp. 12-31-2010	
19 CFR 10.60, 10.61, 123.41, 123.42		22.42	TRANSPORTATION ENTRY AND MANIFEST OF GOODS SUBJECT TO CBP				
Entry No.		INSPECTION	INSPECTION AND PERMIT				
Port		U.S. Customs and		on		1E) (Drawback, etc.)	
Date		PORT FIRS	ST U.S. PORT UNLADING				
					DATE		
Entered or import	ted by		Importer/I	RS#		to be shipped	
in bond via	(C.H.L number)	(Vessel or carrier)	(Car number and Initial	0	(Pier or station)	consigned to	
CBP Port Directo		(vessel or carrier)	Final foreign destin				
Consignee					(For exports	stions only)	
_		(At CE	P port of exit or destination)		D-4635	_	
Foreign port of la	aing	(Above information to be furnished	/L. No. only when merchandise is imp	arted by vessel	Date of Salling	9	
Imported on the		Flag	on		via		
Exported from		sel or carrier and motive power) On	Goods now at	(Date Imported	d)	(Last foreign port)	
	(Count	ry) On (Date)		(Name	of warehouse, station	ı, pier, etc.)	
Marks and Numbers of Packages	Desc (De	oription and Quantity of Merchandise Number and Kind of Packages scribe fully as per shipping papers)	Gross Weight In Pounds	Value (Dollars o		Duty	
G.O. No.		Check if withdrawn for Ve		-			
		FOR TRANSPORTATION IN BOND FOR EXPORTATION FOR	I truly declare that the correct to the best of Entered or withdrawn	my knowled	ge and belief.		
		(Port)					
WITH THE EXCEPTIONS NOTED A WITHIN-DESCRIBED GOODS WERE							
named above, for delivery to the CBP Port Director at destination sealed with CBP		Laden on the-					
		(Vessel, vehicle, or aircraft)	To the Inspector: The above-described goods shall be disposed of				
		which cleared for					
or the packages (were) (were not) labeled, or corded and sealed.					For the Port Dire	ector	
		on	Received from the Port Director of the above CBP location the merchandise described in this manifest for transportation and delivery into the custody of the CBP officers at the port named above, all packages in apparent good order except as noted hereon.				
		(Date) as verified by export records.					
(Inspect	tor)	(Inspector)					
(Date)		(Date)	-		Attorney or Agent of	f Carrier	

CBP Form 7512 (03/08)

Instructions for completing CBPF 7512 For Zone-to-Zone transfers

General instructions for all importers and carriers in preparing this form are found in CD 3240-36 and superseding issuances, including ABI Administrative Message 98-0529. The following instructions should be noted by zone Users filing entries for zone-to-zone transfers.

Upper left corner - After "Entry No.", show the lot number if the inventory is maintained under the lot method.

Upper right corner - Be sure to show class of entry as "IT."

Entered or imported by - Show name of person who is principal on the entry bond guaranteeing transfer.

Importer/IRS# -Show the EIN of the importer.

In-bond via - Usually only the bonded carrier's name would appear here; however, if the merchandise is to be moved to another location for delivery to the bonded carrier, show also the name and EIN of the cartman. In this case, the reverse side of the CBPF 7512 may be filled out in lieu of another CBP form. Also, after the name of the bonded carrier, show in parentheses the EIN to be obligated for transportation.

CBP Port Director - This space must be completed to show the name of the destination port where the receiving zone is located.

Consignee - Show the consignee as the receiving zone.

General export information - Since this form is not designed expressly for export information, placing of the information in the fields entitled "Foreign port of lading," "Date of sailing," "Exported on the," "Flag," "on," "Exported from," and "Goods now at," should all be left blank. This form is a multi-use form and these fields are used for imports rather than for exports.

Open space in the middle of form - As noted, this space is used for information required in 19 CFR 146.62(b)(2) CR, and where applicable, 19 CFR 146.70(c) and (d) CR.

To the Inspector - The dated signature of the CBP personnel in this space usually constitutes the permit to transfer the merchandise for the zone. In rare instances, CBP personnel may physically supervise the transfer, in which case the permit will be shown in the space in the lower left corner beginning with, "Delivered to the Carrier....." In the case of a Weekly Zone-

to-Zone Transfer Permit, no CBP signature is required in this space and a notation should be made as to "FTZ#, In-bond Program, No CBP signature required.

Received from the Port Director - The signature and date of the cartman or bonded carrier in this space relieves the zone Operator of responsibility for the merchandise and transfers that responsibility to the cartman or carrier. Any discrepancies in quantity or condition found upon receipt by the cartman or carrier should be shown in the open space in the middle of the form.

Record of Cartage or Lighterage -This portion of the form may be filled out for the cartage, in lieu of filing other CBP forms. If the CBPF 7512 is used for cartage, 3 extra copies must be prepared as the zone Operator, cartman, and receiving zone copies of the cartage record.

Instructions for completing CBPF 7512 for Immediate or Indirect Exportation or for Transportation and Exportation

General instructions for all importers and carriers in preparing this form are found in CD 3240-36 and superseding issuances, including ABI Administrative Message 98-0529. The following instructions should be noted by zone Users filing entries for direct or immediate exportation and for indirect or transportation exportation.

Upper left corner - After "Entry No.", show the lot number if the inventory is maintained under the lot method.

Upper right corner - For either direct or immediate exportation, show "IE;" for indirect exportation or entry for transportation and exportation show "T&E."

Entered or imported by - Show name of person who is principal on importation or entry bond guaranteeing exportation.

Importer/IRS# - Show the EIN of the importer.

In-bond via - Show name of cartman along with the IRS number (EIN) of the cartman who will deliver merchandise to exporting conveyance, and show location to which merchandise will be delivered. Back of form may also be filled out for the cartage transaction. If T&E, show the name of the bonded carrier, followed in parentheses, by custodial bond of carrier.

CBP Port Director - This space need not be completed for direct or immediate exportation entry, but must be completed to show the name of the destination port for indirect exportation entry or for transportation and exportation entry.

Final foreign destination - Show the destination on the bill of lading or air waybill and not necessarily the foreign port of discharge from the exporting carrier.

Consignee - Show the consignee as designated on the bill of lading or air waybill. If consigned to order, show the name of the person who will make arrangements for disposition at the foreign destination.

General export information - Since this form is not designed expressly for export information, placing of the information in the fields entitled "Foreign port of lading," "Date of sailing," "Exported on the," "Flag," "on," "Exported from," and "Goods now at," should all be left blank. This form is a multi-use form and these fields are used for imports rather than for exports.

Open space in the middle of form - As noted, this space is used for information required in 19 CFR 146.62(b)(2) and where applicable, 19 CFR 146.70(c) and (d), and in the FTZ Manual. Show also here any discrepancies in quantity noted by the cartman or bonded carrier upon pickup, with concurrences and seal notations.

To the Inspector - The dated signature of CBP personnel in this space usually constitutes the permit to transfer the merchandise for the zone. In rare instances, CBP personnel may physically supervise the transfer, in which case the permit will be shown in the space in the lower left corner beginning with, "Delivered to the Carrier....." In the case of a Weekly Export Permit, no CBP signature is required in this space and a notation should be made as to "FTZ #, In-bond Program, No CBP signature required.

Received from the Port Director - The signature and date of the cartman or bonded carrier in this space relieves the zone Operator of responsibility for the merchandise and transfers that responsibility to the cartman or carrier. Any discrepancies in quantity or condition found upon receipt by the cartman or carrier should be shown in the open space in the middle of the form.

Record of Cartage or Lighterage - This portion of the form may be filled out for the cartage, in lieu of filing other CBP forms. If the CBPF 7512 is used for cartage, 3 extra copies must be prepared as the zone Operator, cartman, and exporting carrier copies of the cartage record.

APPENDIX L

Approved OMB No. 1651-0055 Exp. 09-30-2008

DEPARTMENT OF HOMELAND SECURITY U.S. Customs and Border Protection HARBOR MAINTENANCE FEE			Identifying Number				
QUARTERLY SUMMARY REPORT			EIN or	IRS Number Ci	BP Number	SSN	
			2. Company Na	me			
			Individual Na	me			
 Complete Mailing Address 							
			and Street 2)		1000-4		
(Number and Street)		(Number 2	and Street 2)	((City)		
(State) (Zip Code)	(Country)						
(E-mail address)							
(Phone Number)	(Fax Number)		Check here i	f address has changed	since last filing.		
REPORTING PERIOD		1st Quarte	r (Jan. 1 - Mar. 31)	3rd Quarte	er (Jul. 1 - Sep. 30	0)	
Year (One Quarter Only)		, , , , , , , , , , , , , , , , , , , ,			r (Oct. 1 - Dec. 31)		
Type of Shipment With Class Code	5. Value of Shipments	correspor	6. temptions (from ding columns of line 15)	7. Net Value (column 5 less column 6)		8. Itiply the amounts appropriate rate)	
A. Domestic Movements 503							
B. FTZ Admissions 505							
C. Passengers 504							
D. Total Line Value 5, 6 & 7							
9. Total HMF Due (Total of Lines 8	BA through 8C)						
ITEMIZATION OF EXEMPTIONS	A. Domestics	F	B. TZ(s)	C. Passengers		D. Total	
10. Exempt Port				7 4000019			
11. Inland Waterway Fuel Tax							
12. Intraport							
13. U.S. Mainland/State/Possession/ Territory							
14. Other							
15. TOTALS (amounts in 15A thru 15D will automatically fill 6A thru 6D above)							

AVOID CLICKING THE RESET BUTTON. IT WILL CAUSE INVALID FORM DATA.

APPENDIX M

Approved OMB No. 1651-0055 Exp. 09-30-2008

DEDARTMENT OF U.S.	MELAND SECURE	Б,	1. Identifying N	lumber			Exp. 0
DEPARTMENT OF HO U.S. Customs and E							
HARBOR MAINTENANCE FEE AMENDED QUARTERLY SUMMARY REPORT			EIN or IR	S Number	CBP Nun	nber S	SSN
			2. Company Na	ame			
	Individual Na	me					
3. Complete Mailing Address							
71							
(Number and Street)	(Num	iber 2 ani	d Street 2)		(City)		
(State) (Zip Code) (Coun	try)	(E-mall	address)				
		- Ch	eck here if address h	nas changed sir	nce last filing.		
(Phone Number) (Fax i 4. Identifying Number on Origin	Number)	-	5. Reporting P				
4. Identifying Number on Origin	ai Nepoit			-		1	3
EIN or IRS Number	CBP Number	SSN	Year	(One	Quarter Only)	2	4
Reason for Amended Report							
A. Correction of Items 1-4	B. Request for	ra Refu	nd, because:	C. Remit a	Supplement I	Payment, b	ecause:
	Calculation	on/Cleric	cal Error	1. Calcula	tion/Clerical	Error	
	Duplication				on of Shipme		
			of Exemptions		rpretation of I		
	Overvalu				•		'
	5. Other (Pl						
	5. Other (Fi	ease op	edily)	5. Other (i	riease speci	iy)	
7. Type of Shipment With	8. Value of		ue of Exemptions	10. Net \		. HMF Due	
Class Code	Shipments		corresponding	(column to		amounts in by appropri	
		COIGIIII	.57. 0 01 mic 20,	0012111	,	o) appropr	idic rate,
A. Domestic Movements 503				+			
B. FTZ Admissions 505							
B. F12 Admissions 505							
C. Passengers 504							
D. Total Values (Total Column 8, 9,	& 10)			+			
12. Total HMF Due (Total of Lines 1	1A through 11C)						
13. Previous HMF Paid for this Rep	orting Period for this T	Гуре Моч	ement				
14. A. Supplemental Payment. If li	no 12 is greater than li	ne 13					
,,							
B. Refund Due. If line 13 is gr	eater than line 12,						
ITEMIZATION OF EXEMPTIONS	A. Domestics	3	B. FTZ(s)	C. Pas	sengers	D. To	otal
15. Exempt Port	+	\rightarrow		+			
18 Jahred Westerman Fred Terr							
16. Inland Waterway Fuel Tax							
17. Intraport							
18. U.S. Mainland/State/Possession	n/						
Territory		-+					
19. Other							
20. TOTALS (amounts in 20A thru 20C will automatically fill 9A thru 9C							
above)	·						

APPENDIX N

AVOID CLICKING THE RESET BUTTON. IT WILL CAUSE INVALID FORM DATA.

Cargo Security Standards and Specifications (TD 72-56)

Part 1- Physical Security Standards

All cargo handling and storage facilities should provide a physical barrier against unauthorized access to cargo. Usually, this will require a covered structure with walls and apertures, which can be securely closed and locked. In addition, fencing may be needed:

- 1. To prevent unauthorized persons and vehicles from entering cargo storage and handling areas.
- 2. As sole protection for open storage of bulk cargo or large articles, which cannot be easily pilfered or removed without mechanical handling equipment, or which have their own inherent security (containers)?

Buildings

General Standard

All buildings used to house cargo and associated support buildings should be constructed of materials which resist unlawful entry. The integrity of the structure must be maintained by periodic inspection and repair. Security protection should be provided for all doors and windows.

- 1. Equip all exterior doors and windows with locks.
- 2. Protect all windows through which entry can be made from ground level by safety glass, wire mesh or bars.
- 3. Similarly safeguard all glassed-in areas where shipping documents are processed.
- 4. Construct all delivery and receiving doors of steel or other material that will prevent or deter unlawful entry and keep them closed and locked when not in use.
- 5. Where fencing is impractical or guards insufficient, equip the building with an intrusion detection or alarm system.
- 6. Inspections must insure particularly that there are no avenues for surreptitious entry through floors, roofs, or adjacent buildings.

Fencing

General Standard

Where cargo security is dependent upon fencing, it should enclose an area around cargo and support buildings sufficient to provide maneuvering space for pick-up and delivery vehicles and should be set off a sufficient distance on all sides from the building or exterior stored cargo. The fence line must be inspected regularly for integrity and any damage promptly repaired.

Recommended Specifications

- 1. Install chain link fencing with at least nine gauge, two-inch mesh, and at least 8 feet high (not including a barbed wire extension). If the level on which the fence is constructed is lower than the area outside the fence line, increase the height of the fence to provide an effective 8-foot fence at all points.
- 2. Top the fence with a 2-foot barbed wire extension, consisting of 3 strands of barbed wire properly spaced and angled outward.
- 3. Place fence posts on the inside of the fence, and secure them in a cement foundation at least 2 feet deep.
- 4. Ensure that objects or persons cannot pass beneath the fencing by providing;
 - A. Cement aprons not less than 6 inches thick.
 - B. Frame piping.
 - C. U-shaped stakes driven approximately 2 feet into the ground.
- 5. Avoid any conditions that compromise the fence line. Prohibit the placing of containers, dunnage, cargo vehicle or any other item that may facilitate unlawful entry adjacent to the fence line.
- 6. Where necessary, install bumpers or fence guards to prevent damage by vehicles.

Gates

General Standard

The number of gates in fences should be the minimum necessary for access. All fence gates should be at least as substantial as the fence. Gates through which vehicles or personnel enter or exit should be manned or under observation by management or security personnel.

- 1. Equip gates with a deadlocking bolt or a substantially equivalent lock that does not require use of a chain. All hardware connecting the lock to the gate should be strong enough to withstand constant use and attempts to defeat the locking device.
- 2. Construct swing-type gates so that they may be secured to the ground when closed.
- 3. Separate gates for personnel and vehicle traffic are desirable.

Gatehouses

General Standard

Operator s of facilities handling a substantial volume of cargo should maintain a manned gatehouse at all vehicle entrances and exits during business hours.

Recommended Specifications

- 1. Set the gatehouse back from the gate so that vehicles can be stopped and examined on terminal property.
- 2. Equip the gatehouse with a telephone or other communication system.
- 3. Clear the area around the gatehouse of any encumbrances that restrict the guard's line of vision.
- 4. Post prominently on the exterior of all gatehouses signs advising drivers and visitors of the conditions of entry. Include in conditions of entry a notice that all vehicles and personnel entering the area are subject to search.

Parking

General Standard

Private passenger vehicles should be prohibited from parking in cargo areas or immediately adjacent to cargo storage buildings. Access to employee parking areas should be subject to security controls.

Recommended Specifications

- 1. Locate parking areas outside of fenced operational areas or at least a substantial distance from cargo handling and storage areas or buildings and support buildings.
- 2. Require employees exiting to the parking area from the cargo area to pass through an area under the supervision of management or security personnel. Require employees returning to their private vehicles during hours of employment to notify management a
- 3. Allow parking in employee areas by permit only. Maintain a record of each issued permit, listing the vehicle registration number, model, color and year. The permit should consist of a numbered decal, tag, sticker, or sign placed in a uniform location on the vehicle.
- 4. Issue to vendors and other visitors temporary parking permits which allow parking in a designated area under security controls.

Lighting

General Standard

Adequate lighting should be provided for the following areas:

- 1. Entrances, exits and around gatehouses.
- 2. Cargo areas, including container, trailer, aircraft and rail-car holding areas.
- 3. Along fence lines and stringpieces.
- 4. Parking areas.

Recommended Specifications

- 1. The Society of Illuminating Engineers recommends the following light intensities measured at ground level:
 - A. Vehicle and pedestrian areas 2.0 foot candles
 - B. Vital structures and other sensitive areas 2.0 foot candles
 - C. Unattended outdoor parking areas 1.0 foot candles
- 2. Illuminate all vehicle and pedestrian gates, perimeter fence lines, and other outer areas with mercury vapor, sodium vapor, power quartz lamps or substantially similar high intensity lighting, employing a minimum of 400 watts per fixture. Locate lights 30 feet above ground level and properly spaced to provide the appropriate light intensity for the area to be illuminated.
- 3. Establish a system of planned maintenance.
- 4. Protect lighting subject to vandalism by wire screening or other substantially equivalent means.

Locks, Locking Devices, and Key Control

General Standard

Locks or locking devices used on buildings, gates and equipment should be so constructed as to provide positive protection against unauthorized entry. Management or security personnel should control the issuance of all locks and keys.

- 1. Use only locks having (a) multiple pin tumblers, (b) deadlocking bolts, (c) interchangeable cores, and (d) serial numbers.
- 2. To facilitate detection of unauthorized locks, use only locks of standard manufacture displaying the owner's company name.
- 3 Number all keys and obtain a signature from the recipient when issued. Maintain a control file for all keys. Restrict the distribution of master keys to persons whose responsibilities require them to have one.
- 4. Safeguard all unissued keys or duplicate keys.
- 5. Remove and secure keys from cargo handling equipment and vehicle when not in actual use

High Risk Cargo

General Standard

Adequate space capable of being locked, sealed or otherwise secured for storage of high-value cargo and packages, which have been broken prior to or during the course of unloading must be provided at each cargo handling building. When such cargo must be transported a substantial distance from the point of unloading to the special security area, vehicles capable of being locked or otherwise secured must be used. (19 C.F.R. § 4.30(1))

Recommended Specifications

- 1. Construct special security rooms, cribs or vaults so as to resist forcible entry on all sides and from underneath and overhead.
- 2. Locate such special security areas where possible so that management and/or security personnel may keep them under continuous observations. Otherwise, install an alarm system or provide for inspection at frequent intervals.
- 3. Release merchandise from such an area only in the presence of authorized supervisors and/or security personnel
- 4. Log all movements of merchandise in or out of a special security area, showing date, time, condition of cargo upon receipt, name of truckman, name of company making pick-up and registration number of equipment used.

Part 2- Procedural Security Standards

Personnel Screening

General Standard

Operator s of cargo handling facilities should conduct employment screening of prospective employees. (19 C.F.R. §§ 4.30(m), 19.3(d), 111.28(b))

- 1. Require all personnel, including maintenance and clerical personnel, who will have access to cargo areas to submit a detailed employment application that contains a photograph of the applicant and lists his/her residences and prior employment for the prior employment for the preceding 10 years.
- 2. Screen all such employment applications for:
 - A. Verification of address and prior employment
 - B. Credit record
 - C. If possible, criminal record

Security Personnel

General Standard

Operator s of cargo handling facilities should employ security personnel or assign particular personnel of the firm to be responsible for security. All Operator s handling a substantial volume of international cargo should provide guards to protect the cargo.

Recommended Specifications

- 1. Employ the number of guards required to provide adequate security for the size of each facility and the volume of cargo handled. Alarm systems, closed circuit television and other security devices may reduce the number of guards needed.
- 2. Train all company employee guard forces or insure that contract guard forces are trained in:
 - A. Methods of patrolling terminals and warehouses
 - B. Use of firearms and other equipment that may be furnished
 - C. Report writing, log and record keeping
 - D. Identification of security problems and specific trouble areas
- 3. Equip guard forces with uniforms that are complete, distinctive, and authoritative in appearance.
- 4. Provide firearms, vehicles, communications systems, and other equipment deemed necessary for the successful performance of the guard function.
- 5. Insist on physical fitness as a prime consideration in selecting a guard force. Require guards to undergo self-defense training similar to that of police agencies. Require a physical examination at least once a year.
- 6. Furnish each guard a manual covering operating procedures and standards of conduct, and a clear statement of what management expects of him.

Communications

General Standard

Adequate and reliable communications between elements of the terminal security force and from the security force to local police should be provided.

- 1. Provide security personnel with a telephone at fixed posts or two-way radio, intercom, or other type of equipment providing voice communication capability within the company.
- 2. Arrange assured means (telephone, radio or special alarm line) for summoning assistance from local police forces.

Identification System

General Standard

All Operator s of facilities handling a substantial volume of cargo should employ an identification card system to identify personnel authorized to enter cargo and document processing areas.

Recommended Specifications

- 1. Include on the I.D. card: (a) physical description or, preferably, a color photograph of the holder, (b) name and address, (c) social security number, (d) date of birth, (e) employer's Customs license number, if any, (f) signature of holder, (g) reasonable expiration date.
- 2. Laminate all cards to prevent alterations and assign each card a control number.
- 3. Recover I.D. cards from terminated employees.
- 4. Require each employee to display his I.D. card to gain access to the facility, to cargo areas within the facility and to areas where shipping documents are processed. Preferably, the I.D. card should be displayed so that it is visible at all times that the employee is within the facility.

Independent Contractors

General Standard

The background and corporate structure of independent contractors providing janitorial service, refuse disposal, or other services should be verified. Access by independent contractors to the facility should be under security controls.

Recommended Specifications

- 1. Periodically examine independent contractor vehicles that are parked in or near cargo areas.
- 2. Permit independent contractor employees to enter only those areas necessary for their particular work; permit them access to cargo and areas where shipping documents are located only under the supervision of security and/or management personnel.
- 3. Require independent contractors to display identification similar to that required by the facility for its own employees.

Cargo Quantity Controls

General Standard

Cargo should be tallied at time of delivery to the consignee or his agent. In the event of any discrepancies, the discrepancy must be annotated in the proprietor's records and a report pursuant to the appropriate CBP regulations must be filed.

Recommended Specifications

- 1. To facilitate accurate delivery of cargo, terminal Operator s should maintain and continuously update a location chart or list of all cargo received.
- 2. Segregate imported cargo, cargo for export, and domestic cargo.
- 3. Carriers should arrange procedures with each terminal Operator to insure that all overages and shortages are reported to CBP.

Delivery Procedures

General Standard

Gate passes should be issued to truckmen and other onward carriers to control and identify those authorized to enter the facility. Verification of the identity and authority of the carrier requesting delivery of cargo should be made prior to the cargo's release.

- 1. Require truckmen to submit proper personal identification (such as a driver's license) and a vehicle registration certificate before being issued a gate pass and being permitted to enter the facility; require truckmen to surrender the gate pass before leaving the facility.
- 2. Seal containers and trailers and note the seal number on the gate pass before delivery is effected. Verify the seal number when gate pass is surrendered.
- 3. Require the company name of all onward carriers to be clearly shown on all equipment. Do not accept temporary placards or cardboard signs as proper identification of equipment. Require carriers using leased equipment to submit the lease agreement for
- 4. Release cargo only to the carrier specified in the delivery order unless a release authorizing delivery to another carrier, signed by the original carrier, is presented and verified. Accept only original copies of the delivery or pickup orders.
- 5. Personnel processing pre-lodged delivery or pickup orders should verify the identity of the truckman and the trucking company before releasing the pickup order. Limit access to areas where such documentation is processed or held to authorized personnel and rigorously safeguard all shipping documents from theft or unauthorized observation.
- 6. Conduct delivery and receiving operations at separate docks or doors, if feasible.
- 7. Tally salvage and accumulated unclaimed cargo at the time of delivery and have management representatives and/or security personnel verify that only properly released items are included. If a terminal has truck scales, weigh the vehicle used to remove bulk salvage cargo (bales and drums) when empty and loaded.

Containerized Shipments and Seals

General Standard

All containers, trailers, rail cars and air cargo lockers entering or leaving a facility should be sealed. Mounted and high value containerized shipments should receive special security attention.

- 1. Inspect seals whenever a sealed containerized shipment enters or leaves a facility. If the seals are not intact or there is evidence of tampering or the seal numbers are incorrect, notify security and/or management personnel and tally the cargo.
- 2. Seal unsealed containerized shipments at the point of entry to the facility and note the seal number on the shipping documents. Seal all containerized shipments leaving the facility and note the seal number on the shipping documents.
- 3. Release seals to as few persons as possible. Require all persons handling seals to maintain strict control of the seals and to store them in secure place.
- 4. Maintain a seal distribution log that indicates to whom seals have been released.
- 5. Where possible, secure containers by butting or "marrying" their door ends against each other. However, do no butt them against a perimeter fence or building wall if that will compromise the protection provided by the fence or wall. In stacking containers, place those containing high value merchandise on top.
- 6. Locate high value merchandise in mounted containers or trailers in a special security holding area visible to management and/or security personnel.
- 7. When containers are mounted on frames, secure the fifth wheel by a pin-lock that meets the minimum standards for locks and is constructed to withstand normal abuse from equipment. Hold designated management and/or security personnel responsible for storage and control of pin-locks equipment.
- 8. Restrict access to special security holding areas and permit the release of containers or trailers from such areas only in the presence of management representatives and/or security personnel.
- 9. Log movements of containers in or out of a special security holding area, showing: date, time, seal number, name of truckman, name of company making pickup, and registration number of equipment used.

Security Education

General Standard

Management should institute a security awareness program for all personnel.

- 1. Conduct a program of periodic security seminars for all employees involved in cargo handling and documentation processing, stressing the importance of:
 - A. Maintaining legible and accurate cargo tallies
 - B. Processing only legible documents
 - C. Writing only in ink or ballpoint pen
 - D. Completing all information required by shipping documents
 - E. Obtaining clearly written signatures
 - F. Safeguarding the confidentiality of shipping and entry documents

Glossary of Foreign-Trade Zone Terminology

Activation

Approval by the Grantee and CBP Port Director permitting operations to begin which allow the admission and handling of merchandise in zone status.

Admission

The physical arrival of goods into a zone, in a specified zone status, with the appropriate approvals of the zone Grantee and CBP. The word "admission" is used instead of "entry" to avoid confusion with CBP entry processes under Parts 141-144 of CBP Regulations.

Alteration

- a) A change in the boundaries of an activated zone or subzone.
- b) Activation of a separate site of an already activated zone or subzone with the same Operator at the same port.
- c) The relocation of an already activated site with the same Operator .

Compliance Assessment

Is a mechanism by which a team evaluates a company to ensure its CBP transactions are in compliance with the laws and regulations. The objective of a compliance assessment is to access the compliance level in specific trade areas; to determine if the importer had documented CBP related internal controls; and, to evaluate the risk to CBP of the importer's noncompliance.

Customs Territory

Customs territory is the territory of the U.S. in which the general tariff laws of the U.S. apply. Customs territory of the United States includes only the States, the District of Columbia and Puerto Rico. (General Note 2, Harmonized Tariff Schedule of the United States (19 U.S.C. 1202)

Deactivation

Voluntary discontinuation of the activation of an entire zone or subzone by the Grantee or Operator. (Discontinuance of the activated status of only part of a zone is an alteration.)

Direct Delivery

A procedure for delivery of merchandise to a zone without prior application and approval on CBP Form 214; designed for low-risk, repetitive shipments whose ordering and timing are under the control of the Operator. Approval to utilize direct delivery must be obtained from the Port Director.

Domestic Status (D)

Status of zone merchandise grown, produced or manufactured in the U.S. on which all internal revenue taxes have been paid or the status of zone merchandise previously imported on which all applicable duties and internal revenue taxes have been paid.

Drawback

Drawback means the refund or remission, in whole or in part, of a CBP duty, fee or internal revenue tax which was imposed on imported merchandise under Federal law because of its importation, and the refund of internal revenue taxes paid on domestic alcohol as prescribed in 19 U.S.C. §1313(d).

Enforced Evaluation Team

Is a group which evaluates the discrepancy or possible violation of a company to determine if it is an isolated instance of noncompliance or part of a larger pattern or history of violations? The EET considers input from the discovering personnel, the Account Manager/Port Account Team, Compliance Assessment Team Leader, as well as the significance/materiality of the violation. Based upon the nature, extent and impact of the discrepancy/possible violation, the EET selects the appropriate response to the problem.

Entry

Notification to CBP of the arrival of imported goods in the Customs territory of the U.S. Merchandise withdrawn from a zone for consumption in the U.S. is entered when it is removed from the zone. Goods brought into a zone are admitted.

Foreign-First (FOFI)

An accounting method based on the assumption that foreign-status merchandise is decremented first.

General-Purpose Zone

A general-purpose zone is established for multiple activities by multiple Users. Storage, distribution, testing, repackaging and repair are some of the possible activities in a GPZ. Processing or manufacturing in a GPZ requires the permission of the Foreign-Trade Zones Board.

Grantee

A corporation to which the privilege of establishing, operating and maintaining a foreign-trade zone has been granted by the Foreign-Trade Zones Board. Grantee corporations must be either public corporations or private corporations organized for the purpose of establishing a zone project. Examples of public entities that might receive an FTZ grant include: a political subdivision (including a municipality), a public agency, or a corporate municipal instrumentality of one or more states. Qualified private corporations must be chartered for this purpose under a law of the state in which the zone is located.

Harmonized Tariff Schedule of the United States (HTSUS)

Published by the U.S. International Trade Commission, the TSUS is used in the classification of imported merchandise for rates of duty and statistical purposes.

Inverted Tariff Structure

Where imported parts are dutiable at higher rates than the finished product into which they are incorporated.

Manipulation

As defined in Section 562 of the Tariff Act, processing wherein merchandise is packed, unpacked, repacked, cleaned, sorted, graded or otherwise changed in condition. The precise distinction between manipulation and manufacturing is subject to interpretation and enjoys a long history of case law.

Manufacturing

CBP determines what constitutes manufacturing on a case-by-case basis, distinguishing it from other operations such as manipulation, processing, production and blending. The FTZ Board has defined it as any process that results in a change in CBP classification of the merchandise, and therefore, requires prior clearance from the Board pursuant to the manufacturing conditions in specific foreign-trade zone grants.

Merchandise

FTZ merchandise includes goods, wares, and chattels of every description. Not included are prohibited merchandise, building material and supplies for use in the operation of a zone.

Nonprivileged Foreign Status (NPF)

Status of zone merchandise not previously cleared by CBP which is appraised in the condition of the merchandise at the time it enters Customs territory upon exiting the zone. NPF status may be changed upon approval from CBP, provided the merchandise is still in the same condition as when admitted to the zone. While in the zone, NPF status merchandise can be manipulated or manufactured into another commercial item with a different tariff classification. NPF status allows zone Users to pay duty at the rate of the finished product produced in the zone.

Operator

A corporation, partnership or person that operates a zone or subzone under the terms of an agreement with the Grantee. A Grantee may act as its own Operator.

Operator 's Bond

A bond submitted to CBP, on CBP Form 301, to assure compliance with CBP Regulations as set forth at 19 CFR 113.73.

Port of Entry

A place designated by the U.S. Government at which CBP personnel is assigned with authority to accept entries of merchandise, collect duties and enforce the various provisions of CBP laws.

Privileged Foreign Status (PF)

Zone status whereby merchandise is classified and appraised, with duties and taxes determined, at the time the status is elected. Once chosen, the privileged foreign status cannot be changed.

Processing

Any zone activity (other than manufacturing) requiring a change in condition of merchandise which results in a change in the CBP classification of an article or in its eligibility for entry for consumption.

Risk

The degree of exposure to the chance of noncompliance which would result in the loss or injury to the trade, industry or the public.

Subzone

A special-purpose zone established as part of a zone project for a limited purpose that cannot be accommodated within an existing general-purpose zone. Subzones must be sponsored by the Grantee of a general-purpose zone.

User

A person or company using a zone for storage, handling or processing of merchandise. An Operator may authorize a User to maintain its own inventory system and procedures manual. However, the Operator remains responsible to CBP for inventory control unless the User posts its own Operator's bond.

Weekly Entry Procedures

A CBP procedure that permits selected qualified zones and subzones to file a weekly entry on CBPF 3461 for the estimated removals of merchandise destined for domestic consumption during the following business week. Once the Port Director has approved the entry, the Operator may ship the products all week up to the quantity estimated. Weekly entry may be approved for zone operations of a repetitive nature in order to allow for expedited removal of merchandise from the zone.

Zone Lot

A collection of merchandise maintained under an inventory control method based on specific identification of merchandise admitted into a zone by lot and lot number (ZLN).

Zone Restricted Status

Status of zone merchandise transferred to a zone for the sole purpose of exportation or destruction. Zone restricted merchandise cannot be changed or brought into Customs territory without the specific permission of the Foreign-Trade Zones Board on a case-by-case review.

Zone Status

The status of merchandise admitted to a zone, i.e. domestic (D), non-privileged (NPF), privileged foreign (PF) or zone restricted (ZR).

KEY WORD INDEX

F	
FIFO	
N	
NAFTA	
Duty	
Deferral	112, 116, 136, 153, 154
P	
Packages	
Marking	
Zone admission number	
Privileged	
Foreign	
Status	55, 56, 68, 116, 117, 121, 124, 128, 129, 153, 167, 171, 173, 175
S	
Statistical	
Reporting	
Suspense	
Account	64, 65, 76, 77, 89
T	
Time of admission	55, 62, 70, 122, 126, 178, 204
U	
UIN	30, 32, 55, 58, 86, 88, 89, 90, 91, 98, 102
Z	
Zone-Restricted Status	16, 50, 57, 58, 59, 60, 61, 66, 72, 81, 82, 83, 84,
	85, 94, 101, 105, 113, 119, 127, 130, 134, 135,
	136, 137, 138, 144, 145, 146, 147, 148, 149,
	150, 161, 165, 173, 174, 175, 176